CARLOS HERNÁNDEZ MIRELES

Marketing Modeling for New Products



Marketing Modeling for New Products

Marketing Modeling for New Products

Marketing Modellen Voor Nieuwe Producten

Thesis

to obtain the degree of Doctor from the Erasmus University Rotterdam by command of the rector magnificus

Prof.dr. H.G. Schmidt

and according to the decision of the Doctorate Board

The public defense shall be held on

Tuesday 29 June 2010 at 13:30 hours

by

Carlos Hernández Mireles born in Monterrey, México.

zafing ERASMUS UNIVERSITEIT ROTTERDAM

Doctoral Committee

Co-promotor:

Promotor:	Prof.dr. P.H. Franses
Other members:	Prof.dr. B.J. Bronnenberg Prof.dr. P.K. Chintagunta Prof.dr. R. Paap
Co-promotor:	Dr. D. Fok

Erasmus Research Institute of Management - ERIM Rotterdam School of Management (RSM) Erasmus School of Economics (ESE) Erasmus University Rotterdam Internet: http://www.erim.eur.nl

ERIM Electronic Series Portal: http://hdl.handle.net/1765/1

ERIM PhD Series in Research in Management, 202 Reference number ERIM: EPS-2010-202-MKT ISBN 978-90-5892-237-3 ©2010, Carlos Hernández Mireles

Cover design: B&T Ontwerp en advies www.b-en-t.nl Print: Haveka www.haveka.nl

All rights reserved. No part of this publication may be reproduced or transmitted in any form or by any means electronic or mechanical, including photocopying, recording, or by any information storage and retrieval system, without permission in writing from the author.

To My Family Carlos, Leticia, Tanhia and Ileana

Acknowledgements

I started my Ph.D. in November 2006 and this thesis represents the result of three and a half years of joint work with my Ph.D. advisors Philip Hans Franses and Dennis Fok. I thank Philip Hans and Dennis both for their personal support and for the great learning experience that they provided me during this time. Just a couple of minutes of discussion with Philip Hans are enough to gain vital research insights and these minutes will translate into many weeks of work. I am the first Ph.D. student that Dennis supervises and I must say that he did an excellent job and I benefited enormously from our frequent discussions concerning marketing and Bayesian methods. I thank Richard Paap because he was, although unofficially, my third advisor and my thesis benefited from his comments and modeling insights. I also thank Bart Bronnenberg and Pradeep Chintagunta for evaluating this thesis as members of my inner committee.

I am grateful to many people who contributed to my learning experience during my Ph.D. and M.Phil. studies at the Erasmus Research Institute of Management (ERIM). My M.Phil. was sponsored by Cemex and I specially thank Juan Carlos Delrieu and Luis Hernández whose support was key to make the sponsorship possible. I also thank Mauricio Mora and Jorge Martínez who encouraged me to continue my studies. ERIM proved to be the right place to purse my own interests in econometrics and at the same time to gain advanced management and marketing knowledge. I specially enjoyed the courses of Ale Smidts, Alex Koning, Bauke Visser, Daan van Knippenberg, Frans van den Bosch, Patrick Groenen, Philip Hans Franses, Richard Paap, Stefan Stremersch and Stijn van Osselaer. I also enjoyed the summer course of Alan Gelfand and Bradley Carlin that was organized by the Erasmus Medical Center. In addition, I thank the ERIM and Econometric Institute staff who facilitated my research in many different ways. Erasmus University and ERIM host a truly diverse and inspirational research environment. There are many friends and colleagues who have made my doctoral studies an enjoyable experience. Alice, Antonio, Betty, Cecy, Chuy, Erick, Hans, Mariana, Ordener and Roberto always kept me informed and optimistic about Mexico. Agatha, Anna and Hendrik, Annie, Fritz and Anna Margriet, Bettina, Claudio, Esther, Gao, Liz and Karol, Margreet, Nadji, Paula, Roel, Ron, Wessel and Yang will always be in my good memories of Holland. I specially thank Nicolás Gutiérrez who always set some time apart to meet with me during his yearly visits to RSM. I thank Jose and Morteza for their help organizing this day and for being my paranymphs. Finally, I thank my fellow Ph.D. candidates Amir, Andrey, Bram, Cerag, Diana, Ezgi, Francesca, Georgi, Hendrik, Joao, Joost, Kar Yin, Lanah, Merel, Milan, Nima, Nuno, Rene, Ron and Wei. They offered me understanding and we shared all what comes with Ph.D. life.

I completed my Ph.D. largely thanks to the inspiration and unconditional loving support of my family. Carlos, Leticia, Ileana, Tanhia and David are my mentors in all sorts of matters that range from molecular biology and its applications in biotechnology up to much deeper subjects such as hard work, friendship and love.

Carlos Hernández Mireles Rotterdam, June 29th 2010

Contents

Acknowledgements

1 Introduction

1

2	The	Launch Timing of New and Dominant Multi-Generation Technologies	7
	2.1	Introduction	8
	2.2	Literature Review	10
	2.3	A Multi-Product Diffusion Model with Competition	13
	2.4	The Video Game Hardware Market	22
	2.5	Estimation and Parameter Assumptions	26
	2.6	Estimation Results	27
	2.7	Duopoly Case Study: The Portable System Race	30
	2.8	Triopoly Case Study: The Video Game Console Race	34
	2.9	Conclusions and Discussion	40
	2.10	Tables and Figures	43
	2.A	Strategy Simulation Methodology	66
_			
3	The	Timing and Speed of New Product Price Landings	69
	3.1	Introduction	70
	3.2	Literature Review	71
	3.3	Video Game Prices	75
	3.4	Price Landing: Modeling	77
	3.5	Results	87

	3.6	Conclusions	91	
	3.7	Figures and Tables	93	
	3.A	Estimation Methodology	110	
4	Ran	dom Coefficient Logit Models for Large Datasets	117	
	4.1	Introduction	118	
	4.2	Augmented Bayesian BLP Model	120	
	4.3	Bayesian Inference	125	
	4.4	Simulation Experiment	132	
	4.5	Empirical Application	136	
	4.6	Conclusions	139	
	4.7	Tables and Figures	141	
	4.A	Appendix	157	
5	Fine	ling the Influentials that Drive the Diffusion of New Technologies	159	
	5.1	Introduction	160	
	5.2	Literature Review	162	
	5.3	Methodology	164	
	5.4	Data and Modeling Details	168	
	5.5	Results	170	
	5.6	Conclusions	176	
	5.7	Tables and Figures	178	
	5.A	Methodology	208	
N	ederl	andse Samenvatting (Summary in Dutch)	221	
R	esum	en en Español (Summary in Spanish)	223	
Bi	ibliog	raphy	225	
A	Author Index 2:			
\mathbf{A}	About the Author 239			

List of Tables

Chapter 2

2.1	Release Dates of Portable Systems	43	
2.2	Release Time Between Portable Systems (in Years) $\ldots \ldots \ldots \ldots \ldots \ldots$	44	
2.3	Release Dates of Major Video Game Consoles	45	
2.4	Time Between Major VGC Releases (in years).	46	
2.5	Bass Model Estimates for Portable Systems	47	
2.6	Bass Model Estimates for Video Game Consoles	48	
2.7	Video Game Effects on Game Systems	49	
2.8	Multi-Generation Model for Portable Systems	50	
2.9	Multi-Generation Model for Video Game Consoles (Microsoft $\alpha = 1$, Sony $\alpha = 1$,		
	Nintendo $\alpha = 1$)	51	
2.10	Multi-Generation Model for Video Game Consoles (Microsoft $\alpha=0.3,$ Sony $\alpha=$		
	0.1, Nintendo $\alpha=1.1)$	52	
2.11	Competitive Parameters	53	
2.12	Evaluation of Four Launch Strategies	54	
Chapte	Chapter 3		
3.1	Literature Review on New Products Pricing	104	
3.2	Estimation Results Part I	105	
3.3	Estimation Results Part II	106	
3.4	Results of Hierarchical Structure for Mixture Probabilities	107	
3.5	Forecasting Performance	108	

3.6	Comparison with Alternative Model)9
Chapt	er 4	
4.1	Simulation Experiment: Posterior Distribution of the Variance of the Demand	
	Shocks	1
4.2	Simulation Experiment: Posterior Distribution of σ_m^2	12
4.3	Simulation Experiment: Posterior Distribution of the elements of D^2 14	13
4.4	Application: Posterior Mean and HPDR of the τ_m^2	4
4.5	Application: Posterior Mean and HPDR of the σ_m^2	15
4.6	Application: Posterior Mean and HPDR of the Fixed Elements of f_m 14	16
4.7	Application: Posterior Distribution of the Elements of the D^2 matrix 14	! 7
Chapt	er 5	
5.1	R Code to Retrieve Data from VGChartz.com	78
5.2	State Inclusion Probabilities for Each Diffusion Period for the Nintendo Wii 17	' 9
5.3	State Inclusion Probabilities for Each Diffusion Period for the Sony PS3	30

5.3	State Inclusion Probabilities for Each Diffusion Period for the Sony PS3 $\ . \ . \ .$	180
5.4	State Inclusion Probabilities for Each Diffusion Period for the Microsoft Xbox 360	181
5.5	Posterior of MCAR δ coefficients	182
5.6	Posterior of MCAR Spatial Effects	183
5.7	Posterior of MCAR Spatial Effects	184
5.8	Posterior of MCAR Spatial Effects	185
5.9	Posterior of MCAR Spatial Effects	186
5.10	OLS δ coefficients	187
5.11	Posterior of MCAR Λ correlations $\ldots \ldots \ldots$	188
5.12	Highest Posterior Density Region (HPDR) for the ρ coefficient	189
5.13	Aggregate Sales Data Model	190

List of Figures

Chapter 2

	2.1	Interaction Between All Product Generations in Duopoly Model	55
	2.2	Model Relationship with Previous Research $\hdots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	56
	2.3	Multi-Generation Model Fit for Portable Systems	57
	2.4	Multi-Generation Model Fit for Video Game Consoles	58
	2.5	Cumulative Distribution Function of Sales given Different Strategies $\ldots \ldots \ldots$	59
	2.6	Strategy Sales Sensitivity to Competitive Parameters	60
	2.7	What if Scenarios for the Consoles of Nintendo	61
	2.8	What If Scenarios for the Consoles of Sony	62
	2.9	Sensitivity to Launch-Timing and Sales Reaction Surfaces	63
	2.10	Sony PS3 Optimal Launch-Timing Sensitivity to Competitive Parameters	64
	2.11	Sony PS3 Optimal Launch-Timing Sensitivity to Cannibalization and Competi-	
		tive Parameters	65
Cl	napte	er 3	
	3.1	Price Landing Pattern for 50 Randomly Selected Games	93
	3.2	Typical Price Landing Pattern	94
	3.3	The Video Games Market	95
	3.4	What do publishers sell?	96
	3.5	Total Sales Distribution	97
	3.6	Main Pricing Function at Different Parameter Values	98
	3.7	Identification of Triggers	99

	3.8	Histogram of the Posterior Mean of Starting (ρ_i) and Landing Price (κ_i) Parameters 1	.00
	3.9	Histogram of the Posterior Mean of the Threshold (λ_i^k) and Speed (ν_i^k) Parameters1	.01
	3.10	Histogram of the Posterior Mean of the α_i Parameters $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$.02
	3.11	Histogram of the Posterior Mean of Price Triggers $P(S_i = k)$.03
Cl	napte	er 4	
	4.1	Performance of Halton Based Normal Draws versus Normal Draws 1	48
	4.2	Prior Correlations for Different Elements of Ψ	.49
	4.3	Simulation Experiment: Real (Circles) versus the Posterior Distribution (Box-	
		plots) of the Fixed Coefficients	.50
	4.4	Simulation Experiment: Real (Solid Line) versus Posterior Mean (Dots) and the	
		99% HPDR (Dashed Lines) of the Time-Varying Brand Coefficients at Market 5 -1	51
	4.5	Simulation Experiment: Real (Solid Line) versus Posterior $99\%~{\rm HPDR}$ (Dashed	
		Lines) of All Elements in the Correlation matrix R	52
	4.6	Application: Time-Profile Relative to First Period of 8th Market Time-Varying	
		Factors f^m (Solid Lines) and their 99% HPDR (Dashed Lines)	53
	4.7	Application: Distribution of 60 Correlation Elements of the Ψ matrix $\ .$ 1	54
	4.8	Own Price Elasticity for Products at Market 2	55
	4.9	Cross-Price Elasticities at Market 2	56
Cl	napte	er 5	
	5.1	Google Insights for Search	91
	5.2	Model Size (Nintendo Wii)	.92
	5.3	State Inclusion Probabilities for Each Diffusion Period of the Nintendo Wii 1	.93
	5.4	State Inclusion Probabilities for Each Diffusion Period of the Sony PS3 1	.94
	5.5	State Inclusion Probabilities for Each Diffusion Period of the Xbox 360 1	.95
	5.6	Moran's I and Geary's C: Real vs. Simulated 1	.96
	5.7	Distribution of Regression Coefficients for the Wii Model	.97

	0		
5.8	Distribution of Regression	Coefficients for the PS3 Model	198
5.9	Distribution of Regression	Coefficients for the Microsoft Xbox Model	199

5.10	Scatter Plots (Wii): Inclusion Probabilities vs. Search Elasticity $\hfill \ldots \ldots \ldots$.	200
5.11	Scatter Plots (PS3): Inclusion Probabilities vs. Search Elasticity \hdots	201
5.12	Scatter Plots (X360): Inclusion Probabilities vs. Search Elasticity	202
5.13	Spatial Effects of the Nintendo Wii during First Diffusion Period $\ . \ . \ . \ .$	203
5.14	Spatial Effects of the Nintendo Wii during Second Diffusion Period	204
5.15	Spatial Effects of the Nintendo Wii during Third Diffusion Period $\hfill \hfill \hfi$	205
5.16	Spatial Effects of the Nintendo Wii during Fourth Diffusion Period $\ . \ . \ . \ .$	206
5.17	US State Map (Source: Wikipedia)	207

Chapter 1

Introduction

Thesis Road-map

In this thesis we address the marketing of new products using mathematical and econometric models. We present a collection of models that are useful to study the following topics: (1) The optimal launch time of new and dominant technologies, (2) The triggers, speed and timing of new products' *price landings*, (3) The consumer heterogeneity that drives substitutions patterns. And, (4) the influential locations that drive the diffusion of new technologies.

These topics are explored in depth in the next four chapters and the topics of the chapters follow the order introduced above. Each chapter is self-contained and can be read independently from the others. However, the four chapters share a similar structure. That is, each chapter consists of an executive summary, a literature review, the modeling and econometric approach and its own conclusions or discussion.

The econometric approaches that we apply are diverse but they are mainly Bayesian. The exception is the second chapter where we apply non-linear least squares and simulation methods. The third chapter involves Bayesian mixture modeling. In the fourth chapter we present a new Bayesian approach for the random coefficient logit model. Finally, the study in the fifth chapter is based on Bayesian variable selection techniques and Bayesian spatial models.

In the next section we introduce the topics that we will explore in the next four chapters and we aim to give an impression and short overview of some of the important aspects related to the marketing of new products. The overview is based on Apple because the marketing techniques of this company offer a great setting related to the topics covered in this thesis. Note, however, that this thesis' research is not applied to Apple's products. After the overview, we conclude this introductory chapter with a summary of the academic contributions of this thesis.

The perfect marketing for new products?

When will Steve Jobs launch the next generation of the iPhone, the iPhone 4G? Hopefully for those working in marketing, Steve Jobs will prefer to launch the iPhone 4G at the time indicated by Apple's Vice-President (VP) of Marketing and at a time after the engineers and designers at Apple finished its technological development. But what will be the timing suggested by Apple's Marketing VP? Is it likely that the Marketing VP will strive to find the launch date that could result in the greatest consumer demand possible at all dates after the iPhone 4G launch? The question now seems to be when consumers, both current owners and non-owners of the iPhone, will purchase the iPhone 4G. Will they be anxiously waiting to purchase it as soon as it is available online or at their local Apple shop? Or will consumers wait some time after its introduction or will they even wait to leap-forward to a superior iPhone a couple of generations ahead, say, to the iPhone XG?

Currently, the iPhone is the leading and dominant technology in the smart-phone segment. One of the closest competitors of the iPhone is the BlackBerry produced by Research in Motion (RIM). How much do we know about the BlackBerry's "generations"? RIM managers decided to manage their products in a very complex generational series. Consumers have the option to buy the BlackBerry Bold 9700, the BlackBerry Storm2 9550, the Storm2 9530, the Black-Berry Curve 8900, the Curve 8500, the Curve 8300, the Bold 9000, the Tour 9630 and so on. Surprisingly, a similar generational marketing strategy is used by Nokia, Samsung and other phone manufacturers. That is, the current iPhone is competing against dozens of products. Is the communications market the only market where the iPhone is competing? The answer is no. The iPhone is the top ranking camera in Flickr and hence it may be the most popular device to make photos worldwide. The next most popular device in Flickr is the Canon EOS Digital Rebel, that is a Canon digital SLR!¹ Moreover, the iPhone is becoming a popular gaming plat-

¹See http://www.flickr.com/cameras/ for the Flick rankings and http://na.blackberry.com/eng/devices/ for the latest list of RIM devices.

form and it is competing also against the Nintendo DS and the PlayStation Portable. Each new market expands the market potential of the iPhone while at the same time each new market may be a call for tougher competition and retaliation. Later, we will refer to technologies that fight for dominance as *alpha technologies* because these markets resemble the struggle for dominance between, for example, *alpha* chimpanzees. The iPhone faces a market where it may be classified as the dominant and only *alpha* technology in the smart phone segment. However, a common setting consists of several *alpha* technologies all of which have the potential to become the market leader. That is the setting that we study in the second chapter. In the second chapter of this thesis we present a multi-generation model for new and dominant technologies. We specifically focus on the topic of the launch timing of *alpha* technologies and its optimality.

In all ways, Apple is doing a great effort to increase the desirability of its products much before their market launch and in fact, during all their life-cycles. If the marketing strategy is effective then the VP of Marketing could pick a launch date, for example, and then do her best to set an introductory price and launch Apple's product at a good timing relative to its marketing and advertising campaigns. The launch of the iPad has brought attention to Apple's pricing strategy. Not surprisingly, Apple aims to convince its consumers that the iPad is "a magical and revolutionary product at an unbelievable price". That is exactly the current main welcome message at www.apple.com. Of course, prices play an important marketing role and Apple has tried to manage the timing and depth of price cuts carefully. In general, prices of high-tech products show sudden transitions from initial high levels to permanent much lower levels. There may be many different reasons behind a price cut, like demand, competition, products release schedule or seasons, and Apple is adapting each of its products pricing to their specific competitive and demand settings. Later, we will refer to these transitions as price landings. In the third chapter of this thesis we present an empirical study of price landings and their potential triggers. More specifically, we study the heterogeneity of *price landings* and our modeling approach uncovers the relative importance of different landing triggers.

The focus of Apple's marketing efforts varies per product. Recently, the advertising of Mac computers was focused on its product features, the technology. The "Hello, I'm a Mac" ads made special emphasis on the superiority of Mac computers relative to PC's. In contrast, the

marketing for the iPhone was based on its applications ("Apps") while the Apps were not really an Apple's product. However, the flexibility, diversity and immense capabilities of these Apps was featured as the main product to advertise in the marketing campaign "There is an App for That". That is, the Marketing VP might have realized that network effects and the demand for software could increase the demand for the iPhone. The third example is the recent marketing campaign for the iPod and this time the focus were its users. The "dancing silhouettes" campaign featured only color silhouettes of iPod users dancing different types of music or it featured bands and their music, like U2 playing Vertigo. In summary, Apple is addressing consumer heterogeneity with brand-specific campaigns. In the fourth chapter we present a methodology that is useful to capture consumer heterogeneity and preference evolution based on aggregate sales data. Specifically, we present an approach that augments previous Bayesian analysis of the random coefficient logit model. We present a modeling approach that is new because it adds market-specific and global priors, time varying preferences and finally we model heterogeneity with a novel structure.

Overall, Apple's is known as a firm aiming to provide the best consumer experience and it is usually mentioned as a company with great customer service. There are, however, groups of customers that receive greater attention and these are Apple's fans. Steve Jobs manages and talks to this influential and selected group of consumers at different moments. The last time that Steve Jobs appeared on stage as key-note speaker was on January 27th of 2010 and he devoted a complete event to describe the features of the iPad to Apple fans and to the press. In addition, he announced the pricing for the iPad and its launch date. The iPad will be available at Apple stores on April 3rd 2010 and it can be pre-ordered since March 12th 2010. Influentials are people who have a significant effect on the behavior of others and they might be the engine of diffusion at different moments and locations. Hence, it is key to manage influentials and convince them about the marvels of products much before everyone else. Steve Jobs key-notes are always based in San Francisco but Apple fans are everywhere. Are these fans always influential? Do they play different roles during the life-cycle of new technologies? In the fifth chapter of this thesis we present an approach to find the influential locations that drive the diffusion of technologies in aggregate sales data and in location-specific online search data. We further provide insights on how the influential locations distribute in space and how they evolve in time.

Summary and Academic Contributions

The novelty of this thesis consists of the analysis of new or very recent data and the introduction of new marketing models.

The second chapter introduces a new diffusion model that is useful to analyze the optimal introduction timing of multi-generational technologies. Special focus is given to *firms' alpha*, that is the ability of a firm to transfer users of its old technologies to their new generations, and the effects of the *firms' alpha* on the introduction dates of potential dominant technologies. This same chapter's analyses are based on recent weekly data of game consoles and video-games and we provide new insights about the optimality of the launch timing of the Nintendo Wii and the PlayStation 3. Chapter 2 is joint work with Philip Hans Franses.

Next, in the third chapter, we present a new mathematical model for sudden price transitions. Surprisingly, we are the first to empirically model specifically these transitions, what we call *price landings*, and their triggers, timing and speed. Furthermore, our analysis is based in a new dataset containing almost 1200 recently introduced products. Our contribution offer insights into the heterogeneity of *price landings* and the untangling of the most likely triggers of *price landings* based on Bayesian mixture modeling. Chapter 3 is joint work with Dennis Fok and Philip Hans Franses.

The contribution in the fourth chapter is mainly the introduction of an augmented version of recent Bayesian analysis of the random coefficient logit model. The practical application of the Bayesian random coefficient model, specifically to large datasets, requires novel approaches and model formulations. We apply our new approach to both simulated data and to a unique and very large dataset of aggregate sales and our approach proves to be promising. Chapter 4 is joint work with Dennis Fok.

Finally, in the fifth chapter of the thesis we analyze new data collected from Google Insight and we apply recent Bayesian econometric approaches to identify influentials. We focus our analysis on the identification of the influential locations that drive the aggregate sales of new technologies. The specific techniques that we apply in this chapter, Bayesian variable selection and multivariate spatial models, are new to the marketing literature. Hence, our contribution consists of the illustration of how these techniques can be applied to study marketing problems while at the same time we provide insights about the time variation and spatial clustering of influentials.

Chapter 2

The Launch Timing of New and Dominant Multi-Generation Technologies

In this chaper we introduce a model that is suitable to study the diffusion of new and dominant multi-generation technologies. Examples are computer operating systems, mobile phone standards, video game consoles. Our model incorporates three main features. First, we add the ability of a firm to transfer users of its old technologies to the new generations, what we call *firms' alpha*. Second, we add competitive relations between market technologies. Third, the launch strategies diagnosed by our model cover, as special cases, the *now or never* strategies and hence it is suitable to study intermediate launch strategies.

We state the relationship of our model to previous research both in terms of the model formulation and in terms of some of its analytical solutions. Specifically, the model may reduce to the Bass or the Norton and Bass models. Regarding the analytical solutions, we find that the *launch never* strategy arises when there are late product introductions by competitors, when a *firm's alpha* is very low, or when the competition is intense while the *launch now* strategy arises only when a *firm's alpha* is zero.

In addition, we evaluate different launch strategies and the optimality of launch timings in two detailed case studies on the video game systems market. We study the portable systems (PS) and the video game consoles (VGC) industry. We present several insights from our analysis and we find interesting explanations for the pacing strategy in this market, for which we also provide a historical perspective.

We find that the appropriate timing of a new technology depends heavily on both the *firms'* alphas and on the competitive positioning of their products. In the VGC case we find that the Nintendo Wii was launched at an appropriate moment while the Sony PS3 perhaps should have never been launched.

2.1 Introduction

In a well-known study on the behavior of chimpanzees Jane Goodall writes:

"In 1963 Goliath, a powerful and aggressive male in his prime (perhaps about 25 years of age) was the alpha male. He had a spectacular charging display during which he covered the ground very fast indeed, dragging and occasionally hurling branches. Early in 1964, however, Goliath was displaced from his top-ranking position in the community by an older and much less robust male, Mike... Unlike Goliath, who had maintained a very high ranking position for several years after losing his alpha rank, Mike dropped rapidly to a low position in the hierarchy... In chimpanzee society, dominance is something of a conundrum. The usual interpretation of the phenomenon is that it enables a high-ranking individual to have prior access to desirable foods, females, or resting places." (van Lawick-Goodall, 1973)

We believe that Goodall's description of *dominance* in the chimpanzee society directly applies to new technologies and their markets. Specifically, markets of new technologies formed by a few firms and products and by a single or a few dominant *alpha* technologies are analogous to the few chimpanzee males that fight for the *alpha* rank. Examples of products in this type of industries are operating systems, mobile phone standards, video game consoles, smart phones, and so on.

Many technology firms, like Apple or Microsoft, launch several versions of their products, what we know as product generations. Each time a new generation product is introduced to the market some or many of the users of the old generations switch to the new one, at the same time new users may adopt the new generation product while other users may switch from one firm 's products to another firm 's products after a new introduction. That is, each product generation cannibalizes its previous generation and each firm has a different capacity of transferring the users of the old technology to the new one. For example, we know that Apple has been very successful transferring the users of its old technologies to the new ones. Linux, even though it is a smaller player, is a second example of a technology with a high *alpha*. In contrast, it was widely documented how Microsoft users were hesitant to switch from Windows XP to Windows Vista. Some Windows users stickied to Windows XP while others switched to alternative operating systems. In this chapter we will refer to the firms capacity of cannibalizing and transferring users of old technologies to new ones as the *firm's alpha*. In our example, Apple would be the player with a high *alpha*.

In this chapter we extend the Norton and Bass (1987) model by incorporating three new elements that have not been addressed simultaneously in previous literature. These are the firm's ability of transferring its users to new technologies (the *firm's alpha*), the competitive interaction between firms in the market, and a new solution to the timing of new technologies. Our model is suitable to study the timing of new generation products in industries that are characterized by a relatively slow pace of introductions and a few firms launching new technologies. In addition, we test our model empirically under different settings and based on the new model we provide insights into the launch-timing strategies and into the optimality of launch timings.

Previous empirical literature has addressed the diffusion of new multi-generation technologies, like Norton and Bass (1987), Kim and Lee (2005), Danaher et al. (2001) and Kim et al. (2000), but they do not cover the topic of introduction timing. Two exceptions are Norton and Bass (1987) and Mahajan and Muller (1996). These last authors introduce the timing of new products into their models and tested them empirically. However, both the Norton and Bass (1987) and the Mahajan and Muller (1996) models suggest to launch new technology either *now or never*. Other analytical studies have addressed specifically the timing of new technologies, like Wilson and Norton (1989), Joshi et al. (2009), Bayus et al. (1997), Souza et al. (2004) and Morgan et al. (2001), but these later authors models have not been tested empirically and in most cases their models are suitable for industries with a fast pace of technology introductions, an exception being Joshi et al. (2009). More importantly, these studies do not incorporate the three new elements we address simultaneously.

The plan of the chapter is as follows. In Section 2.2 we present our literature review. In Section 2.3 we present our model for the duopoly and triopoly case (sections 2.3.1 and 2.3.2, respectively), we discuss its relationship to previous models (section 2.3.3) and the analytical properties that distinguish it from previous models (section 2.3.4). In Section 2.4 we introduce the market context and our data. In Section 2.5 we motivate the model assumptions and the estimation procedure. In Section 3.5 we discuss the estimation results. In the next two sections we use our model to study the industry. In Section 2.7 we study the portable system market and

we give insights about different launch strategies. Next, in Section 2.8, we study the main video game console market, composed of Microsoft, Sony and Nintendo, and we focus our analysis in the latest console race. We further provide insights into how different introduction timings may be optimal. Finally, in Section 3.6 we present our discussion and conclusions.

2.2 Literature Review

To our knowledge, Wilson and Norton (1989) and Mahajan and Muller (1996) are the two key studies concerned with the question of when it is optimal for a monopoly to launch multigeneration products. According to Wilson and Norton (1989) there are three critical issues which affect the optimal introduction time of a new generation. These are the interrelationship of sales of the two products, their profit margins and the planning horizon. Surprisingly, their model provides two optimal solutions regardless of the relevance of these factors. They conclude that different generations of a product should be introduced either all at the same time or sequentially and not overlapping. In a similar vein, Mahajan and Muller (1996) conclude that a new generation should be introduced as soon as it is available (if its market potential is larger than the preceding one) or it should be delayed to a much later stage, that is, to the maturity of the previous generation. Their findings seem special cases of the solutions proposed by Kamien and Schwartz (1972). Kamien and Schwartz (1972) suggest to *never launch* a technology only under extreme competition and to *launch now* only if the firm needs to take advantage of a profit stream that would otherwise be smaller once competitors come in.

More recently, Joshi et al. (2009) study the problem of product launch timings across different markets. They characterize situations, depending on social influence, where it is optimal to launch before maturity or after the maturity of the first generation product. However, Joshi et al. (2009) do not incorporate competition and their model is only useful to study the interaction of products across markets (same product in two geographies, for example). Souza et al. (2004) study the new product introduction strategy and its relation to industry clock speed. They provide analytical evidence that a time-pacing strategy (launching products every n time periods) performs relatively well compared to the optimal strategy. Their model applies to settings with a high frequency of product introductions. The studies of Morgan et al. (2001) and Bayus et al. (1997) analyze how the trade-offs between quality or product performance (measured by development costs) interact with the introduction timing decision. In contrast, we study the relationship between cannibalization and competition with the introduction timing decisions.

The literature on multi-generation products is very extensive. Padmanabhan and Bass (1993a) and Bayus (1992) propose models to price successive generations of products, Danaher et al. (2001) analyze the relation between the marketing mix and diffusion of multi-generation products, Bucklin and Sengupta (1993) examine the diffusion of complementary innovations, Kim et al. (2001), Chatterjee and Eliashberg (1990), Kim and Srinivasan (2001), Jun and Park (1999), Vakratsas and Bass (2002) and Bayus (1991) study how and when consumers decide to upgrade to improved products' versions. Islam and Meade (2000), Islam and Meade (1997) and Olson and Joi (1985) propose models for diffusion and replacement of products, while Purohit (1994), Robertson et al. (1995) and Prasad et al. (2004) analyze the introduction strategies of multi-generations products or the release of single products in multiple channels. Finally, Kim et al. (2000), Kim and Lee (2005), Peterson and Mahajan (1978) and Islam and Meade (1997) present alternative diffusion models for successive generations of products.

Our contributions to this literature are as follows. First, we propose a model that incorporates competition and cannibalization (*firm's alpha*) based on a duopolistic and triopolistic market. Second, our model parameters are simple to estimate or to calibrate with secondary quantitative or qualitative information and it is possible to find intermediate solutions to the introduction timing problem. Third, we provide two detailed case studies about the timing of game systems that are not documented in the literature. Finally, we present new insights regarding different launch strategies and the optimality of timing decisions.

Next we briefly discuss the Norton and Bass Model (NBM) as it is our departing point and it is essential in our model development.

2.2.1 The Norton and Bass Model

In this chapter we overcome three limitations of the NBM model that have not been jointly addressed in previous research. Denote $S_1(\tau_1, \tau_2)$ as the first generation sales, $S_2(\tau_1, \tau_2)$ as the second generation sales and denote τ_1 and τ_2 as the launch moment of these generations, respectively. The first limitation is that $\partial(S_1(\tau_1, \tau_2) + S_2(\tau_1, \tau_2))/\partial \tau_2 = 0$ is obtained when $\tau_2 = 0$ or when $\tau_2 = \infty$. $S_g(\tau_1, \tau_2)$ are the sales of generation g given the introduction timings of the first and second generation products, τ_1 and τ_2 , respectively. Therefore, the basic Norton and Bass (1987) model is not helpful to derive an intermediate optimal introduction timing apart of these two solutions. The second limitation is that it assumes that all the sales of the previous generation are captured by the second generation. Finally, the NBM does not consider the diffusion of competing products.

In the NBM cumulative sales are proportional to the cumulative distribution function of the adoption rate F(t) and the market potential m. When a second generation is introduced, substitution and adoption effects should be added to the previous equation. For the case of two generations, Norton and Bass posit that the first generation cumulative sales follow

$$S_1(\tau_1, \tau_2) = m_1 F_1(\tau_1) [1 - F_2(\tau_2)], \text{ for } t > 0,$$
(2.1)

and that the second generation follows

$$S_2(\tau_1, \tau_2) = F_2(\tau_2)[m_2 + F_1(\tau_1)m_1], \text{ for } t > \tau_2$$
(2.2)

where we use $S_g(\tau_1, \tau_2)$ to refer to the vector $[S_g(\tau_1, \tau_2; t = 0), \ldots, S_g(\tau_1, \tau_2; t = T_p)]$ and $S_1(\tau_1, \tau_2; t)$ is equal to $m_1F_1(\tau_1; t)[1 - F_2(\tau_2; t)]$ while $S_2(\tau_1, \tau_2; t)$ is equal to $F_2(\tau_2; t)[m_2 + F_1(\tau_1; t)m_1]$. The introduction date of the first generation (g = 1) is τ_1 and the introduction date of the second generation (g = 2) is τ_2 . T_p is the planning horizon, which is set as ∞ in Norton and Bass (1987). $F_i(\tau_i; t)$ is the cumulative sales function of generation g defined as $F_g(\tau_g; t) = [1 - e^{-b_i(t-\tau_g)}/1 + a_i e^{-b_g(t-\tau_g)}]$ for $t > \tau_g$ and $a_g = q_g/p_g$ and $b_g = p_g + q_g$, g = 1, 2. We use $F_g(\tau_g)$ to refer to the vector $[F_g(\tau_g; t = 0), \ldots, F_g(\tau_g; t = T_p)]$. Slightly stricter notation would use $F_g(\tau_g; t, \theta)$ where $\theta = (p_g, q_g, m_g)$ but we use the former as we focus on the timing parameters in this study. Note that in the Norton and Bass (1987) τ_1 is assumed to be fixed at some value (possibly at t = 0) and they do not focus on its value.

The equations of the NBM posit that after the second generation is introduced at time τ_2 , the first generation's cumulative sales $S_1(\tau_1, \tau_2)$ become proportional to its cumulative adoption function $F_1(\tau_1)$, its market potential m_1 , and the sales not captured by the second generation $[1 - F_2(\tau_2)]$ after τ_2 . The sales of the second generation $S_2(\tau_1, \tau_2)$ are proportional to their own market potential m_2 and to the cumulative sales of the first generation $F_1(\tau_1)m_1$ after τ_2 .

If the NBM equation (2.1) would contain only the term $m_1F_1(\tau_1)$, then the sales $S_1(\tau_1, \tau_2)$ will be equivalent to the model of Bass (1969). However, in the Norton and Bass (1987) model a fraction $F_2(\tau_2)$ of $m_1F_1(\tau_1)$ is captured by the second generation. Consequently, there is a moment in time when $F_2(\tau_2)$ will become 1 and all of the first generation sales are transferred to the second generation and the last element of $S_1(\tau_1, \tau_2)$ becomes 0. At the same time $S_2(\tau_1, \tau_2) =$ $m_2F_2(\tau_2) + F_2(\tau_2)F_1(\tau_1)m_1$ and therefore, $m_1 + m_2$ is the last element of the vector $S_2(\tau_1, \tau_2)$, given in equation (2.2).

In the next section we present a model that is a generalized version of the NBM and we believe this new general model overcomes all the three limitations of the NBM.

2.3 A Multi-Product Diffusion Model with Competition

This section is divided in four subsections. In the first (subsection 2.3.1) we extend the NBM to the duopoly case and in the second (subsection 2.3.2) we extend the model to the triopoly case. Both extensions are based on the same assumptions and we present the duopoly case first for ease of exposition. In the third section we present the relationship of our model to previous models proposed in the literature (section 2.3.3). Finally, in the fourth (subsection 2.3.4) we present the intuition and the analytical properties that make our specification suitable to optimize and study the launch timing of new dominant technologies.

2.3.1 Duopoly Multi-Generation Model

In order to expand the Norton and Bass (1987) model and add a second firm or a second competing product, we should make assumptions about the relationship between the firms' products. Here we make the assumption that the relationship between the two generations products of a firm are related in a very similar but more flexible way than in the NBM, and that is where the *alpha* parameter comes in. Additionally, we will assume that the sales that go from one product to a competitor's version are proportional to the cumulative sales function of the competitor's products.

Formally, if the market is composed of two firms s and n, the cumulative sales of firm s are

$$S_1^s(\tau_1^s, \tau_2^s | \tau_1^n, \tau_2^n) = \tilde{S}_1^s(\tau_1^s, \tau_2^s) [1 - \phi_{11}^{sn} F_1^n(\tau_1^n)] [1 - \phi_{12}^{sn} F_2^n(\tau_2^n)]$$
(2.3)

and

$$S_2^s(\tau_1^s, \tau_2^s | \tau_1^n, \tau_2^n) = \tilde{S}_2^s(\tau_1^s, \tau_2^s) [1 - \phi_{21}^{sn} F_1^n(\tau_1^n)] [1 - \phi_{22}^{sn} F_2^n(\tau_2^n)]$$
(2.4)

The cumulative sales of firm n are

$$S_1^n(\tau_1^n, \tau_2^n | \tau_1^s, \tau_2^s) = \tilde{S}_1^n(\tau_1^n, \tau_2^n) [1 + \phi_{11}^{ns} F_1^s(\tau_1^s)] [1 + \phi_{12}^{ns} F_2^s(\tau_2^s)]$$
(2.5)

and

$$S_2^n(\tau_1^n, \tau_2^n | \tau_1^s, \tau_2^s) = \tilde{S}_2^n(\tau_1^n, \tau_2^n) [1 + \phi_{21}^{ns} F_1^s(\tau_1^s)] [1 + \phi_{22}^{ns} F_2^s(\tau_2^s)]$$
(2.6)

where \tilde{S}_1^i and \tilde{S}_2^i are defined as

$$\tilde{S}_1^i(\tau_1^i, \tau_2^i) = m_1^j F_1^i(\tau_1^i) [1 - \alpha_i F_2^i(\tau_2^i)] \text{ for } i = n \text{ or } s$$
(2.7)

and

$$S_2^j(\tau_1^i, \tau_2^i) = F_2^i(\tau_2^i)[m_2^i + \alpha_i F_1^i(\tau_1^i)m_1^i] \text{ for } i = n \text{ or } s$$
(2.8)

Finally we have that

$$F_g^i(\tau_g^i;t) = [1 - e^{-b_g^i(t - \tau_g^i)} / 1 + a_g^i e^{-b_g^i(t - \tau_g^i)}] \times I(\tau_g^i \ge t) \text{ for } t > 0$$
(2.9)

where $S_g^i(\tau_1^i, \tau_2^i | \tau_1^s, \tau_2^s)$ represent the sales of generation g of firm i achieved by launching its first and second generation products at τ_1^i and τ_2^i and given the competing firm s launched its products at τ_1^s and τ_2^s ; $a_g^i = q_g^i/p_g^i$ and $b_g^i = p_g^i + q_g^i$ and $I(\tau_g^i > t)$ is an indicator function that equals 1 when the introduction time of generation g of firm i, τ_g^i , is larger than or equal to t and zero otherwise. The term ϕ_{gk}^{ij} refers to the substitution (or loyalty) parameter between the generation g of firm i and the generation k of firm j. We use $F_g^i(\tau_g^i)$ to represent the vector $[F_g^i(\tau_g^i; t = 0), \ldots, F_g^i(\tau_g^i; t = T_p)]$. Again, stricter notation would use $F_g^i(\tau_g^i; t, \theta)$ where θ is a vector that collects all other parameters in the model. The parameters p_g^i and q_g^i are the innovation and imitation parameters of generation g and firm i, respectively, g = 1, 2 and i = n, s.

We may refer occasionally to ϕ as the vector (ϕ_1, \ldots, ϕ_N) where N is the number of products and to α as the vector $(\alpha_1, \ldots, \alpha_I)$ where I is the number of firms. Equations 2.3 to 2.9 allow for a wide variety of relationships given the sign and size of what we call the loyalty parameters or ϕ and the values of the the *alpha* cannibalization parameters (α) . The role of the α parameter is to relax the assumption of the NBM that all the sales of the first generation of a firm are transferred to the second generation. Note that the last elements of the vector in $\tilde{S}_2^j(\tau_1^i, \tau_2^i)$ will be equal to $m_2^i + \alpha m_1^i$ and the last element of $\tilde{S}_1^i(\tau_1^i, \tau_2^i)$ is equal to $m_1^j - \alpha m_1^j$. Therefore α can be interpreted as the proportion of sales that the first generation transfers to the next when $t = T_p$ and T_p is of course sufficiently long.

In Figure 2.1 we sketch the relationship between product generations in the duopoly model. Basically, there is substitution between all products but substitution starts at different points in time. The first generation is launched at t = 0 and it is the only product in the market up to t = T1. At this moment the first generation of the second firm is launched and the substitution between these two products (represented by the blank continuous line) starts too. The rest of the products are launched at time t = T2 and t = T3 and the substitution between them and the products launched before them start at these times. Note that the model allows for the possibility of *never* launching a product if we set its launch date at $t = T_p$. This figure represents a hypothetical case of launch dates but we can evaluate any launch-timing in the model. For example, we could evaluate the result of launching the products in reverse order or in any order. In practice the second generation arrives after the first one, but any other combination is allowed. Finally, note that there is only one single arrow between the products in the figure. That is, we assume symmetric competitive parameters. If the relationship between products is not symmetric then we would need two arrows connecting any pair of products in Figure 2.1.

Next we present the triopoly model and at the end of next section we discuss how both the duopoly and the triopoly models are related to previous research.

2.3.2 Triopoly Multi-Generation Model

In this section we extend the duopoly model and set the sales equations for firms s, n and x and we hold the assumption that each firm sells two generations of the same product.

The cumulative sales equations for firm x are:

$$S_{1}^{x}(\tau_{1}^{x},\tau_{2}^{x}|\tau_{1}^{s},\tau_{2}^{s},\tau_{1}^{n},\tau_{2}^{n}) = \tilde{S}_{1}^{x}(\tau_{1}^{x},\tau_{2}^{x})[1+\phi_{11}^{xs}F_{1}^{s}(\tau_{1}^{s})] \\ \times [1+\phi_{12}^{xs}F_{2}^{s}(\tau_{2}^{s})][1+\phi_{11}^{xn}F_{1}^{n}(\tau_{1}^{n})][1+\phi_{12}^{xn}F_{2}^{n}(\tau_{2}^{n})]$$
(2.10)

and

$$S_{2}^{x}(\tau_{1}^{x},\tau_{2}^{x}|\tau_{1}^{s},\tau_{2}^{s},\tau_{1}^{n},\tau_{2}^{n}) = S_{2}^{x}(\tau_{1}^{x},\tau_{2}^{x})[1+\phi_{21}^{xs}F_{1}^{s}(\tau_{1}^{s})] \\ \times [1+\phi_{22}^{xs}F_{2}^{s}(\tau_{2}^{s})][1+\phi_{21}^{xn}F_{1}^{n}(\tau_{1}^{n})][1+\phi_{22}^{xn}F_{2}^{n}(\tau_{2}^{n})]$$
(2.11)

The cumulative sales equations for firm s are:

$$S_1^s(\tau_1^s, \tau_2^s | \tau_1^x, \tau_2^x, \tau_1^n, \tau_2^n) = \tilde{S}_1^s(\tau_1^s, \tau_2^s) [1 - \phi_{11}^{sx} F_1^x(\tau_1^x)] \\ \times [1 - \phi_{12}^{sx} F_2^x(\tau_2^x)] [1 + \phi_{11}^{sn} F_1^n(\tau_1^n)] [1 + \phi_{12}^{sn} F_2^n(\tau_2^n)] \quad (2.12)$$

and

$$S_{2}^{s}(\tau_{1}^{s},\tau_{2}^{s}|\tau_{1}^{x},\tau_{2}^{x},\tau_{1}^{n},\tau_{2}^{n}) = \tilde{S}_{2}^{s}(\tau_{1}^{s},\tau_{2}^{s})[1-\phi_{21}^{sx}F_{1}^{x}(\tau_{1}^{x})] \\ \times [1-\phi_{22}^{sx}F_{2}^{x}(\tau_{2}^{x})][1+\phi_{21}^{sn}F_{1}^{n}(\tau_{1}^{n})][1+\phi_{22}^{sn}F_{2}^{n}(\tau_{2}^{n})]$$
(2.13)

And, the cumulative sales equations for firm n are:

$$S_{1}^{n}(\tau_{1}^{n},\tau_{2}^{n}|\tau_{1}^{x},\tau_{2}^{x},\tau_{1}^{s},\tau_{2}^{s}) = \tilde{S}_{1}^{\tilde{n}}(\tau_{1}^{n},\tau_{2}^{n})[1-\phi_{11}^{nx}F_{1}^{x}(t-\tau_{1}^{x})] \\ \times [1-\phi_{12}^{nx}F_{2}^{x}(\tau_{2}^{x})][1-\phi_{11}^{ns}F_{1}^{s}(\tau_{1}^{s})][1-\phi_{12}^{ns}F_{2}^{s}(t-\tau_{2}^{s})]$$
(2.14)

and

$$S_{2}^{n}(\tau_{1}^{n},\tau_{2}^{n}|\tau_{1}^{x},\tau_{2}^{x},\tau_{1}^{s},\tau_{2}^{s}) = \tilde{S}_{2}^{n}(\tau_{1}^{n},\tau_{2}^{n})[1-\phi_{21}^{nx}F_{1}^{x}(\tau_{1}^{x})] \\ \times [1-\phi_{22}^{nx}F_{2}^{x}(\tau_{2}^{x})][1-\phi_{21}^{ns}F_{1}^{s}(t-\tau_{1}^{s})][1-\phi_{22}^{ns}F_{2}^{s}(t-\tau_{2}^{s})]$$
(2.15)

where \tilde{S}_1^i and \tilde{S}_2^i are defined as

$$\hat{S}_{1}^{i}(\tau_{1}^{i},\tau_{2}^{i}) = m_{1}^{i}F_{1}^{i}(\tau_{1}^{i})[1-\alpha_{i}F_{2}^{i}(\tau_{2}^{i})] \text{ for } i = n \text{ or } s \text{ or } x$$

$$(2.16)$$

and

$$\tilde{S}_{2}^{i}((\tau_{1}^{i},\tau_{2}^{i}) = F_{2}^{i}(\tau_{2}^{i})[m_{2}^{i} + \alpha_{i}F_{1}^{i}(t-\tau_{1}^{i})m_{1}^{i}] \text{ for } j = n \text{ or } s \text{ or } x$$

$$(2.17)$$

and

$$F_g^i(\tau_g^i) = [1 - e^{-b_g^i(t - \tau_g^i)} / 1 + a_g^i e^{-b_g^i(t - \tau_g^i)}] \times I(\tau_g^i \ge t) \text{ for } t > 0$$
(2.18)

where $a_g^i = q_g^i/p_g^i$ and $b_g^i = p_g^i + q_g^i$ and $I(\tau_g^i > t)$ is an indicator function that equals 1 when the introduction time of generation g of firm i, τ_g^i , is larger than or equal to t and zero otherwise. The term ϕ_{gk}^{ij} is the competitive parameter that relates the generation g of firm i with the generation k of firm j. The parameters p_g^i and q_g^i are the innovation and imitation parameters of generation g, respectively, g = 1, 2.

The specification of (2.10) to (2.18) is similar to the duopoly case but now we allow for substitution between three market players x, s, and n and each of their products. The duopoly model consists of four launch-timing parameters, eight ϕ parameters, two α parameters, four p and q parameters and four m parameters. That is in total 26 parameters in four equations. The triopoly model consists of 45 parameters (six τ , 24 ϕ , six p and q, three α and six m) in six equations. In the estimation section 2.5 we describe how we calibrate both models and the parameter restrictions and assumptions we use. Next we describe the relationship of our model with previous models.

2.3.3 Links with Other Models

In Figure 2.2 we summarize the relationship of this general NBM with previous models based on different parameter configurations. It is useful to see the nodes at the top of the figure as possible cases for each firm in our model. We start with the left node. If the α parameter, in one of the firm's equations, is equal to zero then there exists no cannibalization between a specific firm generations and the diffusion of each of its generations follows an independent Bass Model. However, in this case if some of the ϕ parameters are different from zero then we have independent Bass Models but we add inter-generation competition (or what is the same as between firms competition); otherwise they follow independent Bass models. On the right hand side of the figure we see the case when the α parameter is set to 1 and this means that the relationship of generations within firms follows the NBM specification. As in the previous node the ϕ parameters may add inter-generation competition between firms (note that is not within the same firm). Finally, in the central node we have the case when α is different from both 0 and 1. In this last case, the model allows cannibalization within a firm's generations but the cannibalization is different from the NBM. Therefore we call this a *second type of cannibalization*. As before, for this node the ϕ parameters may add inter-generation competition between firms.

At the bottom of Figure 2.2 we give three boxes representing firms and the arrows correspond to two hypothetical specifications (case 1 and 2) for each firm. In the first case, firm 1 products follow a NBM with *second type of cannibalization*, firm 2 products follow independent Bass Models while firm 3 products follow the NBM. That is, in this case the only firm facing the effects of competition is firm 2. In the second case we set a different combination and our intention is to illustrate that the model parameters allow a diverse set of diffusion patters among firms and products. A similar specification for the NBM is possible when either the por q of any of the generations is equal to 0. Note that each firm launches two generations of products within the planning horizon but the triopoly model may reduce to the duopoly model in case a firm sets the launch date at the end of the planning horizon (what we refer as T_p) for its two generations. A different specification happens when each firm launches a single product by setting one of its generations launch-timing equal to T_p . Hence, our model is flexible enough to allow different substitution patterns between firms' products and within firm generations. At the same time the triopoly case might reduce to different number of firms or products depending on the parameter values.

2.3.4 Why Our Model Works

In this subsection we present the intuition of why our model is useful to find intermediates dates rather than $\tau = 0$ or $\tau = \infty$ solutions of the NBM. The intermediate solutions are possible due to the trade-off between competitive interaction between products and the cannibalization within a firm's generations. For example, if the firm *n* launches a product at time τ_c and this product might enhance/deter the sales of one of the products of firm *s* after this time. Then the firm *s* has the incentive to advance/postpone the launch of its product relative to the launch of the competing product. In this way, firm *s* could maximize/minimize the positive/negative effects of competition. That is, the timing decision depends on the sign and size of the effect of firm's *n* product on the sales of firm's *s* products. In addition, there is a trade-off between maximizing or minimizing the effect of competition and the effects on firm *s* previous generation product. Therefore, by launching the second generation sooner the previous generation might lose sales to the second generation earlier in time. In summary, the optimization of the competitive effects and the own cannibalization effects is possible in our specification while it is not possible to optimize them in the NBM.

Here we present a simplified version of the duopoly model and assume that one of the competing firms launches only one product at τ_c while the second firm s sells two products and
these are launched at τ_1 and τ_2 . We further assume that the competitive effects are measured by the coefficients ϕ_1 and ϕ_2 . Formally, the equations of firm s are

$$S_1^s(\tau_1, \tau_2 | \tau_c) = m_1 F_1^s(\tau_1) [1 - \alpha_s F_2^s(\tau_2)] [1 - \phi_1 F_1^c(\tau_c)], \text{ for } t > 0,$$
(2.19)

and

$$S_2^s(\tau_1, \tau_2 | \tau_c) = F_2^s(\tau_2) [m_2 + \alpha_s F_1^s(\tau_1) m_1] [1 - \phi_2 F_1^c(\tau_c)], \text{ for } t > \tau_2$$
(2.20)

That is, the first and second generation sales of firm s, $S_1^s(\tau_1, \tau_2 | \tau_c)$ and $S_2^s(\tau_1, \tau_2 | \tau_c)$, are now related to the competing product by the loyalty parameters ϕ_1 and ϕ_2 . It is easy to show that the sales gained or lost by adding competition to the NBM (with cannibalization of type 2) are

$$\Delta_s = \left[\alpha_s(\phi_2 - \phi_1)m_1F_1^s(\tau_1)F_2^s(\tau_2) + \phi_1m_1F_1^s(\tau_1) + \phi_2m_2F_2^s(\tau_2)\right]F_1^c(\tau_c), \text{ for } t \ge \tau_c \quad (2.21)$$

 Δ_s is the sales change due to the introduction of a competing product and it depends on the parameters α_s , ϕ_1 and ϕ_2 and on the introduction timings τ_1 and τ_2 relative to τ_c . The terms $\phi_1 m_1 F_1^s(\tau_1)$ and $\phi_2 m_2 F_2^s(\tau_2)$ measure the share of each product of firm *s* that might be transferred/received to/from a competing product and the shares are ϕ_1 and ϕ_2 . The term $\alpha_s(\phi_2 - \phi_1)m_1F_1^s(\tau_1)F_2^s(\tau_2)$ reflects the share of the cannibalized sales that might be transferred to a competing product and this share is $\alpha_s \times (\phi_2 - \phi_1)$. Note that α_s is the share transferred between generations of the firm *s* while $\alpha_s \times (\phi_2 - \phi_1)$ is the share that might be transfer to a competing product. If $\alpha_s = 0$ this implies no cannibalization and we are back to the NBM specification with competition. Finally, all terms belonging to firm *s* interact with the diffusion of the competing product $F_1^c(\tau_c)$ after τ_c . This last term exists only after $t > \tau_c$ and hence firm *s* decision should take into account that after time τ_c their products will gain or lose some share to the competing product. Note that equation (2.21) uses a simplified version of the duopoly model and that in our application below we use the complete duopoly and triopoly model. The following lemmas cover a few interesting optimal timing scenarios. We include them because they illustrate some extreme cases where the *launch now or never* strategy may be valid and they illustrate the flexibility of our model specification.

Lemma 1 The optimal introduction timing of both the first and second generation products is equal to zero when there is no cannibalization ($\alpha_s = 0$), when the $\phi_1 < 0$ and $\phi_2 < 0$ and there is one competitive introduction at τ_c .

From (2.21) it follows that if $\alpha_s = 0$, one has $\Delta_s = -(\phi_1 m_1 F_1^s(\tau_1) + \phi_2 m_2 F_2^s(\tau_2))F_1^c(\tau_c)$. It is clear that both products should be introduced at t = 0 given that they face competition after τ_c , that is, the earlier they are both introduced, the better. Hence, in the case of no cannibalization with competition the option of *launch now* is the optimal solution. If there is no competition and cannibalization we are back to the solutions of the Norton and Bass model. This lemma is in line with Kamien and Schwartz (1972).

Lemma 2 The optimal introduction timing of the first and second generation products (τ_1 and τ_2) are equal to τ_c when there is no cannibalization ($\alpha_s = 0$), when the $\phi_1 > 0$ or the $\phi_2 > 0$, respectively, and when a competitive introduction happens at τ_c .

Introducing at time τ_c produces a positive Δ_s and it is clear that a firm should choose a time closer to τ_c . If both products are launched before τ_c the sales stream is smaller between τ_1 and τ_c for the first generation, and they are smaller between time τ_2 and τ_c for the second generation. On the other hand, if they are launched after τ_c they do not benefit from competition for $\tau_1 - \tau_c$ or $\tau_2 - \tau_c$ periods, respectively. This lemma implies that *imitation* may be optimal under certain conditions. As before, the strategy of *launch never* is discarded because there are positive returns to launch at dates closer to competitors. This lemma may be modified easily to the situation where imitation is optimal for only one generation, for example if $\phi_1 = 0$ and $\phi_2 > 0$. In our application below we will conduct a numerical exercise (in section 2.7.2) where this lemma is at work.

Lemma 3 It is optimal to never launch the second generation when $S_2^s(\tau_1, \tau_2 | \tau_c) + \Delta_s < 0$.

When the returns on introducing the new product Δ_s outweigh the unit sales of $S_2^s(\tau_1, \tau_2 | \tau_c)$ then it is optimal not to introduce it. Hence, the *launch never* strategy arises when there is stiff competition as in Kamien and Schwartz (1972). In our case study (section 2.8.2) we evaluate the parameter space that leads to this lemma.

There are other interesting possibilities of intermediate launch-timings when there is cannibalization and competition either for the first or second generation given different values for the ϕ_1 , ϕ_2 and α parameters. In our case studies we explore numerically other possibilities for the α parameter and the optimal timing of products and explore the parameter space that may lead to any of these lemmas or to the *launch now or never* strategy.

2.4 The Video Game Hardware Market

The hardware market for video games can be split in two sub-markets: hardware for portable systems (PS) and hardware for video game consoles (VGC). In this chapter we treat these markets to be independent of each other. Indeed, most press articles indicate that the markets of PS and VGC are independent. See for example The Herald (2005), Financial Times (2004), The Economist (2004) and The Washington Post (2008). The reader may be familiar with the video game console wars between Microsoft, Sony and Nintendo (BusinessWeek, 2008b; The Washington Post, 2006). At the moment (September 2009) these three companies are the main market players in the hardware market. Microsoft does not sell any PS while the three companies sell competing video game consoles. Sega stopped producing game consoles in 2001 (San Francisco Chronicle, 2001) and Apple and Microsoft are seen as potential new competitors of Sony and Nintendo in the PS market. (BusinessWeek, 2008a; Wall Street Journal, 2006).

2.4.1 Some Basic Figures

In Table 2.1 we report the release dates of the main PS hardware since 1998 for three main markets: North America, Japan and Europe. The release dates for PS seem almost arbitrary and they occur in months that range from February to December for all three regions. However, when we look at the time between releases within companies we discover a different pattern. Table 2.2 shows an average of two-year intervals between releases.

In Table 2.3 we report the release dates on all major VGC since 1987. Clearly, the VGC market is quite different from the PS market. The release dates in North America are mainly chosen to be close to November while in Japan and Europe most releases occur also in other months of the last quarter of the year. If we look at Table 2.4 we can see that there is an additional regularity around the VGC releases. They occur approximately every five years. Only the Sony PS3 took more than 6 years to be released and this was due to a delay in the development of the blu-ray technology added to the PS3. See The New York Times (2006) for more details on this story.

In Table 2.5 and Table 2.6 we report the estimates of single-generation Bass models for PS and VGC. Portable systems have very similar innovation parameters (p) but quite different imitation parameters (q). We computed simple statistics on the Bass models and in most cases they fit the data quite well. We discuss more details on our data next.

2.4.2 Data and Data Cleaning

Our data for the duopoly and triopoly NBM models consists of weekly time series of sales at the USA for the last two PS of Nintendo and Sony and the last two generations of consoles released by Microsoft, Sony and Nintendo. The portable systems are the Nintendo DS, the Nintendo DS Lite, the Sony PlayStation Portable (PSP) and the Sony PSP Slim. The video game consoles are the Microsoft Xbox, Microsoft Xbox 360, Sony PS2, Sony PS3, Nintendo GameCube and Nintendo Wii. In addition, we obtained the corresponding release dates for all products from different news sources and for all cases the release dates matched the date of the first week that we observed in our data. We used a script to download our data from www.vgchartz.com and the site admins authorized us to use their data. Our data for all systems cover the period since their release week up to January 2009. That is, our data covers a period of almost 9 years and 10 systems.

Before we plug our data into the estimation routines we control for indirect network effects, seasonality and price. It has been documented that indirect network effects might play a role in the video game market (see for example, Chintagunta et al. (2009), Clements and Ohashi (2005) or Shankar and Bayus (2003)). Furthermore, Binken and Stremersch (2009) show that it is mainly *super star software* what drives indirect network effects in the video game systems market. Therefore, in this chapter we use a simplified version of the model proposed by Binken and Stremersch (2009) to clean our data from indirect network effects and price. We use the following equation

$$Y_t = \alpha Y_{t-1} + \sum_{j=1\dots 52} \beta_j W D_j + \sum_{l=t\dots t-L} \lambda_l P C D_l + \sum_{k=t\dots t-K} \delta_k S S I_k + \epsilon_t$$
(2.22)

where Y_t are the system sales at week t; WD_j refers to the week j dummies; PCD_l is the price cut dummies with L total lags and it indicates the week when prices were cut; SSI_k is the total number of *super star software* introduced at week k. To create the independent variables in equation (2.22) we collected release dates and quality ratings on the most popular video games for the systems in our sample. For each system we found approximately 120 video games to construct the SSI variable. In total we collected data for 1200 video games. These data come from many different online sources. Furthermore, we use many different news services to find the price cut timing for all consoles in our sample. We estimated equation (2.22) for each console in our sample and then we subtracted the terms $\sum_{k=t...t-K} \delta_k SSI_k$ and $\sum_{l=t...t-L} \lambda_l PCD_l$ from the consoles sales Y_t only if they are significant. We report in Table 2.7 the sales percentage that indirect network effects represent for each console and the number of lags for the SSI_k variable that we used. We chose the number of lags in the same way as Binken and Stremersch (2009).

Interestingly, despite our model is a much simpler version of that of Binken and Stremersch (2009) we find that indirect network effects represent on average a 13% of the consoles sales while Binken and Stremersch (2009) found that percentage to be 14%. That is, our results confirm their findings. In contrast, we use weekly data, they use monthly, and we find that on average the number of lags correspond to approximately 7 weeks (that is less than 2 months) while they report significant lags up to 5 months. In terms of weeks 5 months represent 20 weeks. We tested lag numbers up to 20 weeks but we did not find significant effects further than 14 weeks (see Table 2.7). Note that the number of lags in the Table should be read with caution because not all lags were found significant and as Binken and Stremersch (2009) we include the last non significant lag to avoid bias. An additional difference is that we estimate the equation

(2.22) separately for each system while they use a panel approach and that their SSI variable is monthly while we trace software introduction per week. Our guess is that they use a panel approach because they consider much shorter time series and the panel approach helped them to identify their model parameters. However, they warn about considerable heterogeneity of the network effects and their result of 14% is therefore close to an average of network effects across systems. Our long time series of weekly data allows us estimate the model for each system and the fit we achieve is very good for all systems (R^2 close to 0.80). A final difference in our approach is that we use the 120 most popular video games per system while they use on average the 10 superstar software video games per system. We estimated a second version of the system models by including only the highly rated video games (the superstars), as do Binken and Stremersch (2009), in the SSI variable. Binken and Stremersch (2009) do not report the percentile they use as a selection heuristic and we selected the video games with a quality rating in the top 25 percentile. In this case, the average network effects jumps up to 15%, while it is also close to their reported number. That is, higher quality video games might have higher network effects although the difference between 13% and 15% can hardly be considered as significant.

The resulting adjusted series without network and price effects still needs to be cleaned from seasonality and for this latter purpose we use the *TRAMO/SEATS* methodology (Gomez and Maravall, 2001, chap. 8). We further control for all major holidays in the USA and for Easter.

In sum, the series we plug in our estimation routine are the seasonally adjusted series without indirect network and price effects. We use this series because the competitive parameters on our model could pick up the correlation caused by indirect network effects, price and seasonality if we do not control for them.

Our data covers 10 gaming systems and therefore we estimated 20 models (10 for the network effects and 10 for the seasonal adjustment). We do not report these results but they are available from the authors upon request. In addition, we estimate both the duopoly and triopoly models with the original data and the parameter estimates remain very similar. However, the fit is better when we use the clean data.

2.5 Estimation and Parameter Assumptions

We use the systems NLS estimator described in Cameron and Trivedi (2005, Chap. 6, page 217) to estimate the parameters.

The duopoly multi-generation model consists of 26 parameters and in our estimation routine we use 16 free parameters. This number reflects the assumptions that the innovation and imitation coefficient, p and q, vary across firms and products and that the loyalty effects are symmetric. That is, we assume that ϕ_{gk}^{ij} is equal to $-\phi_{kg}^{ji}$. The τ_g^i parameters are the introduction date of each product and we keep the real launch dates in our estimation routine.

The triopoly model consists of 45 parameters and in the estimation routine we have 21 free parameters. This number reflects the assumptions that the p and q parameters vary across firms, that the loyalty parameters are symmetric, and that α_i for i = x, s, n are fixed at some value. The main reduction comes from the assumption that $\phi_{gk}^{ij} = -\phi_{kg}^{ji}$ as it reduces the number of free parameters by 12. Note this is the symmetry assumption we described earlier when we discussed Figure 2.1. Finally, we use the real introduction dates as values for the τ_g^i (g = 1, 2and i = x, s, n) parameters.

An important assumption in the estimation routine is the value of the α parameters and we need an assumption on them. As we mentioned earlier, the α parameter is simply the share of the sales that the first generation transfers to the second generation. The reason why we need to make an assumption regarding α is that there is a direct relationship between the α and the m parameters with the realized cumulative sales. We know that the realized cumulative market sales are fixed at some value, call it M, and it depends on both α and m. Of course, the realized M depends on all other parameters but specially the α and the m are very closely related to it. If we increase α then we need a lower m to keep the realized sales at M or if we lower α we need a higher m. This means that we can not simultaneously identify both parameters. This is a limitation and at the same time an advantage of our model because we can obtain the α parameter easily from experts opinions, managers, store sales data, or surveys. All we need to know is what percentage of the first generation sales (of an specific firm) is transferred to its second generation and that is α . However, in case the α is not available then we could make assumptions on the market potentials and estimate the α together with all other free parameters in the model. We know that market potential assumptions are quite common in the new products diffusion literature and they are straightforward to construct.

In the estimation routine first we assume the $\alpha = 1$ for all firms in both the duopoly and triopoly model. Then, as an illustration, we ask an expert opinion on the size of α for each firm in our triopoly model. We contacted a local store manager and asked him about the α parameter of Microsoft, Sony and Nintendo according to his experience. His information is that the α of Microsoft is 0.3, the α of Sony is 0.1 and the α of Nintendo is 1.1. These numbers imply that Nintendo is able to get 1.1 sold unit of Wii for each sold unit of the GameCube, Sony achieves the lowest with a 0.1 of PS2 unit sales going into the PS3, while Microsoft is in between with an α of 0.3.

To estimate both models we use the systems NLS estimator but due to the large number of parameters we split estimation in three steps. First we estimate the six innovation and imitation coefficients p and q given all other parameters fixed. Next we estimate the loyalty coefficients ϕ given all other parameters are fixed at their most recent estimated values. We iterate these two steps until convergence and at the end of the routine we estimate the six market potentials given all other parameters. Chintagunta et al. (2009) apply a similar estimation approach. In the estimation routine we constrained the ϕ coefficients setting their lower and upper limits at -4 and +4, respectively. However, all parameter estimates are within these limits as we report in Section 3.5. All our routines are programmed in R (R Development Core Team, 2005).

2.6 Estimation Results

We report the parameter estimates for the duopoly model in Table 2.8. In this model we consider two companies, Nintendo and Sony, and their portable gaming systems. The systems are the Nintendo DS and DS Lite and the Sony PlayStation Portable (PSP) and PSP Slim. We notice that the parameter estimates for the innovation and imitation parameters, p and q, are lower in the multi-generation model than in the independent Bass model reported in Table 2.5. In addition, the market potentials are remarkably lower in the multi-generation model. Two factors explain the lower estimates. First, the multi-generation model allows the first generation

to transfer a percentage α of its sales to the second generation. Hence, the second generation market potential has a lower *m* estimate but note that the realized market potential in the multi-generation model may be higher than the *m* estimate after adding the competition and cannibalization effects. These results are in line with the findings of Norton and Bass (1987) regarding the size of the market potentials of the second generation products; see (Norton and Bass, 1987, footnote 2, page 1074). Finally, we find significant ϕ parameters and this is evidence supporting the idea that the portable systems compete against each other. For example, we see that the Nintendo DS is losing share to the Sony PSP (see the -0.57 estimate) and it is losing more to the second generation of Sony, the PSP Slim (see the -2.39 estimate). On the other hand, the Nintendo DS Lite is receiving a share from the PSP Slim (see the 0.66 estimate). We report the model fit in Figure 2.3 and we can see the fit is reasonably good.

In Table 2.9 we report the triopoly model parameter estimates with the assumption that all firm's $\alpha = 1$. In Table 2.10 we present the parameter estimates when we use 0.3, 0.1 and 1.1 as the α parameters for Microsoft, Sony and Nintendo, respectively. Finally, in Table 2.11 we present the ϕ and α parameters reported in Table 2.9 in a easy to read format.

For the triopoly case it is the q parameter estimates that are much lower than in the Bass model reported in Table 2.6 while the p parameters remain very similar. An interesting result is that the Microsoft Xbox market potential is around 19 million units while the Xbox 360 market potential is a much lower value of 813 thousand units. A similar drop in market potential occurs from the Sony PS2 to the Sony PS3. The exception is Nintendo. The market potentials for both the Nintendo GameCube and the Nintendo Wii stay around the same level (17 million units). This finding is in line with the results of Shankar and Bayus (2003). Shankar and Bayus (2003) analyze the video game market between 1993 and 1995 and the two main players at that time where Nintendo and Sega. Note that in Table 2.3 we report the history of console releases since 1985 and that they analyzed the last three years of the 4th generation systems. They argue that Nintendo had a higher network strength than Sega and consequently Nintendo sales overtook those of Sega. Recently, the Nintendo Wii is overtaking the sales of the largest player, Sony, and our parameter estimates seem to capture this overtake. In Table 2.10 we report the model with our expert's values on the α parameters. As we anticipated, the parameter estimates of the market potential m are higher for the second generation of Microsoft and Sony because we assumed a much lower α for them (0.1 and 0.3, respectively). The market potential for the Xbox360 goes from 813 thousand units in the first model up to 2,685 thousand units in the second, that is 3.3 times higher. The PS3 m in the second model is 2.14 times higher than in the first. Finally, the market potential of Nintendo's second generation, the Wii, is 1.161 million units lower in the second model relative to the first because of the higher α . Surprisingly, the market potential of both generations of Nintendo are still high relative to each other despite the fact that Nintendo can transfer more consumers from the GameCube to the Wii (it has the highest α among the three companies). The rest of the parameters in Table 2.10, with very few exceptions, remain very close to the model parameters of Table 2.9. We are certain that there are other ways to retrieve the α parameters from experts, surveys or data we stress that this estimation exercise is just an illustration.

In Table 2.11 we arrange the ϕ and α parameters in two six by six tables. We numbered the estimated ϕ parameters in the top table and in the bottom we report their estimates using bold face for parameters with t-values higher than 1. We can see that the Wii is getting some share from the Xbox console (see the 2.39 parameter of the phi[4]) and that is not competing against the PS3 (see the -0.02 of the phi[12]). This confirms what has been argued in the press that these two consoles are not substitutes for each other. A surprising result is that the Wii has a positive influence on PS2 (see the -0.60 estimate of the phi[8]). The PS3 is losing some share to the Xbox 360 and the GameCube (see the phi[6] and phi[11] estimates) according to the sign of the parameters but they are not significant. At the same time, the PS2 received share from the Xbox 360 and the GameCube. Most parameter estimates are in line with our anecdotal evidence and what we read in the press.

Finally, we plot the observed and fitted values of the triopoly model in Figure 2.4 and again the model fits the data reasonably well. Note that the real cumulative sales of the first generation products, the graphs in the left of Figure 2.4, stabilize after they reach their maximum. However, our model forecasts a decline in their number of cumulative units after reaching the maximum and this is a consequence of the substitution that takes place after new generations are introduced. Hence, the fit after the maximum is not really the same as the fit before the maximum of the cumulative sales given that we do not have data on substitution or *un-adoption* of these products. An interesting feature of the left-hand graphs is that the foreseen decline is faster for the Xbox and the GameCube while it is very slow for the PS2.

2.7 Duopoly Case Study: The Portable System Race

In this section we use our model to analyze the portable system market. We take the duopoly model and its parameter estimates and with them we simulate four different strategies for both Nintendo and Sony. We use a planning horizon $T_p = 90$ months and this number is long enough relative to the average pace of two years we report in Table 2.2. Next we describe the strategies we simulate and afterwards we present the insights gained by our numerical exercises. At the end of the section we present the sensitivity analysis to different parameter estimates.

2.7.1 Simulating Plausible Strategies

A strategy is a complete contingent plan for all market players. (Watson, 2002, pg. 26). That is, we define the actions of Nintendo as a response to any of Sony's actions and viceversa. In all of the strategies, except the first, we let Nintendo be the *leader* and Sony the *follower*. We reversed their roles in our numerical exercises and our insights remain without significant changes. Furthermore, the *leader-follower* assumption is common in the literature, see for example Bayus et al. (1997, p. 56). Finally, we assume that the order of entry does not modify the competitive relationship between products, just as in Kamien and Schwartz (1972), but note that we will provide sensitivity analysis to different parameter values in the next section.

The four strategies we consider are:

 Random Date Selection: In this strategy both Nintendo and Sony randomly select a launch date for their two product generations at the beginning of the planning horizon. That is, both firms ignore each other's actions and the interaction among their competing products.

- 2. Imitation: In this strategy Nintendo selects the launch-timing for its two generation products and Sony imitates Nintendo. That is, Sony launches its PS2 console at the same time as the GameCube and it launches the PS3 at the same time as the Nintendo Wii.
- 3. **Pre-commitment and Optimization**: In this strategy Nintendo pre-commits to the launch date of their two generation products while Sony, with perfect foresight, optimizes the launch dates of its two generation products based on Nintendo pre-commitment dates.
- 4. Uncertain Dates and Stochastic Optimization: In this strategy Nintendo does not pre-commit to a launch date for its two generation products. However, Sony assigns a probability to each of the possible launch-timings of the GameCube and the Wii and based on this information it optimizes the launch-timing of the PS2 and the PS3.

We give the details of each strategy in the Appendix 2.A. We simulate these four strategies and we compute the outcome in terms of the maximum cumulative sales of Sony, Nintendo and the sum of both firms' maximum cumulative sales. We repeat the simulation of each strategy until we cover all the combinations possible of the launch-timing selected by Sony and Nintendo that each strategy implies. In this way we recover the distribution of the sales that both players may achieve by following each of the four strategies. We summarize these distributions in Table 2.12 and Figure 2.5.

In Table 2.12 we report six quantiles of the distribution of the sales for Sony, Nintendo and their sum and for each of the four strategies while in Figure 2.5 we plot their percentiles. The purpose of Table 2.12 and Figure 2.5 is to help us rank the strategies in terms of the likelihood of their sales outcomes. For example, in Table 2.12 we see that for Sony the sales achieved by imitating are lower than the sales achieved by randomly selecting its dates, see the second and fourth lines in the table.

In the right-hand side of Figure 2.5 we see that the strategy that results in higher sales for Sony is the third and that is the strategy in which Sony knows the exact launch dates of Nintendo's products. Only at the very first percentiles (from 0 to around 20%) the stochastic optimization strategy is better. In the graph it is clear that the second best strategy results when Sony applies stochastic optimization. As we can notice, this strategy puts a lower and upper limit to the sales of Sony, see the flat areas of the *uncertain dates* line at the first and last percentiles. Surprisingly, imitation is the worst strategy Sony could follow and it performs slightly worse than when Sony randomly selects its dates.

In the left-hand side of Figure 2.5 we see the quantiles of the distribution of sales achieved by Nintendo. Note that Nintendo is the leader and the outcomes are therefore not a mirror of the results obtained by Sony. For Nintendo the results are mixed. We see that before the percentile 50 the best outcome is achieved when Sony is imitating (interestingly this is not a good option for Sony) and that after the 50 percentile the best outcome is achieved by not announcing its launch dates and by not precommiting to them (see the *uncertain dates* line). On the other hand, before the 50 percentile the lowest sales are achieved when Sony uses stochastic optimization and above the 50% the lowest sales are either random selection of dates or pre-commitment. Note that Nintendo does not behave strategically in our simulations. That is, Nintendo does not know that Sony is following one of the four strategies. Given that Nintendo knows which strategy Sony is playing then it is straightforward for Nintendo to strategically select its launch dates and achieve high sales. This implies that if Nintendo strategically chooses its launch dates then playing the *uncertain dates* strategy can result in high sales while if Nintendo acts not strategically then pre-commitment is a reasonable strategy. Of course, we are not using very strict criteria to rank Nintendo's strategies but it is straightforward to rank the strategies using different criteria given we know their corresponding outcomes in terms of sales distributions.

2.7.2 Sensitivity Analysis of the Launch Strategies

The above results are sensitive to the parameter values we plug in the duopoly model. In all previous exercises we used the values we obtained from our estimation routine. To know how the sales outcome may change we compute the expected value of the sales achieved by playing the second strategy (imitation) and the third strategy (optimization) when we plug in a different set of parameter values in the model. First we evaluate the strategy by simulating different combinations for the phi[1] and phi[2] parameters, the phi[3] and phi[4] parameters and finally for the phi[1] and the alpha parameter of Nintendo. The phi[1] and phi[2] are the ϕ parameters between the Nintendo DS and the PSP and the PSP Slim, respectively. The phi[3] and phi[4] are the ϕ parameters between the Nintendo DS Lite and the PSP and the PSP Slim, respectively.

In Figure 2.6 we report the log of the ratio of the expected sales of Nintendo and Sony given all possible combinations of these parameters, take two at a time, for the imitation and pre-commitment strategies. In the ratio Nintendo's expected sales is the numerator. This is a numerical intensive exercise in the sense that for each parameter combination we compute all possible combinations of launch-timings implied by each strategy and based on the outcome (in terms of their maximum cumulative sales) we compute the expected value for the sales of both players. In the graphs we report the log of the ratio of the expected maximum cumulative sales between the two firms. Note that we apply the log transformation to the final values because the log of the expected value is not the same as the expected value of the logs.

The graphs in Figure 2.6 provide a unifying message. Both strategies might yield high sales if a firm's products are superior (in terms of the ϕ) parameters or if a firm's ability to transfer users of old technologies to new ones is high (that is equivalent to a high alpha). If both ϕ parameters tend to be positive the ratio goes up and therefore Nintendo sells more relative to Sony. The ratio increases in a similar way when the alpha of Nintendo is higher. Earlier we concluded that the imitation strategy is the worst among the four strategies we evaluated for Sony. However, if Sony had superior products the imitation strategy may yield high sales, see how the log ratio goes up to -3 and -2 in the left-most and center upper panel graphs. This is evidence supporting Lemma 2. However, we can easily notice that despite the unifying message the surfaces have different slopes. That is, achieving higher sales by raising or decreasing each of the ϕ parameters does not yield the same increase/decrease in expected sales. We conducted the same sensitivity analysis for the random dates and the stochastic optimization strategies and the results are very similar.

The main lesson of this sensitivity analysis is that the outcome of any launch-timing strategy varies radically and it depends heavily on the competitive positioning of the firms' products and on the firms' ability to transfer users of their old technologies to the new ones.

2.8 Triopoly Case Study: The Video Game Console Race

In this section we present a different set of numerical sensitivity analyses and we will focus on the launch-timing of the Sony PlayStation 3 and the Nintendo Wii relative to their previous generations and relative to their competitors. In this section we focus on answering *what if* questions rather than studying the strategic interaction of firms, like in the previous subsection. We use the parameters estimates we obtained from our estimation routine to answer the *what if* questions and we assume a planning horizon $T_p = 150$ months. That is we assume 12.5 years as planning horizon and this is in line with a recent interview statement of the President of Sony Computer Entertainment in America, see Fast Company Blog (2009). In addition, we illustrate the sensitivity of the optimal launch-timing to different competitive and cannibalization parameters.

In Table 2.3 we reported the release dates of all major video game systems. It easy to notice that historically the phenomena of *a launch race* in a single year is relatively a recent experience for system manufacturers. This is interesting given that the number of systems manufacturers has stayed relatively constant since the early nineties. We observed for example that the Nintendo Wii was launched at the same time as the PlayStation 3 in North America three years ago. The GameCube and the Xbox were launched simultaneously in 2001. The other close to simultaneous launch cases occurred between the Wii and PS3 in Japan and between the Xbox and GameCube in Europe in 2006 and 2002, respectively. The average timing between releases is approximately five years (5.09 years), and the standard deviation of this average is almost one year (0.90 years), see Table 2.4. Hence, we believe that there is a need for insights about whether these launch-timing were chosen optimally or what could make them optimal.

The optimization situations that we consider next are much simpler than the optimization situations that we encounter in practice. They are simpler because of mainly two reasons. First, we do not consider the strategic interaction between firms as in previous section. Second, we do not consider price as part of the optimization problem because we focus on analyzing the launch-timing decision relative to different cannibalization and competitive settings. However, the timing decision can be considered as a sub-game of the price and timing game. That is, our analysis has no assumption regarding the price of the consoles and we focus on the effects of timing dates on the unit sales of the systems. This assumption is in line with similar studies to ours, see for example Joshi et al. (2009) and the work cited by (Souza et al., 2004, p. 538) regarding pricing assumptions. However, we do not consider this a very strong assumption in terms of our model estimation because of the cleaning procedure of our data. Nonetheless, if we had a reasonable assumption about the price for all six systems in our triopoly or duopoly model, and how the prices of all systems are strategically related to each other, then it is straightforward to introduce it in the optimization problem. Still, our results will be valid as price would possibly work as a discounting factor in the optimization problem. Of course, the effect of price on demand is not a straightforward introduction into our diffusion model and we consider this an area of further research.

2.8.1 Simulating What If Questions

The first *what if* question we answer is: What would be the maximum cumulative sales of the Nintendo and Sony if they would have launched their consoles at different dates and leaving everything else constant? That is, we answer how either the sum of the maximum of equation (2.12) and (2.13) for Sony and the sum of the maximum of equation (2.14) and (2.15) for Nintendo are maximized. In Figure 2.7 we plot the total sales of Nintendo (summing up the maximum cumulative sales of the Wii and the GameCube) achieved by launching at different dates. The maximum cumulative sales are reached when the Wii is launched at the month 64 (that is April 2005) and the GameCube at month 1 (January 2000). That is 5.33 years between their releases. The real release time between these two consoles was 5.01 years in North America, 5.22 years in Japan and 4.60 in Europe. The real launch dates happened at November 2006 (month 83 in the graph) and November 2001 (month 23 in the graph). Surprisingly, Nintendo is not launching that far from the optimal dates and according to this surface the difference of sales between real and optimal dates is 3,858.62 thousand units (66,431.66 thousand units at the optimal and 62,573.04 at their real launch dates). The story is different for Sony. In Figure 2.8 the maximum is reached when the PS2 is launched at month 1 (January 2000) and with

the PS3 not launched. Note that setting the month of launch equal to the end of the planning horizon is equivalent to not launching. This is a radical scenario but it is explained by the fact that the PS2 is receiving sales from both the Xbox 360 and the Nintendo Wii according to our model estimates while the PS3 competitive parameters are not very favorable, see Table 2.11. The real launch dates of the PS2 and PS3 are the months 10 (October, 2000) and 84 (December, 2006), respectively. The total sales of Sony at these last pair of dates is 59.988 million units, in Figure 2.8 all the sales surface is graphed for all possible launch dates. We know that up to the first week of August 2009 the PS2 has sold 50.767 million units (source vgchartz.com). Hence according to our model the realized sales of PS3 will be around 9.22 (\pm 2.14) million units while up to date the Sony PS3 has sold 9.018 million units. The 2.14 million units is the average derivative of the surface at the real launch dates, the point (10, 84) in Figure 2.8. Therefore, our model is not very optimistic about the PS3.

The next questions we answer are: what is the optimal launch time of the Nintendo Wii given the launch times of the Sony PS3? and what is the optimal time of the Sony PS3 given the launch times of the Wii? We can answer these questions by looking at Figure 2.9. In this figure we present two contour graphs (or heat maps). The lighter (yellow) areas represent higher total sales and the darker (red) areas represent lower sales. We call these graphs sales reaction surfaces because we can derive the best reaction function of either Nintendo or Sony given each other introduction timings. A reaction function maps any launch-timing of a firm to the best launch-timing of a second firm. We use the same definition of *reaction functions* as in Section 2.7. For example, in the left-hand graph we see that the maximum of Nintendo's sales is on month 73 given Sony launched its PS3 in month 1. From Table 2.11 we know that the PS3 and the Wii are not close competitors and not surprisingly the optimal launch date of the Wii given any introduction date of the PS3 remains close to the month 73 (January 2006) for any introduction timing of the PS3. What is surprising is that Nintendo launched 11 months later than its optimal timing. In the right-hand graph we see that the optimal launch dates of Sony are not very sensitive to those of the Nintendo Wii. For example, if the Wii were launched from month 1 up to the month 60 (that is from January 2000 up to December 2004) then the optimal month for the PS3 remains very close to the month 126 (June 2010). However, if the Wii is launched after the month 80 then the optimal action for Sony is to set the introduction date of the PS3 at month 150, the end of the planning horizon. Hence, the best strategy for Sony if the Wii is launched after month 80, is not to launch the PS3.

2.8.2 Sensitivity Analysis of the Optimal Launch-Timing

In the previous subsection we answered *what if* questions assuming our model parameter values are the ones resulting from the estimation routine. However, the optimal timing is sensitive to the parameter values and in this subsection we present how sensitive it is to different competitive and cannibalization settings.

First we present the sensitivity of the optimal launch date of the Sony PS3 to the competitive parameters that relate this console to the Xbox 360 and the Wii, the phi[6] and phi[12] respectively, for six different scenarios. In each of these scenarios we assume an early, a late, and an intermediate introduction timing of the Xbox 360 and the Wii. That is, we present three scenarios for each last generation console that competes against the PS3. Second, we present the sensitivity of the optimal launch date of the Sony PS3 given different cannibalization and competitive parameters using these same six possible scenarios. We present these results in Figure 2.10 and Figure 2.11 respectively.

In Figure 2.10 we present the scenarios for early (month 40), intermediate (month 84) and late (month 120) introduction timings of the Microsoft Xbox 360 at the upper graphs. In the graphs at the bottom we present the scenarios with the Nintendo Wii launched at the same set of introduction timings. For all six scenarios we leave all other introduction timings and parameters at their real or estimated values, respectively. Note that we only use the ϕ parameters that relate the three systems in our scenarios and set the others at their estimated values.

The first lesson we derive from Figure 2.10 is that the optimal timing of the PS3 depends on how it is competitively related to its two main competitors and not to only one of them. The second insight is that there is a parameter space for which it is better not to launch the PS3 (that is the flat top area in all graphs). Therefore, we can visualize the parameter space where Lemma 3 holds, these are the flat top *don't launch areas* in Figure 2.10. Hence, the *launch never* might be optimal depending on the competitive positioning of the PS3. Similarly, there is a parameter space for which there are earlier optimal introduction timings for the PS3. The third insight is that, the parameter space that is suitable for an earlier introduction gets reduced when the competing consoles are launched at later stages. See how the flat surface (the *don't launch area*) is larger for the center and right graphs relative to the left most graph. The fourth insight is that even when the competitive parameters are very favorable for the Sony PS3, its earliest optimal introduction timing happens at the month 60 (December 2004) and that would imply a 4.16 years difference between the PS2 and the PS3. That is, the *launch now* solution is not part of a very favorable set of parameter values. Note that this time difference between consoles is on the low side of the time between actual releases for all the major video game systems reported in Table 2.4.

This last result may point that the 4 year time between releases could be a good introduction pacing strategy when the product is superior relative to its competitors. Interestingly, the time between releases are in the low side for third and fourth generation consoles and they are in the high side for the the six and seventh generation systems. We do not have data on the earlier systems but our intuition is that the fourth generation consoles were superior to the third generation consoles and they were better positioned relative to its competitors. This may be the case, for example, of the Sega Genesis and the Sega Dreamcast launched 4.33 and 3.25 years after their previous generation, respectively. According to our discussions with some hardcore gamers that seems to have been the case indeed. In contrast, we have read in the press that the relative positioning of the Sony PS3 and the Xbox 360, for example, is not very strong relative to each other and this coincides both with longer time between releases diagnosed by our model and with the longer time between releases we document in Table 2.4 for the latest product generations.

In Figure 2.11 we present the sensitivity analysis of the optimal launch-timing of the PS3 to different cannibalization and competitive parameters, that is concerning the α and ϕ parameters. The upper graphs show the optimal timing of the PS3 for three scenarios of the launch-timing of the Xbox 360, similar as previous graphs. In the bottom graphs we present the scenarios with different introduction timings of the Nintendo Wii. The main difference between this and the previous figure is that one of the axis is now replaced by Sony's *alpha*. In the upper graphs we consider the cannibalization parameter of Sony and the ϕ parameter (phi[6]) that relates the Xbox 360 and the PS3. In the bottom graphs we use the same cannibalization parameter of Sony and the ϕ parameter that relates the Wii and the PS3, the phi[12]. The range we use for the α cannibalization goes from 0 up to 3. A higher number than 1 would imply that Sony is able to get more than one unit sale of the PS3 for each PS2 sold.

The first insight we derive from Figure 2.11 is that the optimal introduction timing of the PS3 depends on both the relative positioning to its competitors and to the cannibalization between Sony's generations. The second insight is that, as before, there is a parameter space for which it is optimal not to launch the PS3 (the top flat don't launch areas) and this space seems larger when competitors launch their consoles at late introduction dates. The third new insight is that the larger Sony's α is, the sooner it is optimal to introduce the PS3. If there is little cannibalization, for example for α values between 0 and 0.5, then it is optimal for Sony to set the launch-timing of the PS3 closer to the end of the planning horizon. For example, in the leftmost bottom graph the optimal timing for a low α values ranges between the month 100 (April 2008) and 129 (January 2010), when the phi[12] value is equal to 2. However, if the α value is larger (near 3 in the same graph) the optimal timing stabilizes at 81 (September 2006). The middle bottom graph corresponds to the scenario that considers the real introduction date for the Wii and in this graph the optimal timing stabilizes at month 65 (May 2005) when both the phi[12] and the α parameter are very favorable to Sony. The optimal timing stabilizes in all graphs around the month 64 (April 2005) and this month implies 4.5 years between releases. Therefore, the *launch now* strategy is not a result of very favorable competitive and *alpha* parameters. The real launch of the PS3 occurred in month 84 and this month is optimal only when the α is much larger than 1 and with a phi[12] approximately near 1. This may indicate that Sony's management might have been very optimistic about the PS3 when they chose that month, at least according to our model.

Finally, the last insight is that when there is no cannibalization the optimal timing of the PS3 is at time 0, that is the *launch now* strategy is covered only as a special case when there is no cannibalization between generations, (Lemma 1). In all the graphs of Figure 2.11 we can see that the *don't launch* area does not reach the $\alpha = 0$ and at this parameter value the optimal

timing drops rapidly to the very start of the planning horizon. See the little empty space between the *don't launch area* and the back wall in all graphs. Visually, it is easier to detect how the surface drops to zero in the upper graphs.

To summarize, the *launch now* strategy results only when there is no cannibalization between a firm's product while the *launch never* strategy results when there are late product introductions by competitors, when a *firm's alpha* is very low, or when the competition is intense in terms of the ϕ parameters. In addition, we find that very favorable competitive and alpha parameters do not imply the *launch now* strategy as we discovered that the optimal launch-timing seems to reach a limit of 4 years between generations. Finally, we find that the higher a firm's ability to transfer its old technologies users to the new ones, the earlier it is optimal for it to introduce new generation products.

2.9 Conclusions and Discussion

In this chapter we presented a new model that is helpful to analyze different launch-timing strategies and optimal introduction timings. It is straightforward to estimate the model parameters and to analyze different interesting competitive and *firms' alpha* scenarios. Our model is suitable to study settings where there are just a few market players or products and when there are some dominant *alpha* technologies in the market.

The insights we gained is that the *launch now or never* strategies may arise depending on the competitive parameters and the relationship between the products in the market. Specifically, the *launch now or never* strategies arise when there are late product introductions by competitors, when a *firm's alpha* is very low, or when competition is intense.

For the first time in the academic literature we provide some insights into the introduction strategy of the main players in the studied industry and we document their introductions since the late eighties. We find that the launch strategy of each 4 years seems appropriate when there is a better product positioning or very high *alphas*. That seemed to be the case at the early stages of the game systems industry while it is not any more so now. According to our model, Nintendo launched the Wii at an appropriate moment while the Sony PS3 perhaps should have never been launched. Moreover, we find that different strategic interactions between firms lead to different sales levels and we argue that the strategy should be chosen relative to the firms' *alpha* and relative to the competitive setting that its products face. For example, the imitation strategy returns are higher for certain competitive parameters, specifically when the product is superior.

The managerial implications are clear. According to our insights the managers in industries with *alpha* technologies should pay not only attention to the competition but also to the ability of their firms to transfer users of old technologies to new ones. In our case study we pointed out that the outlook for PS3 is not very promising, it may reach maximum a 12 million unit sales according to our estimates. However, if Sony's managers work in new ways to increase Sony's *alpha* or its competitive positioning the outlook for the PS3 could improve.

The higher a firm's ability to transfer its old technologies users to the new ones, the earlier it is optimal for it to introduce new generation products. Think of the situation where the first generation product of a firm may face stiff competition after a point in time while its second generation is better equipped to fight against the new entrant. In this scenario, the best and perhaps the only surviving strategy would be to transfer its users of old technologies to the new ones as soon as possible and before competitive entry. We speculate then that the ability to survive in such market depends partially but heavily on the *firm's alphas*.

In our view, the technology markets mimic some of the competitive behavior of the *alpha* chimpanzees. The *alpha* rank for a chimpanzee means access to desirable foods, females or resting places while for companies the *alpha* rank means access to the users of their own old technologies. However, note that in the chapter we assumed non-cooperative behavior between firms while it has been documented that *alpha males* in the chimpanzee society may form temporal alliances to overcome the current dominant *alpha male* (Nishida, 1983). This is a situation we do not study and that we may encounter in the future of the game systems markets. For example, the recent search alliance between Yahoo and Microsoft and the alliance between Toshiba and Sony regarding the blu-ray standard seem to be in line with the cooperative behavior of chimpanzees reported by Nishida (1983). On the other hand, the potential entrance of Apple and Microsoft

in the portable gaming systems market points towards the arrival of more *alpha technologies* and hence perhaps more competition. Finally, we left out other aspects of the marketing mix that may prove important in the timing of new dominant technologies. We consider all these extensions interesting avenues for further research.

2.10 Tables and Figures

Firm	Portable System	North America	Japan	Europe
Nintendo	DS Lite	June 11, 2006	March 2, 2006	June 23, 2006
	DS	November 21, 2004	December 2, 2004	March 11, 2005
	GameBoy Advance SP	February 15, 2003	February 14, 2003	March 28, 2003
	GameBoy Advance	June 11, 2001	March 21, 2001	June 22, 2001
	GameBoy Color	November 19, 1998	October 21, 1998	November 23, 1998
	GameBoy	August 15, 1989	April 21, 1989	1990
Sony	PSP Slim Lite	September 5, 2007	September 13, 2007	September 5, 2007
	PSP	March 24, 2005	December 12, 2004	September 1, 2005
Source: VCch	artz Wikipedia & online press art	icles Notes: We report th	a year of introduction wh	en the exact date is not available

Table 2.1: Release Dates of Portable Systems

Firm	Transition to/from	North America	Japan	Europe
Nintendo	DS - DSLite	1.55	1.25	1.28
	GBA SP - DS	1.77	1.80	1.96
	GBA - GBA SP	1.68	1.90	1.76
	GBC - GBA	2.56	2.42	2.58
	GB - GBC	9.27	9.51	-
Sony	PSP Slim - PSP	2.45	2.75	2.01

Table 2.2: Release Time Between Portable Systems (in Years)

7th generation	Nintendo			oupun	Europe
	1 million and 0	Wii	November 19, 2006	December 2, 2006	December 8, 2006
	Sony	PlayStation 3	November 17, 2006	November 11, 2006	March 23, 2007
	Microsoft	Xbox 360	November 22, 2005	December 10, 2005	December 2, 2005
6th generation	Nintendo	GameCube	November 18, 2001	September 14, 2001	May 3, 2002
	Sony	PlayStation 2	October 26, 2000	March 4, 2000	November 24, 2000
	Microsoft	Xbox	November 15, 2001	February 22, 2002	March 14, 2002
	Sega	Dreamcast	September 9, 1999	November 27, 1998	October 14, 1999
5th generation	Nintendo	N64	September 29, 1996	June 29, 1996	March 1, 1997
	Sony	PlayStation	September 9, 1995	December 3, 1994	September 29, 1995
	Sega	Saturn	May 11, 1995	November 22, 1994	July 8, 1995
	Atari	Jaguar	November 18, 1993	-	-
4th generation	Nintendo	Super Nintendo	August 13, 1991	November 21, 1990	April 11, 1992
	Sega	Genesis	September 15, 1989	October 29, 1988	November 30, 1990
3rd generation	Nintendo	Nintendo	October 18, 1985	July 15, 1983	-
-	Sega	Master System	June 15, 1986	1985	1987

Table 2.3: Release Dates of Major Video Game Consoles

Firm	Transition to/from	North America	Japan	Europe
Nintendo	Wii - GameCube	5.01	5.22	4.60
	GameCube -N64	5.14	5.21	5.18
	N64 - SNES	5.13	5.61	4.89
	SNES - Nintendo	5.82	7.36	_
Sony	PS3 - PS2	6.06	6.69	6.33
-	PS2 - PS1	5.13	5.25	5.16
Microsoft	Xbox 360 - Xbox	4.02	3.80	3.72
Sega	Dreamcast - Saturn	4.33	4.02	4.27
-	Saturn - Genesis	5.65	6.07	4.61
	Genesis - Master Sys	3.25	_	-

Table 2.4: Time Between Major VGC Releases (in years).

Video Game Console	m (thousand units)	р	q	Sample
Nintendo DS	6799.3447**	0.0140^{**}	0.1789^{**}	Nov 2004 - June 08
	(1975.3660)	(0.0056)	(0.0543)	
Nintendo DS Lite	27972.9479**	0.0403^{**}	1.9922^{**}	June 2006 - Jan 2009
	(1130.8544)	(0.0146)	(0.2999)	
PSP	9717.9772**	0.0109**	0.1500 * *	Mar 2005 - Sep 2007
	(1525.7210)	(0.0026)	(0.0389)	
PSP Slim Lite	7068.5424**	0.0184^{*}	0.2449*	Sep 2007 - Jan 2009
	(2579.1973)	(0.0091)	(0.1168)	-
Note: standard error in paren	theses: * ** mean that the co	efficient is sign	ificant with 959	% and 99% confidence respectively

Table 2.5: Bass Model Estimates for Portable Systems

Video Game Console	m	р	q	Sample
Xbox	16157.4500 **	0.0058^{**}	0.0993^{**}	Nov 2001 - Aug 2007
	(699.5485)	(0.0009)	(0.0132)	
Xbox 360	16312.2600 **	0.0054^{**}	0.1272^{**}	Nov 2005 - Jan 2009
	(2826.5520)	(0.0012)	(0.0304)	
PlayStation2	47847.1300**	0.0037**	0.0619**	Oct 2000 - Jan 2009
	(4520.1510)	(0.0007)	(0.0149)	
PlayStation3	8190.0120**	0.0075**	0.1789^{**}	Nov 2006 - Jan 2009
	(1173.5730)	(0.0014)	(0.0333)	
GameCube	12716.7600**	0.0058**	0.0959**	Nov 2001 - Apr 2008
	(527.1293)	(0.0009)	(0.0142)	
Wii	23353.9300**	0.0063**	0.1672^{**}	Nov 2006 - Jan 2009
	(4673.3370)	(0.0014)	(0.0340)	
Note: standard error in paren	theses; ** mean that	coefficients are	significant with	n 99% confidence.

Table 2.6: Bass Model Estimates for Video Game Consoles

System	Model Lags	% Network Effects
Nintendo GameCube	4	15.77%
Nintendo Wii	11	23.33%
Sony PlayStation2	_	_
Sony PlayStation3	5	2.28%
Microsoft Xbox	3	3.60%
Microsoft Xbox 360	7	6.69%
Nintendo DS	14	37.24%
Sony PSP	9	2.28%
All Systems	7.57	13.03%

Table 2.7: Video Game Effects on Game Systems

Coefficient	System	Estimate	s.e.	t-value
р	DS	0.01189	(0.0018)	6.56
	DS Lite	0.03243	(0.0040)	8.12
	PSP	0.05556	(0.0106)	5.25
	PSP Slim	0.02843	(0.0076)	3.74
q	DS	0.07897	(0.0406)	1.94
	DS Lite	0.08174	(0.0353)	2.31
	PSP	-0.08717	(0.0918)	-0.95
	PSP Slim	0.12453	(0.0358)	3.48
phis	(1)	-0.57882	(0.6478)	-0.89
	(2)	-2.39516	(1.9112)	-1.25
	(3)	-0.41831	(0.2109)	-1.98
	(4)	0.66457	(0.2196)	3.03
m	DS	15991.8	(530.0378)	30.17
	DS Lite	10980.7	(486.4696)	22.57
	PSP	10498.0	(169.4519)	61.95
	PSP Slim	1012.0	(205.1027)	4.93
Note: phis (1) is the substi	itution coeffic	ient between I	OS and PSP,

phi(2) between DS and PSP Slim, phi(3) between DS Lite and PSP, and phi(4) between DS Lite and PSP Slim; s.e. stands for standard error.

Table 2.8: Multi-Generation Model for Portable Systems

Coefficient	Console	Estimate	s.e.	t-value
р	Microsoft	0.00943	(0.0025)	3.79
	Sony	0.00980	(0.0011)	8.56
	Nintendo	0.01156	(0.0021)	5.63
q	Microsoft	0.06115	(0.0143)	4.27
	Sony	0.03881	(0.0051)	7.55
	Nintendo	0.05381	(0.0144)	3.74
phis	[1]	-0.00512	(0.2087)	-0.02
	[2]	-0.07195	(0.8990)	-0.08
	[3]	0.02003	(0.2270)	0.09
	[4]	-2.39045	(1.0587)	-2.26
	[5]	-0.28843	(0.1261)	-2.29
	[6]	0.31218	(0.8135)	0.38
	[7]	0.62241	(0.2734)	2.28
	[8]	-0.20223	(0.6419)	-0.32
	[9]	0.11293	(0.2238)	0.50
	[10]	0.60116	(0.1576)	3.81
	[11]	-0.42376	(0.5037)	-0.84
	[12]	0.02402	(0.9240)	0.03
m	Xbox	19135.29	(1132.2)	16.90
	Xbox 360	813.10	(1884.7)	0.43
	PS2	41135.91	(848.8)	48.46
	PS3	987.54	(6416.3)	0.15
	GameCube	16382.92	(1194.1)	13.72
	Wii	17385.63	(2029.2)	8.57

Table 2.9: Multi-Generation Model for Video Game Consoles (Microsoft $\alpha=1,$ Sony $\alpha=1,$ Nintendo $\alpha=1)$

Coefficient	Console	Estimate	s.e.	t-value
p	Microsoft	0.01220	(0.0043)	2.82
1	Sonv	0.01142	(0.0021)	5.49
	Nintendo	0.00809	(0.0016)	5.06
q	Microsoft	0.05423	(0.0201)	2.69
-	Sony	0.03216	(0.0078)	4.13
	Nintendo	0.06670	(0.0151)	4.43
phis	[1]	-0.09294	(0.2166)	-0.43
-	[2]	-2.13156	(1.5318)	-1.39
	[3]	0.04654	(0.2377)	0.20
	[4]	-3.08154	(1.6084)	-1.92
	[5]	0.13089	(0.1357)	0.96
	[6]	0.40535	(0.5606)	0.72
	7	0.90676	(0.2681)	3.38
	[8]	0.37551	(0.5191)	0.72
	[9]	0.09675	(0.2246)	0.43
	[10]	0.43943	(0.2651)	1.66
	[11]	0.36342	(0.7003)	0.52
	[12]	0.15694	(1.0118)	0.16
m	Xbox	18908.97	(1346.3)	14.05
	Xbox 360	2685.35	(874.9)	3.07
	PS2	40409.41	(1002.8)	40.30
	PS3	2117.23	(1142.5)	1.85
	GameCube	17665.16	(1570.8)	11.25
	Wii	16224.15	(2706.9)	5.99
Notes: m is in	n thousand units			

Table 2.10: Multi-Generation Model for Video Game Consoles (Microsoft $\alpha=0.3,$ Sony $\alpha=0.1,$ Nintendo $\alpha=1.1)$

	Xbox	X360	PS2	PS3	\mathbf{GC}	Wii
Xbox	х	-1	[1]	[2]	[3]	[4]
X360	1	x	[5]	[6]	[7]	[8]
PS2	-[1]	-[5]	x	-1	[9]	[10]
PS3	-[2]	-[6]	1	x	[11]	[12]
\mathbf{GC}	-[3]	-[7]	-[9]	-[11]	x	-1
Wii	-[4]	-[8]	-[10]	-[12]	1	x
Xbox	х	-1	-0.01	-0.07	0.02	-2.39
X360	1	x	-0.29	0.31	0.62	-0.20
PS2	0.01	0.29	x	-1	0.11	0.60
PS3	0.07	-0.31	1	x	-0.42	0.02
\mathbf{GC}	-0.02	-0.62	-0.11	0.42	x	-1
Wii	2.39	0.20	-0.60	-0.02	1	x
Notes: 1	the numb	ers betwe	en brack	ets repre	esent the	phi coef-
ficients	of the m	ulti-gener	ation me	del repo	orted in	table 2.9.

The bold coefficients have t-values greater than 1.

Table 2.11: Competitive Parameters

Strategy	Player / Quantile	1%	2	10%	20%	%06	95%	%66
[1] Random Selection of Dates	Nintendo	1681.4	6538.5	10933.5	26369.6	39146.8	42941.3	48084.3
:	Sony	1550.5	2528.4	3890.3	10398.8	19962.5	23481.8	29284.3
	Total	13833.1	21187.1	26621.7	45590.7	57653.5	59676.6	62545.3
[2] Pre-commitment / Imitation	Nintendo	3769.5	9998.8	13288.1	28553.4	37470.9	39509.4	41273.4
	Sony	724.0	1431.3	2383.7	9415.3	18372.8	19513.3	20362.9
	Total	7324.8	15961.7	19174.6	38146.2	54316.4	57663.5	60415.2
[3] Pre-commitment / Optimization*	Nintendo	1543.9	4696.4	8540.6	28837.4	35805.5	36944.6	38842.0
	Sony	13890.7	15387.0	16973.4	21908.2	31948.0	33108.3	34149.6
	Total	17887.0	25253.3	30900.2	52132.8	60307.6	62785.7	65845.1
[5] Uncertain Launch Dates / Optimization**	Nintendo	1548.9	3834.1	6769.8	28615.2	36868.9	38215.9	39742.1
	Sony	13285.2	13873.2	14441.2	20579.3	30961.4	31846.6	32695.6
	Total	17335.7	24765.2	30434.0	48845.5	59968.6	62956.7	66081.8
Notes: * Nintendo pre-commits to a date while Son	y optimizes its launch d	lates given	Nintendo	pre-comm	itment dat	es.** Ninte	endo does	not
DIE-COMMINS to any date and SOMY OPPHILIZE SIVEN	DLODGDIIIN PIIGN INTINCIN	uo rauncii	an and can	1.				

Table 2.12: Evaluation of Four Launch Strategies



Figure 2.1: Interaction Between All Product Generations in Duopoly Model






Figure 2.3: Multi-Generation Model Fit for Portable Systems



Figure 2.4: Multi-Generation Model Fit for Video Game Consoles



Figure 2.5: Cumulative Distribution Function of Sales given Different Strategies







Figure 2.7: What if Scenarios for the Consoles of Nintendo



Figure 2.8: What If Scenarios for the Consoles of Sony









2.A Strategy Simulation Methodology

In this section we provide the details of each of the four strategies we use in our application.

Random Date Selection

In this strategy firm 1 selects τ_1^1 and τ_2^1 randomly and firm 2 selects τ_1^2 and τ_2^2 in the same way. We discretize the planning horizon in T_p periods. Hence the possible launch dates τ_i^g (where i stands for firm i while g stands for the system generation) might be at any t within $t = 1, \ldots, T_p$; we denote the length of the planning horizon p. We consider a 90 month planning horizon that is 7.5 years. This time frame is long enough given the average life-cycle of the portable systems is 2.5 years. With this planning horizon we evaluate the maximum of the cumulative sales for each system of both firms given all the feasible launch dates τ_i^g for i = 1, 2 and g = 1, 2. That is, firm 1 might select one out of the p^2 possible launch-timings but we restrict the combinations to the set where $\tau_i^2 \ge \tau_i^1$. This means that we restrict that the second generation product for both firms is launched at a date either at the same time or after the first generation. The feasible set reduces from p^2 to $(p+1) \times p/2$ feasible combinations for each player. Note that we use $T_p = 90$ and we set p = 45. In the duopoly case i = 1 refers to Nintendo and i = 2 refers to Sony.

We evaluate equations (2.3) to (2.6) with the feasible set of launch-timing and we compute the maximum cumulative sales achieved by each product generation for both firms. That is we compute $\max(S_g^{Sony}(\tau_1^{Sony}, \tau_2^{Sony} | \tau_1^{Nin}, \tau_2^{Nin}))$ for g = 1, 2 and $\max(S_g^{Nin}(\tau_1^{Nin}, \tau_2^{Nin} | \tau_1^{Sony}, \tau_2^{Sony}))$ for g = 1, 2. The g = 1 product of Sony is the Sony PSP and the g = 2 product of Sony is PSP Slim; for Nintendo the g = 1 product is the DS and the g = 2 product is the DS Lite. In table 2.12 we report the quantiles of the total sales of Sony achieved by this strategy, that is $\sum_g \max(S_g^{Sony}(\tau_1^{Sony}, \tau_2^{Sony} | \tau_1^{Nin}, \tau_2^{Nin}))$ and $\sum_g \max(S_g^{Nin}(\tau_1^{Nin}, \tau_2^{Nin} | \tau_1^{Sony}, \tau_2^{Sony}))$ and the total sales of both players (the sum of the last two terms).

Imitation

In this strategy firm 1 pre-commits to a launch-timing for its two product generations while firm 2 imitates the launch-timing of firm 1. That is, both firms launch at the same time each of their product generations. In our application we set Nintendo the be the firm that pre-commits to a certain launch date and Sony to be the firm that imitates. We assume that Nintendo pre-commits to a randomly chosen pair of dates τ_1^{Nin} and τ_2^{Nin} and Sony sets $\tau_1^{Sony} = \tau_1^{Nin}$ and $\tau_2^{Sony} = \tau_2^{Nin}$. In this strategy we assume Nintendo ignores that Sony will imitate and we do not assume Nintendo might pre-commit strategically to the best pair of dates. However, it is straightforward to identify the best pre-commitment dates of Nintendo given Sony is imitating.

Pre-commitment and Optimization

In this strategy firm 1 pre-commits to a launch-timing for its two product generations while firm 2 optimizes its launch-timings given the launch dates of firm 1. As before, we set Nintendo the be the firm that pre-commits to a certain launch date and Sony to be the firm that optimizes. We assume that Nintendo pre-commits to a randomly chosen pair of dates τ_1^{Nin} and τ_2^{Nin} and Sony sets τ_1^{Sony} and τ_2^{Sony} such that $\sum_g \max(S_g^{Sony}(\tau_1^{Sony}, \tau_2^{Sony} | \tau_1^{Nin}, \tau_2^{Nin}))$ is maximized. In this strategy we assume Nintendo ignores that Sony will optimize and we do not assume Nintendo might pre-commit strategically to the best pair of dates. However, it is straightforward to identify the equilibrium if both firms are optimizing. Finally, we note that pre-commitment and perfect foresight are usual assumptions in the literature, for examples see Reinganum (1981) and Bayus et al. (1997).

Uncertain Launch Dates and Stochastic Optimization

In this strategy firm 1 selects a pair of launch dates for its two generation products but does not reveal these dates to firm 2. However, we assume firm 2 can derive the best response of firm 1 given any pair of dates assigned by firm 2 to its own products. That is, firm 2 has knowledge on the reaction function of firm 1 however firm 2 does not know which date launch will be picked for certain by firm 1. The reaction function is a function that maps any launch-timing of firm 2 to the best launch-timing of firm 1. In our application, the best reaction function of Nintendo

$$f(\tau_1^{Sony}, \tau_2^{Sony}) = \Omega\left(\max_{\tau_1^{Nin}, \tau_2^{Nin}} (\sum_g S_g^{Nin}(\tau_1^{Nin}, \tau_2^{Nin} | \tau_1^{Sony}, \tau_2^{Sony}))\right)$$
(2.23)

 $\Omega()$ returns a pair of dates τ_1^{Nin} and τ_2^{Nin} that maximize the sales of Nintendo given the launch dates of Sony (τ_1^{Sony} and τ_2^{Sony}). We further assume that firm 2 assigns a probability that firm 1 will launch on dates τ_1^1 and τ_2^1 proportional to the sales achieved by selecting these two dates. That is,

$$p(\tau_1^{Nin}, \tau_2^{Nin}) = \frac{\sum_g \max(S_g^{Nin}(\tau_1^{Nin}, \tau_2^{Nin} | \tau_1^{Sony}, \tau_2^{Sony}))}{\sum_{(\tau_1^{Nin}, \tau_2^{Nin}) \in f(\tau_1^{Sony}, \tau_2^{Sony}) \sum_g \max(S_g^{Nin}(\tau_1^{Nin}, \tau_2^{Nin} | \tau_1^{Sony}, \tau_2^{Sony}))}$$
(2.24)

Note that τ_1^1 and τ_2^1 should belong to the set of dates given by the reaction function of firm 1 and that is why they should be contained in the reaction function $f(\tau_1^{Sony}, \tau_2^{Sony})$; otherwise the strategy is not considered. Given these assumptions the strategy of firm 2 is to select the pair of launch dates that maximize its expected sales. The best reaction function of Sony is

$$f(\tau_1^{Nin}, \tau_2^{Nin}) = \Omega\left(\max_{\tau_1^{Sony}, \tau_2^{Sony}} (\sum_g S_g^{Sony}(\tau_1^{Sony}, \tau_2^{Sony} | \tau_1^{Nin}, \tau_2^{Nin}))\right)$$
(2.25)

and hence Sony selects τ_1^{Sony} and τ_2^{Sony} such that

$$p(\tau_1^{Nin}, \tau_2^{Nin}) \times \sum_g \max(S_g^{Sony}(\tau_1^{Sony}, \tau_2^{Sony} | \tau_1^{Nin}, \tau_2^{Nin}))_{(\tau_1^{Sony}, \tau_2^{Sony}) \in f(\tau_1^{Nin}, \tau_2^{Nin})}$$
(2.26)

is maximized.

Chapter 3

The Timing and Speed of New Product Price Landings

Many high-tech products and durable goods exhibit exactly one significant price cut some time after their launch. We call this sudden transition from high to low prices the *price landing*. In this chapter we present a new model that describes two important features of price landings: their timing and their speed.

Prior literature suggests that prices might be driven by sales, product line pricing, competitor's sales or simply by time. We propose a model using mixture components that identifies which of these explanations is the most likely *trigger* of price landings. We define *triggers* as thresholds after which prices are significantly cut. In addition, price landings might differ across products and therefore we model their heterogeneity with a hierarchical structure that depends mainly on firm, product type and seasonal effects.

We estimate our model parameters applying Bayesian methodology and we use a rich dataset containing the sales and prices of 1195 newly released video games (VG's). In contrast with previous literature, we find that competition and time itself are the main triggers of price landings while past sales and product line are less likely triggers. Moreover, we find substantial heterogeneity in the timing and speed of price landing across firms and product types.

3.1 Introduction

"Don't get us wrong – price cuts are a good thing" Wired.com (2007)

It is well known that prices of new products exhibit one or several important price cuts during their life-cycle. Nowadays, we are witnessing how many new high-technology products are introduced at an initial high price and after a certain moment their prices are cut to a permanent and much lower level. This practice is commonly followed by manufacturers of products like video games, apparel, PCs, movies, and so on. Moreover, scholars have recognized and studied this type of pricing strategy. For example, studies like Feng and Gallego (1995) and Gupta et al. (2006) point that managers at apparel retailers in New York City report the timing and depth of price cuts are important decision variables and the depth of the price cut in this industry is typically between 25 and 50%. In this chapter, we will call this sudden transition from an initially high price to a lower price level the *price landing*.

We are not aware of any empirical study of price landings. This is quite a surprise because the timing of a permanent price cut for a new product is without a doubt an important managerial decision. During the first half of 2007 thousands of American customers purchased Apple's iPhone and they witnessed a \$200 price drop just 66 days after its release. Consumers were outraged by the sudden price drop and Apple apologized and issued a \$100 store credit to everyone who purchased the iPhone before April 2007. More recently, the forthcoming market launch of Apple's iPad has brought attention to the pricing strategy that the Apple Store will apply to e-books. According to journalists, Apple is pushing the industry to apply "variable pricing which apparently is *triggered* by sales volume and not just pricing whim", see Wired Magazine (2010). In some instances Apple's timing of price cuts have been judged too early if they happened short time before the Christmas season and in other instances the price cuts have been judged as occurring too late to stimulate further sales or to fight competition. See BusinessWeek Online (2007) and BusinessWeek Online (2008) for more details on Apple's story.

In this chapter we present a new model for *price landings* and the estimation approach we present is particularly useful to describe the moment and speed at which the *price landing* occurs and to simultaneously find the *triggers* of these sudden price transitions. Our work offers a complement to studies like those of Tellis et al. (2003) and Golder and Tellis (1997) because we characterize and describe pricing patters of new products while these latter authors have studied and characterized new products' sales patterns. On the other hand, our modeling approach goes further than a description of price patterns because it allows us to find what are the most likely *triggers* of price landings. We apply our model to the market of video games and to a rich data set that concerns 1195 newly released video games.

The plan of the chapter is as follows. In Section 3.2 we present our literature review. In Section 3.3 we present our data and market context. Next in Section 3.4 we present our modeling approach and in Section 3.5 we present our results. We present our conclusions in Section 3.6. All figures and tables are presented in Appendix 3.7. We present the estimation approach in Appendix 3.A.

3.2 Literature Review

In this section we review the studies concerned with the video-game industry in subsection 3.2.1 and next in subsection 3.2.2 we review the literature related to new products pricing.

3.2.1 Research on Video-Games

Three empirical studies closely related to our work are Clements and Ohashi (2005), Nair (2007) and Chintagunta et al. (2009).

Clements and Ohashi (2005) study the indirect network effects between video game consoles and video games and the effects of consoles' prices on their own sales. Their findings suggest that price elasticity is low at the beginning and high at the end of the life cycle of video game consoles. Chintagunta et al. (2009) investigate the effects of software availability and prices on the sales of video game consoles. They propose an econometric approach that accounts for the endogeneity of price and sales and they find time varying price elasticities. In contrast with Clements and Ohashi (2005), Chintagunta et al. (2009) find some evidence of both declining and increasing elasticities. Other studies, like Parker (1992) and Simon (1979), report that elasticities may show diverse time profiles across products, like U or inverted U shapes. See Parker (1992, Table 4, page 365).

Nair (2007) studies the video-game software market and he proposes a model that takes into account the interaction between publishers of video-games and two consumer segments formed by high and low valuation gamers. His findings suggest that the optimal pricing by publishers should exhibit declining prices. The price cut rate (that is the slope of the price function) in Nair (2007) depends on the relative size of each of the consumer segments while the overall and initial level of the optimal price depends on the utility discounting factor and the interaction of consumers and firms.

Our study differs markedly from Clements and Ohashi (2005), Nair (2007) and Chintagunta et al. (2009) because our objective is to introduce a model that is flexible enough to capture many different and detailed theoretical features of prices that have been documented in the literature or observed empirically. In this respect, our price model is a generalization based on previous research. In addition, we offer the first empirical study that focuses on price landings and their triggers, timing and speed.

Finally, the methods of Clements and Ohashi (2005), Chintagunta et al. (2009) and Nair (2007) are considered structural while our model may be classified as a reduced form model. A main advantage of our reduced form is that we do not need assumptions regarding supply and demand side interactions or consumer behavior. A disadvantage of our approach is that we can not draw inferences regarding consumer behavior or consumer-firm interactions and that we need assumptions on the form of the price equation. However, the assumptions we will use for the price equation are more flexible than the assumptions of Nair (2007) and Chintagunta et al. (2009). Nair assumes that consumers form expectations based on an auto-regressive process of order one while Chintagunta and colleagues assume that prices are stationary. In contrast, we present a very flexible equation that can capture sudden breaks (non-stationarity) and it allows us investigate what is *triggering* these breaks. Hence, we offer novel findings and we are the first to measure quantitatively empirical features of prices that have not been documented before in the literature. In addition, our econometric approach is computationally simple. Therefore, we can use our method to study relatively large databases of prices. This may be a technical advantage over structural models that are usually much more computationally demanding.

3.2.2 Research on New Products Pricing

The literature dealing with pricing strategy is extensive and in this section we focus our attention to a set of empirical and analytical studies concerned mainly with new product prices. We present the studies we surveyed in Table 3.1.

In Table 3.1, we see that 24 out of 32 studies are analytical while 8 are empirical. Out of these eight empirical studies only Clements and Ohashi (2005), Chintagunta et al. (2009) and Nair (2007) were published recently and only the study of Nair (2007) is focused on pricing policies for new products. To our knowledge, Nair (2007) and our work are the only empirical studies concerned with price patterns. A likely reason of such lack of empirical studies on prices is the scarcity of detailed price data.

We draw the following generalizations the literature in Table 3.1: 1. Prices show gradual or sudden transition from high to low states. Both empirical and theoretical studies have documented such transitions. 2. Prices show transitions that rarely mimic the S-shape of sales or that increase over time (8 studies). 3. Prices respond to competition, changes in consumer valuations across time, consumer heterogeneity, new product releases, learning curves on costs and market saturation.

The first generalization tell us that prices of new products rarely stay constant. We note that some studies, like Schmalen (1982), Ferguson and Koenigsberg (2007) and Eliashberg and Jeuland (1986), have shown when it is optimal to keep prices of new products constant. On the other hand, we could hardly draw a consensus about how fast price transitions should be or how they look empirically. Some studies explicitly report the optimal price decrease rate, like in Dockner and Gaunersdorfer (1996), Raman and Chatterjee (1995) and Bayus (1994) while many other studies give less attention specifically to the speed of price transitions. Much more is known about the shape of price transitions. Many studies, like Robinson and Lakhani (1975), Kalish (1983), Dolan and Jeuland (1981), Bayus (1992), show that the optimal policy is for prices to decline over time. Other studies show the optimal mark-down (or optimal sudden price discount) based on the length of the season, the perishability of the product or drastic seasonal changes in consumer valuations or demand. See for example Ferguson and Koenigsberg (2007), Gupta et al. (2006), Rajan et al. (1992) and Feng and Gallego (1995). Finally, diffusion studies, like Rao and Bass (1985), confirm that the declining pattern is an empirical regularity and recent studies, following Bass et al. (1994), usually incorporate the declining price effect on diffusion.

The literature suggests the generalization that prices should change (in most cases drop) once an event modifies the market and that these price drops occur in synchrony with the movements of price drivers. These events are usually related to the drivers listed in the last column of Table 3.1. In general terms, previous *empirical* literature suggests that x drives y when x is an important underlying variable causing the variance in y. In contrast, many *analytical* studies integrate trigger variables into their models where x is defined as a trigger of y if it has an effect on y only after a certain threshold, for example after $x > x_o$ becomes true where $x > x_o$ might mean, for example, competitive entry, the end of a season or the limit of market potential. For example, Feng and Gallego (1995) and Gupta et al. (2006) incorporate thresholds after which prices should be marked down. We believe there is a disconnect between analytical studies that allow non-linearities and sudden price breaks and empirical studies that assume in most cases linear price functions without structural breaks.

The objective of this chapter is to fill the literature gaps between empirical and analytical studies of new products' prices. First, our model, together with the econometric approach we use, will allow us describe the theoretical features of prices based on a large database of prices. We focus specifically on the speed and timing of sudden price transitions, what we call *price landings.* Second, we test the relative importance of different price triggers suggested by theoretical and empirical studies simultaneously. We test whether saturation, market entry, time (a products' age) or the release schedule of firms trigger the price landing for each of the 1195 products in our data set. In this way, we put to an empirical test the theoretical properties of prices discussed in analytical studies and we connect both streams of research.

3.3 Video Game Prices

In this section we first describe our data and next we present a brief description of the video game market.

3.3.1 Data

The database we analyze consists of monthly time series of unit sales and prices for 1195 PlayStation2 (PS2) video games released between September 1995 and February 2002 in the US. This data was collected by NPD Group from retailers that account for 65% of the US market. We used the first two years of data for each video game and left out VG's with less than 12 monthly observations. This time frame is justified by the fact that most VG's stay on store shelves for less than two years and their sales drop very rapidly to zero afterwards. Binken and Stremersch (2009) use the same data and they assume that a video game is in a so-called *dead regime* after its sales drop below 5000 units. Therefore, Binken and Stremersch (2009) do not use any observation after this cut-off point which leaves out 32 % of their observations. In our case the 24 month cut-off point leaves out 38 % of the observations. We compared our results against a 30 and a 36 month cut-off point that leave out 28 and 20 % of the observations, respectively, and our results are qualitatively the same. Our final sample consists of 1075 video games.

In Figure 3.1 we show the price landing of 50 randomly selected video games. This figure clearly shows the great diversity of price patterns but it is easy to see of the common feature across games: their price drops at a certain moment in time. The introductory prices range from 40 to approximately 60 USD while their landing level is between 15 and 30 USD. Similarly, there is great diversity in the timing of price landings. It is easy to notice that some VG's prices drop right after the second month while others land around the 10th, 12th or 15th month or even later. Finally we notice that some prices drop very fast, see the lines almost parallel to the vertical axis, while in many other cases they land at slower rates and with more noise around them.

In Figure 3.2 we show the price landing of one of the most popular VG's, the Spider-Man game. We plot the price of the Spider-Man game on the vertical axis but in each of the panels we

use a different scale on the horizontal axis. In the upper-left panel we use time on the horizontal axis, in the upper right panel we use the cumulative sales of Spider-Man and in the lower panel we use the cumulative number of VG's launched to the market after the introduction of Spider-Man. We choose these axes because later we will identify each of these variables as a potential trigger of price landings. More details on this are given in Section 3.4.3. These graphs of course show very similar price patterns. That is, we could say that the price cut of the Spider-Man occurred approximately at the 10th month after its introduction (upper-left panel); or just after reaching 600 thousand unit sales (upper-right panel); or after 250 VG's were launched (lower panel). The price landings in these figures are similar but the interpretation of the different thresholds is very different. In all cases, these thresholds represent an event after which prices drop, that is the timing of price landings. Finally, if we look closely at the different price landing patterns we discover that the speed of landing varies across these panels. Prices seem to drop much faster when we use cumulative sales than when we use time on the horizontal axis.

In the analysis that follows we show how we select one of these potential price landing triggers for each of the products in our sample. Specifically, in Section 3.4.3 we present how we use our mixture specification and the underlying distributions of price landings to select among potentially correlated price landing triggers. Developing a joint model for prices, sales and competitive entry is beyond the scope of this chapter and we consider it as an area for future research. We explain more details on our modeling approach in Section 3.4 and in this section we continue with a presentation of the market context of our application.

3.3.2 The Video Game Market

The video game market is highly competitive and there are 78 video game publishers who design games for PS2. On average, they released 29 new video games per month between 1992 and 2005. The main publisher of these VG's is Sony and it has a market share of 16%. Acclaim and Electronic Arts follow Sony with market shares of 11% and 6%, respectively. In the upper left panel of Figure 3.3 we present the distribution of the market shares across all publishers. We notice that 20 publishers have about 80% of the market while the 58 remaining publishers cover the next 20% of the market. In the upper right panel of Figure 3.3 we depict the monthly time series of the number of newly released video games. There is an upward trend in the number of VG's being released. In 1996 less than 11 VG's were released per month while in 2002 this volume has increased to 40 monthly releases.

The bottom left panel of Figure 3.3 shows the industry's sales pattern. Total VG's sales are extremely seasonal and they peak every December when they may reach numbers like 14 million copies. This last number is especially high if we compare it against the 24.1 million units of PS2 consoles sold between 1995 and 2002. Finally, in the lower right panel we show the average number of video games released from 1995 to 2002 and the average sales per month. An interesting fact is that most new VG's are released during November and January but sales peak in between these two months. From 1995 to 2002, December VG's unit sales are on average 14 million and in January sales decrease to less than 3 million copies while on average 18 new VG's are released on December, 27 in November and 34 in January. In Figure 3.4 we can see the distribution of the type of video games sold. For example, sports games account for 21.5 %, Action 14 % while Strategy games account for 4 % of all VG's in our data.

The consumers in this market concern 40 million US-based consumers who buy video games each year. Figure 3.5 shows a histogram of the total sales across all VG's. Preferences clearly differ across VG's as we observe substantial heterogeneity in the market potential across the video games. We follow the tradition of diffusion research by labeling the cumulative sales reached by a video game as the market potential. From Figure 3.5 we can learn that sales above one million units for a single game seem to occur only rarely. The average market potential for the video games in our sample is around 254.75 thousand units. However, approximately for half of the VG's in our sample (to be precise: for 504 video games) the market potential is less than 66 thousand units.

3.4 Price Landing: Modeling

In this section we present our modeling approach. The model and econometric approach we present allow us measure quantitatively the theoretical features of prices discussed in our literature review. Specifically, the equation we propose allows us describe the speed and timing of price transitions while we use mixture modeling to test the relative importance of different price landing triggers. Finally, we apply a hierarchical structure to describe the empirical distributions of the timing and speed of price transitions and at the same time to identify the most likely price triggers.

Our model consists of two parts. First we present an equation to describe the price landing, that is the underlying price of product i at time t, which we call $P_i^*(t)$. Next we specify an equation that relates the pricing landing to the actually observed prices, what we call $P_i(t)$. As we observe in Figure 3.1, prices follow a general inverse S-shape but they do not follow it very smoothly and in most cases the prices we observe are noisy. Hence, in the first equation we capture the price landing and its main two features (timing and speed) and in the second we capture deviations from it. In Section 3.4.1 we present these two equations. Each video-game is allowed to have its own price landing speed, timing, initial price and landing price parameters. In Section 3.4.2 we therefore specify how we model this heterogeneity. In Section 3.4.3 we briefly discuss the mixture specification that allows us to identify the trigger of the price landing for each video game. In Section 3.4.4 we discuss heterogeneity in mixture probabilities. In Section 3.4.5 we present details regarding the co-variates in the hierarchical structure of the model.

3.4.1 Price Landing Model

The price landing of game *i* is $P_i^*(t)$ and we assume it depends on a trigger denoted by $D_i(t)$. That is, prices change according to

$$\frac{dP_i^*(t)}{dD_i(t)} = \frac{(P_i^*(t) - \kappa_i)(\rho_i - P_i^*(t))}{(\kappa_i - \rho_i)\nu_i},$$
(3.1)

where ρ_i is the starting price level, κ_i is the final pricing level, and ν_i a constant that moderates the rate of change $dP_i^*(t)/dD_i(t)$. For ease of interpretation, $D_i(t)$ might be for example time and then $dP_i^*(t)/dD_i(t) = dP_i^*(t)/dt$. $D_i(t)$ can be set to be any trigger variable that we are interested in, like sales or competition. From (3.1) we see that a smaller ν_i implies a faster rate of change. Here, the time index t will in each case be relative to the launch date of the particular product. In other words for each product t = 0 corresponds to the time of launch. In the numerator of (1) we have that the closer $P_i^*(t)$ is to its initial or final levels, the slower prices would change and that if $P_i^*(t) < \rho_i$, $\nu_i > 0$, $P_i^*(t) > \kappa_i$, $\rho_i > \kappa_i$ for all t then $dP_i^*(t)/dD_i(t) < 0$. These last conditions describe very closely the price patterns that are common among high-tech products.

Equation (3.1) may be unusual in the sense that it models dP/dD instead of dD/dP. However, in our application we will use different trigger variables for D and hence dD/dP would not have the common interpretation we find in the literature when D are sales; for example, Dcould be competitive introductions. The former is the typical solution proposed by analytical studies while the latter is the typical form assumed in empirical studies. One of the possible reasons why empirical studies have assumed this latter form is that many of them focus on a single firm, usually a monopolist that sets prices. In contrast, in our study we observe the $\frac{dP_i^*(t)}{dD_i(t)}$ for hundredths of products launched by 78 firms that are price setters. Hence, our objective is to characterize the heterogeneity of $\frac{dP_i^*(t)}{dD_i(t)}$ across products and to capture two of its features, the timing (λ_i) and speed (ν_i) of significant price cuts. In addition, the advantage of equation (3.1) is that we can solve it analytically and test it empirically. In fact, it can be shown that (3.1) is a separable differential equation and that its solution is

$$P_i^*(t) = \kappa_i + (\rho_i - \kappa_i)h_i(t), \qquad (3.2)$$

with

$$h_{i}(t) = 1 - \frac{e^{\left(\frac{D_{i}(t) - \lambda_{i}}{\nu_{i}}\right)}}{1 + e^{\left(\frac{D_{i}(t) - \lambda_{i}}{\nu_{i}}\right)}}.$$
(3.3)

That is, we propose that the price of product i is composed of two parts, a fixed landing price (κ_i) plus a mark-up $(\rho_i - \kappa_i)$ that evolves over time proportionally to $h_i(t)$. The function $h_i(t)$ gives the percentage of the markup at time t and it is bounded between 0 and 1. The function (3.3) for $h_i(t)$ follows a logistic shape and λ_i can be interpreted as the location of the price landing for product i in terms of the trigger $D_i(t)$ while ν_i is the speed at which the landing occurs. That is, we observe a price drop after $D_i(t)$ reaches its threshold λ_i and this is why we call $D_i(t)$ the trigger variable.

The advantage of a logistic function for the pricing equation is that we can interpret its parameters in a natural way in our application. We plot equation (3.1) for $D_i(t) = t$ and different values of λ_i and ν_i in Figure 3.6. As can be noticed from the graph, the effect of an increase (decrease) of λ_i is to shift the complete function to the right (left) and ν_i has the role of smoothing the function or steepening the function. That is, ν_i is a parameter that determines how fast prices are falling and λ_i captures the moment (event) when prices are dropping.

In principle, $D_i(t)$ can be any variable that increases monotonously. The simplest choice for $D_i(t)$ is simply time $(D_i(t) = t)$. It is important to notice that the interpretation of λ_i and ν_i depend on the choice of $D_i(t)$. If we set $D_i(t)$ to be the cumulative sales of product *i* then λ_i is simply the number of items sold at high prices. We might interpret this limit as a proxy for the size of the segment that buys at high prices; what some call the hard-core gamer segment. This is a natural interpretation for λ_i but we do not claim that this model really identifies who and how many are the real hard-core gamers. Furthermore, if we define $D_i(t)$ as the number of products introduced after launch of product *i* then λ_i becomes a competitive threshold after which prices are cut. In all cases ν_i is a scaling constant that marks the transition speed of prices as we set in equation (3.1) and it of course depends on the scale of $D_i(t)$. Notice that $D_i(t)$ might be a combination of different trigger variables. The interpretation of the λ_i parameters then becomes troublesome with such specification.

As discussed above, $P_i^*(t)$ aims to capture the underlying price pattern of product *i*, that we call price landing. For actual data we observe this pattern plus noise. The observed prices may therefore differ from $P_i^*(t)$. Furthermore, we only observe the prices at regularly spaced intervals. We adopt the convention that we observe the prices for product *i* at $t = 0, 1, 2, ..., T_i$. We denote the observed price at time *t* by $P_i(t)$. We model the relation between the observed prices and price landing pattern using a first order auto-regressive specification. In terms of the observed price this gives

$$P_i(t) = P_i^*(t) + \alpha_i [P_i(t-1) - P_i^*(t-1)] + \varepsilon_i(t) \qquad t = 1, 2, 3, \dots, T_i,$$
(3.4)

where $\varepsilon_i(t)$ denotes the source of the random deviation at time t from the underlying price landing pattern, and α_i determines the memory in the deviations from the underlying pattern. We assume that $\varepsilon_i(t) \sim N(0, \sigma_i^2)$ for $t = 0, 1, \ldots, T_i$. If $\alpha_i = 0$ there is no memory, and (3.4) then states that the deviations are independent over time. If $\alpha_i > 0$, a positive deviation at time t is likely to induce a positive deviation at time t + 1. For the first observation we set

$$P_i(0) = P_i^*(0) + \sqrt{\frac{1}{1 - \alpha^2}} \times \varepsilon_i(0).$$
(3.5)

The variance factor is set such that the variance of the random term equals the unconditional variance of $P_i(t)$ in (3.4).

3.4.2 Heterogeneity in Main Parameters

In the above discussion of the model we have explicitly allowed for heterogeneity, that is, all parameters and the price cut trigger $D_i(t)$ are product-specific. In this section we discuss how we model the heterogeneity in all parameters.

In the model we will allow for K different triggers, which are denoted by $D_{1i}(t)$, $D_{2i}(t)$, ..., $D_{Ki}(t)$. The relationship between the observed price and the price landing in (3.2) remains unchanged. In addition, we define a different price landing equation $P_{ki}^*(t)$ for each trigger variable k, that is,

$$P_{ki}^{*}(t) = \kappa_{i} + (\rho_{i} - \kappa_{i})h_{ki}(t)$$

$$h_{ki}(t) = 1 - \frac{e^{\left(\frac{D_{ki}(t) - \lambda_{ki}}{\nu_{ki}}\right)}}{1 + e^{\left(\frac{D_{ki}(t) - \lambda_{ki}}{\nu_{ki}}\right)}}.$$
(3.6)

Note that this definition is very similar to that in (3.2) and (3.3). However, the parameters λ_{ki} and ν_{ki} are now trigger (k) and product (i) specific. Note that the price starting and landing level ρ_i and κ_i are the same across all k possible triggers.

The landing level (κ_i) , the initial price level (ρ_i) , the threshold value (λ_{ki}) and the speed of adjustment (ν_{ki}) are defined to vary across products. For each of these parameters we specify a

second-level model. For the price landing level and the launch prices we specify

$$\kappa_{i} = Z_{i}^{\prime} \gamma^{\kappa} + \omega_{i}^{\kappa}$$
with $(\omega_{i}^{\kappa}, \omega_{i}^{\rho}) \sim N(0, \Sigma),$

$$\rho_{i} = Z_{i}^{\prime} \gamma^{\rho} + \omega_{i}^{\rho}$$
(3.7)

where Z_i denotes a vector of dimension M of product specific characteristics, γ^{κ} and γ^{ρ} are coefficient vectors (dimension $M \times 1$) common across all i products. The error terms ω_i^{κ} and ω_i^{ρ} are assumed to be normal with mean 0 and covariance matrix Σ . The Z_i in our model will include mainly product type, manufacturer variables and seasonal dummies. We define the Z_i variables with more detail in Section 3.4.5. We specify a similar form for the speed and timing parameters. That is, for each trigger variable k we define

$$\ln \lambda_{ki} = Z'_i \gamma^{\lambda}_k + \eta^{\lambda}_{ki}$$

with $(\eta^{\lambda}_{ki}, \eta^{\nu}_{ki})' \sim N(0, \Omega_k).$ (3.8)
$$\ln \nu_{ki} = Z'_i \gamma^{\nu}_k + \eta^{\nu}_{ki}$$

where η_{ki}^{λ} and η_{ki}^{ν} are the error terms and they are assumed to be normal with mean 0 and covariance matrix Ω_k . The γ_k^{λ} and γ_k^{ν} are coefficients vectors (dimension M) and Z_i are the same group of group of covariates as in the equations for κ_i and ρ_i . The log transformation in (3.8) is used to ensure that λ_{ki} and ν_{ki} are positive. If it is the case that the timing and the speed of price landings are correlated we will capture this correlation with the matrix Ω_k . For example, it might be that when prices fall at a slower rate (ν_i^k) they are cut at an earlier time (λ_i^k) .

3.4.3 Choice of Trigger and Mixture Specification

The actual trigger of the price landing for each product is of course unobserved to the *researcher*. We denote this (unobserved) variable as S_i , that is, we denote $S_i = k$ if the trigger variable k is selected for product *i*. We complete this part of the model by specifying probabilities for each trigger, that is, the trigger k is selected with probability π_k for k = 1, 2, ..., K. In our application k = 1 would mean that *time* is the trigger, k = 2 means that *cumulative sales* are the trigger and k = 3 means that *cumulative competitive introductions* are the main trigger of equation (3.2). In the four-trigger version of our model k = 4 means that the release schedule of firms are the main trigger. We provide more details on how we measure each trigger variable in subsection 3.4.5. The probabilities π_k will reflect the overall likelihood of each of the different triggers. However, note that conditionally on the observed prices, the probability of $S_i = k$ is different across games.

In Figure 3.7 we describe the intuition about how triggers are selected and statistically identified. For this purpose we need two main elements. The first element consists of the distributions of the threshold parameters for each of the different triggers. That is, the distribution of λ_{ki} and ν_{ki} across all *i* and for each *k*. For example, if we collect the parameter λ_{1i} for all *i* we obtain the distribution of λ for the first trigger variable. As we defined in equation (3.8), the distribution of λ_{ki} and ν_{ki} depend on co-variates Z_i and hyper-parameters γ_k and the variance term associated to them. The second element we need is the match between the price landing of game *i* and the distributions of λ_{ki} and ν_{ki} for $k = 1, \ldots, K$.

In Figure 3.7 we plot again the price of the Spider-Man. In addition, we plot a hypothetical distribution of the threshold parameters λ_{ki} for each of the mixture components k. The distribution of λ_{1i} in the upper left panel, λ_{2i} in the upper right panel and λ_{3i} in the lower left panel. Note that λ_{1i} is the time (in months) after which the price drops (if $D_i(t)$ is time, that is when k = 1). In the same way, if $D_i(t)$ is cumulative sales then λ_{2i} is the cumulative number of sales after which the price drops and λ_{3i} is the cumulative number of competitive introductions after which the price drops when $D_i(t)$ amounts to competitive introductions. We notice that the $\hat{\lambda}_{1i} \approx 11$ months, that $\hat{\lambda}_{2i} \approx 600$ thousand units and that $\hat{\lambda}_{3i} \approx 250$ units. Given the λ_{ki} thresholds we can now compare them against the corresponding distributions. In this case we see that the $\hat{\lambda}_{2i}$ is the closest to the mode of its corresponding distribution. Hence, the most likely trigger of the Spider-Man price landing is sales. The least likely trigger is competition and next is time. Of course, in our model we take into account the distribution of λ_{ki} and ν_{ki} technical details about trigger selection are given in the Appendix 3.A. Next we describe how we model heterogeneity in the mixture components.

3.4.4 Heterogeneity in Mixture Probabilities

We suspect that there also might be heterogeneity in the mixture probabilities across games. For example, the games of some publishers may be more likely to belong to the time mixture. Hence, as an extension to the model we allow the probabilities of $S_i = k$ to depend on a set of product specific variables. To model this dependence we specify a Multinomial Probit Model for S_i . Hence, we introduce additional latent variables y_i^* for i = 1, ..., N and k = 1, ..., K. These latent variables are related to S_i by

$$S_i = k \quad \text{if and only if} \quad y_{ki}^* = \max(y_{li}^*). \tag{3.9}$$

We specify y_{ki}^* as

$$y_{ki}^* = Z_i' \delta_k + \vartheta_{ik} \quad \text{with} \quad \vartheta_i \sim N(0, I),$$

$$(3.10)$$

where $\vartheta_{\mathbf{i}} = (\vartheta_{1i}, \vartheta_{2i}, ..., \vartheta_{Ki})$ and we set $\delta_1 = 0$ for identification. In principle the set of variables used in this specification may differ from that in (6) and (7). The probability that the trigger k is used for product *i* now becomes

$$\pi_{ki} = \Pr[y_{ki}^* = \max_{l=1...K} (y_{li}^*)].$$
(3.11)

This concludes our model specification. For inference we will rely on MCMC and Bayesian analysis and treat all product specific parameters as latent variables and we sample these together with the parameters in (3.6), (3.7) and (3.8). A complete description of the sampling steps in this Markov Chain can be found in the Appendix 3.A.

3.4.5 Model Specifics for Video Games Pricing Model

We consider two versions of our model. The first version uses three trigger variables and the second uses four trigger variables. We define $D_{ki}(t)$, for k = 1, 2, 3, 4 where $D_{1i}(t) = A_i(t)$,

 $D_{2i}(t) = C_i(t)$ and $D_{3i}(t) = I_i(t)$ and $D_{4i}(t) = R_i(t)$. $A_i(t)$ is defined as the age of a video game in months, that is, the time between launch and t. $C_i(t)$ is the cumulative sales of video game i between release date and t. $I_i(t)$ is defined as the cumulative number of video games introduced between the launch date of video game i and t. $R_i(t)$ is defined as the release schedule of the firm that released product i. We know the number of games a firm released at every point in time. To create $R_i(t)$, we use a time window that sums the introductions from the introduction of game i up to the next three months after t.

The interpretation of λ_{ki} and ν_{ki} varies depending on the trigger k. Hence, λ_{1i} can be interpreted as the price landing time, λ_{2i} as a competitive threshold, λ_{3i} as the hard-core gamer segment size and λ_{4i} as a release limit after which we observe a price drop. For each of these triggers, the parameter ν_{ki} for k = 1, 2, 3, 4 can be interpreted as a scaling constant that changes the speed at which the price landing occurs.

In all what follows in this section we focus on the model with three triggers, that is k = 1, 2, 3and we leave out $R_i(t)$. The reason for this is that $R_i(t)$ is selected with a probability very close to zero when we include it as the fourth trigger variable. We present the discussion regarding the fourth trigger in our results in section 3.5.2.

The hierarchical structure of the corresponding threshold λ_{ki} , speed ν_{ki} and ρ_i and κ_i parameters for each mixture component will depend on a set of Z_i variables that contain game type, publisher and seasonal effects plus the launch price and the time to the introduction of a new game consoles as co-variates. Seasonal dummies are defined by the month of launch of each video game *i*. The launch price is the observed price of video game *i* at launch time, that is at t = 0. We include this variable in order to test if our co-variates remain significant after including past prices in the equation for the timing and speed of launch. It might be that the price at launch of a VG might contain information regarding the timing of the price landing and its speed. In addition, we believe it is reasonable to include the launch price because of the very likely uni-directional relationship between launch price and timing of price landing. That is, it is very hard to argue that a firm decides how to price a VG's based on its decision on when to permanently cut its price; on the other hand, it might be that firms decide to cut prices based on the launch price. For example, firms might cut the price of expensive games after longer time

than the time they wait to cut the price of cheaper VG's. Moreover, the launch price is a proxy for quality and hence we test if our covariates remain significant after we control for them.

The time to console launch measures the time (in months) between a video game release and the launch of the VG's console that is being released after the video game introduction. For example, the PlayStation2 with DS controllers was introduced in June 1998 and other versions of the PS2 console were released in February 1999 and January 2002. This means that a video game released in January 1998 will face a console introduction after 6 months; a video game released in January 1997 will face a release in 18 months, and so on. For each video game we calculate the time between its release and the forthcoming console at the video game release date. We include this variable to test whether the price landing pattern varies relative to the release date of video game consoles. Our results do not significantly change if we leave both time to console launch and launch price out of the Z_i covariates.

From the seasonal fixed effects we excluded January, from the game types we excluded Adventure games. The remaining game type categories are: Action, Arcade, Children, Driving, Family, Fighting, Role playing, Shooter, Sports, Strategy and Compilations. The remaining publisher dummies are Electronic Arts, Acclaim, Infogames, Konami, Activision, Midway, Eidos Interactive, THQ, Capcom, Namco, Agetec, Interplay, Hasbro, 2nd group, 3rd group and 4th group. The 2nd group is composed by six publishers that each have at least 1% market share, the 3nd group is composed by 14 publishers that account for the next 10% market share and the 4th group is composed by 43 publishers that account for less than 1% of the market share in total. In all our tables we sorted publishers by their market share and in descending order. The main publishers (EA, Acclaim, etc.) account for 80% of the VG's in our sample while the dummies for 2nd, 3rd and 4th publishers group the next 20% of the market share. We set Sony as the reference publisher.

3.5 Results

In this section we present our results in three subsections. In the first we present results regarding the heterogeneity of the parameters. Next we present the results regarding trigger selection and finally we discuss the model performance.

3.5.1 Heterogeneity of Landing Time and Speed

Our results indicate that there indeed exists heterogeneity in the model parameters. The first contribution we have to offer is that we find significant firm effects on both the timing and speed parameters across all mixtures. That is, firms might be deciding not only on when to cut the price but also on how fast to cut it. To our knowledge, this result is new and we are the first to show it empirically. In Table 3.2 and in Table 3.3 we can see the different firm effects across mixtures and model parameters. For example, Acclaim's landing time (λ_i) coefficient in the time mixture is -0.196 and this means that VG's of Acclaim face a price drop 17.8% earlier relative to Sony. In addition, we find several of the firm effects on the landing speed (ν_i) to be significant. For example, Electronic Arts has a ν that is 91.7% higher than Sony while Agetec has a slower landing speed with a ν parameter that is 3.81 times higher than Sony. An interesting feature of the time mixture parameters is that most of the firm effects $log(\lambda_{1i})$ are negative while the firm effects for the $log(\nu_{1i})$ are positive. That is, it seems that the video game prices of most firms are cut at earlier dates than Sony but most firms cut prices at slower speed relative to Sony.

In the last four columns of Table 3.3 we report the results for the hierarchical specification of the initial and landing price levels, (3.7). In both cases we observe very important firm effects. For example, Konami sets the landing prices 2.535 USD above the landing prices of Sony, 17.34 USD, while the launch prices of Konami are not significantly different than those of Sony that start at 40.49.

We give a histogram of the posterior mean of the game-specific parameters of the threemixture model in Figure 3.8, Figure 3.9 and for the auto-regressive term of equation (3.4) in Figure 3.10. The dispersion in the timing and speed parameters is reported in Figure 3.9. We can see that each mixture has quite different thresholds and speeds. For example, the time mixture mean is around 7 months. That is, firms cut VG's prices mainly in the 7th month after their release. The timing parameters for all mixtures are graphed in the left frames while in the right frames we present the speed parameter distribution. We note that if the speed parameter ν_i is close to zero then prices fall more steeply. From the histograms in the right panels of Figure 3.9 we see that several products face sharp price cuts. In addition, in Figure 3.8 we see the distribution of the starting price level ρ_i for all *i* in the left frame and the distribution of the κ_i in the right frame. These parameters show that the starting level might be as low as 20 USD and as high as 70 USD while the landing level is as low as 5 USD and as high as 35 USD.

In summary we find that firm effects are important to describe the price landing timing and to describe its speed, the launch and the landing prices of the VG's in our sample. Seasonality is more important for the starting and landing levels of prices and less so for the price landing timing and speed. We also find that for some mixtures the effect of the launch price and the time to launch a new console are significant for some of the main parameters.

3.5.2 Triggers of Price Landings

Our second contribution is that we find that the triggers that best describe price landings are competitive introductions and time and not cumulative sales. In Figure 3.11 we report a histogram of the posterior probability of each of the triggers across all games in the three-mixture version of our model.

The academic convention is that sales should be an important price driver. In contrast, we find that the sales indicator is the least likely trigger variable of price landings and it is useful to explain only a 12% of the video-games in our sample. Note that we do not go against the academic convention that posits that sales are a price *driver*. Our results only indicate that sales are not the main price landing *trigger*. Furthermore, we find that the competition indicator, measured by competitive VG's introductions, is the likely trigger of price landings of approximately 25.7% of the VGs in our sample. The study of Nair (2007) finds no evidence of important substitution patterns between video-games and hence he suggests that competition, at the game-specific level, is not important to explain video-game prices. Our model cannot provide insights regarding the individual level competition between different video-games but

we find that competition, measured as the cummulative sum of VG's introductions, is a likely trigger of price landings. Finally, we find that the most likely trigger is time itself or in other words, the most probable trigger is simply the *age* of a video-game. The time mixture has a posterior mean probability of 62.21%.

In the fourth-mixture version of our model we tested a fourth trigger without much success. The additional mixture included the release schedule of firms as trigger. The idea was to test whether firms release schedule could have a large probability relative to the other three trigger variables. Firms usually have information on the dates that their new VG's are to be released and therefore the prices of their previously released VG's could depend on the dates of these new introductions. Our data include the number of games each firm released at every point in time and therefore we also know the number of games each firm will release after each point in time. Hence, we sum the VG's introductions before time t up to the introductions in the next three months after t and this sum is the value of $D_i(t)$. Note that $D_i(t)$ is then the release schedule of the manufacturer of the video game i at time t. We decided to use a three-months time window because most online sources of VG's releases cover, as a maximum, the upcoming three months. Of course, in our database we just know the release schedule perfectly. However, our results indicate that the probability of this latter trigger mixture is on average very close zero. Our conclusion is that price landings are better described by the entire market introductions rather than the release schedule of any single firm. This makes some sense given that the 78 VG's firms in our sample face on average 29 releases per month. Consequently, firms might be more likely to monitor all market introductions rather than their own product introductions.

3.5.3 Model with Hierarchical Specification in the Mixture Probabilities

We estimate the same specification of our model but now we add a hierarchical specification in the mixture probabilities. In this section we discuss the estimates of this hierarchical specification and in the Appendix 3.A we provide its technical details. The estimates of the parameters in the hierarchical structure of the mixture probabilities are reported in Table 3.4. In contrast with the heterogeneity in the main parameters we do not find substantial heterogeneity in the mixture probabilities. For example, we find only three significant publisher effects (Konami, Activision, Midway) in the latent utility of sales and two significant publisher effects (Capcom and Interplay) in the latent utility of the time mixture. That is, we know that there is heterogeneity in the timing and speed of price landings but we do not know why a trigger is more likely than the others. We consider this an area for further research.

3.5.4 Model Performance

We compare the out-of-sample performance of our model against two models: A naive model for prices, that is an AR(1), and against an alternative version of our model. In this alternative model we use the same specification and parameters and the same number of mixtures as our model but we replace all triggers with time. That is, $D_i(t) = time$ for all k mixture components. We randomly selected 50 video-games and used their first six observations to forecast their complete series. That is, only the first six observations of these 50 games were used for parameter estimation while we continue to use all observations for all other games. These comparisons are reported in Table 3.5 and in Table 3.6.

Our model preforms extremely well when compared against the AR(1) model and reasonably well when compared against the restricted model. In Table 3.5 we see that our model forecasts prices better than a naive AR(1) model for 40 out of the 50 randomly chosen games. We report the root mean square forecast error and the log of the predictive density for all 50 VG's. More details on how we compute the predictive density are given in the Appendix 3.A. Moreover, our model performs better than the model with three time mixtures for 19 out of the same 50 games and in 18 other cases it performs equally well as the alternative specification. In total 37 out of 50 games our specification performs at least as well as the alternative or better. This means that there is information contained in past sales and the competition mixtures that increase our model fit. The AR(1) model does not capture the timing of significant price cuts and the speed at which the price cut occurs while our model captures these significant price cuts. Nonetheless, the assumption that prices follow an AR(1) pattern is common in previous marketing literature and our evidence suggests that this model performs poorly. The main reason is that new products face significant price cuts during their life-cycle and hence the AR(1) is not a suitable specification for such price patterns. At the moment and to our knowledge, we are the first to propose an empirical model that captures these price dynamics.

The results we presented in the previous sections are robust to different model specifications. For example, we estimated the model without the hierarchical specifications of all its parameters and the price landing timing and speed parameters stay about the same. Furthermore, we estimated the model without the auto-regressive structure in (3.4) and (3.5), and again the main parameters are estimated similarly. A reason why our results stay the same is that the pricing equation in (3.2) can accommodate many different pricing patterns with only four parameters and that we let these parameters to be product-specific. These four parameters are the initial and landing price levels, ρ_i and κ_i , and the timing and speed of price landings, the λ_i and ν_i .

3.6 Conclusions

Our aim with this chapter was to model the dynamics of new product price landing patterns. Price landings usually follow the inverse of the well known S-shape of sales. Nonetheless, we found no empirical studies dealing with these regularities of new product prices.

In this chapter we were concerned with products that face one significant price cut during their life cycle. Several online price trackers report similar dynamics to a wider range of products like mobile phones, cameras, storage media, books, etc. Our data was collected by NPD Group but several websites like www.pricescan.com or www.streetprices.com let their users plot price trends and indeed it is relatively easy to find many other products facing a single and significant price drop during their lifetime. We believe that knowing when a price is cut or when to significantly cut the price of a product permanently is an exciting area of further research and one with wide managerial implications across different industries.
In this chapter we provided evidence that there is heterogeneity both in the timing and speed of price landings. We found that most of this heterogeneity is driven by firm effects. Our model captures this heterogeneity and it is flexible and useful to forecast and describe the price landing patterns in our data. Finally, we found that it is the age of a video game that is triggering the price landings. The next most likely trigger is competition and the least likely is cumulative sales. This latter finding goes against the academic convention that sales are the main *driver* of prices. At least for our application we found convincing evidence that sales are not the most likely *trigger* of significant price cuts.

3.7 Figures and Tables



Figure 3.1: Price Landing Pattern for 50 Randomly Selected Games



Figure 3.2: Typical Price Landing Pattern



Figure 3.3: The Video Games Market



Distribution of Game Types 1995-2002

Figure 3.4: What do publishers sell?



Distribution of Market Potential

Figure 3.5: Total Sales Distribution



Figure 3.6: Main Pricing Function at Different Parameter Values



Figure 3.7: Identification of Triggers



Figure 3.8: Histogram of the Posterior Mean of Starting (ρ_i) and Landing Price (κ_i) Parameters



Figure 3.9: Histogram of the Posterior Mean of the Threshold (λ_i^k) and Speed (ν_i^k) Parameters



Distribution of Autoregressive Coefficients

Figure 3.10: Histogram of the Posterior Mean of the α_i Parameters



Figure 3.11: Histogram of the Posterior Mean of Price Triggers $P(S_i = k)$

		NEW PRODUC	TS PRICING S	TUDIES	
Author (Journal, Year)	Approach	Price Change Speed	Price Mimics Diffusion	When to cut prices	Main Price Driver
Bass and Bultez (1982)	Analytical	I	Yes	No	Saturation
Bayus (1994)	Empirical	Gradual	No*	No	Saturation
Bayus (1992)	Analytical	Gradual	$Maybe^*$	No	Learning Curve + Consumer Heterogeneity + Entry
Clements and Ohashi (2005)	Empirical	Gradual	No	No	Saturation + Indirect Network Effects
Chintagunta et al. (2009)	Empirical	Gradual	No	No	Saturation + Marketing Mix
Dockner and Gaunersdorfer (1996)	Analytical	Gradual	No*	No	Saturation + Entry
Dockner and Jorgensen (1988)	Analytical	Gradual	Yes^*	No	Learning Curve + Saturation
Dolan and Jeuland (1981)	Analytical	Gradual	$Maybe^*$	No	Learning Curve
Eliashberg and Jeuland (1986)	Analytical	Jumps	No*	No	Entry
Feng and Gallego (1995)	Analytical	Jumps	No	Yes	Saturation
Ferguson and Koenigsberg (2007)	Analytical	Jumps	No	Yes	Deteriorating Inventory
Franza and Gaimon (1998)	Analytical	Gradual	No	Yes	Entry Timing + Saturation + Learning Curve
Gupta and Di Benedetto (2007)	Analytical	Gradual	No	Yes	Entry + Advertising
Gupta et al. (2006)	Analytical	Jumps	No	Yes	Deteriorating Inventory + Consumer Valuations
Horsky (1990)	Empirical	Gradual	Yes^*	No	Saturation
Kalish (1985)	Empirical	Gradual	No	No	Advertising
Kalish (1983)	Analytical	Gradual	Yes^*	No	Learning Curve + Saturation
Kalish and Lilien (1983)	Analytical	Gradual	Yes^*	No	Saturation
Kornish (2001)	Analytical	Gradual	No^*	No	Entry (New Product Generations)
Krishnan et al. (1999)	Analytical	Gradual	No	Yes	Saturation
Nair (2007)	Empirical	Jumps	No	Yes	Consumer Heterogeneity + Expectations
Nascimento and Vanhonacker (1993)	Analytical	Gradual	No	No	Consumer Heterogeneity + New Product Generations
Padmanabhan and Bass (1993b)	Analytical	Jumps	No	\mathbf{Yes}	Entry + New Product Generations
Parker (1992)	$\operatorname{Empirical}$	I	No	No	Saturation
Rajan et al. (1992)	Analytical	Jumps	Yes	γ_{es}	Saturation
Rao and Bass (1985)	Analytical	Gradual	Yes	I	Learning Curves + Saturation
Raman and Chatterjee (1995)	Analytical	Gradual	Yes	No	Saturation
Robinson and Lakhani (1975)	Analytical	Gradual	No*	No	Saturation
Schmalen (1982)	Analytical	Jumps	No	No	Entry
Simon (1979)	Empirical	Gradual	No	Yes	Saturation
Teng and Thompson (1996)	Analytical	Jumps	No	Yes	Saturation + Quality
Zhao and Zheng (2000)	Analytical	Jumps	No	Yes	Consumer Heterogeneity
Note: *The study supports price skimming					

Table 3.1: Literature Review on New Products Pricing

		Mixture (1) I	O(t) = VG's Age		Mixture (2) D(t) = Cumul	ative VG's Introd	uctions
	Landing Time		Landing Speed		Competitive (Threshold <i>loa</i> (λ_{\pm})		Landing Speed	ł
Intercept	1.887***	(0.266)	-0.575	(0.627)	1.273***	(0.389)	2.419**	(0.914)
$Game \ Type$		~		~		~		~
Action	-0.234^{*}	(0.122)	-0.172	(0.307)	-0.050	(0.205)	-0.197	(0.378)
Arcade	-0.248	(0.186)	-0.715	(0.500)	-1.909	(1.517)	-2.508	(2.354)
Children	-0.266	(0.217)	-0.250	(0.526)	0.380	(0.286)	-1.575**	(0.695)
Driving	-0.350**	(0.133)	-0.400	(0.334)	-0.308	(0.195)	-0.200	(0.393)
Family	-0.378**	(0.178)	0.459	(0.417)	0.504^{*}	(0.281)	-0.602	(0.529)
Fighting	-0.295^{*}	(0.145)	-0.120	(0.347)	-0.218	(0.224)	-0.607	(0.543)
Role playing	-0.056	(0.143)	-0.493	(0.381)	-0.060	(0.207)	-0.060	(0.502)
Shooter	-0.646^{***}	(0.134)	-0.006	(0.334)	-0.182	(0.188)	-0.085	(0.357)
Sports	-0.276^{**}	(0.123)	-0.249	(0.313)	-0.096	(0.179)	-0.056	(0.382)
Strategy	-0.269	(0.176)	-0.383	(0.393)	-0.142	(0.212)	-0.329	(0.470)
Compilations	-0.342	(0.220)	1.116^{**}	(0.503)	-2.450^{***}	(0.473)	-5.903^{***}	(1.389)
Publisher								
Electronic Arts	-0.014	(0.079)	1.226^{***}	(0.211)	-0.220	(0.133)	-0.022	(0.298)
Acclaim	-0.196^{*}	(0.105)	1.106^{***}	(0.288)	-0.507***	(0.162)	0.253	(0.328)
Infogames	-0.234	(0.147)	1.016^{***}	(0.327)	-0.690***	(0.196)	-0.020	(0.345)
Konami	-0.387^{***}	(0.122)	0.446	(0.344)	-0.042	(0.179)	0.144	(0.412)
Activision	-0.280 * *	(0.111)	0.693^{**}	(0.291)	-0.020	(0.201)	0.146	(0.445)
Midway	0.039	(0.124)	1.665***	(0.345)	-0.281	(0.221)	0.115	(0.423)
Fidos Interactive	-0 701***	(0 193)	1 434***	0.308)	-0 607***	0 187)	-0.948	(0.599)
THO	-0.320**	(0.157)	1.621***	(0.414)	-0.693*	(0.336)	0.338	(0.520)
Cancom	-0.109	(0.128)	0.657	(0 410)	-0.108	(0 1 9 9)	0.356	(0.416)
Vameo	-01.0 0 918	0.150)	0 1 00***	(0.457)	-0.100 116**	(906 0)	-0.648	(0 065)
Ametoo	0.120	0.102)	0 1 1 7 * * *	064.0)		(007.0)	N 010***	(1620)
Tutourlou	- 760%**	(061.0)	1 170***	(0.210)	-4.000***	(061.1)	0T0.6-	(1020)
IIIter play	-0.007	(001-0)	T.4/U	(016.0)	-1.1 90 1 016	(020.0)	-1.430 0.010	(1.014)
	-0.09/	(net.u)	U.941	(260.0)	010.1-	(0.970)	016.0	(070.1)
Znd Publishers	-0.461^{++}	(0.099)	L.335***	(0.254)	-0.388++	(0.182)	0.154	(0.488)
3rd Publishers	-0.299^{***}	(0.096)	0.942^{***}	(0.269)	-0.636***	(0.142)	0.066	(0.336)
4th Publishers	-0.399***	(0.109)	1.334^{***}	(0.267)	-0.386^{**}	(0.176)	0.410	(0.382)
Season								
Feb	-0.183	(0.150)	-0.561	(0.429)	0.014	(0.235)	-0.364	(0.767)
Mar	-0.111	(0.140)	-0.564	(0.389)	-0.125	(0.223)	-0.403	(0.627)
Apr	-0.056	(0.162)	-0.218	(0.452)	-0.421	(0.267)	-0.514	(0.810)
May	-0.142	(0.154)	-0.549	(0.456)	-0.199	(0.247)	0.110	(0.701)
Jun	0.050	(0.154)	-0.947**	(0.443)	-0.259	(0.253)	-1.087	(0.718)
Jul	-0.224	(0.173)	-0.354	(0.530)	-0.249	(0.265)	-1.023	(0.907)
Aug	0.045	(0.153)	-0.474	(0.397)	-0.154	(0.276)	-0.550	(0.741)
Sep	-0.014	(0.130)	-0.469	(0.387)	0.023	(0.213)	-0.215	(0.673)
Oct	-0.129	(0.137)	-0.402	(0.393)	-0.042	(0.211)	-0.358	(0.677)
Nov	-0.109	(0.128)	-0.393	(0.373)	0.054	(0.194)	-0.309	(0.678)
Dec	-0.229*	(0.140)	-0.420	(0.421)	-0.287	(0.231)	-0.198	(0.789)
Other conariates		(0000)		()		()	0	(00.00)
Lannch Price	0.018***	(0.005)	0.003	(0.012)	0.048***	(0.007)	-0.025	(0.019)
Time to Console Launch	-0.005	(0.004)	-0.002	(0.00)	-0.021^{***}	(0.006)	0.027**	(0.011)
Notos: Destanion standard darieti	ion botwoon norm	****	*** indicate zono is	not containe	d in the 00 05 and 000	Z hickort n	octorior doneity ve	(
INDICS: I OSICITOI SIGNIATA ACTAN	INTERNATION PART	nulleace. ; ;	OT A TAT AND THIN	TION COTINGTIN	ת דוד הדוב מה, מה מדות <i>הם</i> י	ל יפטוואוון 0/	nerenan meranan	10m

	Mixtu	re $(3) D(t) =$	Cumulative Sales			IA	Mixtures	
	Sales Threshold Log(\\.)		Landing Speed		$\operatorname{Landing}_{\operatorname{Price} K_{i}}$		Starting Price of	
Intercept	2.579	(1.810)	4.133***	(1.363)	17.34^{***}	(1.510)	40.79***	(2.046)
Game Type		~		~		~		~
Action	-0.431	(1.179)	0.161	(0.829)	0.751	(0.987)	-1.793	(1.297)
Arcade	-1.171	(1.567)	-1.866	(1.145)	0.543	(1.492)	-5.244^{***}	(1.969)
Children	-1.696	(1.887)	0.551	(1.245)	-1.128	(1.456)	-10.46^{***}	(2.092)
Driving	0.342	(1.203)	1.040	(0.718)	-0.243	(1.007)	-1.363	(1.299)
Family	0.705	(1.641)	-0.012	(1.243)	-1.111	(1.225)	-5.322***	(1.627)
Fighting	-1.416	(1.129)	-0.974	(0.814)	2.061^{*}	(1.109)	0.378	(1.443)
Role playing	2.021	(1.730)	0.067	(1.356)	3.162^{***}	(1.180)	1.033	(1.460)
Shooter	-0.823	(1.230)	0.170	(0.901)	-0.310	(1.056)	2.050	(1.372)
Sports	-1.147	(1.301)	-1.369*	(0.845)	-0.987	(0.983)	-1.450	(1.301)
Strategy	-1.545	(1.640)	-0.462	(1.271)	2.364^{*}	(1.240)	1.290	(1.592)
Compilations	-0.603	(1.732)	2.599^{**}	(1.185)	-7.490^{***}	(1.534)	8.880***	(3.272)
Publisher								
Electronic Arts	3.395^{***}	(0.647)	2.077***	(0.442)	1.135*	(0.671)	1.712^{**}	(0.807)
Acclaim	0.904	(0.782)	0.039	(0.619)	-1.572*	(0.865)	1.617	(1.124)
Infogames	0.049	(0.664)	0.119	(0.577)	-1.766^{**}	(0.888)	1.262	(1.289)
Konami	2.240^{**}	(1.090)	-0.037	(0.656)	2.535 * * *	(0.979)	0.586	(1.199)
Activision	2.886**	(1.078)	1.683 * * *	(0.651)	-1.105	(0.962)	-0.439	(1.196)
Midway	1 346	(1 018)	-0.030	(0.875)	-1 617	(1 054)	9 344*	(1 330)
Fidos Interactive	1.674	(1 142)	1.537***	(0.747)	-2.009**	(1 013)	5.029***	(1 421)
THO	F 710***	(1 196)	2 3.41 ***	0 888)	-0.010	(1179)	9 005	(1 656)
Concorn	0.15A	(1 296)	0 476***	0.000)	1 KAD	(1 020)	0.074	1 204
Nomeo	101-2	(070-1)	0120	(600-0)	1 499	(1 664)	-10.0-	(1 E07)
A+	-2.402	(1,1,0)	010.2-	(0417)	1.400	(1.407)	007.7	(1.097)
Agerec	0.000	(701.1)	-0.40***	(0.114) (0.014)	-07.6-	(17471)	0.40/	(001.2)
Interplay	1.134 0.010	(1.241)	2.043***	(0.814)	-0.778	(1.127)	9.382***	(1.741)
Hasbro	0.816	(1.009)	-1.433	(1.460)	-3.545	(1.289)	-4.731***	(1.720)
2nd Publishers	0.398	(0.933)	-0.035	(0.909)	-3.134***	(0.767)	1.892^{+}	(1.081)
3rd Publishers	-0.212	(0.597)	0.007	(0.454)	-1.124	(0.702)	1.947^{**}	(0.943)
4th Publishers	1.356^{*}	(0.779)	0.408	(0.567)	-3.216^{***}	(0.815)	0.771	(1.212)
Season								
Feb	-1.274	(2.237)	-0.508	(1.474)	-0.121	(1.254)	-2.178	(1.747)
Mar	0.364	(1.099)	-0.344	(0.890)	2.038^{*}	(1.123)	-2.663^{*}	(1.559)
Apr	-10.56^{***}	(3.330)	-8.255***	(2.689)	2.306^{*}	(1.375)	0.148	(1.878)
May	-12.00^{***}	(2.095)	-8.820***	(1.870)	1.499	(1.281)	-1.118	(1.811)
Jun	-11.45***	(2.627)	-8.422***	(1.864)	3.648 * * *	(1.230)	-3.058*	(1.717)
lul.	-3.430^{**}	(1.819)	-3.950 ***	(1.306)	1.227	(1.367)	-1.847	(1.926)
Allo	1.129	(1.519)	0.173	(1 202)	0.236	(1, 235)	-3.719**	(1.655)
Sen	0 739	(0.877)	0.840	(00.709)	1 397	(1 088)	-2.537	(1.536)
-t-C	0.840	0.805	0.659	(0.761)	1 119	1194	-0.832	(1 573)
Nov	2500	(2000)	0.507	(062.0)	0.617	(1 072)	0.065	(1 408)
Dec	0.100	(066.0)	0.001	(07170)	110'0	(610.1)	0 694	(1 600)
Louish Lafa	60T'0	(206.0)	000.0	(001.0)	001.2	(011.1)	1.004	(700.1)
	9991 1 1 1	1010/0/	***00000	1410 07				
Time to Console I annot	0.100***	(910.0)	0.060***	(010.0)		- (0.00E)	***•••	(0.00)
TILLE to Collsole Launch	77T-0-	(070.0)		(otn.u)	10 00 11 .	(070.0)	TU4.0	(een.u)
INDES: POSTETIOT STANDARD DEVIATION	on between parentn	eses. , ,	" Indicate zero is	not contained	in the 90, 95 a	nd 99% nigi	test posterior de	ensity region

Table 3.3: Estimation Results Part II

The Timing and Speed of New Product Price Landings

3.7 Figures and Tables

	Tetert		Tatant	
	Latent		Latent	
	Othity		othity	
	or Sales		or rime	
	Mixture	(0	Mixture	(0.1.2)
Intercept	3.727***	(0.579)	0.549	(0.445)
Game Type	0 000M	(0.000)		(0.000)
Action	0.699*	(0.360)	0.260	(0.269)
Arcade	-0.032	(0.543)	0.404	(0.451)
Children	0.481	(0.479)	0.005	(0.456)
Driving	0.708*	(0.402)	0.288	(0.278)
Family	-0.100	(0.451)	-0.577*	(0.333)
Fighting	1.153^{***}	(0.427)	0.357	(0.372)
Role playing	-0.304	(0.552)	-0.146	(0.355)
Shooter	0.809	(0.567)	0.716^{**}	(0.297)
Sports	-0.010	(0.372)	0.057	(0.252)
Strategy	0.342	(0.613)	-0.055	(0.362)
Compilations	0.930^{*}	(0.530)	-0.258	(0.438)
Publisher				
Electronic Arts	-0.689	(0.514)	-0.137	(0.239)
Acclaim	-0.723	(0.583)	-0.365	(0.284)
Infogames	0.723	(0.456)	-0.332	(0.360)
Konami	1.110^{**}	(0.469)	0.068	(0.343)
Activision	1.060 * *	(0.450)	0.356	(0.325)
Midway	0.884^{*}	(0.454)	-0.119	(0.384)
Eidos Interactive	0.002	(0.668)	0.265	(0.365)
THQ	-0.287	(0.531)	-0.017	(0.459)
Capcom	-0.355	(0.451)	-0.821**	(0.357)
Namco	-0.043	(0.649)	-0.503	(0.426)
Agetec	0.592	(0.558)	0.592	(0.512)
Interplay	0.648	(0.626)	0.800*	(0.430)
Hasbro	-0.364	(0.512)	0.303	(0.380)
2nd Publishers	0.203	(0.427)	0.387	(0.305)
3rd Publishers	0.324	(0.402)	-0.025	(0.272)
4th Publishers	0.510	(0.374)	0.164	(0.301)
Season	01010	(0.011)	01101	(0.001)
Feb	0.159	(0.467)	-0.026	(0.361)
Mar	0.002	(0.451)	0.441	(0.201)
Apr	0.448	(0.101) (0.575)	0.296	(0.201) (0.431)
May	-0.309	(0.373)	-0.361	(0.344)
Jup	0.067	(0.433)	0.501	(0.344) (0.381)
Jul	0.140	(0.533) (0.547)	0.007	(0.331) (0.417)
Aug	0.149	(0.347) (0.470)	0.007	(0.417) (0.267)
Son	-0.197	(0.479) (0.272)	0.333	(0.307)
Sep	0.008	(0.373)	-0.214	(0.259)
Nee	0.150	(0.410)	0.285	(0.293)
Nov	0.070	(0.357)	0.069	(0.243)
Dec	0.000	(0.418)	0.329	(0.307)
Launch Injo	0.100***	(0.012)	0.004	(0,007)
Launch Price	-0.162****	(0.013)	0.004	(0.007)
11me to Launch	0.010	(0.015)	-0.012	(0.008)
Notes: Posterior standar	d deviation in	1 parentheses.	*,**,*** ind	licate zero
is not contained in the 90	J, 95, and 99	% highest pos	terior density	region.

Table 3.4: Results of Hierarchical Structure for Mixture Probabilities

					Log of	Log Like-
	F (II (Forecast	Pre-	lihood of
Come Title	Forecaste	a St. Dev.	Forecast	RMSE	Donaitu	predicted
NHL 2001	10	0 18	0.17	AR(1)	Density	An (1)
IVE VD ECOTRALL OS	10	0.10	1.76*	0.15 5 10	-0.69	-3.47
JJ 5 VR FOOTBALL 90	0	2.39	1.70*	0.19	-0.00	-0.79
MADDEN NEL 08	10	1.14	2.10	11.92 E 02	-10.25*	-142.23
MR DOMINO	12	4.00 8.22	2.00*	0.00 19.94	-11.00*	-27.01
THE CROW CITY ANGELS	18	14 53	3.24*	26.27	-12.15*	-360.10
DITRALI	18	19.00	3.24	20.27	-12.10	-309.19
FILDALL FROCCER 9	10	10.49	3.12	21.92	-12.00*	-204.09
BIC OL' BASS 2	18	14.45	3.00*	22.00	-12.30	485 70
MK & ASHI EV WINNED'S	18	11.40	0.52 4 11*	17.46	-10.22	402.25
CIVILIZATION 2	18	0.72	4.11	19.59	-11.50	1000.22
PONC 2	18	5.75 11.42	4.25	20.81	-0.94	-1009.23
PORG POCHE TRIP	10	11.45	4.57	1 00	-14.04	-040.32
DESIDENT EVIL 2-NEMES	19	10.16	4.30	10.34	15.01*	-0.58
RESIDENT EVIL SINEMES	10	10.10 8 21	4.70*	0.45	-10.91	-71.04
TERNAL LILO	10	0.31 7.20	4.92 5 19*	9.40	-24.10	-02.00
TERREN Z	10	11.59	5.13	28.20	10 5 2*	-00.20
E1 WDID CDAND DDIV 00	10	7 11	0.09	20.29	-12.05	-011.42
FI WALD GRAND FAIX 00	10	0.00	5.05	0 50	-29.70*	-00.07
CHEED DAIDED	10	9.09	0.00*	0.00	-21.99	-30.33
C DOLICES WDN IUSTICE	18	9.35	6.03	11.20	-81.57	-112.20
G POLICEZ:WPN JUSTICE	9	10.79	6.04 6 55*	24.87	-8.00	-380.84
RISK SVNDICATE WADS	10	9.50	0.55	12.39	-13.02	-207.31
SINDICALE WARS	18	8.93	0.00	12.83	-10.71	-00.07
JUGGERNAU I VICC DINDALI	18	9.33	0.71	10.33	-01.02	-00.97
KI55 PINDALL	10	8.47	0.73	13.79	-11.21	-128.41
OLYMPIC SUMMED CAMES	10	10.59	7.09*	20.00	-24.94	-052.10
NECTADIC MILITARY MAD	18	8.07	7.02*	12.52	-10.79	-38.77
T CLANCYC DOCUE CDEAD	18	13.34	7.00	19.03	-10.75	-292.34
TOCA 2 CAD CHALLENCE	18	0.08	1.88	4.54	-21.11	-10.02
NEL XEDEME 2	18	13.75	1.97*	13.93	-23.30	-1/1.80
ADENA FOOTDALL	18	14.30	8.27	24.55	-20.09	-407.21
ARENA FOOTBALL	17	3.40	8.30	4.08	-14.70*	-23.33
FINAL FANTASY IX	13	0.23	8.43**	12.81	-10.78*	-32.38
SHEEP	18	3.47	8.83	3.54	-22.08	-4.11
SIMPSON'S WRESTLING	12	8.66	8.87*	12.12	-16.69*	-31.70
POCKET FIGHTER	18	10.51	9.02*	17.25	-17.09*	-169.35
POWERBOAT RACING	18	10.50	9.19**	24.08	-14.13*	-466.81
GRAND SLAM 97	18	11.09	9.53*	11.69	-24.66*	-121.27
RAMPAGE WORLD TOUR	6	2.88	9.67	4.58	-1245.7	-6.05
EAGLE ONE: HARRIER	13	11.43	10.5*	13.02	-50.33*	-847.83
STRIKER PRO 2000	9	10.66	10.6*	20.87	-10.52*	-139.91
NEWMAN/HAAS RACING	16	3.88	11.31	4.63	-38.44	-4.55
DISCWRLD 2:MRTLY BYTE	18	6.31	11.42	5.95	-87.15*	-41.79
CROSSROAD CRISIS	18	9.77	12.4*	15.33	-93.93*	-458.48
SLAM N JAM 96	18	10.23	13.3*	19.53	-18.82*	-280.41
NBA LIVE 2002	18	9.23	14.4*	21.76	-80.22*	-466.39
ARMD COR 2 PRJ PNTSMA	15	8.71	15.0*	16.21	-11.77*	-263.19
CRASH TEAM RACING	18	15.92	15.8*	17.91	-342.07	-117.05
DISNEY'S DINOSAUR	18	5.00	16.35	5.68	-27.82	-15.57
NFL BLITZ 2000	18	5.63	17.57	6.41	-30.46*	-130.64
Notes: * Means the RMSE or th	ne predictive l	likelihood is s	smaller in our	model than	in the $AR(1)$	

Table 3.5: Forecasting	Performance
Table 3.5: Forecasting	Performance

3.7 Figures and Tables

			Log of Predictive	Log of Predicted
	Forecast	St Dev	Density (LPD)	Density (LDP)
Game Title	Horizon	\mathbf{Price}^{a}	Original Model	Alt Model
NHL 2001	10	0.18	_0.89 *	-0.90
LI'S VE FOOTBALL 98	8	2.39	-5.06 *	-4.89
HIGH HEAT BSBALL 2002	18	7.14	-10.25 *	-0.02
MADDEN NEL 98	12	4.56	-11.05 *	-11.19
MR DOMINO	12	8.33	-17.22	-15.86
THE CROW CITY ANGELS	18	14.53	-12.15 *	-11.20
DITRALI	18	12.40	12.10	12.06
FROCCER 2	18	10.48	12.00 *	12.00
BIC OL' BASS 2	18	14.45	-12.00	16.20
MK & ASHI EV WINNED'S	18	11.40	-10.22	-10.23
CIVILIZATION 2	18	0.72	-11.50 8 04 **	-12.11
PONC	18	9.75 11.49	-0.54	-11.30
PORG TRUE	10	11.45	-14.04	-13.10
ROGUE IRIF DESIDENT EVIL 2.NEMES	14	1.00	15 01 **	-30.14
RESIDENT EVIL SINEMES	10	10.10	-10.91	-17.40
EIERNAL EIES TEKKEN 9	18	8.31	-24.18	-20.30
TECT DDD/E 4	18	1.39	-10.13	-10.17
IESI DRIVE 4	18	11.50	-12.53	-12.40
FI WRLD GRAND PRIX 00	18	7.11	-29.76	-20.22
FADE TO BLACK	18	9.09	-21.99	-23.04
SHEEP RAIDER	18	9.35	-81.57	-121.60
G POLICE2: WPN JUSTICE	9	10.79	-8.60 **	-8.50
RISK GVNDLCATE WADG	10	9.50	-13.02	-33.94
SYNDICATE WARS	18	8.93	-10.71	-13.63
JUGGERNAUT	18	9.33	-51.62 **	-98.83
KISS PINBALL	10	8.47	-11.21 *	-11.05
BACKYARD SOCCER	18	16.59	-24.94	-21.56
OLYMPIC SUMMER GAMES	18	8.57	-16.79	-14.71
NECTARIS: MILITARY MAD	18	13.34	-16.75 *	-16.17
T.CLANCYS ROGUE SPEAR	18	5.38	-21.11 **	-22.90
TOCA 2 CAR CHALLENGE	18	13.75	-23.36 **	-26.08
NFL XTREME 2	18	14.35	-20.69	-17.40
ARENA FOOTBALL	17	3.40	-14.76 *	-14.56
FINAL FANTASY IX	13	6.23	-10.78	-9.54
SHEEP	18	3.47	-22.08 **	-23.30
SIMPSON'S WRESTLING	12	8.66	-16.69 *	-16.84
POCKET FIGHTER	18	10.51	-17.09 *	-16.58
POWERBOAT RACING	18	10.50	-14.13 *	-14.76
GRAND SLAM 97	18	11.09	-24.66 *	-23.81
RAMPAGE WORLD TOUR	6	2.88	-1245.7 **	-2072.64
EAGLE ONE: HARRIER	13	11.43	-50.32 **	-77.20
STRIKER PRO 2000	9	10.66	-10.52 *	-11.26
NEWMAN/HAAS RACING	16	3.88	-38.44 **	-54.93
DISCWRLD 2:MRTLY BYTE	18	6.31	-87.15	-38.19
CROSSROAD CRISIS	18	9.77	-93.93	-34.08
SLAM N JAM 96	18	10.23	-18.82	-15.18
NBA LIVE 2002	18	9.23	-80.22 **	-82.06
ARMD COR 2 PRJ PNTSMA	15	8.71	-11.77 **	-14.41
CRASH TEAM RACING	18	15.92	-342.07 **	-397.18
DISNEY'S DINOSAUR	18	5.00	-27.82	-18.46
NFL BLITZ 2000	18	5.63	-30.46 **	-34.48
Notes ** means that the Origina	l Model LPD	is grantor th	on the Alternative I Pl	D by more then

Notes:** means that the Original Model LPD is greater than the Alternative LPD by more t 1 unit, * means that the difference between the original and alternative are less than 1 unit. Alt. stand for Alternative and St.Dev for Standard Deviation.

Table 3.6: Comparison with Alternative Model

3.A Estimation Methodology

To draw inference on the parameters we will rely on a Bayesian analysis and more specifically the Gibbs sampler. Whenever possible we use Gibbs sampling with block updating and when there are no closed form sampling distributions we rely on the Metropolis algorithm. We run a Markov Chain for 200 thousand iterations of which the first 100 thousand are discarded for burnin and we keep each tenth remaining draws. This Markov Chain has the posterior distribution of the parameters and the latent trigger variable indicators S_i i = 1, ..., N as the stationary distribution. We programmed all our routines in Ox (see Doornik (2007)) and our graphs in R (see R Development Core Team (2005)).

In all what follows we collect the first level model parameters in the blocks: $\tau_i = (\rho_i, \kappa_i, \alpha_i, \sigma_i, \lambda_k, \nu_k), \ \rho = (\rho_1, ..., \rho_N), \ \kappa = (\kappa_1, ..., \kappa_N), \ \alpha = (\alpha_1, ..., \alpha_N), \ \sigma^2 = (\sigma_1^2, ..., \sigma_N^2), \ \lambda_k = (\ln(\lambda_{ki}), ..., \ln(\lambda_{kN}))$ and finally $\nu_k = (\ln(\nu_{ik}), ..., \ln(\nu_{Nk})).$

We further collect all hyper-parameters in the following blocks: $\theta = (\gamma^P, \gamma^L, \Pi, \Omega)$, where $\Omega = (\Omega_1, ..., \Omega_K)$, $\Pi = (\pi_1, ..., \pi_K)$. We have that $\gamma^P = (\gamma^\kappa, \gamma^\rho)$ where $\gamma^\kappa = (\gamma^\kappa_1, ..., \gamma^\kappa_M)$ and $\gamma^\rho = (\gamma^\rho_1, ..., \gamma^\rho_M)$. Finally, $\gamma^L = (\gamma^L_1, ..., \gamma^L_K)$ where $\gamma^L_k = (\gamma^\lambda_k, \gamma^\nu_k)$ and $\gamma^\lambda_k = (\gamma^\lambda_{k1}, ..., \gamma^\lambda_{kM})$ and $\gamma^\mu_k = (\gamma^\nu_{k1}, ..., \gamma^\nu_{kM})$. *M* refers to the number of variables in *Z*, *K* refers to the number of mixtures (same as number of triggers), and *N* refers to the total number of products. Next $\mathbf{Z} = (Z_1, ..., Z_M)$ and $\phi(x; \mu, \sigma^2)$ denotes the normal pdf distribution with mean μ and variance σ^2 evaluated at *x*. Finally, p() denotes a general density function and $IW(\widehat{\Omega}, N)$ denotes the inverted Whishart distribution with scale matrix $\widehat{\Omega}$ and *N* degrees of freedom.

Note that in this context we treat the product specific parameters τ_i as latent variables. We consider the log of λ_{ki} and ν_{ki} k = 1, ..., K, i = 1, ..., N as focal parameters strictly for convenience and to impose that λ_{ki} and ν_{ki} are positive. This has no impact on the results. In this Markov Chain we will sample the latent variables alongside the parameters.

The complete data likelihood for product i is

$$p(\mathbf{P}_i, S_i, \tau_i | \theta) = \pi_{S_i} \times p(\mathbf{P}_i | S_i, \tau_i, \theta) \times p(\tau_i | \theta),$$
(3.12)

where $\mathbf{P}_i = (P_i(0), ..., P_i(T))$ and $p(\mathbf{P}_i | S_i, \tau_i, \theta)$ is equal to

$$p(P_i(0)|S_i, \tau_i, \theta) \times \prod_{t=1}^{t=T} p(P_i(t)|P_i(t-1), S_i, \tau_i, \theta).$$
(3.13)

Furthermore, we have that the first observation likelihood is

$$p(P_i(0)|S_i, \tau_i, \theta) = \phi\left(P_i(0); P_i^*(0), \frac{1}{1 - \alpha^2}\sigma_i^2\right),$$
(3.14)

and all other observations have as likelihood

$$p(P_i(t)|P_i(t-1), S_i, \tau_i, \theta) = \phi\left(P_i(t); P_i^*(t) + \alpha_i [P_i(t-1) - P_i^*(t-1)], \sigma_i^2\right).$$
(3.15)

Next, we have

$$p(\tau_i|\theta) = p(\rho_i, \kappa_i|\theta) \prod_{k=1}^{K} p(\lambda_{ki}, \nu_{ki}|\theta), \qquad (3.16)$$

where

$$p\left((\rho_i,\kappa_i)|\theta\right) = \phi\left((\rho_i,\kappa_i)';\gamma^{P'}Z_i,\boldsymbol{\Sigma}\right),\tag{3.17}$$

and

$$p\left((\lambda_{ki},\nu_{ki})|\theta\right) = \phi\left((\lambda_{ki},\nu_{ki})';\gamma_k^{L'}Z_i,\mathbf{\Omega}_k\right).$$
(3.18)

We impose flat priors on all almost all parameters, for α_i we set a uniform prior on the interval (-1,1) to impose stationarity. This completes the main model specification and next we discuss how we sample from the posterior distribution for all parameters.

Sampling distributions

If π_k is fixed across products, the density of S_i conditional on \mathbf{P}_i , τ_i , and θ equals a Multinomial distribution with probabilities proportional to

$$\pi_{S_i} \times p(\mathbf{P}_i | S_i, \tau_i, \theta). \tag{3.19}$$

The full conditional distribution for α_i is a truncated normal on the interval [-1,1], where the mean and variance are given by applying the Ordinary Least Squares formulas to a regression of $P_i(t)$ - $P_i^*(t)$ on its lag with known variance of the disturbance term σ_i^2 . A draw for σ_i^2 can be obtained using the Metropolis-Hastings sampler and taking as candidate

$$\sigma_{i_{cand}}^{2} = \frac{\sum_{t=1}^{T} (\hat{\varepsilon}_{i}(t))^{2}}{w} \quad \text{where} \quad w \sim \chi_{(T-1)}^{2}, \tag{3.20}$$

where $\hat{\varepsilon}_i(t)$ is the residual of equation (3.4) given all other parameters. We evaluate this candidate and the current draw of σ_i^2 in the conditional distribution of the first observation given in equation (3.14). Hence we take the candidate as the next drawn value of σ_i^2 with probability

$$\min\left(1, \frac{\phi\left(P_i(0); P_i^*(0), \frac{1}{1-\alpha^2}\sigma_{i_{cand}}^2\right)}{\phi\left(P_i(0); P_i^*(0), \frac{1}{1-\alpha^2}\sigma_{i_{current}}^2\right)}\right).$$
(3.21)

To derive the full conditional distribution of κ_i and ρ_i we first rewrite equations (3.4) and (3.5) as

$$\sqrt{1-\alpha_i^2}P_i(0) = \left[\sqrt{1-\alpha_i^2}h_{S_i}(0)\right] \times \kappa_i + \left[\sqrt{1-\alpha_i^2}h_{S_i}(0)\right] \times \rho_i + \varepsilon_i(0), \tag{3.22}$$

and

$$P_{i}(t) - \alpha_{i}P_{i}(t-1) = [1 - h_{S_{i}}(t) - \alpha_{i}(1 - h_{S_{i}}(t))] \times \kappa_{i} + [h_{S_{i}}(t) - \alpha_{i}h_{S_{i}}(t)] \times \rho_{i} + \varepsilon_{i}(t).$$
(3.23)

These equations should be combined with the specification of the hierarchical layer in (3.7) as follows:

$$\begin{pmatrix} Y_i \\ \gamma^{\rho'} Z_i \\ \gamma^{\kappa'} Z_i \end{pmatrix} = \begin{pmatrix} X_i^A & X_i^B \\ 1 & 0 \\ 0 & 1 \end{pmatrix} \begin{pmatrix} \rho_i \\ \kappa_i \end{pmatrix} + \begin{pmatrix} \varepsilon_i \\ \omega^{\rho} \\ \omega^{\kappa} \end{pmatrix}, \quad (3.24)$$

where we define X_i^A and X_i^B as

$$X_{i}^{A} = \begin{pmatrix} \sqrt{1 - \alpha_{i}^{2}}(1 - h_{S_{i}}(0)) \\ 1 - h_{S_{i}}(1) - \alpha_{i}(1 - h_{S_{i}}(1)) \\ \vdots \\ 1 - h_{S_{i}}(T_{i}) - \alpha_{i}(1 - h_{S_{i}}(T_{i})) \end{pmatrix} \quad \text{and} \quad X_{i}^{B} = \begin{pmatrix} \sqrt{1 - \alpha_{i}^{2}}h_{S_{i}}(0) \\ h_{S_{i}}(1) - \alpha_{i}h_{S_{i}}(1) \\ \vdots \\ h_{S_{i}}(T_{i}) - \alpha_{i}h_{S_{i}}(T_{i}) \end{pmatrix}, \quad (3.25)$$

and Y_i as

$$Y_{i} = \begin{pmatrix} \sqrt{1 - \alpha_{i}^{2}} P_{i}(0) \\ P_{i}(1) - P_{i}(0) \\ \vdots \\ P_{i}(T) - P_{i}(T - 1) \end{pmatrix}.$$
(3.26)

Finally, we can draw κ_i and ρ_i from

$$N\left((W_i'\Gamma_i^{-1}W_i)^{-1}W_i'\Gamma_i^{-1}Y_i,(W_i'\Gamma^{-1}W_i)^{-1}\right),$$
(3.27)

where

$$W_{i} = \begin{pmatrix} X_{i}^{A} & X_{i}^{B} \\ 1 & 0 \\ 0 & 1 \end{pmatrix} \quad \text{and} \quad E \left(\begin{pmatrix} \varepsilon_{i} & \omega^{\rho} & \omega^{\kappa} \end{pmatrix} \begin{pmatrix} \varepsilon_{i} \\ \omega^{\rho} \\ \omega^{\kappa} \end{pmatrix} \right) = \begin{pmatrix} \sigma_{i}^{2}I & 0 \\ 0 & \Sigma \end{pmatrix} = \Gamma_{i}. \quad (3.28)$$

Due to the non-linearity in the price patterns, the conditional distributions of λ_k and ν_k are not of a known form. We will sample each parameter one at a time using a random walk Metropolis Hastings sampler. Given the current draw of one of these parameters we draw a candidate by adding a draw from a normal with mean zero and a fixed variance. This candidate draw for λ_k and ν_k is accepted with probability

$$\min\left(1, \frac{p(\lambda_{ki}^{cand}|\nu_{ki})}{p(\lambda_{ki}^{current}|\nu_{ki})}\right) \quad \text{and} \quad \min\left(1, \frac{p(\nu_{ki}^{cand}|\lambda_{ki})}{p(\nu_{ki}^{current}|\lambda_{ki})}\right), \tag{3.29}$$

respectively. The posterior of the i'th element of λ_k is

$$p(\lambda_{ki}|\nu_{ki}) = p(P_i(0)|S_i, \tau_i, \theta) \prod_{t=1}^{t=T} p(P_i(t)|P_i(t-1)S_i, \tau_i, \theta) \phi\left(\lambda_{ki}; \lambda_{ki}|\nu_{ki}, \Omega_k^{\lambda_{ki}|\nu_{ki}}\right), \quad (3.30)$$

and the posterior of the i'th element of ν_k is

$$p(\nu_{ki}|\lambda_{ki}) = p(P_i(0)|S_i, \tau_i, \theta) \prod_{t=1}^{t=T} p(P_i(t)|P_i(t-1)S_i, \tau_i, \theta) \phi\left(\nu_{ki}; \nu_{ki}|\lambda_{ki}, \Omega_k^{\nu_{ki}|\lambda_{ki}}\right).$$
(3.31)

Here x|y refers to the conditional mean of x given y and $\sigma^{x|y}$ refers to the conditional variance of x given y. These are conditional posterior distributions because we allow λ_k and ν_k to be correlated to each other. In other words, the timing of the price cut and the speed of the price cut might be correlated and these correlation is different across mixtures. The variance of the proposal density is chosen such that we obtain an acceptance rate close to approximately 25%, that is the optimal rate for high-dimensional models (see Robert and Casella (2004, page 316), Carlin and Louis (2000, page 154) or Gamerman and Lopes (2006, page 196)).

The conditional distribution of π_1, \ldots, π_K is a Dirichlet distribution with parameters $1 + \sum_i \mathbb{1}[S_i = 1], \ldots, 1 + \sum_i \mathbb{1}[S_i = K]$; that is, we draw each π_k proportional to the number of products assigned to mixture k, that is $\sum_1 \mathbb{1}[S_i = k]$, and naturally restrict $\sum_k \pi_k = 1$.

Given the latent variables in τ_i sampling the hyper-parameters of the hierarchical part for the marginal costs, launch price, and price landing characteristics is relatively straightforward. We draw γ^P from a normal

$$\gamma^{P} \sim N \begin{pmatrix} (\mathbf{Z}'\mathbf{Z})^{-1} Z' \kappa \\ (Z'\mathbf{Z})^{-1} \mathbf{Z}' \rho \end{pmatrix}, \mathbf{\Sigma} \otimes (\mathbf{Z}'\mathbf{Z})^{-1} \end{pmatrix},$$
(3.32)

and $\gamma_k^L | \Omega_k$ from

$$N\left(\begin{array}{c}\frac{1}{1+g}(\mathbf{Z}'\mathbf{Z})^{-1}\mathbf{Z}'\lambda_{ki}\\\frac{1}{1+g}(\mathbf{Z}'\mathbf{Z})^{-1}\mathbf{Z}'\nu_{ki}\end{array},\frac{1}{1+g}\mathbf{\Omega}_k\otimes(\mathbf{Z}'\mathbf{Z})^{-1}\right).$$
(3.33)

The factor g comes from the g-prior which states that the variance of (λ_{ki}, ν_{ki}) is proportional to the variance of the data. See Fernandez et al. (2001) for a detailed discussion.

Finally, we draw $\boldsymbol{\Sigma} \sim IW\left(\widehat{\boldsymbol{\Sigma}}, N\right)$ where

$$\widehat{\boldsymbol{\Sigma}} = \begin{pmatrix} \widehat{\omega}^{\kappa} \\ \widehat{\omega}^{\rho} \end{pmatrix} (\widehat{\omega}^{\kappa}, \widehat{\omega}^{\rho})$$
(3.34)

and $\widehat{\omega}_{i}^{\kappa} = \kappa_{i} - Z_{i}^{\prime} \gamma^{\kappa}$, $\widehat{\omega}_{i}^{\rho} = \rho_{i} - Z_{i}^{\prime} \gamma^{\rho}$ and $\widehat{\omega}^{\kappa} = (\widehat{\omega}_{1}^{\kappa} \dots \widehat{\omega}_{N}^{\kappa})$ and $\widehat{\omega}^{\rho} = (\widehat{\omega}_{1}^{\rho} \dots \widehat{\omega}_{N}^{\rho})$. Next, we draw $\Omega_{\mathbf{k}} \sim IW\left(\widehat{\Omega}_{\mathbf{k}} + \mathbf{G} + I_{2}, 7 + N\right)$ where

$$\widehat{\mathbf{\Omega}}_{\mathbf{k}} = \begin{pmatrix} \widehat{\eta}_{k}^{\lambda} \\ \widehat{\eta}_{k}^{\nu} \end{pmatrix} (\widehat{\eta}_{k}^{\lambda}, \widehat{\eta}_{k}^{\nu}), \qquad (3.35)$$

and $\widehat{\eta}_k^{\lambda} = \log(\lambda_k) - Z' \gamma_k^{\lambda}$ and $\widehat{\eta}_k^{\nu} = \log(\nu_k) - Z' \gamma_k^{\nu}$. finally \mathbf{G}_k is defined as

$$\widehat{\mathbf{G}}_{k} = \begin{pmatrix} \widehat{\gamma}_{k}^{\lambda} \\ \widehat{\gamma}_{k}^{\nu} \\ \widehat{\gamma}_{k}^{\nu} \end{pmatrix} g(\mathbf{Z}'\mathbf{Z})^{-1}(\widehat{\gamma}_{k}^{\lambda}, \widehat{\gamma}_{k}^{\nu})$$
(3.36)

and I_2 is an identity matrix size 2×2 .

Hierarchical Structure in the Mixture Probabilities

The previous steps give the methodology to analyze our model without a hierarchical specification on the mixture probabilities π_k . As discussed in this chapter, the model can be easily expanded to include a hierarchical specification on the mixture probabilities. As before, we will assume that π_{ki} differs across products but here we test if a multinomial probit specification that depends on **Z** is useful to explain their heterogeneity. For that we need to define first *K* latent variables for each product *i*

$$y_{ki}^* \sim N(Z_i'\delta_k, 1) \tag{3.37}$$

where $\delta_1 = 0$ for identification. Product i belongs to mixture *m* if y_{mi}^* is the largest of all y_{ki}^* $k = 1, \ldots, K$. Given (3.37), we can write the conditional distribution of y_{mi}^* given the other latent utilities (-m), denoted as $y^*_{-m,i}$, as follows:

$$p(y_{mi}^{*}|y_{-m,i}^{*},\theta,\tau_{i},S_{i}) = p(y_{mi}^{*} > \max(y_{-m,i}^{*})) \times p(\mathbf{P}_{i},S_{i} = m,\tau_{i}|\theta)$$

+
$$p(y_{mi}^{*} < \max(y_{-m,i}^{*})) \times p(\mathbf{P}_{i},S_{i} = m^{*},\tau_{i}|\theta)$$
(3.38)

where $m^* = \underset{m \neq k}{\operatorname{argmax}}(y_{ki}^*)$. Based on (3.38) we can apply the inverse cdf technique to draw y_{mi}^* from its full conditional distribution. Note that in this specification the indicator variable S_{ki} is determined based on y_{mi}^* and the δ_m parameters can be obtained from a normal with mean $(Z'_i Z_i)^{-1} Z'_i \delta_k$ and variance $(Z'_i Z_i)^{-1}$ for $m = 2, \ldots, K$.

Posterior Predictive Density

We used two measures to compare predictive performance in Table 3.5: the root mean squared error and the log of the posterior predictive density for observations after t = 7. The predictive density $\log(p(P_i(7),...,P_i(T)|P_i(1),...,P_i(6)))$ is defined as:

$$\log \int \int \int p(P_i(7),...,P_i(T)|P_i(1),...,P_i(6),S_i,\tau_i,\theta) \times p(S_i,\tau_i,\theta|P_i(1),...,P_i(6))dS_id\tau_id\theta \quad (3.39)$$

That is, we compute the log of the density for the forecast sample given the six observations included in the model and the posterior of all model parameters given these latter observations. The posterior predictive density can easily be obtained from the MCMC output by taking the log of the average out-of-sample likelihood over all draws.

Chapter 4

Random Coefficient Logit Models for Large Datasets

We present an approach for analyzing market shares and products' price elasticities based on large datasets containing aggregate sales data for many products, several markets and for relatively long time periods. We consider the recently proposed Bayesian approach of Jiang et al Jiang, Renna, Machanda, Puneet and Peter Rossi, 2009. Journal of Econometrics 149 (2) 136-148] and we extend their method in four directions. First, we reduce the dimensionality of the covariance matrix of the random effects by using a factor structure. The dimension reduction can be substantial depending on the number of common factors and the number of products. Second, we parametrize the covariance matrix in terms of correlations and standard deviations, like Barnard et al. Barnard, John, McCulloch, Robert and Xiao-Li Meng, 2000. Statistica Sinica 10 1281-1311] and we present a Metropolis sampling scheme based on this specification. Third, we allow for long term trends in preferences using time-varying common factors. Inference on these factors is obtained using a simulation smoother for state space time series. Finally, we consider an attractive combination of priors applied to each market and globally to all markets to speed up computation time. The main advantage of this prior specification is that it let us estimate the random coefficients based on all data available. We study both simulated data and a real dataset containing several markets each consisting of 30 to 60 products and our method proves to be promising with immediate practical applicability.

4.1 Introduction

A growing number of scholars is developing estimation methods for random coefficient logit models based on aggregate sales data. Currently, the estimation methods are based on the generalized method of moments [GMM], as in Nevo (2001) and Berry et al. (1995) (hereafter BLP), or based on likelihood or Bayesian approaches, as in Jiang et al. (2009) (hereafter Bayesian BLP or BBLP), Yang et al. (2003), and Park and Gupta (2009). The choice of the estimation method depends on the modeling assumptions regarding aggregate demand shocks, consumer heterogeneity, stability of preferences, price endogeneity and on the size and type of data available.

Recent Bayesian and maximum likelihood-based approaches have been successfully applied to data containing relatively long time series of weekly data (ranging from one to six years) concerning a small number of products (usually less than 6 products) sold in a single market (Jiang et al., 2009; Musalem et al., 2006; Yang et al., 2003). The GMM approach has been applied to similar sized data, like in Goeree (2008). Recently, Berry and Pakes (2007) use GMM and apply an extension of the BLP model to data consisting of both small (between 2 and 10) and large (100) number of products. The extension of Berry and Pakes (2007) is mainly focused on relaxing the assumption of non-zero demand shocks specifically when the market is saturated with many products. Their specification of null demand shocks may decrease the precision of the BLP contraction mapping and they present new complementary routines that overcome this issue.

One of the most challenging aspects for all methods is the estimation of the underlying distribution of the random effects that describe individual level consumer heterogeneity. As only aggregate data is available, the heterogeneity needs to be identified based on switching patterns. The simulation results of Jiang et al. (2009) suggest that their Bayesian method performs well and makes a more efficient use of the data relative to a GMM estimator. Nonetheless, today still little is known about the scalability (that is the performance and adaptability) of current methodologies to settings with many products and markets.

In this chapter we investigate the scalability of the Bayesian approach proposed by Jiang et al. (2009) and we extend their method in four directions. First, we propose a factor structure for the covariance matrix of the random effects. We will assume that the covariance matrix between J products can be modeled by a group of K factors, where the factor loadings are based on observable characteristics. Such a structure helps to keep the dimension of the heterogeneity structure under control. That is, we make the same distributional assumptions as in Jiang et al. (2009) regarding the heterogeneity and aggregate demand shocks but we strongly reduce the dimension of the covariance matrix. This reduction will be especially important in applications with a large number of products.

Second, we specify the covariance matrix following Barnard et al. (2000) as a function of correlations and standard deviations and we propose a Metropolis sampling scheme based on this parametrization. This parametrization has two main advantages. A technical advantage is that splitting the covariance in variances and correlations allows for a more efficient sampling scheme. A practical advantage is that the correlation structure of the heterogeneity itself may be most informative for managers.

The third extension in our approach is that we allow for time variation in preferences. Preference fluctuations are likely to occur over long periods of time and over seasons. In the currently considered Random Coefficient Logit Models such developments are often ignored. One exception we are aware of is Chintagunta et al. (2005), who show that allowing for time variation in preferences is beneficial to reduce both the uncertainty regarding brand preferences and the uncertainty regarding the sensitivity of products' shares to marketing efforts.

Finally, we consider an attractive combination of priors applied to each market and globally to all markets. This prior specification let us analyze all data simultaneously and it facilitates the estimation of the underlying distribution of the random coefficients based in all the data.

The Bayesian approach we use in this chapter allows for an efficient implementation of the four extensions mentioned above. One main advantage of the Bayesian approach over simulated maximum likelihood and GMM is that inference over any function of the parameters is straightforward because we obtain the posterior distribution of all parameters as the MCMC output. This for example allows for a straightforward assessment of the uncertainty in (cross) price elasticities. A second main advantage of the Bayesian approach is that we can incorporate efficient sampling of time-varying parameters alongside the other model parameters. Chintagunta et al.

(2005) use MLE and specify brand-specific time-fixed effects to account for time variation in preferences. Their specification of brand and time-specific fixed effects is an attractive formulation but the number of fixed effects to estimate may increase rapidly as the number of brands and time periods increases. As Chintagunta et al. (2005), we allow for time-variation in preferences but we use the simple and efficient simulation smoother of Durbin and Koopman (2002) to sample the time-varying parameters. The smoother is flexible because it let us reduce the model to the setting where brand preferences are fixed in time and this reduction may depend on the model's parameter estimates or it can easily be specified a priori.

We illustrate our approach using both simulated data and a real dataset that contains sales data for more than 20 markets each with a different, large, number of products and brands. The remainder of the chapter is structured as follows. In the next section we discuss the model. Next we present the Bayesian inference (some technical details are discussed in the appendix). Section 4.4 shows the results of a simulation experiment. In Section 4.5 we show detailed results of the application of the model to actual data. We conclude the chapter with a discussion.

4.2 Augmented Bayesian BLP Model

In this section we present our approach and we discuss how we augment the BBLP model in the directions discussed earlier. First in subsection 4.2.1 we present the model specification. Next in subsection 4.2.2 we discuss the share inversion method and the integration of the share function.

4.2.1 Model Specification

Consider consumers who make purchases from a set of J products during T time periods in M different markets. In general not all products will be available in all markets. We will use the letter J to refer to the total number of unique products available across all markets. \mathcal{J}^m denotes the set of products that are available in market m. The size of this set, that is, the number of products available in market m is denoted by J^m . In each period a consumer in market m can either choose to purchase one of the products in \mathcal{J}^m or choose an outside good, that is, he buys a product outside the set \mathcal{J}^m .

The purchase behavior of individual i in market m is based on utility maximization. We assume that the (latent) utility for consumer i for product j at time t in market m (denoted by u_{ijt}^m) contains three parts, (i) an "explained" part (w_{ijt}^m) , (ii) a market level aggregate demand shock (η_{jt}^m) , and (iii) an individual level random effect (ϵ_{ijt}^m) , that is, we specify

$$u_{ijt}^{m} = w_{ijt}^{m} + \eta_{jt}^{m} + \epsilon_{ijt}^{m}, \qquad j \in \mathcal{J}^{m}, t = 1, \dots, T.$$
 (4.1)

We make the standard assumption of a type-I extreme value distribution for ϵ_{ijt}^m and we assume $\eta_{jt}^m \sim N(0, \tau_m^2)$. We use a factor structure to further model w_{ijt}^m , that is we use

$$w_{ijt}^m = f_{it}^{m'} \lambda_{jt}^m, \qquad j \in \mathcal{J}^m, t = 1, \dots, T,$$
(4.2)

where f_{it}^m denotes an individual-specific K^m dimensional dynamic factor, and λ_{jt}^m is a $(K^m \times 1)$ vector containing the factor loadings for product j in market m. The factor loadings are based on observable product characteristics, such as, packaging and brand name, but also (log) price and promotional indicators may be part of the factor loading vector. In general λ_{jt}^m will contain constant as well as time-varying elements. In principle the same factors will be used in all markets, however, in some cases some factors may not be present in a market. For example, a particular package may not yet be available in a market. Therefore we need to specify the number factors to be dependent on the market.

The factor f_{it}^m gives the importance of a particular product characteristic for individual *i* in market *m* at time *t*. We split this factor into a time-varying part, which is the same across the population, and a heterogeneous part, which is constant over time, that is,

$$f_{it}^m = \bar{f}_t^m + v_i^m, \text{ where } v_i^m \sim \phi(0, A^m \Psi A^{m\prime}), \tag{4.3}$$

where Ψ denotes the variance matrix of all individual level random effects and A^m denotes a selection matrix. This matrix selects the rows and columns of the variance matrix that correspond to factors that are relevant for market m. The matrix A^m can be obtained by deleting all rows from the K dimensional identity matrix that correspond to irrelevant factors. Note that the variance of the random effects is in principle common across markets. Together with the factor loadings in λ_{jt}^m the covariance matrix $A^m \Psi A^{m'}$ gives a flexible but parsimonious specification for the variance structure of the preference heterogeneity.

Note that we can write the covariance matrix of the utilities for all products in market m, call this matrix Σ^m , as a function of Ψ , the selection matrices A^m and the factor loadings Λ^m_t where $\Lambda^m_t = \{\lambda^m_{jt}\}_{j \in \mathcal{J}^m}$. That is,

$$\Sigma^m = \Lambda_t^{m\prime} A^{m\prime} \Psi A^m \Lambda_t^m. \tag{4.4}$$

Next we assume a particular law of motion for \bar{f}_t^m , the common dynamic component of the factor. We use the state space specification

$$\bar{f}_{t+1}^m = \Gamma_t^m \bar{f}_t^m + \Pi_t^m \omega_t^m, \tag{4.5}$$

where $\omega_t^m \sim N(0, \Omega^m)$ and Γ_t^m is a known matrix. In the state space literature, Ω^m and Γ_t^m are usually set to be diagonal. Furthermore, if we additionally restrict the k-th diagonal element of Γ_t^m to be 1, we obtain a random walk for the k-th factor, that is, $\bar{f}_{kt+1}^m = \bar{f}_{kt}^m + \omega_{kt}^m$. If the variance of ω_{kt}^m is set to zero (or the corresponding element of Π_t^m), we obtain a constant specification for the factor, $\bar{f}_{kt}^m = \bar{f}_{k1}^m$. If we instead set the diagonal element of Γ_t^m to zero and the corresponding variance to a non-zero value, we obtain independent random effects over time, $\bar{f}_{k,t+1}^m = \omega_{kt}^m$.

We complete the model by normalizing the utility of the outside good to be 0. Based on the complete utility specification we can derive the purchase probabilities, or consumption share for individual i, s_{ijt}^m as a function of $(f_{it}^m \Lambda_t^m, \eta_t^m)$, where η_t^m is a vector with elements $\{\eta_{jt}^m\}_{j \in \mathcal{J}^m}$ and Λ_t^m is a vector with elements $\{\lambda_{jt}^m\}_{j \in \mathcal{J}^m}$. We use $\{x_{jt}^m\}_{j \in \mathcal{J}^m}$ to refer to a vector containing the elements $(x_{1t}^m, \ldots, x_{jt}^m)$ and we use $j \in \mathcal{J}^m$ to denote that the product index j is market-specific and hence it covers only the products in the set \mathcal{J}^m . Using the properties of the extreme value distribution we obtain

$$s_{ijt}^{m}(f_{it}^{m'}\Lambda_{t}^{m},\eta_{t}^{m}) = \frac{\exp(f_{it}^{m'}\lambda_{jt}^{m}+\eta_{jt}^{m})}{1+\sum_{h\in\mathcal{J}^{m}}\exp(f_{it}^{m'}\lambda_{ht}^{m}+\eta_{ht}^{m})}.$$
(4.6)

The overall market share, denoted by s_{jt}^m , of product j and time t in market m, measured over the entire population, is obtained by integrating $s_{ijt}^m(f_{it}^m'\Lambda_t^m,\eta_t^m)$ over the individual-specific parameters in f_{it}^m . Therefore we have that

$$s_{jt}^{m} = \int \frac{\exp(f_{it}^{m'}\lambda_{jt}^{m} + \eta_{jt}^{m})}{1 + \sum_{h \in \mathcal{J}^{m}} \exp(f_{it}^{m'}\lambda_{ht}^{m} + \eta_{ht}^{m})} \phi(f_{it}^{m}; \bar{f}_{t}^{m}, A^{m}\Psi A^{m'}) df_{it}^{m}$$
(4.7)

If we use $f_{it}^m = \bar{f}_t^m + v_i^m$ and $v_i^m \sim \phi(0, A^m \Psi A^{m\prime})$ we can write equation (4.7) as

$$s_{jt}^{m} = \int \frac{\exp(\mu_{jt}^{m} + \lambda_{jt}^{m'} v_{i}^{m})}{1 + \sum_{h \in \mathcal{J}^{m}} \exp(\mu_{ht}^{m} + \lambda_{ht}^{m'} v_{i}^{m})} \phi(v_{i}^{m}; 0, A^{m} \Psi A^{m'}) dv_{i}^{m},$$
(4.8)

where $\mu_{jt}^m = (\bar{f}_t^m)' \lambda_{jt}^m + \eta_{jt}^m$. Note that the share s_{jt}^m inherits randomness only from the term η_{jt}^m as we integrate over v_i^m .

Following Jiang et al. (2009) we denote the relationship between the shares vector $s_t^m = \{s_{jt}^m\}_{j \in \mathcal{J}^m}$ and the vector with aggregate demand shocks η_t^m in (4.7) as

$$s_t^m = h(\eta_t^m | \Lambda_t^m, \bar{f}_t^m, \Psi).$$

$$(4.9)$$

Based on the relation in (4.9) and the distribution of η_t^m , the joint density of the shares at time t is

$$\pi(s_t^m | \Lambda_t^m, \bar{f}_t^m, \Psi, \tau_m^2) = \phi(h^{-1}(s_t^m | \Lambda_t^m, \bar{f}_t^m, \Psi) | 0, \tau_m^2) | J_{s_t^m \to \eta_t^m} |^{-1},$$
(4.10)

for t = 0, ..., T and m = 1, ..., M and where we use $\pi(.)$ to denote a generic density and $\pi(y|x)$ the density of y given x. In addition, the Jacobian $J_{s_t^m \to \eta_t^m}$ is defined as the $(J^m \times J^m)$ matrix with elements

$$\partial s_{jt}^{m} / \partial \eta_{kt}^{m} = \begin{cases} -\int s_{ijt}^{m} s_{ikt}^{m} \phi(v_{i}^{m}; 0, A^{m} \Psi A_{m}') dv_{i}^{m} & \text{if } k \neq j \\ \int s_{ijt}^{m} (1 - s_{ikt}^{m}) \phi(v_{i}^{m}; 0, A^{m} \Psi A_{m}') dv_{i}^{m} & \text{if } k = j, \end{cases}$$
(4.11)

where the arguments of the functions s_{ijt}^m and s_{ikt}^m are dropped for convenience, see (4.8) and $j, k \in \mathcal{J}^m$. Given equation (4.10) the joint conditional density for the shares, or the likelihood,

for market m is given by

$$\pi(s^m | \Lambda^m, \bar{f}^m, \Psi, \tau_m^2) = \prod_{t=1}^T \pi(s_t^m | \lambda_t^m, \bar{f}_t^m, \Psi, \tau_m^2),$$
(4.12)

where $\Lambda^m = (\Lambda^m_1, \dots, \Lambda^m_T)$, $s^m = (s^m_1, \dots, s^m_T)$ and $\bar{f}^m = (\bar{f}^m_1, \dots, \bar{f}^m_T)$.

Two difficulties in this model are the inversion of the share function h() in equation (4.9) and the evaluation of the integrals in equations (4.8) and (4.11). The inversion and the integration are required to obtain the aggregate shocks η_t^m and hence to evaluate the density in equation (4.10). We discuss these two issues next.

4.2.2 Share inversion method and integral approximation

To calculate the joint density in (4.12) we need to take two hurdles. First we need to solve the integrals in (4.8) and (4.11). Next, we need to obtain the inverse of the function h() in (4.9).

We apply the contraction mapping of Berry et al. (1995) to obtain the inverse in terms of μ_{jt}^m for all necessary m, j and t. Within this procedure we need to calculate the market shares given μ_{jt}^m, Λ_t^m and $\Psi, j \in \mathcal{J}^m, t = 1, ..., T$ and m = 1, ..., M by integrating equation (4.8) with respect to v_i . We numerically approximate this integral by averaging over H draws from the distribution of v_i that is $N(0, A^m \Psi A^{m'})$. Jiang et al. (2009) report that H ranges from 20 to 50 in previous literature and they show that their Bayesian estimator has the same performance for H = 50 and H = 200. However, in our case we may need more draws as we develop the model for many more parameters.

A common approach to obtain each of the H draws of v_i is based on the product of the Cholesky decomposition of $A^m \Psi A^{m'}$ and draws from a standard normal, that is $v_i^d = (A^m \Psi A^{m'})^{1/2} \zeta^d$ where $\Sigma^{1/2}$ denotes the Cholesky decomposition of Σ and $\zeta^d \sim N(0, \mathbf{I})$ for $d = 1, \ldots, H$, where \mathbf{I} denotes an identity matrix. A more efficient approximation of the integral may be obtained by using a quasi-random scheme to generate the ζ^d . Train (2003, chap. 9, page 236) suggests scrambled Halton sequences for integrals of large dimensions and his suggestion, we believe, is motivated by the same family of logit models that we are concerned with here. In Figure 4.1 we compare the integration results based on scrambled Halton sequences versus the integration results based on regular normal draws. We consider the scenario where the parameters are known and we use the approximation method discussed above to obtain the market shares. In the top panel we report the performance when the integral has only three dimensions and in the lower panel we report the performance when the integral has 30 dimensions. In both panels we report the market share for only one of the products. This simple exercise suggests that the market shares are much better approximated by integrating with Halton draws regardless of the dimension of the integral. For large dimensions the approximation of the normal draws seems to converge to the approximation of the Halton draws after the number of draws (H) is higher than 400 while the approximation based on Halton draws performs well for H > 100.

4.3 Bayesian Inference

In this section we discuss the priors we choose to complete the model specification. Specifically, we present in subsection 4.3.1 a prior for the matrix Ψ that is simple to calibrate when analyzing many products and markets and at the same time the prior will let us treat the scale and the correlation structure of Ψ separately. Next in subsection 4.3.2, we discuss the market-specific priors. Finally in subsection 4.3.3, we discuss the MCMC sampling scheme.

4.3.1 Prior and Structure for Ψ

Jiang et al. (2009) specify the covariance matrix Σ^m in terms of the unique elements of its Cholesky root. They set $\Sigma^m = U'U$ where

$$U = \begin{pmatrix} e^{r_{11}} & r_{12} & r_{13} & \dots & r_{1J} \\ 0 & e^{r_{22}} & r_{23} & \dots & r_{2J} \\ 0 & 0 & \ddots & \ddots & \vdots \\ \vdots & \vdots & \ddots & e^{e_{J-1,J-1}} & r_{J-1,J} \\ 0 & 0 & \dots & 0 & e^{r_{JJ}} \end{pmatrix},$$
(4.13)

and they choose to set separate normal priors for the diagonal and off-diagonal elements of U. That is, Jiang et al. (2009) set $r_{jj} \sim N(0, \sigma_j^2)$ for the diagonal elements $j = 1, \ldots, J$ and $r_{jk} \sim N(0, \sigma_{od}^2)$ for the off-diagonal elements $j \neq k$. Note that Jiang et al. (2009) deal with only one market m and that in our approach we model the heterogeneity through Ψ (that is a $K \times K$ matrix) and not through Σ^m (that is a $J \times J$ matrix) where K is the number of factors while J is the number of products.

This prior specification enforces the positive-definitiveness of Σ^m and in addition the priors on the r_{ik} elements used by Jiang et al. (2009) are symmetric and this matched well with the random walk Metropolis Hastings [MH] sampling scheme they propose to sample the rparameters. A second advantage of Jiang et al. (2009) prior is that it can be set to be relatively uniform on the correlation range (-1, 1). Overall, this specification is attractive and simple but it also has a number of shortcomings. First, Jiang et al. (2009) note that to obtain a plausible (implied) prior on the variances in Σ^m the prior variances σ_j^2 should be decreasing with j and they provide a particular relation between σ_i^2 and j. However, the prior in one of the elements of (4.13) may affect many of the elements in Σ^m and this complicates the prior interpretation. specially when J is large. Second, this prior imposes a correlation structure simultaneously with the overall scale of the heterogeneity. Other studies point out that it may be relatively hard to identify the heterogeneity (Bodapati and Gupta, 2004) and the uncertainty related to the Σ^m elements is therefore usually large, see for example Jiang et al. (2009) and Musalem et al. (2006). However, we do not know if the large uncertainty reported in previous studies is due to the uncertainty on the overall scale of Σ^m or if it is due to the correlation structure in Σ^m . Finally, it is well known that the correlation structure of Σ^m is very important in order to obtain different substitution patterns far from the IIA assumption of the logit. Therefore, we would like to use a prior that can let us deal with the scale and correlations separately.

Finally, changing one element of U may lead to a very different Σ^m . This fact makes the implementation of an efficient MH sampler difficult if J is large. More precisely, in their MH scheme Jiang et al. (2009) choose to draw the candidate elements of U from a multivariate normal that is calibrated based on a short chain of their model MCMC output. The length of the chain needed for the calibration of the multivariate normal needs to be longer when

the number of dimensions is large. When dealing with large dimensions, the step size in the random walk MH sampling, for each of the r_{jk} elements, needs to be set smaller as the dimension increases in order to keep a good acceptance rate in the estimation algorithm. Although this last issue always arises whenever many elements are updated simultaneously, here it can be more dramatic as "local" changes in U lead to "global" changes in Σ^m .

Summarizing, we would like to use a prior specification that is simpler to calibrate when dealing with large dimensions and at the same time we like to treat the scale and the correlation structure of the heterogeneity separately.

We choose to use the prior specification of Barnard et al. (2000) for Ψ . We define $\Psi = DRD$ where D is a diagonal matrix with K elements (standard deviations) and R is a $K \times K$ correlation matrix. For the variances in D we set the prior $\log(\operatorname{diag}(D)) \sim N(0, \Delta)$. Formulating a prior on R is not straightforward because we need a prior that deals with all the elements of R and the restrictions on them. We need to assure the positive definitiveness of R, the range of its elements must be (-1, 1) and all the elements together should satisfy triangularity restriction inherent to any correlation matrix. In addition, we need to update all the elements of R simultaneously to ease the computational burden. However, based on any variance matrix Σ one can obtain the corresponding correlation matrix by standardization. Hence, we assume $R = f_c(S)$ and we specify an Inverted Wishart prior for S with parameters (G, v). The function $f_c()$ transforms a covariance matrix to a correlation matrix. The location matrix G is set such that the expected value of S is an identity matrix; that is $G = (v - K - 1) \times \mathbf{I}$, v is the degrees of freedom of the Inverted Wishart and \mathbf{I} is an identity matrix of size K. Note that our variance matrix Ψ is now actually a function of D and S. In the MCMC sampling below we will actually sample these two matrices.

Barnard et al. (2000) set a prior directly on R while we set a prior on R implicitly by the prior on S. The main reason why we deviate from them is that evaluation of the posterior is very costly in our application and hence we need to use a proposal for S that updates all the correlations in R in a single step. In contrast, the computation time of the application in Barnard et al. (2000) allows for a relatively fast element by element update of the matrix R.
In Figure 4.2 we report the implied correlation distributions for two different degrees of freedom and for two elements of a Ψ matrix of size 10×10 . The implied correlations can be set to be roughly uniform on the (-1, 1) interval depending on the degrees of freedom set on the Wishart distribution. Therefore, the implied correlations of this prior are very similar to the implied correlations of the specification used by Jiang et al. (2009). However, the parameters in our suggested priors are easier to interpret.

4.3.2 Market-Specific Priors and Joint Posterior

We presented the likelihood for each of the m markets in equation (4.12) and we presented the structure and prior for Ψ in the previous section. This variance matrix applies to all markets. What is left to specify are the priors for τ_m^2 , Ω^m and the initial state distributions for all common dynamic factors f_t^m . In addition the matrix Γ_t^m needs to be defined. We define $f_1^m \sim N(0, P^m)$ where we set P^m reasonably large and non-informative, $P^m = 100$ I for all m. We assume $\Omega^m = \sigma_m^2 \mathbf{H}_m$ where \mathbf{H}_m is a diagonal matrix of size J^m and $\sigma_m^2 \sim v_o s_o^2/\chi_{v_o}^2$. The diagonal elements of \mathbf{H}_m are equal to one for the factors f_t^m that are time-varying and equal to zero for the factors that are fixed over time. We set Γ_t^m equal to an identity matrix \mathbf{I}_m of size J^m and for τ_m^2 we do not set any prior.

The joint posterior is proportional to the product of the likelihood and priors for each market times the prior distribution of Ψ that apply to all markets. Note that the factor loadings are assumed to be given, as they represent observed product characteristics. The posterior becomes

$$\pi(\bar{f}^{*}, \tau_{*}^{2}, \sigma_{*}^{2}, D, S|s^{*}, \Lambda) \propto \pi(\log(\operatorname{diag}(D)); 0, \Delta)\pi(S; G, v) \times \left(\prod_{m} \pi(s^{m}|\bar{f}^{m}, \Psi, \tau_{m}^{2})\pi_{m}(\bar{f}^{m}_{1}; 0, P^{m}) \left[\prod_{t=1}^{T-1} \pi(\bar{f}^{m}_{t+1}|\bar{f}^{m}_{t}, \sigma_{m}^{2}\mathbf{I}_{m})\right] \pi(\sigma_{m}^{2}; v_{o}, s_{o}^{2})\right), \quad (4.14)$$

where $s^* = (s_1, \ldots, s_M), \ \bar{f}^* = (\bar{f}^1, \ldots, \bar{f}^M), \ \tau^2_* = (\tau^2_1, \ldots, \tau^2_M), \ \sigma^2_* = (\sigma^2_1, \ldots, \sigma^2_M), \ \Psi = Df_c(S)D.$

In addition, the priors for f_1^m , σ_m^2 , D and S are defined as follows

$$\pi(\bar{f}_{1}^{m}; 0, P^{m}) \sim N(\bar{f}_{1}^{m}; 0, P^{m})$$

$$\pi(\sigma_{m}^{2}; v_{o}, s_{o}^{2}) = (\sigma^{2})^{-(v_{o}/2+1)} e^{-v_{o}s_{o}^{2}/2\sigma^{2}}$$

$$\log(\operatorname{diag}(D)) \sim N(0, \Delta)$$

$$\pi(S; G, v) \propto \frac{|G|^{v/2}}{|S|^{(v+K+1)/2}} e^{-1/2tr(S^{-1}G)}$$
(4.15)

4.3.3 MCMC algorithm

The approach we follow is a combination of the sampler proposed in Jiang et al. (2009) with simulation smoother of Durbin and Koopman (2002) and a Metropolis Hastings sampler for Ψ . We use the following steps: (i) conditional on Ψ we use the contraction mapping to obtain the (implied) μ_{jt}^m , for $m = 1, \ldots, M$, $j \in \mathcal{J}^m$, $t = 1, \ldots, T$; (ii) conditional on Ψ (and μ_{jt}^m) we use the simulation smoother to sample \bar{f}^* , the μ_{jt}^m values appear as dependent variables in this smoother; (iii) conditional on \bar{f}^* and all μ_{jt}^m we sample τ_*^2 and σ_*^2 ; (iv) finally we use a Metropolis Hastings sampler to draw the elements of D and S which determine Ψ .

More specifically, we use the following three set of conditionals

$$\bar{f}^{*} | \Psi, \sigma_{*}^{2}, \tau_{*}^{2}, s^{*}, \Lambda
\sigma_{*}^{2}, \tau_{*}^{2} | \Psi, \bar{f}^{*}, s^{*}, \Lambda
D, S | \sigma_{*}^{2}, \tau_{*}^{2}, \bar{f}^{*}, s^{*}, \Lambda.$$
(4.16)

We draw the first set of conditionals using the simulation smoother of Durbin and Koopman (2002). That is, given μ_{jt}^m for all m, j and t we can draw the parameters of the following measurement and state equations

$$\mu_t^m = \Lambda_t^{m'} \bar{f}_t^m + \eta_t^m \qquad \text{with} \qquad \eta_{jt}^m \sim N(0, \tau_m^2)$$

$$\bar{f}_{t+1}^m = \Gamma_t^m \bar{f}_t^m + \Pi_t^m \omega_t^m \qquad \text{with} \qquad \omega_t^m \sim N(0, \sigma_m^2 \mathbf{I}_m),$$
(4.17)

where μ_t^m is defined as the $(J_m \times 1)$ vector with elements μ_{jt}^m , $j \in \mathcal{J}^m$. This specification is attractive because we can set some of the common factors \bar{f}_t^m to be fixed in time while others can remain time-varying. This is done simply by setting some of the elements in the diagonal matrix Π_t^m equal to zero. The simulation smoother of Durbin and Koopman (2002) gives a draw from the joint posterior of \bar{f}_t^m , for t = 1, ..., T. For details we refer to their paper. Conditional on μ_t^m and \bar{f}^m sampling the variances is straightforward. Given our priors they can be sampled from Inverted Gamma distributions. That is, $\tau_m^2 \sim IG(n_\tau^m, s_\tau^2)$ and $\sigma_m \sim IG((v_o + n_\sigma^m), (s_\sigma^2 + s_o^2))$. The n_τ^m are the number of observations available for the measurement equation at market mfor m = 1, ..., M and s_τ^2 is the sum of squared residuals of the measurement equation. The n_σ^m is the number of observations available in the state equation at market m and s_σ^2 is the sum of squared residuals of the measurement equation. The v_o and s_o^2 are the parameters of the prior for the variance of the state equation, see the priors in equation (4.15).

For the third set of conditionals we use a Metropolis Hastings algorithm. For the (log of the) elements of D we use a standard random walk as candidate distribution. For comparison with the second part of this step we write the proposal as

$$\log(\operatorname{diag}(D^{\operatorname{candidate}})) \sim N(\log(\operatorname{diag}(D^{\operatorname{current}})), \zeta^{2}\mathbf{I}).$$
(4.18)

For S we also propose a random walk candidate distribution. However, for efficiency in the total sampler we which to have a candidate that can generate matrices close to the current value. As a candidate distribution we use an inverted Wishart distribution which has the current value as expected value, that is,

$$S^{\text{candidate}} \sim IW((v_1 - K - 1)S^{\text{current}}, v_1).$$

$$(4.19)$$

We choose v_1 and ζ^2 to achieve between 20% and 50% acceptance rate in the Metropolis steps. In the MCMC we use two Metropolis steps to update D and S separately.

To sample D and S we evaluate the model posterior in equation (4.14) in two Metropolis steps, the first for D and the second for S. We set

$$D^{new} = D^{cand} \text{ with probability } \min\{\frac{p^*(D^{cand}|S,\bar{f}^*,\tau_*^2,\sigma_*^2,s^*,\Lambda)}{p^*(D^{prev}|S,f^*,\tau_*^2,\sigma_*^2,s^*,\Lambda)},1\},$$
(4.20)

and we set

$$S^{new} = S^{cand} \quad \text{with probability} \quad \min\{\frac{p^*(S^{cand}|D, f^*, \tau_*^2, \sigma_*^2, s^*, \Lambda)}{p^*(S^{prev}|D, f^*, \tau_*^2, \sigma_*^2, s^*, \Lambda)}, 1\}.$$
(4.21)

The candidate and previous posterior density in equation (4.20) are given by

$$p^{*}(D^{cand}|S, \bar{f^{*}}, \tau_{*}^{2}, \sigma_{*}^{2}, s^{*}, \Lambda) = \\\pi(\log(\operatorname{diag}(D^{cand})); 0, \Delta) \times \prod_{m} \pi(s^{m}|\Lambda^{m}, \bar{f^{m}}, \Psi^{cand}, \tau_{m}^{2}), \quad (4.22)$$

and by

$$p^{*}(D^{prev}|S, \bar{f^{*}}, \tau_{*}^{2}, \sigma_{*}^{2}, s^{*}, \Lambda) = \pi(\log(\operatorname{diag}(D^{prev})); 0, \Delta) \times \prod_{m} \pi(s^{m}|\Lambda^{m}, \bar{f^{m}}, \Psi^{prev}, \tau_{m}^{2}). \quad (4.23)$$

where $\Psi^{cand} = D^{cand} f_c(S) D^{cand}$ and $\Psi^{prev} = D^{prev} f_c(S) D^{prev}$ while the terms in the Metropolis step in equation (4.21) are given by

$$p^{*}(S^{cand}|D, \bar{f}^{*}, \tau_{*}^{2}, \sigma_{*}^{2}, s^{*}, \Lambda) = \\\pi(S^{cand}; G, v) \times \pi(S^{prev}; S^{cand}, v_{1}) \times \prod_{m} \pi(s^{m}|\Lambda^{m}, \bar{f}^{m}, \Psi^{cand}, \tau_{m}^{2}), \quad (4.24)$$

and by

$$p^{*}(S^{prev}|D, \bar{f}^{*}, \tau_{*}^{2}, \sigma_{*}^{2}, s^{*}, \Lambda) = \\ \pi(S^{prev}; G, v) \times \pi(S^{cand}; S^{prev}, v_{1}) \times \prod_{m} \pi(s^{m}|\Lambda^{m}, \bar{f}^{m}, \Psi^{prev}, \tau_{m}^{2}), \quad (4.25)$$

where now $\Psi^{cand} = Df_c(S^{cand})D$, $\Psi^{prev} = Df_c(S^{prev})D$ and $\pi(s^m|\Lambda^m, \bar{f}^m, \Psi, \tau_m^2)$ is defined in equation (4.12). Note that the proposal distribution for S is not symmetric and hence its distribution ($\pi(S; S', v_1)$) also appears in the acceptance probability. We use the proposal distributions in equation (4.18) and equation (4.19) to draw the candidate matrices D^{cand} and S^{cand} based on their previous values D^{prev} and S^{prev} , respectively. The Metropolis steps are very costly in terms of computation time in the MCMC algorithm. This is the only step in the algorithm where we need to use the BLP contraction mapping and where we need to evaluate the Jacobian in equations (4.10) and (4.11). Some time may be saved by jointly updating these matrices. However, the joint updating of D and S will not let us distinguish what is driving the acceptance rate in the Metropolis steps. Moreover, the separate updating of these matrices let us distinguish if a candidate matrix Ψ^{cand} ($\Psi = Df_c(S)D$) is rejected because of its correlation structure or because of its overall scale.

4.4 Simulation Experiment

We test our modeling approach on simulated data and in this section we discuss the data generation process and the results of the MCMC estimation procedure.

4.4.1 Data Simulation

In this section we describe how we create synthetic data and we consider a setting where we have data for many products and markets. This setting is not typical in the literature but it that corresponds with the setting that we deal with in the application.

We assume products are sold in 10 markets and we simulate 4 years of monthly data for each market. Each market will be assigned a specific number of products and these products will be assigned to 10 different brands.

All 10 brands are available in each market and we assign 5, 6, 8 or 10 products to each brand at each market. Hence, the number of products assigned to a brand varies per market and each market consists of a specific number of products. The probability of a brand to be assigned 5 or 6 products at each specific market is 90% while the probability of being assigned 8 or 10 products is 10%. That is, the expected number of products per brand is 5.85 and the expected number of products per market is 58.5. This is a large number of products relative to previous studies. For example, Jiang et al. (2009) and Yang et al. (2003) study one market that consists of 3 products and one outside good while Musalem et al. (2006) apply their model to a setting with four products and one outside good. The mean utilities (the μ_{jt}^m) for products are market-specific. The mean products' utilities are assumed to depend on the products' brands, the products' attributes and the products' prices and promotions. We define 5 attributes and we assign only one attribute per product and all attributes are equally likely per product. Note that this last assumption implies that each brand may have a certain number of products that share the same attributes. We set one of the attribute coefficients as the base and equal to 0 while the rest is generated from a normal distribution with mean 0 and variance 5.

Next, we define the price and promotion coefficients and these are -5 and -2, respectively, and these coefficients are the same across markets. The price series for each product follows a sine curve (with a very long cycle) plus normal noise with mean 2 and variance 1. To create the promotional series we use a uniform distribution with range (0, 1). We assume that there is a 30% chance of a promotion and the range of promotions is between 0 and 30%. That is, when we draw a promotional index value (from the uniform) lower than 0.70 then the promotional index is equal to 1 otherwise the promotion index is equal to the drawn value.

Attribute, price and promotions coefficients will be fixed in time while the brand coefficients will be time-varying. We generate 10 brand coefficients using the recursion in equation (4.17) and we set σ_m to be equal to 0.40. We set the initial values for the brand coefficients f_1^m based on a normal distribution with mean -3 and variance 0.16. We use the same recursion to generate the attribute, price and promotion coefficients and their initial value is assigned as we discussed in the previous paragraph. We further need to set Π_t^m to be a diagonal matrix with the first 10 elements equal to 1 and the remainder 6 elements of the diagonal are equal to 0. The Γ_t^m is equal to an identity matrix of size 16.

The factor loadings Λ_t^m will consist of brand and attribute dummies for all products at time t plus the products prices and promotions at time t. That is, Λ_t^m is a $J \times K$ matrix, J is the number of products and K is equal to 16 (the number of brands, attribute, price and promotion coefficients). Finally, we assume that the variance of product demand shocks τ_m^2 are equal to 0.8 for all m.

We use the specification of $\Psi = DRD$ to draw the random coefficients v_i^m . We first draw a matrix P based on a $IW(I_{16}, 21)$ and we set $R = f_c(P)$. The implied range of the correlations

in R goes from -1 to 1 but the extremes of the range are not common. Further, we assume the scale of the heterogeneity depends on both small and large elements with the purpose of checking whether their size affects their retrieval from the synthetic data. That is, we set $D^2 =$ (2, 2, 2, 2, 8, 8, 8, 8, 4, 4, 4, 4, 2, 2, 2, 2). Finally, we use 3000 draws to approximate the integral in equation (4.8) and we generate the draws of the random coefficients based on the Cholesky decomposition of Ψ and normal draws generated with scrambled Halton sequences. The 16 factors in Ψ are available at every market and therefore the A^m matrix is the same for all markets and it is equal to I_{16} .

4.4.2 MCMC Setup

We use a hybrid Metropolis Gibbs sampler to estimate the parameters of the model in equation (4.7). The sampler iterates over the conditionals in equation (4.16). The first set of conditionals concerns the \bar{f}^* . We set the prior on the initial values as $f_1^m \sim N(0, 100I)$ for all m and we use the simulation smoother of Durbin and Koopman (2002) to sample all elements of \bar{f}^* .

The second set of conditionals samples the variances of equation (4.17). We did not set any prior information on σ_m^2 and τ_m^2 for all m. Hence, $\tau_m^2 \sim IG(n_m, s_m)$ where n_m are the number of observations in the measurement equation in (4.17) and s_m are the sum of squared residuals in the same equation. In a similar fashion, $\sigma_m^2 \sim IG(n_m^s, s_m^s)$ where n_m^s are the number of observations and s_m^s are the sum of squared residuals of the state equation in (4.17).

The third set of conditionals concerns the sampling of the D and S matrices. We set the v parameter in the prior $\pi(S; \mathbf{I}, v)$ equal to 21 and we use $\Delta = 10\mathbf{I}$ in the normal prior of the log of the diagonal elements of D.

We use the proposal distributions in equation (4.18) and equation (4.19) to draw the candidate matrices D^{cand} and S^{cand} , respectively. In these proposals we set ζ^2 equal to 0.01 and v_1 equal to 10000. The large number in v_1 corresponds to steps of approximately 0.05 in the elements of the correlation matrix R where $R = f_c(S)$. We calibrated ζ^2 and v_1 to achieve an acceptance rate between 20% and 50% for both Metropolis steps.

We let the Gibbs-Metropolis sampler to run for 20 thousand iterations. However, we do oversampling of Ψ . We use 4 updates of S and one of D at every iteration. That is, we generate 100 thousand candidates for the matrix Ψ . The matrix Ψ contains 136 unique elements and our purpose with the oversampling is to let all these elements to move at larger steps at every iteration and let them adjust better to the rest of the model parameters drawn at every iteration. In this way, the oversampling may compensate for the small moving steps that we need to achieve a good acceptance rate in the Metropolis algorithm. Haran et al. (2003) also consider the oversampling of parameters to accelerate their computation in the MCMC algorithm.

4.4.3 Results of the Simulation Experiment

In Table 4.1 we present the posterior mean and the 99% Highest Posterior Density Region (HPDR) of the demand shocks for every market. The true value of τ_m^2 is equal to 0.66 for all markets. Note that we generated data for 10 markets. In most cases, the posterior mean is very close to its true value. The maximum absolute deviation of the posterior mean from the true value is approximately 0.06, see the $\tau_{m=6}^2$ that is equal to 0.580.

In Table 4.2 we present the posterior mean and HPDR of the variance term in the state equation (4.17), that is σ_m^2 . The true value of this parameter is 0.16 while in most cases the posterior mean is close to 0.12. That is, we are finding a small negative bias that is close to 0.04 for most cases.

In Figure 4.3 we present the estimates of the fixed coefficients (in circles) and the box-plots of their posterior distribution. Note that we specified 4 attribute coefficients and one price and promotion coefficient that vary across markets. That is a total of 50 coefficients in all markets. We see that for 30 out of the 50 coefficients the circles (true values) overlap with the position of their distribution in the box-plot. In the same figure, we see that there is a systematic positive bias in the posterior distribution of the price coefficients. The true value of the price coefficient is equal to -5 while the posterior distribution is higher than -5. In contrast, the posterior distribution of the promotion coefficients overlap with its true value (-2) for all markets.

In Figure 4.4 we present the distribution of the time-varying brand coefficients for the 5th market. We see that the overall time profile is well retrieved by the estimation algorithm. In most cases the true value is inside the 99% HPDR. The results for the other markets are very similar.

In Table 4.3 we report the posterior mean and HPDR for the elements of the D^2 matrix. We see that the 99% HPDR contains the true value for 7 out of the 16 elements. The deviation of the posterior mean from its true value, when the true value is not contained in the HPDR, may be as small as 1 or as large as 6 variance points. That is, we find large uncertainty regarding the scale of the heterogeneity driven by the random coefficients. Jiang et al. (2009), Musalem et al. (2006) and Yang et al. (2003) find similar levels of uncertainty.

In Figure 4.5 we report the 99% HPDR (in dashed lines) and the true value (solid line) of the 120 unique elements of the correlation matrix R ($f_c(S)$). We find that the HPDR contains the true value for 57 out of 120 elements, that is 47.5% of the elements. However, we find that the posterior mean of the correlations is on average 0.16 points far from its true level. Hence, our results suggest that the uncertainty regarding the scale of the heterogeneity (the elements of D) is much larger than the uncertainty in the elements of the correlation matrix R.

4.5 Empirical Application

In this section we apply our estimation approach to a real dataset and we analyze the substitution patterns between a large number of products. Next we provide a description of the data (in subsection 4.5.1), the modeling details (in subsection 4.5.1) and the estimation results (in subsection 4.5.3).

4.5.1 Data

Our dataset contains sales, price and promotion data for all the products of one supermarket food category. The data is monthly and it covers a period of four years and 18 different regions. Consumers at each region may have available a minimum of 25 up to a maximum of 65 products of 20 different brands. Each brand has its own positioning in terms of calories, taste and labeling while each brand may offer products of the same size and packaging. Therefore, we can describe each product in terms of its brand, size and packaging attributes and its price and promotion data. There are brands with similar attributes both in terms of calories and taste and in terms of packaging and size and these brands are usually produced by different companies. Our data contains products sold by all major companies at each market and very few firms compose the market.¹ Depending on the market, the size of the outside good varies from 20 up to a maximum of 50%. The calculation of the outside good share is region-specific and it varies according to the share of the closest and competing food categories.

4.5.2 Modeling Details and MCMC Setup

The MCMC setup for the application is very similar to the MCMC setup we use for the simulated data. An important distinguishing feature is that the matrix Ψ consists of 32 rows and columns. This number corresponds to 20 brands, 11 size and packaging attributes, one price and one promotion factor. We leave one attribute as reference and this results in 32 random coefficients. In the application the A^m matrices select the appropriate elements of the Ψ matrix relevant for the market m. That is, some attributes or brands may not be available in all markets.

We will assume that all coefficients are fixed with the exception of the brand coefficients that will be specified as time-varying. We use the priors in equation (4.15) where we set $P^m = 100$ I for all m, $v_o = 1$ and $s_o = 0.01$. The Δ matrix is equal to 25I and v = 35. We did not set a prior on τ_m^2 parameters. The proposal distributions in equation (4.18) and equation (4.19) have the parameters $v_1 = 30000$ and $\zeta^2 = 1/200$. This configuration achieves between 30% and 50% of acceptance rate in the Metropolis updates of S and D. We sample D and S separately in the same way as we did in the simulation experiment.

The matrix Π_t^m in equation (4.17) is set equal to an identity matrix of size K^m (K^m is the number of factors at each m) and we set some of its diagonal elements equal to 0 and these zeros correspond to the factors related to size, packaging and to price and promotions. The matrix Γ_t^m is of size J^m (the number of products available at market m) times K^m and it is also set to be an identity matrix.

We ran the MCMC chain for 50 thousand iterations and we discarded the first 10 thousand with a thin value of 20. The computation time was of approximately five days. The number of

 $^{^{1}}$ Because of our confidentiality agreement we can not reveal the companies names, brands or any other product or market information in the chapter.

draws that we used for approximating integrals was 200 and we use draws based on scrambled Halton sequences.

4.5.3 Estimation Results

We present the posterior mean and HPDR of the τ_m^2 parameters in Table 4.4. The uncertainty of the demand shocks is very large for six of the eighteen markets, see the τ_m^2 for *m* equal to 1, 2, 12, 13 and 17. The uncertainty in the demand shock for the remaining markets seems small relative to these six markets.

The posterior mean of HPDR of the σ_m^2 parameters can be read in Table 4.5. These are the variances of the time-varying coefficients and we see that they are very small as we expected. The variance of time-varying parameters in state space models is usually small (Fruhwirth-Schnatter, 2004) and this indicates slowly evolving factors.

In Table 4.6 we present the posterior mean and HPDR for the fixed coefficients at three markets. We notice that price and promotion coefficients have the expected negative signs. The promotional index is a number that takes a value between 0 and 1 and it indicates the percentage of the regular price level that is observed. We notice that the uncertainty related to the price coefficients varies across markets while at the same time they remain negative. The preference for size also varies per market and we find that for each market there are only two sizes with a positive posterior mean that may be larger than the base category. In Figure 4.6 we report the evolution of the time-varying brand coefficients. We report the time profiles of the time-varying factors relative to their starting point and their corresponding 99% HPDR. This transformation is useful to illustrate how some brands' preferences (measured by the time-varying factors) face large variations relative to their starting position, like brands C, E, F or L, while we see other brands like J or B with much smaller time variation. Note that this figure does not show the level uncertainty around the time-varying brand coefficients. Their level uncertainty, however, is similar to the uncertainty of the fixed coefficients. We see also that the coefficients for different types of packaging show significant time variation relative to their starting point, see the bottom row in Figure 4.6.

In Figure 4.7 we report the distribution of 60 elements of the Ψ matrix. The Ψ matrix size is 32×32 and therefore it contains 528 unique elements. We notice that the uncertainty varies per element but overall the uncertainty is relative small for most correlations. The element 39 in the lower panel has the largest uncertainty and its range goes from -0.75 up to 0 while there are other cases like the element 12 in the upper panel with very tight posterior distributions.

In Table 4.7 we present the posterior mean and the 99% HPDR of the matrix D^2 . Some of the elements of the matrix are retrieved with a lot of uncertainty. For example, the posterior mean of the D_9^2 is 4.64 but its HPDR includes values close to 10 while the posterior mean of D_{15}^2 is equal to 10.625 and its HPDR includes values as high as 22. These rest of the elements in the D^2 matrix, and the mayority, show a much smaller uncertainty relative to these high values in D_9^2 and D_{15}^2 . Previous studies, like Jiang et al. (2009), Musalem et al. (2006) and Yang et al. (2003), report simular range of both the scale of the heterogeneity and its uncertainty.

Finally, in Figure 4.8 and Figure 4.9 we present the own-price and cross price elasticities for the products in market 2. We computed the elasticities as we describe in the Appendix 4.A. The price elasticities have a range that goes from -1 to -3.5 while the cross-price elasticities range goes from 0 up to 1.6. In this last figure light (white) colors represent high values while darker (dark red) colors represent lower cross price elasticities. In Figure 4.9 we notice that many products respond to the price changes of a relatively small set of products. For example, a price change in the 10^{th} product affects almost all products in this market and their cross price elasticity is close to 1.66. Finally, we notice that substitution patterns (measured by cross price elasticities) are stronger among a small subset of products.

4.6 Conclusions

The estimation of aggregate share models based on the random coefficient logit specification presents different challenges. The scalability of models and estimation algorithms is one of these main challenges. Berry and Pakes (2007) is a recent paper with a similar concern as ours and that is the practical application of this family of models to larger and more comprehensive datasets. In this chapter we investigate the scalability of the BBLP approach and we successfully

applied our method to simulated data and to a relative large real dataset. It is large in terms of the number of products, brands and markets that it includes while it is still small relative to the time periods we have available.

Our specification is based on the recent advances of Jiang et al. (2009), Durbin and Koopman (2002) and Barnard et al. (2000). These advances all put together allow us to model time variation in preferences and to separate the uncertainty of the random coefficients in terms of their scale and in terms of their correlation. In addition, our mode specification combines global and market specific priors and this allows us pool information across several markets.

We believe that the uncertainty related to the random coefficients is a great challenge. In contrast with previous studies we report the uncertainty related to the correlation and the scale of the random coefficients separately. Our results point that the overall scale of the covariance matrix of the random coefficients may present a larger uncertainty relative to the uncertainty present in their correlation structure. This last result is an initial step towards the untangling and modeling of the sources of uncertainty in the random coefficients of the BBLP approach and we consider that this is a promising area for further research.

We present an approach that is the "augmented" version of the BBLP and it should be considered whenever there is a large dataset of market shares available for analysis. Large datasets, particularly of shares, are rarely collected but they are becoming increasingly common and more detailed. Therefore, approaches like ours may be needed more often in the future.

We presented our results to managers and they showed a great interest in understanding the uncertainty regarding the correlations between a reduced number of key product factors. Their immediate questions concerned what factors are "competing" between each other and to what extent. Moreover, their intuition and knowledge of the market supports the idea that preferences for key factors, like brands, are evolving in time. However, they usually measure these time variations based on market wide "top of mind" surveys while the use of sales data for this type of analysis is rare. Hence, the modeling of the evolution in brands-preferences based in market shares data, they argue, is one of the key and most valuable aspects of our approach.

4.7 Tables and Figures

	Posterior	HP	DR
	Mean	1%	99%
$\tau_{m=1}^{2}$	0.636	0.618	0.698
$\tau_{m=2}^2$	0.640	0.624	0.670
$\tau_{m=3}^{2}$	0.595	0.580	0.626
$\tau_{m=4}^2$	0.659	0.641	0.717
$\tau_{m=5}^2$	0.618	0.598	0.657
$\tau_{m=6}^2$	0.580	0.565	0.598
$\tau_{m=7}^{2}$	0.657	0.640	0.711
$\tau_{m=8}^{2}$	0.662	0.647	0.755
$\tau_{m=9}^{2}$	0.641	0.624	0.682
$\tau_{m=10}^{2}$	0.630	0.615	0.680
Notes:	The true val	lue of τ_m^2	is equal
to 0.64	for all m . I	IPDR st	ands for
Highes	t Posterior I	Density R	legion.

Table 4.1: Simulation Experiment: Posterior Distribution of the Variance of the Demand Shocks

	Posterior	HP	DR
	Mean	1%	99%
$\sigma_{m=1}^2$	0.107	0.082	0.148
$\sigma_{m=2}^2$	0.120	0.096	0.157
$\sigma_{m=3}^2$	0.118	0.088	0.144
$\sigma_{m=4}^2$	0.120	0.094	0.155
$\sigma_{m=5}^2$	0.136	0.104	0.182
$\sigma_{m=6}^2$	0.134	0.102	0.170
$\sigma_{m=7}^2$	0.128	0.096	0.170
$\sigma_{m=8}^2$	0.116	0.090	0.153
$\sigma_{m=9}^2$	0.121	0.090	0.160
$\sigma_{m=10}^2$	0.128	0.098	0.170
Notes: 7	The true val	ue of σ_m^2	is equal
to 0.16	for all m . H	IPDR sta	ands for
Highest	Posterior D	ensity R	egion.

Table 4.2: Simulation Experiment: Posterior Distribution of σ_m^2

	Posterior	HP	DR	Real
	Mean	1%	99%	Value
Brand A	1.950^{*}	1.081	4.089	2.0
Brand B	0.435^{*}	0.209	3.446	2.0
Brand C	3.036^{*}	1.958	3.755	2.0
Brand D	1.127^{*}	0.890	3.514	2.0
Brand E	7.890^{*}	5.126	8.983	8.0
Brand F	3.929	3.119	4.702	8.0
Brand G	3.066	2.421	4.500	8.0
Brand H	1.992	1.776	3.789	8.0
Brand I	2.295	1.817	3.617	4.0
Brand J	2.219	1.938	3.603	4.0
Attribute b	4.827	4.270	5.316	4.0
Attribute c	1.990	1.463	3.636	4.0
Attribute d	1.473^{*}	0.732	3.202	2.0
Attribute e	2.988	2.693	3.490	2.0
Price	0.866	0.543	1.510	2.0
Promotion	1.533^{*}	1.112	2.550	2.0
Note: * mean	s that the re	eal value	is inclue	led in
the HPDR. H	IPDR stands	s for Hig	hest Pos	terior

Density Region.

Table 4.3: Simulation Experiment: Posterior Distribution of the elements of D^2 .

	Posterior	HF	PDR
	Mean	1%	99%
$\tau_{m=1}^2$	5.329	0.606	53.233
$\tau_{m=2}^2$	7.518	4.782	11.322
$\tau_{m=3}^2$	0.814	0.659	1.034
$\tau_{m=4}^2$	0.895	0.684	1.172
$\tau_{m=5}^2$	1.500	1.127	2.042
$\tau_{m=6}^{2}$	1.284	0.989	1.715
$\tau_{m=7}^{2}$	0.831	0.577	1.281
$\tau_{m=8}^{2}$	0.638	0.574	0.734
$\tau_{m=9}^{2}$	0.838	0.754	1.013
$\tau_{m=10}^{2}$	0.467	0.395	0.589
$\tau_{m=11}^{2}$	1.015	0.805	1.322
$\tau_{m=12}^{2}$	5.664	1.632	65.241
$\tau_{m=13}^{2}$	2.631	1.906	3.652
$\tau_{m=14}^{2}$	0.752	0.672	0.901
$\tau_{m=15}^{2}$	1.917	1.490	2.572
$\tau_{m=16}^{2}$	1.439	1.316	1.607
$\tau_{m=17}^{2}$	4.144	0.602	44.067
$\tau_{m=18}^{2}$	1.612	1.328	2.101
Note:	HPDR sta	nds for	Highest
Posteri	or Density F	Region.	

	Posterior	HP	DR
	Mean	1%	99%
$\sigma_{m=1}^2$	0.0149	0.0090	0.0249
$\sigma_{m=2}^2$	0.0200	0.0102	0.0397
$\sigma_{m=3}^2$	0.0628	0.0357	0.1213
$\sigma_{m=4}^2$	0.0384	0.0206	0.0650
$\sigma_{m=5}^2$	0.0128	0.0080	0.0276
$\sigma_{m=6}^2$	0.0180	0.0101	0.0293
$\sigma_{m=7}^2$	0.0361	0.0208	0.0600
$\sigma_{m=8}^2$	0.0322	0.0190	0.0563
$\sigma_{m=9}^2$	0.0390	0.0202	0.0654
$\sigma_{m=10}^2$	0.0339	0.0196	0.0611
$\sigma_{m=11}^2$	0.0557	0.0272	0.1021
$\sigma_{m=12}^2$	0.0131	0.0072	0.0236
$\sigma_{m=13}^{2}$	0.0204	0.0113	0.0425
$\sigma_{m=14}^2$	0.0252	0.0150	0.0450
$\sigma_{m=15}^{2}$	0.0132	0.0088	0.0230
$\sigma_{m=16}^{2}$	0.0255	0.0139	0.0429
$\sigma_{m=17}^{2}$	0.0123	0.0073	0.0249
$\sigma_{m=18}^2$	0.0147	0.0087	0.0202
Note:	HPDR star	nds for H	Iighest
Posterie	or Density R	legion.	

Table 4.5: Application: Posterior Mean and HPDR of the $\sigma_m^2.$

		Posterior	HPDR	
		Mean	1%	99%
Market 1	Size A	0.811	-0.098	1.766
	Size B	-1.835	-3.025	-0.577
	Size C	0.268	-0.643	1.164
	Size D	-0.513	-2.131	1.174
	Size E	-2.042	-3.053	-1.017
	Price	-2.684	-3.925	-1.529
	Promotion	-2.380	-6.455	1.856
Market 2	Size A	0.120	-0.207	0.438
	Size B	-0.934	-1.204	-0.669
	Size D	0.414	0.059	0.739
	Size E	-0.218	-0.660	0.191
	Price	-0.904	-1.472	-0.309
	Promotion	-2.714	-4.229	-0.894
Market 3	Size A	0.749	0.397	1.093
	Size B	0.237	-0.087	0.430
	Size C	-0.594	-1.015	-0.224
	Size D	-0.641	-0.877	-0.367
	Size E	-0.545	-0.821	-0.271
	Price	-0.593	-0.924	-0.211
	Promotion	-3.388	-4.764	-2.079
Note: HPDR stands for Highest Posterior Density Region.				

Table 4.6: Application: Posterior Mean and HPDR of the Fixed Elements of f_m (size and price and promotion coefficients) for 3 out 18 markets

	Posterior	HPDR	
	Mean	1%	99%
D_{1}^{2}	1.436	0.801	2.493
D_2^2	0.798	0.653	0.896
D_{3}^{2}	1.341	1.184	1.619
D_4^2	0.831	0.628	1.461
D_5^2	0.293	0.249	0.335
D_6^2	0.874	0.739	1.022
D_{7}^{2}	1.853	0.970	4.412
$D_8^{\dot{2}}$	0.500	0.443	0.575
$D_9^{\tilde{2}}$	4.648	1.509	9.801
D_{10}^{2}	0.395	0.327	0.457
D_{11}^2	1.197	0.787	1.718
D_{12}^2	0.569	0.466	0.727
D_{13}^{2}	0.556	0.478	0.603
$D_{14}^{\tilde{2}}$	0.628	0.522	0.700
$D_{15}^{\hat{2}}$	10.625	3.911	21.916
$D_{16}^{\hat{2}}$	0.432	0.329	0.502
$D_{17}^{\hat{2}}$	0.195	0.160	0.238
D_{18}^2	0.990	0.750	1.257
D_{19}^2	3.249	1.824	5.073
D_{20}^2	0.221	0.182	0.276
$D_{21}^{\bar{2}}$	0.337	0.248	0.378
$D_{22}^{\overline{2}}$	0.602	0.540	0.698
$D_{23}^{\overline{2}}$	7.294	5.028	12.136
D_{24}^{2}	0.700	0.623	0.794
D_{25}^{2}	2.361	1.697	2.710
D_{26}^2	0.754	0.654	0.955
D_{27}^2	0.643	0.522	0.732
D_{28}^2	0.558	0.472	0.667
D_{29}^{2}	0.579	0.468	0.672
D_{30}^2	0.587	0.474	0.797
D_{31}^{2}	0.590	0.443	0.777
D_{32}^{2}	0.624	0.490	0.811
Note:	HPDR st	tands fo	r Highest
Posterior Density Region.			

Table 4.7: Application: Posterior Distribution of the Elements of the D^2 matrix, where $\Psi = DSD$.



Integral with 3 dimensions



Figure 4.1: Performance of Halton Based Normal Draws versus Normal Draws



Figure 4.2: Prior Correlations for Different Elements of Ψ . The degrees of freedom for the Wishart Distribution v are set to 11 for the left panel and 20 for the right panel.





Figure 4.4: Simulation Experiment: Real (Solid Line) versus Posterior Mean (Dots) and the 99% HPDR (Dashed Lines) of the Time-Varying Brand Coefficients at Market 5



Figure 4.5: Simulation Experiment: Real (Solid Line) versus Posterior 99% HPDR (Dashed Lines) of All Elements in the Correlation matrix R.





Figure 4.6: Application: Time-Profile Relative to First Period of 8th Market Time-Varying Factors f^m (Solid Lines) and their 99% HPDR (Dashed Lines).







Figure 4.9: Cross-Price Elasticities at Market 2

4.A Appendix

Iterative BLP Procedure

We use the iterative procedure proposed in BLP to obtain the $\mu_t = (\mu_{1t}, \ldots, \mu_{Jt})$. Note that, for convenience, we omit the market indicators m here. The procedure consists of the following steps. First we obtain H draws of v_i . To this end we write the d-th draw as $v_i^d = \Psi^{1/2} \zeta^d$ where ζ^d is draw from a joint normal distribution which we obtain using scrambled Halton draws. Given these draws and some initial value for μ_t we can compute the implied market shares \hat{s}_t . Given the shares, both real s_t and implied \hat{s}_t , we can use the contraction mapping

$$\mu_t^{new} = \mu_t^{old} + \log(s_t) - \log(\hat{s}_t)$$
(4.26)

to obtain a new value for μ_t . We repeat the contraction mapping computing the implied shares \hat{s}_{jt} as

$$\left(\sum_{i=1}^{H} \frac{\exp(\mu_{jt}^{old} + \lambda'_{jt}v_i)}{\exp(\mu_{0t}) + \sum_k \exp(\mu_{kt}^{old} + \lambda'_{kt}v_i)}\right) / H$$
(4.27)

for j = 1, ..., J and t = 1, ..., T and we stop the contraction mapping when the values of μ_t^{new} and μ_t^{old} converge.

Note that we include and solve for the outside good utility μ_{0t} in the contraction mapping iterations. We discovered that the precision of the contraction mapping is higher when we iterate over the utilities of all products together with the utility of the outside good.

Computing Elasticities

We use the following definition to compute the price elasticities φ_{jl}^m between product j and l in market m conditional on all model parameters:

$$\varphi_{jl}^{m} = \frac{p_{lt}^{m}}{\mathbf{E}[s_{jt}^{m}]} \frac{\partial \mathbf{E}[s_{jt}^{m}]}{\partial p_{lt}^{m}} = \frac{p_{lt}^{m}}{\int s_{jt}^{m} \pi(\eta_{t}; 0, \tau_{m}^{2}\mathbf{I}) d\eta_{t}^{m}} \times \int \frac{\partial}{\partial p_{lt}^{m}} s_{jt}^{m} \pi(\eta_{t} | \tau_{m}^{2}) d\eta_{t}^{m}$$
(4.28)

where s_{jt}^m is defined in (4.7) and

$$\frac{\partial}{\partial p_{lt}^m} s_{jt}^m = \begin{cases} -\beta^m \int s_{ijt}^m s_{ikt}^m \phi(v_i^m; 0, A^m \Psi A'_m) dv_i^m & \text{if } k \neq j \\ \\ \beta^m \int s_{ijt}^m (1 - s_{ikt}^m) \phi(v_i^m; 0, A^m \Psi A'_m) dv_i^m & \text{if } k = j, \end{cases}$$
(4.29)

and β^m is the price coefficient in market m. Finally, to obtain the posterior distribution of the price elasticities we average (4.28) over the posterior draws for all parameters.

Chapter 5

Finding the Influentials that Drive the Diffusion of New Technologies

In this chapter we consider the diffusion of similar technologies in a single market composed of many locations. We address the identification of the influential locations that drive the aggregate sales of these new technologies based on aggregate sales data and location specific online search data.

In this chapter we put forward a model where aggregate sales are a function of the online search of potential consumers at many locations. We argue that a location may be influential because of its power to drive aggregate sales and this power may be dynamic and evolving in time. Second, the influential locations may produce spillover effects over their neighbors and hence we may observe clusters of influence. We apply Bayesian Variable Selection (BVS) techniques and we use Multivariate Conditional Autoregressive Models (MCAR) to identify influentials locations and their clustering.

We apply our methodology to the video-game consoles market and to new search data of Google Insight. More precisely, we study the influential locations that drive the sales growth of the Nintendo Wii, the Sony PS3 and Microsoft Xbox 360. Specifically, we study the diffusion of these technologies at four different stages of their life-cycle. In this way, we can identify the group of influential locations and its composition in different sub-periods.

Our results indicate that the influential locations and their economic value (measured by search elasticities) vary over time. Moreover, we find significant geographical clusters of influential locations and the clusters composition varies during the life-cycle of the consoles. Finally, we find weak evidence that demographics explain the probability of a location to be influential. The main managerial implication of our results is the notion that the group of influential locations and their clustering varies during the life-cycle of a technology. Hence, managers should aim to identify the identity plus the locations and the dynamics of influentials.

5.1 Introduction

An important topic related to the diffusion of new technologies is the identification of influentials. Influentials play an important role as opinion leaders and trend setters and they critically affect the speed of adoption of new technologies (van den Bulte and Joshi, 2007).

Recent attention is being given to the identification of the location and identities of these influentials. In the literature, influentials are defined as individuals or groups of individuals that influence the behavior of others in a significant way. Their influence has been studied at the individual level (Trusov et al., 2010), at the firm level (Albuquerque et al., 2007) and at the country level (van Everdingen et al., 2009). Influentials may have a specific location in a social network (Trusov et al., 2010; Christakis and Fowler, 2009; Cho and Fowler, 2007) or a specific physical location (Choi et al., 2009; Goldenberg et al., 2009). Their influence can be limited to a few others (Christakis and Fowler, 2009, page 28) but their impact may also exceed national boundaries (van Everdingen et al., 2009).

In this article we study the diffusion of a number of similar and competing technologies and we address the identification of the influential locations that drive the aggregate sales of these new technologies. We put forward a model where sales are a function of the online search registered at many different locations. We will refer to this model as the sales-search model. We know that consumers search for technologies (or products) online and we posit that online search should be a good predictor of sales. However, people in many different locations search for products while only the consumers living in a subset of these locations may be the key groups driving the sales of new technologies. Moreover, the influential locations may not always be the same. And, the cross-influence among locations may be important and time-varying or fixed in time.

We present an approach that is new to the marketing literature and we study new search data obtained from Google Insight. Our novelty is that we use the sales-search model together with Bayesian Variable Selection techniques to select the locations that are most likely driving the aggregate sales of these three new technologies. We use this methodology because there are many possible important locations and a straightforward choice between them is not possible. In addition, we present a second model with Multivariate Conditional Autoregressive priors (known as MCAR priors) to study the cross-location influence, the significance of spatial clustering of influential locations and the competing relationships between technologies. We will refer to this model as the spatial model.

Our data consists of the aggregate weekly sales of the Nintendo Wii, the PlayStation 3 and the Microsoft Xbox 360 for the entire US market and online search data for each of these products. The online search data were obtained from Google Insight and these data consist of weekly indicators of online search for each of these technologies in each US state. The data cover a period from the launch time of each technology up to February 2010 (approximately four years) for both the sales and the online search data. This dataset is attractive because it allows us study three very successful technologies that receive worldwide interest. These three products were marketed simultaneously in all US states and this fact allows us to discard the explanation that a region may become influential because its products were available at an earlier introduction date relative to other regions.¹ Moreover, these technologies have unique names and they have kept these unique names for long periods of time and therefore we can obtain reliable online search data for all US states.² The sales data we observe can be easily classified in different periods of the products' life-cycle and we will identify the influential locations at these product life-cycle stages. We base these life-cycle stages on Rogers (2003) who suggests that innovations are characterized by five periods when different groups of people (innovators, early adopters, early majority, late majority and laggards) adopt an innovation. In this way we will be able to uncover the location of influential groups of adopters at different life-cycle phases of the products. Our results suggest that the influential regions driving aggregate sales differ across the life-cycle of a technology. Moreover, our approach uncovers geographical clustering of both influential and not influential regions. Influential regions seem to be close to each other but we find that their influence and the geographical clustering varies over time. In addition, we find only a weak association between demographic information and the probability that a region is influential. Finally, our results indicate that a 10% increase in local online search translates

 $^{^{1}}$ For example, the launch time of the products studied by van Everdingen et al. (2009) differs across countries.

²Note that it is impossible to obtain state level sales data. We made inquiries at different market research firms, including NPD group, and to our knowledge there are no firms collecting these data.

on average into a 1.5% percent increase in global sales but this number varies across regions and diffusion periods and its range goes from 0 up to 3%.

The plan of the chapter is as follows. In Section 5.2 we discuss previous literature and its relationship to our work. In Section 5.3 we present our methodology. Later in Section 5.4 we give details about our data and some specific details regarding our model. In Section 5.5 we present our results and finally in Section 5.6 we conclude the chapter. The statistical methodology that we use is presented in detail in Section 5.A and Section 5.A.

5.2 Literature Review

The literature related to our work can be classified into micro-studies of adoption, like Choi et al. (2009), Goldenberg et al. (2009), Trusov et al. (2010), Garber et al. (2004) and Jank and Kannan (2005), and into macro-studies of technology diffusion, like van Everdingen et al. (2009), Albuquerque et al. (2007) and Putsis Jr et al. (1997).

van Everdingen et al. (2009) examine the global spillover effects of product introductions and take-offs. They find that the product take-off in a country can help to predict the takeoff of the same product in different countries. In addition, they report asymmetric patterns of influence and foreign susceptibility. The heterogeneity in the spill-over effects is significantly explained by economic and demographic characteristics. Moreover, van Everdingen et al. (2009) discuss briefly the time dimension of influence. Their results suggest that there are countries that have a large impact on others late in the diffusion process, while other countries may have a smaller influence but sooner. Albuquerque et al. (2007) study the global adoption of two ISO certification standards and their results indicate that cross-country influence is important and it improves the fit of their model. They find that the role of culture, geography and trade in the adoption process is different across the ISO standards. They use a multi-country diffusion model and therefore they assume that a firm's adoption is influenced by previous cumulative number of adoptions by other firms in different countries. Therefore, the global cumulative adoptions of ISO standards foster more adoptions. Albuquerque et al. (2007) also find that the influence of cumulative past adoptions is stronger among firms close to each other or between firms in neighboring countries. Finally, Putsis Jr et al. (1997) study cross-country and inter-country diffusion patterns and they report important cross-country influence on diffusion. Their findings suggest that each country's influence varies from product to product.

The micro diffusion studies have documented the role and economic value of influential people in a social network (Trusov et al., 2010; Goldenberg et al., 2009) and the formation of spatial clusters (Garber et al., 2004; Choi et al., 2009; Jank and Kannan, 2005). The study of Garber et al. (2004) deals with the spatial distribution of adoption. They discovered that the spatial pattern at early stages of the diffusion of a technology is an accurate predictor of new product success. They argue that spatial clustering is a sign of imitation and therefore if the spatial distribution of adoption shows clusters it is very likely that the diffusion process will continue and sales will eventually take off. They compare the spatial distribution of adoption against a uniform distribution of adoption and they find that successful products show an early spike of divergence between these two distributions (cross-entropy) while the cross-entropy of product failures remains relatively constant and low.

More recently, Choi et al. (2009) studied the temporal and spatial patterns of adoption in Pennsylvania and they discovered that the spatial clusters of adoption change over time and that the cross-region (cross zip code) influence decays over time. In the same way, Jank and Kannan (2005) report spatial clusters of customers with the same price sensitivity and preferences and they use spatial random effects to capture the geographical variation in preferences. The study of Hofstede et al. (2002) is focused in identifying spatial country and cross-country segments and they find evidence of contiguous and spatial clustering of consumer preferences. They argue that the spatial dependence in preferences should be useful to define distribution and marketing decisions across countries. Bradlow et al. (2005) provide an overview of spatial models and their relationship to marketing models. Finally, Trusov et al. (2010) and Goldenberg et al. (2009) suggest that influentials can have a significant economic value and they may foster the diffusion of new technologies.

In this chapter we explore the time dimension and the spatial structure of influence at the level between micro and macro, that is at the regional level within a country. The objective of van Everdingen et al. (2009) and Albuquerque et al. (2007) is to identify the cross-country
influence while our objective is to discover whether a region is influential and when it is influential. In contrast with previous research, in our study a region may be influential initially while later it may exert no influence at all or the other way around. That is, we consider the influence across the life-cycle of the products' diffusion while previous research has not focused particularly on this aspect. Moreover, the Bayesian Variable Selection technique that we use to detect influentials also distinguishes our study from previous work at a technical level. Finally, the visual inspection of our results suggests important geographical clusters of influential regions and we study whether these geographical clusters of influence are statistically relevant. For this latter purpose, we fit a spatial model with MCAR priors and perform tests to detect spatial clusters. It is the univariate version of this prior that has recently been applied in some marketing studies, an example is Duan and Mela (2009). The MCAR prior can incorporate both the spatial structure of the data as well as the relationship between technologies. To our knowledge, we are the first to use an MCAR prior on a marketing application while it must be mentioned that this prior is frequently used in bio-statistics and environmental studies.

5.3 Methodology

The approach we use consists of two main parts. First, in Section 5.3.1 we describe how we use Bayesian Variable Selection techniques to identify the regions and the sub-periods during which each region is likely to drive aggregate sales. The Bayesian Variable Selection technique will let us compute the posterior probability that a region is influential for any given sub-period. In Section 5.3.2 we specify a second model to study these posterior probabilities and our main objective in this section is to test whether there are important spatial clusters or demographic variables explaining these inclusion probabilities.

5.3.1 The Sales-Search Model

We observe the aggregate sales y_{it} of i = 1, ..., M technologies at time t = 1, ..., T. We also observe the online search s_{ijt} for each of these *i* technologies at *J* different locations for j = 1, ..., J and time periods t = 1, ..., T. In addition, s_{ijtn} will refer to the search observed at location j at a time t that is included in sub-period n, for n = 1, ..., N. We define subperiods of diffusion because we are interested in studying the early, mid and late diffusion of the technologies.

The sales equation is specified as

$$y_{it} = \sum_{j} \sum_{n} \beta_{ijn} s_{ijtn} + \epsilon_{it} \text{ where } \epsilon_{it} \sim N(0, \sigma_i^2).$$
(5.1)

where both y_{it} and s_{ijtn} are in logs; sales are measured in hundred thousands and search is measured as an "interest indicator" and its range goes from 10 to 110. We give more details about the data in Section 5.4. We specify a technology *i*, sub-period *n* (for n = 1, ..., N) and region *j* specific coefficient β_{ijn} and the error term ϵ_{jt} is assumed to be normal with zero mean and variance σ_i^2 .

This specification sums over all sub-periods n and locations j but estimating such a model may be impossible when the total number of regressors $J \times N$ is large relative to T. Note that in practice $J \times N$ can be even much larger than T. Moreover, it is very likely that many of the $\beta_{ijn} = 0$ because of the likely correlation among the s_{ijn} and the fact that some locations may simply do not drive sales. Hence, we need to select a subset location specific regressors that consists of the best set of all possible regressors. We will call the set of all possible regressors Xand we will use X_{γ} to refer to the subset of best regressors. We will call q_{γ} to the total number of elements in X_{γ} and p to the total number of elements in X. That is, $X_{\gamma} \subset X$ and X is a set containing s_{ijn} for $j = 1, \ldots, J$ and $n = 1, \ldots, N$. The purpose is to select a model that sums only over this subset. Therefore we specify

$$y_{it} = \sum_{j} \sum_{n} \gamma_{ijn} \beta_{ijn} s_{ijn} + \epsilon_{it} \text{ where } \epsilon_{it} \sim N(0, \sigma^2)$$
(5.2)

as the sales equation where γ_{ijn} is a technology and region sub-period specific indicator that takes the value of 1 if s_{ijn} is in the subset X_{γ} and zero otherwise. Note that JN potential regressors result in 2^{JN} possible subsets and vectors γ_i where $\gamma_i = (\gamma_{i11}, \ldots, \gamma_{iJN})'$.

One could suggest for equation (5.2) that we could also sum over i on the right hand side and not only j and n. That is, the sales of a technology could be a function of the search for all technologies in the market. However, in our application there are over 2.57×10^{61} (that is $2^{51\times4}$ where 51 is the number of locations and 4 is the number of sub-periods) possible subsets of regressors and if we were to sum over *i* there would be more than 1.69×10^{184} (that is $2^{3\times52\times4}$) subsets of models. That is 6.61×10^{122} more subsets. Therefore, we study the relationship between technologies with a different model and we discuss this second model later in this section. A second issue is that sales are a function of search while at the same time search may be a function of sales. We are aware of this possible endogeneity of sales and search but as we are using local indicators for search and aggregate measures for sales we believe the endogeneity between them should be relatively weak. Finally, the right hand side could contain lags of the search indicators. However, the inclusion of lags forbids us to compare the inclusion reason across locations. For example, a location may be selected because it has an important lagged effect while another location because of its contemporaneous effect on sales. We restrict the model to a contemporaneous relationship between sales and search to be able to use the probability of a location regressor to be in X_{γ} at a later stage in the spatial model.

We use Bayesian Variable Selection (BVS) as presented in George and McCulloch (1997) and Chipman et al. (2001) to select the best subset of regressors. To use BVS we need proper priors, we specify $\pi(\beta_i | \sigma_i, \gamma_i)$ as in Equation (5.5) and $\pi(\sigma_i^2 | \gamma_i)$ as in equation (5.7); these are the prior distributions of β_i coefficients and the variance σ_i^2 where $\beta_i = (\beta_{i11}, \ldots, \beta_{iJN})$ and we specify the prior distribution of the indicators $\pi(\gamma_i)$. We use equations (5.9) and (5.10) to define the prior on γ . BVS is an attractive technique because we can draw inferences on the probability of inclusion for each potential regressor in model (5.2). That is, we can draw inferences on the posterior distribution of the indicators given the data $\pi(\gamma_i | y_i)$ where $y_i = (y_{it}, \ldots, y_{iT})'$. We estimate model (5.2) for each of the technologies separately and details of our estimation approach are presented in the Appendix. In the Appendix we drop the sub-index *i* because we use the same prior specification for all technologies.

5.3.2 The Spatial Model

The indicator vector γ_i is composed of location and sub-period indicators and based on BVS we can compute for each element of the vector γ_i the probability that it equals one. That is, we can compute each region's posterior probability to be included at any sub-period and this posterior is available for each of the technologies. We will refer to the logit transformation³ of this posterior probability as \bar{p}_{ijn} where as before *i* refers to the technology, *j* to the location and *n* is the sub-period index.

Our objective is to test whether the variation in inclusion probabilities is explained by demographic variables and whether there are significant spatial effects in these inclusion probabilities. Hence, we propose a model where the posterior probabilities of inclusion depend on a set of covariates Z_n and their corresponding coefficients δ_n plus spatial effects Φ_n and some noise ε_n . We propose that

$$\bar{P}_n = Z_n \theta_n + \Phi_n + \varepsilon_n \tag{5.3}$$

where $\bar{P}_n = (\bar{p}'_{1n}, \dots, \bar{p}'_{Mn})$, $\bar{p}'_{in} = (\bar{p}_{i1n}, \dots, \bar{p}_{iJn})$. That is, \bar{P}_n is a $J \times M$ matrix with the inclusion probabilities of each of the J locations for each technology in M columns. Z_n are covariates available for period n where Z_n is a $J \times K$ matrix where K is the number of covariates. We assume $\theta_n = \iota \otimes \delta_n$ is a $K \times M$ matrix with coefficients where ι is a row vector of ones of size M and δ_n is a $K \times 1$ vector of coefficients. $\Phi_n = (\phi'_{1n}, \dots, \phi'_{Mn}), \phi'_{in} = (\phi_{i1n}, \dots, \phi_{iJn})$ and $\varepsilon_n = (\varepsilon'_{1n}, \dots, \varepsilon'_{Mn})$ with $\varepsilon'_{in} = (\varepsilon_{i1n}, \dots, \varepsilon_{iJn})$. Both Φ_n and ε_n are $J \times M$ matrices.

The spatial effects Φ are a function of the relationships between technologies and the neighborhood structure of the market. The Φ matrix is composed of one spatial effect for each location and technology. Each spatial effect, in general terms, depends on the spatial effects of all technologies at neighboring locations. Hence, the spatial effects reflect spatial clustering but they do not detect the direction of influence between locations. This property of the spatial effects is specified in a prior distribution that depends on Λ , Ψ and ρ where Λ is a $M \times M$ matrix with the covariance structure between the technologies, Ψ is a $J \times J$ matrix that measures the neighborhood or the spatial structure of the market and ρ is a parameter that measures spatial auto-correlation. The element Ψ_{kl} is either a fixed distance between location k and l or an indicator that takes a value of 1 if the location k is a neighbor of l and zero otherwise. In the Appendix we provide details on how we draw inference about ρ , Λ , δ_n and the covariance matrix

³The function is $\log(p/(1-p))$. A second transformation may be $\log(-\log(p))$. We tested both transformations and our results are similar.

associated with ε_n . Note that Ψ is a fixed matrix with the neighborhood structure and hence we do not estimate it. We give more details about Ψ in the next section.

Next, we use this specification to explore if there are significant spatial effects Φ in the posterior probabilities of inclusion for each region during each sub-period n and if there is a relationship between the inclusion probabilities between technologies after controlling for the covariates in Z_n . Note that in the equation (5.3) we are pooling all technologies $i = 1, \ldots, M$ together. The reason we pool technologies together is that their inclusion probabilities may be related to each other. For example, Texas could be the driver of growth for one technology but not for all technologies. That is, technologies may be competing against each other when the sign of the covariance terms in the Λ matrix are negative.

5.4 Data and Modeling Details

Weekly search indicators are available online from Google Insight for all US states and the weekly series of sales data for the video-game consoles were obtained from VGchartz.com. The data of VGchartz follows very closely the monthly figures of the NPD group. We use the latest (year 2000) demographic information of the US Census Bureau for all US states.

In Figure 5.1 we present a printed screen with the exact keywords that we used to retrieve the search data from Google Insights for Search (http://www.google.com/insights/search/). In Table 5.1 we provide the R code to automatically retrieve the data from http://www.vgchartz.com/.

To estimate the parameters of equation (5.2) we used MCMC and the chain ran for 210 thousand iterations and we discarded the first 10 thousand. The equation that we used includes a spline term that captures the seasonal fluctuation of y_i and its overall level. We fit a smoothing spline of y_i as a function of time and we use 10 degrees of freedom as the smoothing parameter; we refer to Hastie et al. (2001, page 127-137) for mode details on fitting smoothing splines. Sloot et al. (2006) also use spline terms to capture seasonal fluctuations. The spline term is always included on the right-hand side of the model and we do not use BVS on this term. Finally, note that we used the logs of y_i and the s_{ijn} and that y_{it} are the sales of the technology

i at the end of week t and s_{ijt} is the online search index for the technology *i* at state *j* during the week t.

Next, we use MCMC to estimate the parameters of equation (5.3) and the chain ran for 2000 iterations and we discarded the first 1000. We used much less draws than before because convergence for a linear model is quite fast. We run the estimation for each sub-period separately and therefore we estimated the parameters of equation (5.3) for each period.

We divide the sales data of each consoles in four periods of equal length. These periods roughly correspond with the first four stages of adoption proposed by Rogers (2003). It is likely that in practice the length of each period varies per product or industries. For example, we know that the time to take-off is different across countries while within a country the take offs tend to occur at a systematic time difference relative to other countries (van Everdingen et al., 2009; Golder and Tellis, 1997; Tellis et al., 2003). Additionally, we choose periods of equal length to be able to compare the influential locations across products for exactly the same period of time. In this way we can naturally make cross-product comparisons.

We estimate equation (5.2) and equation (5.3) separately because we prefer not to impose any spatial structure on the prior probability of including regressors in the prior for the indicator variables, that is $\pi(\gamma)$. We estimate equation (5.3) for each life-cycle stage. The disadvantage of treating equations (5.2) and (5.3) separately is that the uncertainty of the first model is not taken into account in the second model. A technical reason to keep the estimation of these equations separately is that the posterior probabilities of inclusion are computed using the full MCMC chain and therefore we know them only at the end of the estimation. However, the most important reason to keep the estimation in two steps is not to impose a priori a spatial structure in the inclusion probabilities. In this way, we leave the task of testing for spatial clustering as a second step and we may be able to provide stronger evidence of any spatial structure.

We checked for convergence of the MCMC chains visually. We give more details about the estimation approach and about the MCAR models in the Appendix.

5.5 Results

In this section we first discuss the results for the sales-search model in equation (5.2) and then for the spatial model in equation (5.3).

5.5.1 Sales-Search Model Results

In Figure 5.2 we report the posterior distribution of the number of regressors included in the model, that is q_{γ} . The average number of regressors included in the model is around 17 with a minimum near 5 and a maximum of 35 regressors. If the regressors were uniformly distributed among diffusion periods this would mean an average of 4 regressors per diffusion period.⁴

In Figure 5.3, 5.4 and 5.5 we graphically report the posterior means of the inclusion probabilities for all US states and for the Nintendo Wii, the Sony PS3 and the Xbox 360, respectively. All these probabilities are also reported in Table 5.2, 5.3 and 5.4. In Figures 5.3, 5.4 and 5.5 the lighter (green) colors represent high posterior probabilities while the darker (red) colors represent low inclusion probabilities. We include a map of the USA including state names in Figure 5.17 to facilitate the reading of these figures.

In Figure 5.3 we can observe that the states with the higher inclusion probabilities during the first diffusion period of the Nintendo Wii are Washington, Texas, Alabama, Wyoming, Kansas and New Hampshire. So, this means that these states are more likely to drive the sales of the Wii at an early stage of the Wii's life-cycle. It is noticeable too that the Western states are more likely to be included in the first diffusion period while the North-Eastern states have very low probability of inclusion. However, during the third diffusion period the Western states are not likely to be included in the model while it is more likely to include states in the center and North-East of the US. In the last diffusion period we find that very few states have high probabilities and these are Montana, North Dakota and New Hampshire. That is, there are many locations driving the growth of the Wii at early life-cycle stages and relatively few engines of growth at the end.

⁴Note that we chose $v_1 = 7$ and a = 50 and b = 100 (the parameters of the distribution of the prior inclusion probability w, see equations (5.9) and (5.10)) and this set-up results in a relatively small number of selected regressors q_{γ} .

The geographical pattern for the Sony PS3 is slightly similar to the pattern of the Nintendo Wii. However, we find that during the first diffusion period there are many more states (relative to the Wii) with high probability of inclusion. Again, all states in the West (California, Nevada, Oregon and Washington) have higher inclusion probabilities but for the PS3 many states in the East and North-East also have high probabilities during this first period. In fact, there are very few states with low probability of inclusion during the first period and these are North and South Dakota and Minnesota together with Kentucky and West Virginia. The opposite happens during the last diffusion period where many states have low probability of being included in the model. The probability of the West Coast states is high at the beginning and their influence seems to diminish in subsequent periods. The maps seem to be revealing a *boom bust* pattern. That is, many states may be influential during the first diffusion period but of this first set of countries very few remain influential in the last diffusion period and other states take the influential position.

The geographical pattern for the Microsoft Xbox 360 is very different from the other two consoles. The states with higher probabilities at each diffusion period are fewer than for the other two consoles and the influential states seem to be far from each other. However, for all regions, with the exception of Washington and Oregon, the states that seem more likely to be included in the model are in the North and North-East of the US.

An immediate question about these results is whether there is evidence of geographical clusters. At first glance, influential regions seem to be neighbors of other influential regions while not influential regions seem to be clustered together too. However, we may have some bias when judging probability distributions (Kahneman et al., 1982, page 32) and therefore we need some formal way to measure spatial association. Two statistics that can measure spatial association in aereal data are the Moran's I and the Geary's C (Banerjee et al., 2004, page 71).

We computed both the Moran's I and Geary's C for all sub-periods and technologies and we compared these two statistics, computed with the estimated inclusion probabilities, against the distribution of these two statistics when we assume that the probability of inclusion is uniformly distributed. Garber et al. (2004) also compare the observed spatial distribution of adoption against the uniform distribution. High spatial association is indicated by high Moran's I or by low Geary's C statistics. In Figure 5.6 we report the statistics computed with the real inclusion probabilities (in vertical dashed lines) and the distribution of both statistics (in the histograms) assuming the inclusion probabilities follow a uniform distribution.⁵ As we can observe in Figure 5.6, when the inclusion probabilities are uniformly distributed the chances are very low to obtain the statistics in the extremes where the Moran's I and Geary's C based on the estimated inclusion probabilities appear. In the next section we discuss the results regarding the spatial model (equation (5.3)) where we further investigate the significance of the spatial clustering.

In the left panel of Figure 5.7, 5.8 and 5.9 we report the histogram of the posterior mean of the β coefficients for all sub-periods of the Nintendo Wii, the PS3 and Xbox 360, respectively. We report the distribution of the $\beta | \gamma = 1$ coefficients. That is, we report their distribution given that their corresponding regressor was included in the selected subset of regressors X_{γ} and we refer to these coefficients simply as β . In the right hand panel of the same figures we report the distribution of the posterior mean of the β coefficients divided by their posterior standard deviation. As we can see, the size of the β coefficients seems to be centered around 0.15 for the Nintendo Wii and the Xbox 360 and around 0.12 for the PS3. This means that on average a local (state) increase of 10% in search translates into a 1.5% or 1.2% increase in the global (nation) sales. The significance of the β coefficients varies from 1 up to 2 and there are approximately 25 regressors with a ratio (posterior mean over posterior standard deviation) higher than 1.5 and this number is quite satisfactory for a model with an average number of 17 regressors included.

In Figures 5.3, 5.4 and 5.5 we noticed that the probability of inclusion of different regions varies depending on the time period. In Figures 5.10, 5.11 and 5.12 we draw a scatter plot between the posterior mean of the search elasticity (the β coefficients) for each state and their probability of inclusion for the Nintendo Wii, the PS3 and Xbox 360, respectively. The vertical and horizontal lines correspond with the average inclusion probability and the average search elasticity, respectively. What we see in all three figures is that the place where states appear varies not only relative to their inclusion probabilities but also relative to the search elasticities. For example, in Figure 5.10 we see that the states with above average search elasticity and above

⁵We assume that the inclusion probability of each state is independent and identically distributed from other states and they follow a uniform with range [0,1]. We draw the probability for every state from the uniform and then we compute the Moran's I and Geary's C for L number of draws to obtain the probability distribution of these two statistics.

average inclusion probability (upper right quadrant) during the first period are Kansas, New Hampshire, Delaware, New Mexico, Nebraska, Arizona, New Jersey and California. However, the upper right quadrant states that appear in the following periods are different. For example, during the forth period the upper-quadrant states are North Dakota, Montana, Maine and New Hampshire. The Figures 5.11 and 5.12 for the PS3 and Xbox 360 confirm the same pattern, different groups of states appear at each quadrant of the scatter plots at each sub-period. These results point that some states may be important earlier in the diffusion of a technology while other states become important during later states of the diffusion. Note that this result is not explained by different introduction dates as the three consoles were launched simultaneously in all US states.

The sales-search model takes into account the relationship between aggregate sales and the online search at many different locations. This provides with interesting inclusion probabilities and we can rank the states according to their power to drive the aggregate sales. If we were to ignore all these details and we run a simple regression between aggregate sales and aggregate online search we obtain the results reported in Table 5.13. The overall sensitivity of sales to aggregate search (an indicator of search for all US) is larger than the sensitivity of sales to state-specific search. The estimates range from 0.17 up to 0.46, see the coefficient of search in this table. These last results seem intuitive but we miss the detailed region-specific analysis and a possible spatial story behind the results of the sales search model.

5.5.2 Results of the Spatial Model

In Table 5.5 we report the posterior mean and the posterior standard deviation of the δ coefficients of the spatial model (5.3). In the Table we report the δ coefficients for a set of seven variables. We tested other demographic variables measuring the ethnic origin and age distribution but we did not find them as significant and they were highly correlated with the set of seven variables that we kept in the model.

As we can observe, our results indicate that there is not a very strong association between demographic variables and the inclusion probabilities at each state. The reason why the posterior standard deviations might be large is because we have only 48 states in the probability model and therefore we have very few observations to estimate the coefficients. A second reason may be that we observe a relatively small variation in our dependent variable. Nonetheless, we find some interesting features in the δ_n coefficients.

The variables that seem to be relevant are the percentage of the population in college dorms and the percentage of the population that is married (percentage of households with married couples). Both of these variables are somewhat significant during the first and second diffusion periods. The effect of travel time to work is not significant but it is most of the time negative, as we would expect given than longer commuting time reduces leisure time to play video games or to search for consoles. Population density and income per capita seem to be slightly more important in the last diffusion stage while in the first stages of diffusion they are not. A last important feature to notice is that in many cases the size and sign of the δ_n coefficients may vary according to the diffusion stage of the products. For example, it may be that students and married couples tend to buy more video-game consoles at an early stage, as a high proportion of these groups increases the chance of a state being influential, while these groups may not buy at the end of the diffusion when we see that other parameters like population density and income per capita are slightly more important.

We estimate the spatial random effects Φ_n along side with the δ_n coefficients and we report their posterior mean and their posterior mean divided by their posterior standard deviation in Tables 5.6, 5.7, 5.8 and 5.9 for the first, second, third and fourth diffusion periods, respectively. In contrast with the δ_n coefficients, several of the spatial effects are significant. For example, in Table 5.6 we see that the spatial effect of Texas is significant both for the Nintendo Wii and the PS3 while it is not for the Xbox 360. This means that Texas is more likely to be driving the sales of the Wii and PS3 relative to the Xbox 360 during the first diffusion period. In the same table we notice that Ohio, South Dakota and Washington are positive and significant for the Xbox 360. The spatial effect of Washington is significant for all three technologies. Tables 5.7, 5.8 and 5.9 show similar many significant spatial effects during the rest of the diffusion periods.

In Figures 5.13, 5.14, 5.15 and 5.16 we report the distribution of the spatial effects for the Nintendo Wii and the first, second, third and fourth diffusion periods, respectively. In Figure 5.13 we can observe that for the first diffusion period the states with higher posterior spatial effects are Alabama, Delaware, Kentucky, Texas, Washington and Wyoming. The states with the lowest spatial effects are Georgia, Massachusetts, Missouri and Rhode Island. Texas and Wyoming continue to have a high spatial effect in the next diffusion period, see Figure 5.14 but the other states that had a high spatial effect in the first period no longer continue to be high in the second. In general, the spatial effect for each state varies according to the diffusion time of the technologies. For example, according to our results Texas is very influential for the Nintendo Wii at an early stage of its life-cycle while this state is not influential at the end of the life-cycle of the Wii.

We are finding significant spatial random effects for several states and all diffusion periods. However, a natural concern is whether the δ_n coefficients may have a different level of significance if we were to exclude the spatial effects from equation (5.3). In Table 5.10 we report the same δ_n coefficients estimated with ordinary least squares and their level of significance is relatively the same as before. Again, the population in college dorms and the percentage of households with married couples seem to be the more important variables. That is, the spatial effects explain geographical variation without affecting the inference we draw from the posteriors of the δ_n coefficients.

In Table 5.11 we present the posterior distribution of the correlations derived from the matrix Λ . The matrix Λ is a 3 × 3 covariance matrix and it measures the covariance between the spatial effects of different technologies. In the first diffusion period, for example, we find that the correlation of the spatial effects of the Xbox 360 are negatively correlated with the spatial effects of the PS3. The posterior mean of the correlation is -0.257 and the association is significant (zero is almost excluded from the 95% highest posterior density region). This negative correlation implies that if a state is likely to drive the sales of the Xbox 360 then it is not likely to drive the sales of the PS3. The association between the spatial effects of the Wii and those of the Xbox and PS3 are not different from zero (in these cases 0 is almost in the middle of the highest posterior density region) during the first diffusion period. We find some other significant associations during the third and fourth periods while in the second period we find no association between the spatial effects of the different technologies. The variation in correlation structure shows that at an early stage there is some degree of competition only between the

PS3 and Xbox 360 (because of the negative correlation in their spatial effects) while at later stages technologies seem to nurture each other (because we find significant positive correlations in their spatial effects).

Finally, in Table 5.12 we report the highest posterior density region for the ρ coefficients. We find roughly the same spatial decay (or spatial correlation) during all diffusion periods. The posterior mean of the ρ_n for all n is around 0.82. This number should be between 0 and 1 and numbers close to 1 indicate high spatial correlation between a state and its neighbors. The estimate of the ρ coefficient together with the Φ_n spatial effects are evidence of significant clusters of spillover effects between states. We do not know the direction of influence between the states but the model parameters capture significant spatial dependence among neighboring states.

5.6 Conclusions

We applied Bayesian variable selection methods to identify the influential locations for the diffusion of new technologies. We define influential locations as those that are more likely to drive the aggregate sales of the technologies. For our particular data on game consoles, we find that the influential locations change over time and that there is geographical clustering that is significantly captured by the spatial random effects in the probability model and by different measures of spatial association.

Moreover, we find variation in the groups of influential locations over time and the size of their associated search elasticity varies over time too. The search elasticity for the technologies at influential locations is on average 0.15. That is, an increase of 10% in local (state) search translates into a 1.5% increase in country level sales. Finally, we find some evidence of time variation in the association between spatial affects. Our results suggest that the geographical clustering is not driven by demographic heterogeneity and we find some evidence that suggests that the demographic effects vary over time.

In summary, our results suggest that influential locations may change over time together with the relationship between technologies and the relevance of demographics. The main managerial implications of this research is the notion that the group of influential locations is not fixed and therefore when a manager is looking to identify influentials, she or he should expect influentials to play a role at different locations and at different times. If managers were to ignore the spatial heterogeneity they will miss the valuable insights of how to allocate their marketing efforts based on the important locations for their products. The relevant question is not only who is influential but where and when and for how long a consumer or a group of consumers is influential.

5.7 Tables and Figures

```
library(RCurl)
library(XML)
wii_sales<-rep(0,416)</pre>
week.numbers<-seq((39838)-2184,40358,by=7)
for(i in 1:416)
ſ
part1<-"http://vgchartz.com/hwtable.php?cons[]=Wii&reg[]=America&start="</pre>
part2<-"&end="
week<-week.numbers[i]
url.dir<-paste(part1,week,part2,week,sep="")</pre>
url.text <- getURL(url.dir)</pre>
doc <- htmlParse(url.text,useInternalNodes=TRUE, error=function(...){})</pre>
x = xpathSApply(doc, "//table//td|//table//th", xmlValue)
wii_sales[i] <- as.numeric(gsub(",", ".", x[12]))</pre>
}
write.csv(wii_sales,file="wii_data.csv")
```

Note that the keyword Wii should be changed to PS3 or X360 to retrieve the data for each of these consoles.

Table 5.1: R Code to Retrieve Data from VGChartz.com

	Posterior Inclusion Probabilities						
	1st Period	2nd Period	3rd Period	4th Period			
Alabama	0.111	0.083	0.080	0.117			
Alaska	0.074	0.069	0.091	0.115			
Arizona	0.092	0.110	0.070	0.075			
Arkansas	0.091	0.093	0.103	0.089			
California	0.093	0.093	0.081	0.116			
Colorado	0.076	0.069	0.098	0.079			
Connecticut	0.074	0.057	0.103	0.071			
Delaware	0.105	0.058	0.079	0.103			
District of Columbia	0.076	0.108	0.077	0.083			
Florida	0.096	0.061	0.090	0.077			
Georgia	0.056	0.079	0.089	0.076			
Hawaii	0.072	0.096	0.077	0.099			
Idaho	0.086	0.082	0.075	0.112			
Illinois	0.073	0.092	0.080	0.066			
Indiana	0.079	0.059	0.065	0.082			
Iowa	0.077	0.077	0.083	0.125			
Kansas	0.108	0.085	0.088	0.083			
Kentucky	0.075	0.000	0.003	0.000			
Louisiana	0.102	0.122	0.081	0.065			
Maine	0.102	0.080	0.001	0.005			
Maryland	0.079	0.088	0.057	0.079			
Massachusette	0.084	0.119	0.096	0.073			
Michigan	0.034	0.079	0.086	0.086			
Minnesota	0.078	0.008	0.088	0.030			
Mississippi	0.058	0.002	0.105	0.060			
Missouri	0.036	0.032	0.105	0.000			
Montana	0.000	0.084	0.000	0.173			
Nebraeka	0.033	0.073	0.093	0.000			
Novodo	0.092	0.006	0.068	0.090			
New Hampshire	0.000	0.097	0.076	0.154			
New Jersey	0.105	0.127	0.103	0.073			
New Mexico	0.035	0.026	0.103	0.105			
New York	0.033	0.050	0.080	0.054			
North Carolina	0.096	0.071	0.083	0.066			
North Dakota	0.081	0.086	0.082	0.000			
Ohio	0.078	0.000	0.102	0.089			
Oklahoma	0.078	0.090	0.081	0.003			
Oregon	0.002	0 144	0.063	0.055			
Pennsylvania	0.058	0.081	0.005	0.055			
Rhode Island	0.086	0.074	0.005	0.101			
South Carolina	0.000	0.074	0.082	0.007			
South Dakota	0.098	0.079	0.035	0.098			
Tennessee	0.000	0.073	0.110	0.058			
Torse	0.032	0.075	0.086	0.000			
Utah	0.097	0.007	0.089	0.097			
Vermont	0.076	0.073	0.005	0.091			
Virginia	0.100	0.070	0.079	0.086			
Washington	0.126	0.073	0.065	0.000			
West Virginia	0.073	0.075	0.005	0.119			
Wisconsin	0.072	0.002	0.131	0.060			
Wyoming	0.072	0.115	0.107	0.119			
Note: In hold probabi	lities larger t	han 0 10	0.101	0.110			

Table 5.2: State Inclusion Probabilities for Each Diffusion Period for the Nintendo Wii

	Posterior Inclusion Probabilities						
	1st Period	2nd Period	3rd Period	4th Period			
Alabama	0.088	0.073	0.094	0.086			
Alaska	0.081	0.094	0.084	0.185			
Arizona	0.081	0.063	0.091	0.057			
Arkansas	0.101	0.090	0.098	0.093			
California	0.096	0.096	0.088	0.080			
Colorado	0.106	0.092	0.092	0.099			
Connecticut	0.104	0.102	0.093	0.076			
Delaware	0.086	0.078	0.118	0.090			
District of Columbia	0.088	0.099	0.098	0.075			
Florida	0.100	0.079	0.091	0.083			
Georgia	0.095	0.097	0.103	0.067			
Hawaii	0.098	0.080	0.094	0.088			
Idaho	0.092	0.085	0.080	0.076			
Illinois	0.091	0 107	0.082	0.088			
Indiana	0.085	0.081	0.104	0.085			
Iowa	0.087	0.102	0.087	0.000			
Kansas	0.079	0.094	0.083	0.080			
Kentucky	0.070	0.094	0.084	0.085			
Louisiana	0.087	0.093	0.004	0.000			
Maina	0.086	0.033	0.031	0.115			
Maryland	0.080	0.110	0.075	0.113			
Massachusette	0.095	0.115	0.085	0.093			
Michigan	0.095	0.095	0.081	0.071			
Minnocoto	0.039	0.105	0.081	0.086			
Mississinni	0.075	0.005	0.081	0.030			
Missouri	0.084	0.003	0.087	0.078			
Mantana	0.004	0.095	0.087	0.004			
Nohroako	0.091	0.009	0.089	0.103			
Neoraska	0.092	0.109	0.069	0.095			
Nevada	0.096	0.087	0.090	0.072			
New Hampshire	0.091	0.087	0.090	0.140			
New Jersey	0.090	0.094	0.072	0.071			
New Wexico	0.085	0.097	0.009	0.105			
New IOIK	0.090	0.069	0.095	0.004			
North Dalasta	0.103	0.085	0.084	0.071			
North Dakota	0.070	0.076	0.084	0.094			
Ohio	0.088	0.097	0.105	0.073			
Oklanoma	0.080	0.091	0.091	0.084			
Demoscie	0.104	0.077	0.101	0.102			
Pennsylvania	0.089	0.091	0.079	0.074			
Knode Island	0.081	0.087	0.082	0.130			
South Carolina	0.090	0.092	0.076	0.090			
South Dakota	0.066	0.068	0.079	0.094			
rennessee	0.089	0.087	0.095	0.091			
lexas	0.108	0.093	0.113	0.065			
Utan	0.101	0.072	0.109	0.097			
vermont	0.090	0.086	0.100	0.141			
Virginia	0.096	0.083	0.062	0.073			
Washington	0.101	0.081	0.095	0.069			
west Virginia	0.074	0.083	0.106	0.097			
Wisconsin	0.089	0.087	0.092	0.095			
Wyoming	0.092	0.086	0.094	0.090			
Note: In bold probabi	lities larger th	nan 0.10					

Table 5.3: State Inclusion Probabilities for Each Diffusion Period for the Sony PS3

	Posterior Inclusion Probabilities						
	1st Period	2nd Period	3rd Period	4th Period			
Alabama	0.085	0.091	0.090	0.079			
Alaska	0.104	0.188	0.080	0.199			
Arizona	0.077	0.074	0.080	0.054			
Arkansas	0.098	0.082	0.087	0.075			
California	0.099	0.082	0.075	0.074			
Colorado	0.078	0.088	0.087	0.084			
Connecticut	0.078	0.081	0.091	0.101			
Delaware	0.116	0.136	0.075	0.204			
District of Columbia	0.091	0.096	0.097	0.071			
Florida	0.102	0.066	0.100	0.065			
Georgia	0.084	0.073	0.115	0.092			
Hawaii	0.089	0.115	0.055	0.076			
Idaho	0.087	0.109	0.076	0.137			
Illinois	0.086	0.075	0.105	0.100			
Indiana	0.086	0.087	0.074	0.064			
Iowa	0.097	0.126	0.083	0.100			
Kansas	0.114	0.082	0.087	0.081			
Kentucky	0.103	0.078	0.101	0.109			
Louisiana	0.067	0.074	0.082	0.058			
Maine	0.097	0.113	0.097	0.095			
Maryland	0.087	0.066	0.085	0.087			
Massachusetts	0.095	0.100	0.085	0.079			
Michigan	0.092	0.076	0.096	0.082			
Minnesota	0.097	0.092	0.073	0.095			
Mississippi	0.096	0.062	0.131	0.080			
Missouri	0.079	0.087	0.098	0.073			
Montana	0.071	0.059	0.087	0.096			
Nebraska	0.084	0.067	0.071	0.095			
Nevada	0.093	0.074	0.071	0.084			
New Hampshire	0.089	0.098	0.089	0.119			
New Jersev	0.085	0.110	0.095	0.071			
New Mexico	0.091	0.112	0.071	0.100			
New York	0.083	0.106	0.101	0.093			
North Carolina	0.091	0.103	0.090	0.066			
North Dakota	0.129	0.082	0.113	0.113			
Ohio	0.099	0.094	0.083	0.079			
Oklahoma	0.086	0.085	0.084	0.094			
Oregon	0.116	0.081	0.087	0.081			
Pennsylvania	0.096	0.087	0.093	0.085			
Rhode Island	0.102	0.113	0.092	0.127			
South Carolina	0.090	0.081	0.082	0.073			
South Dakota	0.132	0.097	0.118	0.102			
Tennessee	0.084	0.096	0.134	0.089			
Texas	0.093	0.073	0.078	0.063			
Utah	0.085	0.088	0.077	0.059			
Vermont	0.082	0.110	0.152	0.082			
Virginia	0.103	0.074	0.078	0.064			
Washington	0.111	0.101	0.100	0.079			
West Virginia	0.085	0.065	0.140	0.114			
Wisconsin	0.090	0.087	0.093	0.086			
Wyoming	0.070	0.125	0.105	0.084			
Note: In bold probabi	lities larger t	nan 0.10					

Table 5.4: State Inclusion Probabilities for Each Diffusion Period for the Microsoft Xbox 360

	MCAP First Diffusion Poriod							
	Coefficient	St Dev	t_value					
Intercent	2 2684	0.0162	145 9191					
Mala Fomala Patio	-2.3064	0.0105	-140.2101					
Population Dongity	0.0103	0.0320	0.3132					
Population in Collage Dorma	0.0042	0.0270	1.6920					
Manniad Counts	0.0337	0.0200	1.0620					
Travel Time to Werk	0.0230	0.0171	0.0751					
Inaver Time to Work	-0.0015	0.0194	-0.0751					
income per Capita	0.0100	0.0157	0.0725					
	MCAR S	econd Diffus	ion Period					
	Coefficient	St. Dev	t-value					
Intercept	-2.4029	0.0177	-135.4291					
Male Female Ratio	0.0345	0.0361	0.9561					
Population Density	0.0357	0.0285	1.2506					
Population in College Dorms	0.0304	0.0232	1.3138					
Married Couple	-0.0208	0.0202	-1.0332					
Travel Time to Work	-0.0183	0.0221	-0.8307					
Income per Capita	-0.0185	0.0185	-1.0007					
meenie per cupita	0.0100	010100	110001					
	MCAR 7	Third Diffusi	on Period					
	Coefficient	St. Dev.	t-value					
Intercept	-2.3715	0.0214	-110.8737					
Male Female Ratio	-0.0192	0.0409	-0.4694					
Population Density	-0.0294	0.0343	-0.8562					
Population in College Dorms	-0.0245	0.0251	-0.9758					
Married Couple	0.0231	0.0251	0.9208					
Travel Time to Work	0.0163	0.0258	0.6329					
Income per Capita	-0.0103	0.0199	-0.5181					
· · · ·								
	MCAR F	ourth Diffusi	ion Period					
	Coefficient	St. Dev.	t-value					
Intercept	-2.3987	0.0283	-84.6213					
Male Female Ratio	-0.0305	0.0589	-0.5171					
Population Density	0.0464	0.0474	0.9795					
Population in College Dorms	-0.0218	0.0339	-0.6436					
Married Couple	-0.0013	0.0324	-0.0402					
Travel Time to Work -0.0157 0.0352 -0.4457								
Income per Capita 0.0191 0.0267 0.7141								
Note: The first column reports	s the posterio	r mean of th	e coefficient.					
The second column reports the	e posterior sta	andard devia	tion and the					
third column reports the ratio of	of the posterio	r mean over	the posterior					
standard deviation, called here t-value.								

Table 5.5: Posterior of MCAR δ coefficients

5.7 Tables and Figures

			MCAR First	Diffusion	Period	
	Wii	t-value	PS3	t-value	Xbox360	t-value
Alabama	0.260	3.990	0.025	0.380	-0.011	-0.196
Arizona	0.053	0.780	-0.066	-1.203	-0.131	-2.336
Arkansas	0.041	0.770	0.141	2.329	0.110	2.093
California	-0.038	-0.481	-0.001	-0.035	0.023	0.356
Colorado	-0.164	-2.412	0.154	2.322	-0.141	-2.268
Connecticut	-0.213	-2.762	0.135	1.903	-0.155	-2.167
Delaware	0.201	3.464	-0.005	-0.103	0.302	5.033
Florida	0.081	1.306	0.132	2.304	0.150	2.406
Georgia	-0.462	-6 174	0.063	1.322	-0.055	-1 172
Idaho	-0.243	-3 420	0.053	0.952	-0.034	-0.525
Illinois	-0.070	-1.349	-0.007	-0.261	-0.053	-1 149
Indiana	-0.232	-4 463	-0.014	-0.237	-0.066	-1 417
Iowa	-0.138	-2.370	-0.058	-1 276	-0.051	-0.861
Kansas	-0.162	-3.003	-0.037	-0.728	0.074	1 578
Kentucky	0.205	3 492	-0.101	-2.010	0.255	4 228
Louisiana	-0.097	-1 705	-0.174	-2.826	0.200	3.517
Maine	0.183	2.815	0.010	0.151	-0.254	-4 285
Maryland	-0.099	-1 467	-0.020	-0.460	0.105	1 764
Margachusette	-0.443	-5.979	0.020	0.278	-0.064	-0.882
Michigan	-0.078	-1.512	0.021	0.270	0.004	0.052
Minnocoto	-0.078	4 806	0.041	0.751	0.047	0.352
Mississippi	-0.204	-4.800	-0.037	-0.701	0.009	2 150
Missouri	-0.008	5 600	-0.137	-2.282	0.152	1.520
Montana	-0.013	-0.253	-0.020	-0.700	-0.095	-1.832
Nohnadia	-0.013	1.959	-0.037	-0.700	-0.035	-1.052
Novodo	0.009	0.620	0.020	0.343	-0.217	-3.822
New Hampshire	0.050	0.029	0.055	1.082	-0.030	0.455
New Iorgov	0.000	1 449	0.000	0.121	0.027	0.400
New Merrice	0.134	0.990	-0.011	1 222	-0.034	-0.400
New Wexico	0.129	2.000	0.000	1.220	0.012	0.145
New IOIK	0.052	2 2 47	-0.110	-1.550	-0.029	-0.410
North Dalvata	-0.100	-3.347	0.040	0.900	-0.099	-1.794
Obio	0.069	0.179	0.155	2.331	0.037	0.008 E 94E
Oldehome	-0.120	-2.172	-0.200	-3.950	0.342	0.240
Okianoma	-0.125	-2.007	0.003	1.695	0.110	2.305
Dregon	-0.005	-1.420	-0.081	-1.080	-0.020	-0.305
Pennsylvania Dhada Island	0.027	0.433	0.075	1.114	0.187	2.729
Rhode Island	-0.289	-3.190	0.059	0.587	0.133	1.393
South Carolina	0.035	0.472	0.044	0.780	0.043	0.771
South Dakota	0.108	2.093	-0.276	-4.334	0.409	0.108
Tennessee	0.034	0.688	0.006	0.111	-0.054	-1.247
Texas	0.314	4.926	0.128	2.476	-0.020	-0.262
Utah	0.009	0.147	0.047	0.716	-0.124	-1.475
vermont	-0.139	-2.786	0.037	0.498	-0.057	-1.164
v irginia	0.082	1.547	0.038	0.831	0.109	2.372
Washington	0.353	6.191	0.121	2.318	0.224	4.035
West Virginia	-0.160	-2.826	-0.142	-2.508	-0.011	-0.243
Wisconsin	-0.242	-4.815	-0.024	-0.626	-0.013	-0.288
Wyoming	0.180	2.908	0.033	0.611	-0.239	-3.781
Note: The numbe	ers corresp	pond to th	e Φ paramete	ers of the N	ACAR model fo	r the first

diffusion period. We report the posterior mean of the spatial effects for the Wii, PS3 and X360 and the ratio of the posterior mean over the posterior standard deviation.

|--|

Wii t-value PS3 t-value Xbox360 t-value Alabama 0.005 0.082 -0.106 -1.394 0.096 1.139 Arizona 0.231 2.843 -0.278 -2.827 -0.148 -1.867 Arkansas 0.096 1.393 0.061 0.972 -0.024 -0.365 California -0.004 -0.039 0.019 0.185 -0.134 -1.459 Colorado -0.211 -2.324 0.053 0.683 0.005 0.091 Connecticut -0.392 -4.157 0.172 1.713 -0.061 -0.559 Delaware -0.350 -4.939 -0.020 -0.334 0.217 2.856 Illinois -0.063 -0.859 -0.020 -0.334 0.217 2.803 Indian 0.046 5.992 0.180 2.409 -0.163 2.212 Iowa -0.349 -4.061 -0.047 -0.740 0.017 0.274 Kas			M	CAR Secon	d Diffusio	n Period	
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		Wii	t-value	PS3	t-value	Xbox360	t-value
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	Alabama	0.005	0.082	-0.106	-1.394	0.096	1.139
Arkansas 0.096 1.393 0.061 0.972 -0.024 -0.365 California -0.004 -0.039 0.019 0.185 -0.134 -1.459 Colorado -0.211 -2.324 0.053 0.683 0.005 0.091 Connecticut -0.365 -4.093 -0.080 -1.367 -0.284 -3.924 Georgia -0.072 -1.085 0.118 1.746 -0.153 -2.014 Idaho 0.113 1.363 -0.057 -0.792 0.288 3.083 Illinois -0.063 -0.859 -0.202 -0.334 0.217 2.850 Indiana 0.046 0.592 0.180 2.409 -0.152 -2.212 Iowa -0.349 -4.061 -0.047 -0.740 0.017 0.274 Kansas -0.100 -1.419 0.159 2.145 0.382 4.666 Kentucky 0.005 0.129 0.095 1.494 -0.030 -0.495 Louisiana 0.072 0.934 0.127 1.718 -0.089 -0.191 Maine 0.406 5.044 0.123 1.628 -0.104 -1.115 Maryland -0.055 -0.321 -0.321 -3.198 Michigan 0.306 4.063 0.053 0.796 0.129 1.833 Minesota -0.084 -1.101 0.223 2.738 -0.111 -1.645 Missigipi 0.139 1.745 -0.203 $-2.$	Arizona	0.231	2.843	-0.278	-2.827	-0.148	-1.867
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Arkansas	0.096	1.393	0.061	0.972	-0.024	-0.365
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	California	-0.004	-0.039	0.019	0.185	-0.134	-1.459
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Colorado	-0.211	-2.324	0.053	0.683	0.005	0.091
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Connecticut	-0.392	-4.157	0.172	1.713	-0.061	-0.599
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Delaware	-0.365	-4.093	-0.080	-0.950	0.482	5.568
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Florida	-0.350	-4.939	-0.090	-1.367	-0.284	-3.924
Idaho0.1131.363-0.057-0.7920.2883.083Illinois-0.063-0.859-0.020-0.3340.2172.850Indiana0.0460.5920.1802.409-0.152-2.212Iowa-0.349-4.061-0.047-0.7400.0170.274Kansas-0.100-1.4190.1592.1450.3824.606Kentucky0.0050.1290.0951.494-0.030-0.495Louisiana0.0720.9340.1271.718-0.089-1.073Maine0.4065.4040.1231.628-0.104-1.115Maryland-0.035-0.398-0.140-1.6700.3053.512Massachusetts-0.054-0.5810.2322.750-0.321-3.198Michigan0.3064.0630.0530.7960.1291.833Minnesota-0.0821.1560.0180.254-0.278-3.091Montana-0.163-2.1170.0440.563-0.017-0.213Nebraska-0.020-0.3010.0340.612-0.331-3.717Nevada-0.2121.8930.1611.571-0.299-2.607New Hampshire0.1762.2730.0740.955-0.086-1.124New Jersey0.0910.830-0.013-0.1460.1081.013Neth Carolina-0.259-3.5580.0100.1280.1772.434 <t< td=""><td>Georgia</td><td>-0.072</td><td>-1.085</td><td>0.118</td><td>1 746</td><td>-0.153</td><td>-2.014</td></t<>	Georgia	-0.072	-1.085	0.118	1 746	-0.153	-2.014
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Idaho	0.113	1 363	-0.057	-0 792	0.288	3 083
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Illinois	-0.063	-0.859	-0.020	-0.334	0.217	2.850
Inimit0.0349-4.061-0.0470.7400.0170.274Kansas-0.100-1.4190.1592.1450.3824.606Kentucky0.0050.1290.0951.494-0.030-0.495Louisiana0.0720.9340.1271.718-0.089-1.073Maine0.4065.4040.1231.628-0.104-1.115Maryland-0.035-0.398-0.140-1.6700.3053.512Massachusetts-0.064-0.6310.2322.750-0.221-3.198Michigan0.3064.0630.0530.7960.1291.833Minnesota-0.084-1.1010.2232.738-0.111-1.645Missouri0.0821.1560.0180.254-0.278-3.091Montana-0.163-2.1170.0440.563-0.017-0.213Nebraska-0.020-0.3010.0340.612-0.351-3.717Nevada-0.212-1.8930.1611.571-0.299-2.607New Hampshire0.1762.2730.0740.955-0.086-1.124New Jersey0.0910.830-0.013-0.1460.1081.013New Mexico0.3724.6610.0691.0760.2342.975New York0.0290.2720.0320.3910.1821.950North Dakota-0.259-3.5880.0100.1881.736Ohio<	Indiana	0.046	0.592	0.180	2 409	-0.152	-2.000
Kansas 0.0100 1.419 0.059 0.140 0.011 0.012 0.030 0.0495 1.030 0.0495 1.013 1.011 0.035 0.0305 3.512 0.0355 0.0111 0.111 1.645 0.035 0.011 0.0117 0.0218 0.017 0.0218 0.017 0.0218 0.017 0.0218 0.017 0.0218 0.017 0.0239 0.025 0.0134 0.029 0.272 0.032 0.032	Iowa	-0.349	-4.061	-0.047	-0.740	0.017	0.274
Minust0.1001.1150.1031.4940.0301.003Kentucky0.0050.1290.0951.494-0.030-0.495Louisiana0.0720.9340.1271.718 -0.089 -1.073Maine0.4065.4040.1231.628-0.104-1.115Maryland-0.035-0.398-0.140-1.6700.3053.512Massachusetts-0.054-0.5810.2322.550-0.321-3.198Michigan0.3064.0630.0530.7960.1291.833Minnesota-0.084-1.1010.2232.738-0.111-1.645Mississippi0.1391.745-0.203-2.3840.0851.070Missouri0.0821.1560.0180.254-0.278-3.091Montana-0.163-2.1170.0440.563-0.017-0.213Nebraska-0.020-0.3010.0340.612-0.351-3.717Nevada-0.212-1.8930.1611.571-0.299-2.607New Hampshire0.1762.2730.0740.955-0.086-1.124New Mexico0.3724.6610.0691.0760.2342.975New York0.0290.2720.0320.3910.1821.950North Dakota-0.237-2.899-0.075-1.0590.1341.736Ohio-0.036-0.403-0.137-1.909-0.287-1.222	Kansas	-0.100	-1 419	0.159	2 145	0.382	4 606
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Kontucky	0.005	0.120	0.105	1 494	-0.030	-0.495
Dominanta 0.012 0.124 1.116 -0.035 -1.015 Maine 0.406 5.404 0.123 1.628 -0.104 -1.115 Maryland -0.035 -0.398 -0.140 -1.670 0.305 3.512 Massachusetts -0.054 -0.581 0.232 2.550 -0.321 -3.198 Michigan 0.306 4.063 0.053 0.796 0.129 1.833 Minnesota -0.084 -1.101 0.223 2.738 -0.111 -1.645 Mississippi 0.139 1.745 -0.203 -2.384 0.085 1.070 Missouri 0.082 1.156 0.018 0.254 -0.278 -3.091 Montana -0.163 2.117 0.044 0.563 -0.017 -0.213 Nebraska -0.020 -0.301 0.034 0.612 -0.351 -3.717 Newda -0.212 -1.893 0.161 1.571 -0.299 -2.607 New Hampshire 0.176 2.273 0.074 0.955 -0.086 -1.124 New York 0.029 0.272 0.032 0.391 0.182 1.950 North Carolina -0.257 -3.588 0.010 0.128 1.77 2.434 North Dakota -0.237 -2.899 -0.75 -0.087 -1.222 Okiahoma 0.052 0.714 0.102 1.671 0.087 1.135 Orio -0.36 -0.403 -0.137	Louisiana	0.005	0.934	0.035	1.434	-0.089	-1.073
$\begin{array}{llllllllllllllllllllllllllllllllllll$	Maine	0.012	5 404	0.127	1.628	-0.104	-1.115
Massachusetts -0.054 -0.581 0.232 2.550 -0.321 -3.198 Michigan 0.306 4.063 0.0232 2.570 -0.321 -3.198 Minesota -0.084 -1.101 0.223 2.738 -0.111 -1.645 Mississippi 0.139 1.745 -0.203 -2.384 0.085 1.070 Missouri 0.082 1.156 0.018 0.254 -0.278 -3.091 Montana -0.163 -2.117 0.044 0.563 -0.017 -0.213 Nebraska -0.020 -0.301 0.034 0.612 -0.351 -3.717 Nevada -0.212 -1.893 0.161 1.571 -0.299 -2.607 New Hampshire 0.176 2.273 0.074 0.955 -0.086 -1.124 New Jersey 0.091 0.830 -0.013 -0.146 0.108 1.013 New Mexico 0.372 4.661 0.069 1.076 0.234 2.975 New York 0.029 0.272 0.322 0.391 0.182 1.950 North Dakota -0.237 -2.899 -0.075 -1.059 0.134 1.736 Ohio -0.036 0.714 0.102 1.671 0.087 1.135 Oregon 0.128 1.746 0.050 0.816 -0.012 -1.268 Rhode Island -0.133 -1.648 -0.226 -0.79 -0.691 South Carolina -0.137 <td>Maryland</td> <td>-0.035</td> <td>-0.398</td> <td>-0.140</td> <td>-1.670</td> <td>0.305</td> <td>3 512</td>	Maryland	-0.035	-0.398	-0.140	-1.670	0.305	3 512
Michigan 0.306 4.063 0.053 0.796 0.129 1.833 Minnesota -0.084 -1.101 0.223 2.738 -0.111 -1.645 Mississippi 0.139 1.745 -0.203 -2.384 0.085 1.070 Missouri 0.082 1.156 0.018 0.254 -0.278 -3.091 Montana -0.163 -2.117 0.044 0.563 -0.017 -0.213 Nebraska -0.020 -0.301 0.034 0.612 -0.351 -3.717 Nevada -0.212 -1.893 0.161 1.571 -0.299 -2.607 New Hampshire 0.176 2.273 0.074 0.955 -0.086 -1.124 New Jersey 0.091 0.830 -0.013 -0.146 0.108 1.013 New Mexico 0.372 4.661 0.069 1.076 0.234 2.975 New York 0.029 0.272 0.032 0.391 0.182 1.950 North Carolina -0.259 -3.58 0.010 0.182 1.772 2.434 North Dakota -0.237 -2.899 -0.075 -1.059 0.134 1.736 Ohio -0.036 -0.403 -0.137 -1.999 -0.087 1.222 Oklahoma 0.052 0.714 0.102 1.671 0.877 1.355 Oregon 0.128 1.746 0.050 0.816 -0.012 -0.710 South Carolina $-0.$	Margachusette	-0.054	-0.581	0.140	2 550	-0.321	-3 108
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Michigan	0.306	4.063	0.252	0.796	0.120	1 833
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Minnesota	-0.084	-1.101	0.000	2 738	-0.111	-1.645
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Mississippi	0.130	1 745	-0.223	-2.384	0.085	1.070
$\begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$	Missouri	0.155	1.140	0.018	0.254	-0.278	-3.001
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Montana	-0.163	-2 117	0.013	0.563	-0.017	-0.213
Nevada -0.212 -1.893 0.054 0.012 -0.291 -5.111 Nevada -0.212 -1.893 0.161 1.571 -0.299 -2.607 New Hampshire 0.176 2.273 0.074 0.955 -0.086 -1.124 New Jersey 0.091 0.830 -0.013 -0.146 0.108 1.013 New Mexico 0.372 4.661 0.069 1.076 0.234 2.975 New York 0.029 0.272 0.032 0.391 0.182 1.950 North Carolina -0.259 -3.558 0.010 0.128 0.177 2.434 North Dakota -0.237 -2.899 -0.075 -1.059 0.134 1.736 Ohio -0.036 -0.403 -0.137 -1.909 -0.087 -1.222 Oklahoma 0.052 0.714 0.102 1.671 0.087 1.135 Oregon 0.128 1.746 0.050 0.816 -0.012 -0.202 Pensylvania 0.442 4.291 -0.143 -1.726 -0.105 -1.710 South Carolina -0.133 -1.663 0.080 1.223 -0.055 -0.710 South Dakota -0.102 -1.486 -0.216 -2.726 0.096 1.267 Texas -0.171 -2.121 0.043 0.518 -0.207 -2.642 Utah 0.160 1.554 -0.216 -2.726 0.096 1.267 Texas $-0.$	Nobrocko	0.020	0.201	0.034	0.612	0.251	2 717
New Hampshire0.1762.2730.1011.371 -0.239 -2.001 New Hampshire0.1760.2730.0740.955 -0.086 -1.124 New Mexico0.3724.6610.069 1.076 0.2342.975New York0.0290.2720.0320.3910.1821.950North Carolina -0.259 -3.558 0.0100.1280.1772.434North Dakota -0.237 -2.899 -0.075 -1.059 0.1341.736Ohio -0.036 -0.403 -0.137 -1.909 -0.087 -1.222 Oklahoma0.0520.7140.1021.6710.087 -1.135 Oregon0.1281.7460.0500.816 -0.012 -0.202 Pennsylvania0.4424.291 -0.143 -1.726 -0.105 -1.268 Rhode Island -0.133 -1.663 0.0801.223 -0.055 -0.710 South Carolina -0.133 -1.663 0.0801.223 -0.055 -0.710 South Dakota -0.102 -1.486 -0.216 -2.726 0.096 1.267 Texas -0.171 -2.121 0.0430.518 -0.207 -2.642 Utah0.160 1.554 -0.118 -1.298 0.076 0.766 Virginia -0.131 -1.648 0.025 0.432 0.269 3.278 Virginia -0.251 -2.873 0.018 -0.207 -2.642 <td>Novodo</td> <td>-0.020</td> <td>1 802</td> <td>0.054</td> <td>1 571</td> <td>-0.331</td> <td>-3.717</td>	Novodo	-0.020	1 802	0.054	1 571	-0.331	-3.717
New Jersey0.1012.2130.0140.505-0.060-1.124New Jersey0.0910.830-0.013-0.1460.1081.013New Mexico0.3724.6610.0691.0760.2342.975New York0.0290.2720.0320.3910.1821.950North Carolina-0.259-3.5580.0100.1280.1772.434North Dakota-0.237-2.899-0.075-1.0590.1341.736Ohio-0.036-0.403-0.137-1.909-0.087-1.222Oklahoma0.0520.7140.1021.6710.0871.135Oregon0.1281.7460.0500.816-0.012-0.202Pennsylvania0.4424.291-0.143-1.726-0.005-1.268Rhode Island-0.133-1.6630.0801.223-0.079-0.691South Carolina-0.137-1.9540.0270.5020.1081.570Texas-0.137-1.9540.0270.5020.1081.570Texas-0.137-1.6480.0250.4320.2693.278Virginia-0.181-2.476-0.013-0.137-0.125-1.923Washington-0.165-1.911-0.051-0.7600.1641.966West Virginia-0.251-2.8730.0180.230-0.211-2.454Wisconsin-0.143-1.8580.000-0.016-0.011	New Hampshire	-0.212	-1.033	0.101	0.055	-0.233	-2.007
New Mersico 0.372 4.661 0.169 1.076 0.234 2.975 New York 0.029 0.272 0.032 0.391 0.182 1.950 North Carolina -0.259 -3.558 0.010 0.128 0.177 2.434 North Dakota -0.237 -2.899 -0.075 -1.059 0.134 1.736 Ohio -0.036 -0.403 -0.137 -1.909 -0.087 1.222 Oklahoma 0.052 0.714 0.102 1.671 0.087 1.135 Oregon 0.128 1.746 0.050 0.816 -0.012 -0.202 Pennsylvania 0.442 4.291 -0.143 -1.726 -0.079 -0.691 South Carolina -0.133 -1.663 0.080 1.223 -0.055 -0.710 South Dakota -0.102 -1.486 -0.216 -2.726 0.096 1.267 Texas -0.171 -2.121 0.043 0.518 -0.207 -2.642 Utah 0.160 1.554 -0.118 -1.298 0.078 0.766 Vermont -0.137 -1.648 0.025 0.432 0.269 3.278 Virginia -0.251 -2.873 0.016 -0.216 -2.2454 Wisconsin -0.143 -1.858 0.000 -0.016 -0.011 -0.187	New Jersey	0.170	0.830	-0.013	-0.146	0.108	1 013
New Mexico 0.312 4.001 0.009 1.010 0.234 2.345 New York 0.029 0.272 0.032 0.391 0.182 1.950 North Carolina -0.259 -3.558 0.010 0.128 0.177 2.434 North Dakota -0.237 -2.899 -0.075 -1.059 0.134 1.736 Ohio -0.036 -0.403 -0.137 -1.909 -0.087 -1.222 Oklahoma 0.052 0.714 0.102 1.671 0.087 -1.222 Pensylvania 0.442 4.291 -0.143 -1.726 -0.105 -1.268 Rhode Island -0.139 -1.234 -0.022 -0.172 -0.079 -0.691 South Carolina -0.133 -1.663 0.801 -2.2726 0.096 1.267 Tenessee -0.137 -1.954 0.027 0.502 0.108 1.570 Texas -0.171 -2.121 0.043 0.518 -0.207 -2.642 Utah 0.160 1.554 -0.113 -1.298 0.078 0.766 Vermont -0.137 -1.648 0.025 0.432 0.269 3.278 Wirginia -0.251 -2.873 0.016 -0.011 -0.125 -1.923 Washington -0.156 -1.911 -0.051 -0.760 0.164 1.966 West Virginia -0.244 2.904 -0.031 -0.453 0.333 3.692	New Movico	0.031	4 661	-0.013	1.076	0.108	2.075
New Tork 0.029 0.272 0.032 0.037 0.182 1.900 North Carolina 0.259 0.259 0.358 0.010 0.128 0.177 2.434 North Dakota -0.237 -2.899 -0.075 -1.059 0.134 1.736 Ohio -0.036 -0.403 -0.137 -1.909 -0.087 -1.222 Oklahoma 0.052 0.714 0.102 1.671 0.087 1.135 Oregon 0.128 1.746 0.050 0.816 -0.012 -0.202 Pennsylvania 0.442 4.291 -0.143 -1.726 -0.105 -1.268 Rhode Island -0.133 -1.663 0.080 1.223 -0.055 -0.710 South Carolina -0.102 -1.486 -0.216 -2.726 0.096 1.267 Tennessee -0.137 -1.954 0.027 0.502 0.108 1.570 Texas -0.171 -2.121 0.043 0.518 -0.207 -2.642 Utah 0.160 1.554 -0.118 -1.298 0.078 0.766 Virginia -0.137 -1.648 0.025 0.432 0.269 3.278 Virginia -0.181 -2.476 -0.013 -0.137 -0.125 -1.923 Washington -0.156 -1.911 -0.051 -0.760 0.164 1.966 West Virginia -0.251 -2.873 0.000 -0.016 -0.011 -0.187 <	New Wexico	0.372	4.001	0.009	1.070	0.234	2.975
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	North Carolina	0.029	2 558	0.032	0.391	0.132	2 424
$\begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$	North Dakota	-0.209	-3.558	0.010	1.050	0.124	1 726
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Obio	-0.237	-2.033	-0.075	-1.009	0.134	1.730
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Oklahoma	-0.050	-0.403	-0.137	-1.909	-0.037	-1.222
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Onagon	0.052	1.746	0.102	0.816	0.007	0.202
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Dependencie	0.120	1.740	0.050	1.726	-0.012	-0.202
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Pennsylvania Dhodo Island	0.442	4.291	-0.145	-1.720	-0.105	-1.200
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	South Carolina	-0.139	1.662	-0.022	1 222	-0.075	-0.031
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	South Dakota	-0.133	-1.003	0.080	2 726	-0.000	-0.710
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Tourn Dakota	-0.102	-1.400	-0.210	-2.720	0.090	1.207
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Tennessee	-0.137	-1.954	0.027	0.502	0.108	1.570
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Itab	-0.171	-2.121	0.043	1.208	-0.207	-2.042 0.766
Vermon -0.151 -1.048 0.025 0.452 0.209 5.276 Virginia -0.181 -2.476 -0.013 -0.137 -0.125 -1.923 Washington -0.156 -1.911 -0.051 -0.760 0.164 1.966 West Virginia -0.251 -2.873 0.018 0.230 -0.211 -2.454 Wisconsin -0.143 -1.858 0.000 -0.016 -0.011 -0.187 Wyoming 0.254 2.904 -0.031 -0.454 0.333 3.692	Vormont	0.100	1.649	-0.118	-1.290	0.076	2.278
virginia -0.161 -2.470 -0.031 -0.137 -0.125 -1.923 Washington -0.156 -1.911 -0.051 -0.760 0.164 1.966 West Virginia -0.251 -2.873 0.018 0.230 -0.211 -2.454 Wisconsin -0.143 -1.858 0.000 -0.016 -0.011 -0.187 Wyoming 0.254 2.904 -0.031 -0.453 0.333 3.692	Virginia	-0.137	-1.040 9.476	0.020	0.432	0.209	0.270 1.099
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	v irginia Woolden at an	-0.181	-2.470	-0.013	-0.137	-0.120	-1.923
west virginia -0.251 -2.873 0.018 0.230 -0.211 -2.454 Wisconsin -0.143 -1.858 0.000 -0.016 -0.011 -0.187 Wyoming 0.254 2.904 -0.031 -0.453 0.333 3.692 Note: The number commendate the normalization of the MCAP with the state of the MCAP withe state of the MCAP with the state of the MCAP with the	wasnington	-0.156	-1.911	-0.051	-0.760	0.164	1.900
wisconsin -0.143 -1.858 0.000 -0.016 -0.011 -0.187 Wyoming 0.254 2.904 -0.031 -0.453 0.333 3.692 Note: The number commendate the non-metation of the MCAP with the theorem -0.415 -0.421 -0.421	west virginia	-0.251	-2.8/3	0.018	0.230	-0.211	-2.454
wyonning 0.254 2.904 -0.051 -0.455 0.353 3.692	wisconsin Watering	-0.143	-1.898	0.000	-0.010	-0.011	-0.187
where the manufacture composition of the compositions of the additional terms the	vv yoming	0.254	2.904	-0.031	-0.405	U.333	3.092

Note: The numbers correspond to the Φ parameters of the MCAR model for the second diffusion period. We report the posterior mean of the spatial effects for the Wii, PS3 and X360 and the ratio of the posterior mean over the posterior standard deviation.

Table 5.7:	Posterior	of MCAR	Spatial	Effects

5.7 Tables and Figures

	MCAR Third Diffusion Period							
	Wii	t-value	PS3	t-value	Xbox360	t-value		
Alabama	-0.155	-2.445	0.003	0.000	-0.045	-0.739		
Arizona	-0.247	-4.132	0.003	0.112	-0.122	-2.098		
Arkansas	0.085	1.788	0.038	0.831	-0.075	-1.489		
California	-0.016	-0.192	0.060	0.748	-0.093	-1.187		
Colorado	0.111	1.571	0.056	0.795	0.002	0.000		
Connecticut	0.216	2.797	0.107	1.416	0.093	1.206		
Delaware	-0.120	-2.013	0.292	4.726	-0.169	-2.349		
Florida	0.033	0.730	0.040	0.932	0.141	3.158		
Georgia	-0.003	-0.161	0.134	2.776	0.256	5.227		
Idaho	-0.209	-2.808	-0.011	-0.166	-0.548	-5.979		
Illinois	-0.143	-3.097	-0.077	-1.455	-0.135	-2.773		
Indiana	-0.104	-2.571	-0.069	-1.580	0.176	3.609		
Iowa	-0.343	-5.740	0.131	2.517	-0.213	-3.606		
Kansas	-0.089	-1 768	-0.039	-0.748	-0.090	-1 730		
Kentucky	-0.055	-1.379	-0.114	-2.935	-0.068	-1 564		
Louisiana	0.013	0.159	-0.083	-1 145	0.102	1 350		
Maine	-0.147	-2 509	-0.026	-0.453	-0.136	-2 225		
Maryland	0.029	0.381	-0.183	-2.747	0 101	1 594		
Massachusetts	-0.336	-3.908	0.069	0.842	0.068	0.830		
Michigan	0.107	2 373	-0.049	-1 242	-0.023	-0.579		
Minnesota	-0.028	-0.632	-0.045	-1.840	0.081	1.874		
Mississippi	-0.020	-0.737	-0.129	-1.040	-0.244	-3 254		
Missouri	0.140	2 612	-0.041	-0.930	0.369	5 740		
Montana	-0.035	-0.683	-0.039	-0.789	0.084	1 485		
Nobrocko	0.000	1 574	-0.035	0.577	0.004	0.086		
Nevada	0.074	0.727	0.020	0.368	-0.194	-1.727		
New Hampshire	-0.316	-5.361	-0.027	-0.516	-0.273	-4.767		
New Jersey	-0.070	-0.704	0.021	0.850	0.074	0.646		
New Movice	0.118	2.261	0.030	2.083	0.074	0.626		
New Wexico	0.110	2.201	-0.233	-3.365	0.030	1 207		
North Carolina	0.040	2 102	-0.147	-1.705	-0.120	2 221		
North Dakota	-0.032	1 262	0.059	1.367	0.142	0.152		
Obio	-0.019	-1.203	-0.082	-1.330	0.007	4 128		
Oklahoma	-0.040	-0.701	-0.010	2.062	0.280	2 205		
Orianoma	0.100	2.012	0.127	0.421	-0.109	-2.290		
Dependencie	-0.098	-2.221	0.018	0.431	-0.001	-1.229		
Pennsylvania Dhodo Island	-0.260	-3.205	0.170	2.233	0.035	1.246		
South Coroling	-0.222	-1.949	-0.023	-0.192	0.144	2.070		
South Daliota	-0.069	-1.495	-0.170	-2.974	-0.105	-2.070		
Journ Dakota	-0.252	-4.376	-0.132	-2.201	0.201	4.065		
Tennessee	0.270	0.944	0.047	1.130	0.396	0.843		
Iexas	-0.020	-0.361	0.250	4.013	-0.115	-1.882		
Utan V	-0.094	-0.951	0.098	0.941	-0.242	-2.204		
vermont	0.411	1.003	0.088	1.529	0.530	9.155		
v irginia	-0.113	-2.047	-0.343	-0.009	-0.124	-2.007		
washington	-0.321	-5.615	0.069	1.359	0.119	2.074		
west Virginia	0.117	1.873	0.100	1.595	0.386	5.353		
Wisconsin	0.404	7.422	0.040	0.977	0.056	1.251		
wyoming	0.180	2.606	0.042	0.677	0.154	2.353		
Note: The numbe	rs corresp	ond to the	Ψ paramete	ers of the M	UAR model for	tne third		

diffusion period. We report the posterior mean of the spatial effects for the Wii, PS3 and X360 and the ratio of the posterior mean over the posterior standard deviation.

Table 5.8: Posterior of MCAR Spatial Effe

Wiit-valuePS3t-valueXbox360t-valueAlabama0.3113.294-0.008-0.171-0.092-1.108Arizona-0.092-1.076-0.367-4.305-0.412-4.807Arkansas0.0440.6130.0781.217-0.123-1.864California0.3773.187-0.005-0.083-0.080-0.759Colorado-0.040-0.4490.1781.7380.0220.201Connecticut-0.386-3.484-0.306-2.844-0.017-0.144Delaware0.0790.999-0.045-0.6310.8069.027Florida-0.136-2.313-0.065-1.059-0.313-5.411Georgia-0.103-1.827-0.228-3.5860.0881.323Idaho0.1971.8420.0720.719-0.077-0.745Ilinois0.2774.295-0.109-1.9780.4857.363Indiana-0.258-4.5410.022-0.211-0.293-4.108Kansas0.3956.1890.0791.3200.1562.558Kentucky-0.027-0.567-0.062-1.127-0.049-0.780Louisiana0.1551.626-0.006-0.1620.2542.682Maine-0.3773.821-0.149-1.267Missouri-0.33-2.048Maryland0.3773.823-0.065-0.597-0.140-1.277			М	CAR Fourt	h Diffusior	n Period	
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		Wii	t-value	PS3	t-value	Xbox360	t-value
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Alabama	0.311	3.294	-0.008	-0.171	-0.092	-1.108
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Arizona	-0.092	-1.076	-0.367	-4.305	-0.412	-4.807
	Arkansas	0.044	0.613	0.078	1.217	-0.123	-1.864
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	California	0.377	3.187	-0.005	-0.083	-0.080	-0.759
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Colorado	-0.040	-0.449	0.178	1.738	0.022	0.201
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Connecticut	-0.386	-3.484	-0.306	-2.844	-0.017	-0.144
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Delaware	0.079	0.999	-0.045	-0.631	0.806	9.027
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Florida	-0.136	-2.313	-0.065	-1.059	-0.313	-5.411
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Georgia	-0.103	-1.827	-0.228	-3.586	0.088	1.323
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Idaho	0.197	1.842	0.072	0.719	-0.077	-0.745
	Illinois	0.277	4.295	-0.109	-1.978	0.485	7.363
	Indiana	-0.258	-4.541	0.022	0.413	0.158	3.015
Kansas 0.395 6.189 0.079 1.320 0.156 2.558 Kentucky -0.027 -0.567 -0.062 -1.127 -0.049 -0.780 Louisiana 0.155 1.626 -0.006 -0.162 0.254 2.682 Maine -0.307 3.621 -0.149 -1.864 -0.412 -4.844 Maryland 0.372 3.837 0.191 2.154 0.002 -0.065 Massachusetts -0.233 -2.048 -0.065 -0.597 -0.140 -1.277 Michigan -0.155 -2.323 -0.191 -3.465 -0.098 -1.414 Minnesota 0.006 0.061 0.007 0.100 0.113 1.63 Missouri -0.374 -4.820 -0.103 -1.523 -0.076 -1.138 Montana 0.118 1.436 0.006 0.001 -0.132 -1.815 Nebraska 0.729 8.505 0.187 2.936 0.118 1.865 Nevada 0.132 0.819 0.160 1.043 0.180 1.101 New Hampshire 0.066 0.846 -0.196 -2.587 -0.054 -0.703 New Hacico -0.151 -2.031 -0.166 -2.577 -0.181 -2.489 New York 0.219 1.758 0.203 1.710 0.162 1.303 North Dakota -0.247 -3.025 -0.166 -2.722 -0.071 -1.272 Oregon -0	Iowa	-0.063	-0.839	-0.021	-0.271	-0.293	-4.108
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Kansas	0.395	6.189	0.079	1.320	0.156	2.558
Louisiana 0.155 1.626 -0.006 -0.162 0.254 2.682 Maine -0.307 -3.621 -0.149 -1.864 -0.412 -4.844 Maryland 0.372 3.837 0.191 2.154 0.002 -0.065 Massachusetts -0.233 -2.048 -0.065 -0.597 -0.140 -1.277 Michigan -0.155 -2.323 -0.191 -3.465 -0.089 -1.414 Minnesota 0.006 0.061 0.007 0.110 -0.038 -0.671 Mississippi -0.374 -4.820 -0.103 -1.523 -0.076 -1.138 Montana 0.118 1.436 0.006 0.001 -0.132 -1.815 Nebraska 0.729 8.505 0.187 2.936 0.118 1.865 Nevada 0.132 0.816 -0.196 -2.587 -0.054 -0.703 New Hampshire 0.066 0.846 -0.196 -2.587 -0.054 -0.703 New Jersey 0.394 2.581 0.285 1.962 0.124 0.840 New Mexico -0.151 -2.031 -0.168 -2.577 -0.181 -2.489 New York 0.219 1.758 0.235 0.260 3.472 North Dakota -0.247 -3.025 -0.166 -1.989 -0.251 -3.191 Ohio 0.805 8.115 0.075 0.965 0.260 3.472 Oregon -0.703 <	Kentucky	-0.027	-0.567	-0.062	-1.127	-0.049	-0.780
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Louisiana	0.155	1.626	-0.006	-0.162	0.254	2.682
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Maine	-0.307	-3.621	-0.149	-1.864	-0.412	-4.844
$\begin{array}{llllllllllllllllllllllllllllllllllll$	Marvland	0.372	3.837	0.191	2.154	0.002	-0.065
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Massachusetts	-0.233	-2.048	-0.065	-0.597	-0.140	-1.277
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Michigan	-0.155	-2.323	-0.191	-3.465	-0.089	-1.414
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Minnesota	0.006	0.061	0.007	0.110	-0.038	-0.671
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Mississippi	-0.137	-1.491	0.007	-0.006	0.113	1.163
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Missouri	-0.374	-4.820	-0.103	-1.523	-0.076	-1.138
$\begin{array}{llllllllllllllllllllllllllllllllllll$	Montana	0.118	1.436	0.006	0.001	-0.132	-1.815
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Nebraska	0.729	8,505	0.187	2.936	0.118	1.865
$\begin{array}{llllllllllllllllllllllllllllllllllll$	Nevada	0.132	0.819	0.160	1.043	0.180	1.101
New Jersey 0.394 2.581 0.285 1.962 0.124 0.840 New Mexico -0.151 -2.031 -0.168 -2.577 -0.181 -2.489 New York 0.219 1.758 0.203 1.710 0.162 1.303 North Carolina -0.462 -7.091 -0.277 -4.951 0.081 1.415 North Dakota -0.247 -3.025 -0.166 -1.989 -0.251 -3.191 Ohio 0.805 8.115 0.075 0.965 0.260 3.472 Oklahoma 0.050 1.012 -0.146 -2.722 -0.071 -1.272 Oregon -0.073 -1.372 -0.004 -0.084 0.109 1.615 Pennsylvania -0.443 -3.824 0.165 1.630 -0.566 -0.544 Rhode Island -0.566 -3.806 -0.380 -2.586 -0.243 -1.660 South Carolina 0.121 1.680 0.038 0.472 -0.175 -2.585 South Dakota 0.149 2.292 0.100 1.566 0.191 2.710 Tennessee -0.235 -3.317 0.053 0.929 0.029 0.457 Texas 0.163 1.913 -0.198 -2.502 -0.239 -3.037 Utah 0.773 0.289 0.011 -0.060 -0.965 Virginia -0.011 -0.170 -0.166 -2.960 -0.304 -5.075 Washington 0.1	New Hampshire	0.066	0.846	-0.196	-2.587	-0.054	-0.703
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	New Jersev	0.394	2.581	0.285	1.962	0.124	0.840
$\begin{array}{llllllllllllllllllllllllllllllllllll$	New Mexico	-0.151	-2.031	-0.168	-2.577	-0.181	-2.489
North Carolina -0.462 -7.091 -0.277 -4.951 0.081 1.415 North Dakota -0.247 -3.025 -0.166 -1.989 -0.251 -3.191 Ohio 0.805 8.115 0.075 0.965 0.260 3.472 Oklahoma 0.050 1.012 -0.146 -2.722 -0.071 -1.272 Oregon -0.073 -1.372 -0.004 -0.084 0.109 1.615 Pennsylvania -0.443 -3.824 0.165 1.630 -0.056 -0.544 Rhode Island -0.566 -3.806 -0.380 -2.586 -0.243 -1.660 South Carolina 0.121 1.680 0.038 0.472 -0.175 -2.585 South Dakota 0.149 2.292 0.100 1.566 0.191 2.710 Texnessee -0.235 -3.317 0.053 0.929 0.029 0.457 Texas 0.163 1.913 -0.198 -2.502 -0.239 -3.037 Utah 0.778 1.235 0.169 1.186 -0.324 -2.171 Vermont 0.052 0.736 0.489 6.011 -0.060 -0.965 Virginia -0.011 -0.171 -0.166 -2.960 -0.344 -5.075 Washington 0.133 1.688 -0.188 -2.728 -0.060 -0.839 West Virginia 0.355 3.908 0.144 1.711 0.302 3.604 Wisco	New York	0.219	1.758	0.203	1.710	0.162	1.303
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	North Carolina	-0.462	-7.091	-0.277	-4.951	0.081	1.415
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	North Dakota	-0.247	-3.025	-0.166	-1.989	-0.251	-3.191
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Ohio	0.805	8.115	0.075	0.965	0.260	3.472
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Oklahoma	0.050	1.012	-0.146	-2.722	-0.071	-1.272
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Oregon	-0.073	-1.372	-0.004	-0.084	0.109	1.615
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Pennsvlvania	-0.443	-3.824	0.165	1.630	-0.056	-0.544
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Rhode Island	-0.566	-3.806	-0.380	-2.586	-0.243	-1.660
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	South Carolina	0.121	1.680	0.038	0.472	-0.175	-2.585
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	South Dakota	0.149	2.292	0.100	1.566	0.191	2.710
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Tennessee	-0.235	-3.317	0.053	0.929	0.029	0.457
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Texas	0.163	1.913	-0.198	-2.502	-0.239	-3.037
Vermont 0.052 0.736 0.489 6.011 -0.060 -0.965 Virginia -0.011 -0.171 -0.166 -2.960 -0.304 -5.075 Washington 0.133 1.688 -0.188 -2.728 -0.060 -0.839 West Virginia 0.355 3.908 0.144 1.711 0.302 3.604 Wisconsin -0.375 -5.133 0.099 1.862 0.008 0.121 Woming 0.375 3.677 0.088 0.941 0.018 0.213	Utah	0.178	1.235	0.169	1.186	-0.324	-2.171
Virginia -0.011 -0.171 -0.166 -2.960 -0.304 -5.075 Washington 0.133 1.688 -0.188 -2.728 -0.060 -0.839 West Virginia 0.355 3.908 0.144 1.711 0.302 3.604 Wisconsin -0.375 -5.133 0.099 1.862 0.008 0.121 Wyoming 0.375 3.677 0.088 0.941 0.018 0.213	Vermont	0.052	0.736	0.489	6.011	-0.060	-0.965
Washington 0.133 1.688 -0.188 -2.728 -0.060 -0.839 West Virginia 0.355 3.908 0.144 1.711 0.302 3.604 Wisconsin -0.351 -5.133 0.099 1.862 0.008 0.121 Wyoming 0.375 3.677 0.088 0.941 0.018 0.213	Virginia	-0.011	-0.171	-0.166	-2.960	-0.304	-5.075
West Virginia 0.355 3.908 0.144 1.711 0.302 3.604 Wisconsin -0.351 -5.133 0.099 1.862 0.008 0.121 Wyoming 0.375 3.677 0.088 0.941 0.018 0.213	Washington	0.133	1.688	-0.188	-2.728	-0.060	-0.839
Wisconsin -0.351 -5.133 0.099 1.862 0.008 0.121 Wyoming 0.375 3.677 0.088 0.941 0.018 0.213	West Virginia	0.355	3.908	0.144	1.711	0.302	3.604
Wyoming 0.375 3.677 0.088 0.941 0.018 0.213	Wisconsin	-0.351	-5.133	0.099	1.862	0.008	0.121
	Wyoming	0.375	3.677	0.088	0.941	0.018	0.213

Note: The numbers correspond to the Φ parameters of the MCAR model for the fourth diffusion period. We report the posterior mean of the spatial effects for the Wii, PS3 and X360 and the ratio of the posterior mean over the posterior standard deviation.

Table 5.9: Post	erior of	MCAR	Spatial	Effects
-----------------	----------	------	---------	---------

	OLS F	iret Diffueior	Period
	Coefficient	St Dev	t_value
Intercept	-2.3730	0.0131	-180 7930
Male Female Batio	0.0185	0.0195	0.9480
Population Density	0.0100	0.0228	0.1720
Population in College Dorms	0.0000	0.0164	1 6040
Married Couple	0.0203	0.0161	1.0040
Travel Time to Work	0.0130	0.0101	0.6740
Income per Capita	-0.0028	0.0134	-0.1300
meome per cupita	0.0020	0.0211	0.1000
	OLS Sec	cond Diffusio	on Period
	Coefficient	St. Dev.	t-value
Intercept	-2.4053	0.0159	-151.0310
Male Female Ratio	0.0218	0.0237	0.9230
Population Density	0.0239	0.0277	0.8620
Population in College Dorms	0.0257	0.0199	1.2910
Married Couple	-0.0092	0.0196	-0.4710
Travel Time to Work	-0.0136	0.0235	-0.5780
Income per Capita	-0.0194	0.0263	-0.7390
· · · ·			
	OLS Th	nird Diffusion	n Period
	Coefficient	St. Dev.	t-value
Intercept	-2.3730	0.0131	-180.7930
Male Female Ratio	0.0185	0.0195	0.9480
Population Density	0.0039	0.0228	0.1720
Population in College Dorms	0.0263	0.0164	1.6040
Married Couple	0.0198	0.0161	1.2250
Travel Time to Work	0.0131	0.0194	0.6740
Income per Capita	-0.0028	0.0217	-0.1300
	OLS For	urth Diffusic	n Period
	Coefficient	St. Dev.	t-value
Intercept	-2.4008	0.0203	-118.0540
Male Female Ratio	-0.0197	0.0302	-0.6520
Population Density	0.0355	0.0354	1.0030
Population in College Dorms	-0.0237	0.0254	-0.9310
Married Couple	0.0210	0.0250	0.8400
Travel Time to Work	0.0189	0.0300	0.6290
Income per Capita	0.0182	0.0336	0.5410
Note: These are parameter est	timates of the	e model in e	quation (5.3)
obtained by OLS and with no	spatial effects		

Table 5.10: OLS δ coefficients

	MC.	AR First 1	Period
	Mean	5%	95%
Λ_{12} (Wii-PS3)	0.075	-0.185	0.337
Λ_{13} (Wii-Xbox)	0.115	-0.139	0.352
Λ_{23} (PS3-Xbox)	-0.257	-0.488	0.036
		<u>.</u>	D 4 1
	MCA	R Second	Period
	Mean	5%	95%
Λ_{12} (Wii-PS3)	-0.082	-0.344	0.210
Λ_{13} (Wii-Xbox)	0.096	-0.156	0.354
Λ_{23} (PS3-Xbox)	-0.103	-0.377	0.179
	MCA	AR Third	Period
	Mean	5%	95%
Λ_{12} (Wii-PS3)	0.061	-0.204	0.302
Λ_{13} (Wii-Xbox)	0.401	0.145	0.600
Λ_{23} (PS3-Xbox)	0.117	-0.137	0.368
	MCA	R Fourth	Period
	Mean	5%	95%
Λ_{12} (Wii-PS3)	0.409	0.156	0.601
Λ_{13} (Wii-Xbox)	0.349	0.090	0.552
Λ_{23} (PS3-Xbox)	0.311	0.058	0.534
Note: We present	the post	erior meai	and the
posterior 95% high	nest densi	ty region o	of the cor-
relation matrix o	btained f	rom the /	A matrix.
The Λ matrix mea	sures the	covariance	e between
the spatial effects	of the th	ree produ	cts.

Table 5.11: Posterior of MCAR Λ correlations

		HPDR	
	95%	50%	5%
MCAR 1st period ρ	0.975	0.805	0.150
MCAR 2nd period ρ	0.975	0.825	0.150
MCAR 3rd period ρ	0.975	0.825	0.150
MCAR 4th period ρ	0.975	0.815	0.200
Note:			

Table 5.12: Highest Posterior Density Region (HPDR) for the ρ coefficient.

	Aggreg	ate Model for	the Wii
Variable	Estimate	Std. Error	t-value
spline	0.663	0.088	7.554
Search Wii	0.468	0.121	3.862
	Aggreg	ate Model for	the PS3
Variable	Estimate	Std. Error	t-value
spline	0.862	0.108	7.974
Search PS3	0.171	0.133	1.287
	Aggrega	ate Model for	the X360
Variable	Estimate	Std. Error	t-value
spline	0.722	0.122	5.916
Search X360	0.375	0.163	2.304
Note: The dep	pendent var	iable is aggreg	ate sales for
each of the co	nsoles (in le	ogs). The right	nt hand side
includes a splir	ne term and	the logs of the	search index
for the console	e. The R^2	is higher than	0.95 for all

Table 5.13: OLS Regressions between Aggregate Sales Data and Aggregate Online Search Data

three regressions.

ompare by	Search terms		Filter	
Search terms	Tip: Use the plus sign to indicate OR. (tennis + squash)		Web Search	-
Locations	nintendo wii + wii + wii fit + wii sport	×	United States	subragions + All metros
Time Ranges	playstation 3 + PS3	×	2004	
	microsoft xbox 360 + xbox 360 + x360	×	2004 - present	
	+ Add search term	_	All Categories	(•) (Sea
h Google Insight nes and propertie Categories Narrow data	s for Search, you can compare search volume patterns acro ss. See <u>examples</u> of how you can use Google insights for S to specific categories, like finance, health, and sports.	ss specific earch.	regions, categories, tin	ne More examples man, machine flickr, webshots
h Google Insight mes and propertie Categories Narrow data Examples: II	s for Search, you can compare search volume patterns acro ss. See <u>examples</u> of how you can use Google Insights for S to specific categories, like finance, health, and sports. the top vehicle brands in France (last 30 days) Soccer in 20 month for university business on you can budget and alon according	ss specific earch. 108. 2007.	regions, categories, tin	ne More examples man, machine filickr, webshots defroit auto show, chicago auto show iron maiden, slayer
th Google Insight mes and propertie Categories Narrow data Examples: 11 Seasonality Anticipate de Examples: d	s for Search, you can compare search volume patterns acro ss. See <u>examples</u> of how you can use Google Insights for S to specific categories, like finance, health, and sports, he top vehicle brands in France (tast 30 daws) Soccer in 20 mand for your business so you can budget and plan accord raid camera in 2007, 2006	ss specific earch. 08. 2007. ingly.	regions, categories, tin	ne More examples man, machine flickr, webshots detroit auto show, chicago auto show iron maiden, slaver mike tyson, muhammad ali
h Google Insight nes and propertie Categories Narrow data Examples: 1 Seasonality Anticipate de Examples: di	s for Search, you can compare search volume patterns acro s. See <u>examples</u> of how you can use Google Insights for S to specific categories, like finance, health, and sports. the top vehicle brands in France (last 30 days) Soccar in 20 mand for your business so you can budget and plan accord gliat camera in 2007, 2006	ss specific earch. 108, 2007. ingly.	regions, categories, tin	ne More examples man, machine filick, webshots detroit auto show, chicago auto show irron maiden, slaver mike tyson, muhammad ali summer camps The scenes in camps (mat 90 data)
h Google Insight mes and propertie Categories Narrow data Examples: 11 Seasonality Anticipate de Examples: di Geographic Know where	s for Search, you can compare search volume patterns acro ss. See <u>examples</u> of how you can use Google Insights for S to specific categories, like finance, health, and sports. the top vehicle brands in France (last 30) days) Soccerin 20 mand for your business so you can budget and plan accord plai camera in 2007, 2006 apple in 2006 vs. 2007 distribution to find your customers. See how search volume is distribute	ss specific earch. 08. 2007. ingly. ed across i	regions, categories, tin	ne More examples man, machine filickr, webshots defroit auto show, chicago auto show iron maiden, slaver mike lyeon, muhammad ali summer camps Top searches in France (last 30 days) red hat dehian centoo, slackware
th Google Insight mes and propertia Categories Narrow data Examples: 1 Seasonality Anticipate de Examples: di Geographic Know where Examples: tid	s for Search, you can compare search volume patterns acro ss. See <u>examples</u> of how you can use Google Insights for S to specific categories, like finance, health, and sports, the top vehicle brands in France (tast 30 days) Soccer in 20 mand for your business so you can budget and plan accord fail camera in 2007, 2006	ss specific earch. 108, 2007. ingly. ingly.	regions, categories, tin	ne More examples man, machine flickr, webshots detroit auto show, chicago auto show iron maiden, slaver mike lyaon, muhammad ali summer camps Top searches in France (tast 30 days) red hat, debian, gentoo, slackware oundam, star wars
th Google Insight mes and propertie Categories Narrow data Examples: 11 Seasonality Anticipate de Examples: di Geographic Know where Examples: It	s for Search, you can compare search volume patterns acro ss. See <u>examples</u> of how you can use Google Insights for S to specific categories, like finance, health, and sports. ne top vehicle brands in France (tast 30 days) Soccar in 20 mand for your business so you can budget and plan accord gital camera in 2007, 2006 apple in 2006 vs. 2007. distribution to find your customers. See how search volume is distribute kets in different US metro areas recipes in different US metro set of the second second second second second second second second second to find your customers. See how search volume is distribute	ss specific earch. 108, 2007, ingly. Id across i etro areas	regions, categories, tin	ne More examples man, machine flickr, webshots detroit auto show, chicago auto show iron maiden, slayer mike tyson, muhammad ali summer camps Top searches in France flast 30 days) red hat, deblan, gentoo, slackware gundam, star wars firefox, internet exolorer, safari, opera
h Google Insight mes and propertie Categories Narrow data Examples: 11 Anticipate de Examples: de Geographic Know where Examples: de Properties Exes search properties	s for Search, you can compare search volume patterns acro ss. See <u>examples</u> of how you can use Google Insights for S to specific categories, like finance, heath, and sports. ne top vehicle brands in France (last 30 days) Soccerin 20 mand for your business so you can budget and plan accord fails camera in 2007, 2005 apple in 2006 vs. 2007 distribution to find your customers. See how search volume is distribute kets in different US metro areas recipes in different US me statemen in other Google properties.	ss specific earch. 108, 2007, ingly. Id across i <u>stro areas</u>	regions, categories, tin agions and cities.	More examples man, machine filekr, webshols defroit auto show, chicago auto show iron maiden, slaver mike tyson, muhammad all summer camos Top searches in France (last 30 days) red hat, debian, sentos, slackware gundam, slar, wars firefox, internet explorer, safari, opera, chrome
h Google Insight nes and properties Narrow data Examples: 11 Seasonality Anticipate de Examples: 16 Geographic Know where Examples: 16 See search p Examples: 18	s for Search, you can compare search volume patterns acro ss. See <u>examples</u> of how you can use Google Insights for S to specific categories, like finance, heath, and sports. The top vehicle brands in France (tast 30 days) Soccer in 20 mand for your business so you can budget and plan accord fait camera in 2007, 2006	ss specific earch. 108, 2007, ingly. Id across i etro areas	regions, categories, tin segions and cities. <u>re last 7 days (USA)</u>	ne More examples man, machine flickr, webshots detroit auto show, chicago auto show iron maiden, slaver mike lyaon, muhammad ali summer camps Top searches in France (tast 30 days) red hat, debian, gentoo, slackware gundam, star wars firefos, internet explorer, safari, opera, chrome doctor who, battlestar galactica

Figure 5.1: Google Insights for Search



Figure 5.2: Model Size: Posterior Distribution of the Number of Regressors Included in the Model for the Nintendo Wii















Figure 5.6: Moran's I and Geary's C for Uniform Probabilities (Histogram) and Moran's I and Geary's C for all Diffusion Periods and Technologies (Vertical Lines)



Figure 5.7: Nintendo Wii Model: Histogram of the Posterior Mean of the Regression Coefficient for all US States and All Time Periods (Left Panel) and Posterior Mean Over Posterior Standard Deviation (Right Panel)



Figure 5.8: Sony PS3 Model: Histogram of the Posterior Mean of the Regression Coefficient for all US States and All Time Periods (Left Panel) and Posterior Mean Over Posterior Standard Deviation (Right Panel)



Figure 5.9: Microsoft Xbox Model: Histogram of the Posterior Mean of the Regression Coefficient for all US States and All Time Periods (Left Panel) and Posterior Mean Over Posterior Standard Deviation (Right Panel)


Figure 5.10: Scatter Plots between Inclusion Probabilities for Each Diffusion Period and Search Elasticity (Nintendo Wii)





















5.A Methodology

In this appendix we discuss the BVS method and the MCAR model estimation we use to study the probabilities of inclusion of the different regions and locations.

Bayesian Variable Selection

In what follows we follow closely the presentation of George and McCulloch (1997) section 4. In Section 4 they discuss the specification of conjugate priors for β and σ . We chose to use conjugate priors because it facilitates the integration of β and σ out of the posterior distribution of the indicators γ and hence the computation of the posterior of γ becomes simple and fast.

The likelihood is specified as

$$f(Y|\beta,\sigma) = \phi(Y; X_{\gamma}\beta_{\gamma}, \sigma^2 I)$$
(5.4)

where $Y = y_i = (y_{i1}, \dots, y_{iT})$, X_{γ} is a subset of potential regressors for which $\gamma = 1$, I is an identity and $\phi(y; x, \Sigma)$ is the Normal distribution density with mean x and variance Σ evaluated at y. The prior for β is

$$\pi(\beta|\sigma,\gamma) = \phi(\beta;0,\sigma^2 D_\gamma R D_\gamma),\tag{5.5}$$

where D_{γ} is a diagonal matrix with elements

$$D_{\gamma}^{kk} = \begin{cases} \upsilon_0 & \text{when } \gamma_k = 0\\ \upsilon_1 & \text{when } \gamma_k = 1, \end{cases}$$
(5.6)

and R is a correlation matrix. $R \propto I$ or $R \propto (X'_{\gamma}X_{\gamma})^{-1}$ are attractive choices when $v_0 = 0$. The scalars v_0 and v_1 are chosen according to the objectives of the modeler. The choice of v_0 and v_1 affect the number of regressors included in the subset X_{γ} and the threshold after which an element of β is distinguished from zero. See George and McCulloch (1997, page 346-347) for more details.

George and McCulloch (1997) discuss how different choices of v_0 and v_1 affect the selection of variables and the size of the β coefficients that are included in the model. The suggestion is to set v_0 small and v_1 large such that when the posterior supports that $\gamma_k = 0$ then the prior specification is narrow enough to keep β_k close to zero. A popular choice in the literature is to set $v_0 = 0$ and to specify $\pi(\beta|\gamma) = \pi(\beta_{\gamma}|\gamma) \times \pi(\beta_{\bar{\gamma}}|\gamma)$ where $\pi(\beta_{\gamma}|\gamma) = \phi(\beta_{\gamma}; 0, \sigma^2 \Sigma_{\gamma})$ and $\pi(\beta_{\bar{\gamma}}|\gamma) = 1$ being β_{γ} and $\beta_{\bar{\gamma}}$ the coefficients included and excluded in the model, respectively. The attractiveness of this last specification is that we can select β_k depending on how significantly they are different from zero rather than selecting them depending on their relative size when $v_0 \neq 0$.

The prior for σ^2 is

$$\pi(\sigma^2) = IG(\nu/2, \nu\lambda/2) \tag{5.7}$$

where ν are the degrees of freedom and λ is the scale of the inverse gamma (IG) distribution. What is left to specify is the prior for the indicators γ . They are usually specified as

$$\pi(\gamma) = \prod_{k} w_{k}^{\gamma_{k}} (1 - w_{k})^{1 - \gamma_{k}}, \qquad (5.8)$$

where w_k is the probability of including the k regressor in the model. A popular choice in the literature is to use $w_k = w$ and therefore

$$\pi(\gamma) = w^{q_{\gamma}} (1-w)^{p-q_{\gamma}},\tag{5.9}$$

where q_{γ} is the number of regressors included out of a total set of size p. This last prior can be combined with a conjugate prior on w and set $w \sim Beta(a, b)$ and the prior becomes

$$\pi(\gamma) = \frac{B(a+q_{\gamma}, b+p-q_{\gamma})}{B(a,b)},\tag{5.10}$$

where B(x, y) is the beta function with x and y parameters. See Chipman et al. (2001) for other choices of $\pi(\gamma)$. Careful selection should be given to the scalars v_1 and w (or a and b) as they directly affect model size. Large v_1 and small w concentrate the prior on parsimonious models with large coefficients while large w and small v_1 concentrate the prior on saturated models with small coefficients (Clyde and George, 2004, page 86). The joint density $\pi(Y, \beta, \sigma^2 | \gamma) = \pi(Y | \beta, \sigma^2, \gamma) \pi(\beta | \sigma, \gamma) \pi(\sigma^2 | \gamma)$ has a closed form expression when $v_0 = 0$ and after integrating over β and σ^2 and that is

$$\pi(Y|\gamma) \propto |X'_{\gamma}X_{\gamma} + \Sigma_{\gamma}^{-1}|^{-1/2} |\Sigma_{\gamma}|^{-1/2} (\nu\lambda + S_{\gamma}^2)^{-(T+\nu)/2},$$
(5.11)

where

$$S_{\gamma}^{2} = Y'Y - Y'X_{\gamma}(X_{\gamma}'X_{\gamma} + \Sigma_{\gamma}^{-1})X_{\gamma}'Y, \qquad (5.12)$$

and $\Sigma_{\gamma} = D_{\gamma}RD_{\gamma}$. The posterior of the indicators is straightforward to compute as $\pi(\gamma|Y) \propto \pi(Y|\gamma)\pi(\gamma)$ and the Metropolis-Gibbs sampler is straightforward and it proceeds by sampling $\pi(\gamma|Y)$, $\pi(\beta_{\gamma}|Y, \sigma^2, \gamma)$ and $\pi(\sigma^2|Y, \beta_{\gamma}, \gamma)$ sequentially.

We use a = 50 and b = 100 for the prior on w (in equation (5.10)). The prior of σ^2 has $\nu = 1000$ and $\lambda = 0.30$. We follow the recommendation of George and McCulloch (1997, page 341) who suggest to set λ such that the posterior of σ^2 assigns substantial probability to an interval close to the sample variance of Y and the variance of the residual of a saturated model. The prior on β in equation (5.5) and (5.6) has $v_0 = 0$ and $v_1 = 7$ and we use $R = (X'_{\gamma}X_{\gamma})^{-1}$.

A short review of aereal data models

Aereal data usually refers to cross sectional or panel data collected across different regions or areas with well defined boundaries. Therefore aereal data consists of aggregate or summary measures at different locations. The CAR and SAR models are among the most popular models applied to aereal data but there are many other popular approaches like *kriging* or spatial interpolation. In this review we focus on the CAR model and its multivariate extensions.

CAR stands for *Conditional Autoregressors* and SAR stands for *Simultaneous Autoregressors* and hence CAR models are usually referred as Conditionally Autoregressive models and the SAR as Simultaneous Autoregressive models.

The CAR and SAR models are discussed in several sources. A basic reference is Cressie (1992). Cressie covers topics that range from model specification, classical and Bayesian estimation to the theoretical foundations of the CAR and SAR models. Many other topics in spatial analysis are discussed in Cressie (1992). Banerjee et al. (2004) focus on Bayesian analysis and

estimation of spatial models. Held and Rue (2002) review many of the computational methods and sampling techniques usually applied to the Bayesian analysis of CAR models and to more general spatial models referred to as *Gaussian Markov Random Fields*.

Wall (2004) compares the CAR and SAR models and offers some insights about the different correlation between locations implied by these two models. The CAR and SAR models might be equivalent under certain conditions, for example see Assunçao (2003) or Banerjee et al. (2004, page 86). We intend to apply spatial priors to the distribution of model parameters. Therefore, in what follows we focus on the CAR model as it is better suited than the SAR both as a hierarchical prior specification on a model's parameters and for Bayesian modeling (Banerjee et al., 2004, page 86).

The main assumption of the CAR model is that a measurement at a location has a conditional distribution with a mean that is proportional to a weighted sum of the measurement at neighboring locations. Both the joint distribution and the conditional distribution of the spatial effects given all other spatial parameters can be derived in closed form and they are presented in Banerjee et al. (2004, page 79) and in the references therein. However, there are alternative specifications to the joint distribution of the spatial effects and a common approach is to use the pairwise difference specification (Besag et al., 1991). Haran et al. (2003) present how to use block updating when some of the coefficients in a linear regression follow the pairwise difference prior.

The CAR is suited for univariate aereal data and Mardia (1988) presents an extension to the multivariate case, usually referred to as *multivariate CAR* or simply as MCAR. It is common to have more than one measurement at each location and the MCAR allows to model both the correlation among measurements of neighboring sites and the correlation among the different measures across sites. Gelfand and Vounatsou (2003) and Carlin and Banerjee (2003) apply Bayesian analysis to the MCAR of Mardia (1988) and present applications with two and up to five dimensional data. On the other hand, Gamerman et al. (2003) present a multivariate version of the pairwise difference specification (used as a prior) and its sampling schemes.

Other extensions of the CAR model incorporate dynamics into its spatial coefficients. Waller et al. (1997), Nobre et al. (2005) and Gelfand et al. (2005) propose models that use a random walk specification for the mean or for the variance of the spatial effects. Gelfand et al. (2005) provide a review of spatio-temporal models.

Linear Model with CAR Prior

Next we work out the specification and sampling for the model

$$y_i = x_i\beta + \phi_i + \epsilon_i, \tag{5.13}$$

where y_i is measured at *i* locations for i = 1, ..., p, x_i is a set of *k* covariates at *i* and β is a coefficient column vector $k \times 1$ while ϵ_i and ϕ_i are random effects meant to capture overall variability and spatial heterogeneity, respectively. We define $y' = (y_1, ..., y_p)$, $\phi' = (\phi_1, ..., \phi_p)$ and $X = (x_1, ..., x_k)$. The distribution of ϵ_i is

$$\epsilon \sim N(0, \Sigma),$$
 (5.14)

where $\epsilon' = (\epsilon_1, \ldots, \epsilon_p)$, $\Sigma = \sigma^2 I$ and σ^2 is the variance of ϵ . $N(\mu, \Sigma)$ refers to a normal distribution with mean μ and covariance matrix Σ . We define $\lambda_{\epsilon} = 1/\sigma^2$. The prior distribution of the spatial effects ϕ_i follows

$$\phi_i | \phi_{j \sim i} \sim N\Big(\sum_{j \sim i} c_{ij} \phi_j, \tau_i^2\Big).$$
(5.15)

This form states that the distribution of ϕ_i given its *j* neighbors, denoted as $j \sim i$, has a normal distribution with a mean that is a weighted sum (using weights c_{ij}) of the neighboring values and variance τ_i^2 . Besag (1974) shows that the joint distribution of the spatial effects in (5.15) can we written in the form

$$\phi \sim N(0,\Omega),\tag{5.16}$$

where $\phi = (\phi_1, \ldots, \phi_p)$ and Ω is a $p \times p$ symmetric and positive semi-definite or positive definite matrix. In the literature it is common to define the elements of Ω^{-1} directly instead of specifying Ω . For example, Banerjee et al. (2004, page 79) assume that $\tau_i^2 = \tau^2/w_{i+}$ and that $c_{ij} = w_{ij}/w_{i+}$ where w_{ij} takes the value of 1 if $j \sim i$ and zero otherwise and where w_{i+} is the total number of neighbors of *i*. Given these assumptions $\Omega^{-1} = T^{-1}(I - C)$ and given that *T* is a diagonal matrix with elements $T_{ii} = \tau^2/w_{i+}$ and $C_{ij} = c_{ij}$ then Ω^{-1} can be written as

$$\Omega^{-1} = \frac{1}{\tau^2} (I_{w_{i+}} - W), \qquad (5.17)$$

where $I_{w_{i+}}$ is a diagonal matrix with elements w_{i+} and $W_{ij} = w_{ij}$. This last specification for Ω results in an improper distribution given that the rows of $(I_{w_{i+}} - W)$ sum to zero. A solution to this issue is to specify Ω as

$$\Omega^{-1} = \frac{1}{\tau^2} (I_{w_{i+}} - \rho W), \qquad (5.18)$$

where ρ takes a value (between 0 and 1) that makes Ω^{-1} positive definite and consequently the distribution of ϕ becomes proper. For a discussion on the *impropriety* of the CAR distribution and the role of the ρ parameter see Banerjee et al. (2004, page 163), Eberly and Carlin (2000), Sahu and Gelfand (1999) or Best et al. (1999). This latter form implies that

$$\phi_i | \phi_{j \sim i} \sim N\left(\rho \sum_{j \sim i} c_{ij} \phi_j, \tau_i^2\right).$$
(5.19)

The distribution of ϕ is usually referred as $CAR(\tau^2)$ when the conditional distributions of the spatial effects are defined as in equation (5.15) and it is referred as $CAR(\rho, \tau^2)$ when its conditional distribution follows (5.19). In what follows we use $\Omega^{-1} = \lambda_{\phi}Q$ with $Q = I_{w_{i+}} - \rho W$ and $\lambda_{\phi} = 1/\tau^2$. To carry out Bayesian inference and to complete the model specification we need to define the priors for β , λ_y , λ_{ϕ} and ρ . We specify them as

$$p(\beta) \propto 1$$

$$p(\lambda_y) \propto \lambda_y^{a_y} e^{-b_y \lambda_y}$$

$$p(\lambda_{\phi}) \propto \lambda_{\phi}^{a_{\phi}} e^{-b_{\phi} \lambda_{\phi}}$$

$$p(\rho) \propto \text{ discretized prior}$$
(5.20)

We use $p(\cdot)$ generically to denote a probability density. That is, the prior for β is noninformative, the priors for λ_y and λ_{ϕ} have the form of a Gamma distribution. Finally, for ρ we give probability mass to a discrete set of values with a high proportion of them near 1. Gelfand and Vounatsou (2003) suggest the use of discretized priors for ρ . The model specification is now complete and next we describe the sampling steps to estimate equation (5.13).

Sampling Steps for the CAR

To sample the parameters of the model in equation (5.13) we can apply the Gibbs sampler and MCMC. To derive the posterior of β we can write the likelihood of equation (5.13) as

$$L(y|\beta, \lambda_y) \propto |M|^{-1/2} e^{-\frac{1}{2}(y-X\beta)'M^{-1}(y-X\beta)},$$
 (5.21)

where $M = (\frac{1}{\lambda_{\phi}}Q^{-1} + \frac{1}{\lambda_{\epsilon}}I)$. The posterior of β is then

$$p(\beta|y,\lambda_y,\lambda_\phi) \propto |M|^{-1/2} e^{-\frac{1}{2}(\beta-b)'(X'M^{-1}X)^{-1}(\beta-b)},$$
 (5.22)

with $b = (X'M^{-1}X)^{-1}X'M^{-1}y$. Therefore β can be sampled from $N(b, (X'M^{-1}X)^{-1})$.

Next we derive the posterior distribution of the spatial effects ϕ . To do so we write the density of y conditional on β . That is

$$L(y|\beta,\phi,\lambda_y) \propto \lambda_y^{p/2} e^{-\frac{\lambda_y}{2}(\tilde{y}-\phi)'(\tilde{y}-\phi)},$$
(5.23)

with $\tilde{y} = y - X\beta$. Therefore, the posterior of ϕ is

$$p(\phi|\tilde{y},\lambda_y,\lambda_\phi) \propto \lambda_y^{p/2} e^{-\frac{1}{2}((\phi-a)'R^{-1}(\phi-a))},$$
(5.24)

where $a = (\lambda_y I + \lambda_\phi Q)^{-1} \lambda_y \tilde{y}$ and $R^{-1} = (\lambda_y I + \lambda_\phi Q)$. That is ϕ can be sampled form N(a, R).

The posterior of λ_y and λ_{ϕ} are

$$p(\lambda_{\phi}|\tilde{y},\phi,\lambda_{y}) \propto \lambda_{\phi}^{p/2+a_{\phi}} e^{-\lambda_{\phi}(\frac{1}{2}\phi'Q\phi+b_{\phi})}$$

$$p(\lambda_{y}|\tilde{y},\phi,\lambda_{\phi}) \propto \lambda_{y}^{p/2+a_{y}} e^{-\lambda_{y}(\frac{1}{2}(\tilde{y}-\phi)'(\tilde{y}-\phi)+b_{y})}.$$
(5.25)

That is $\lambda_{\phi} \sim \Gamma(p/2 + a_y, b_y + 1/2\phi'Q\phi)$ and $\lambda_y \sim \Gamma(p/2 + a_{\phi}, b_{\phi} + 1/2(\tilde{y} - \phi)'(\tilde{y} - \phi))$.

Finally we need to sample the ρ in the Q matrix. We know that

$$p(\rho|\phi, y, \lambda_y, \lambda_\phi) \propto |Q|^{1/2} e^{-\frac{1}{2}\phi' Q\phi} p(\rho).$$
(5.26)

A common method to sample ρ is to assume that $p(\rho)$ is a uniform distribution with range (0,1) and to sample it with the Metropolis-Hastings algorithm. A second popular choice is to discretize ρ in a set of values and to draw them proportional to their posterior probability. We use the following set $0.01, 0.10, 0.20, 0.30, \ldots, 0.70, 0.71, 0.72, \ldots, 0.99$.

In summary we use the next steps in the Gibbs sampler

1. $\beta \sim N((X'M^{-1}X)^{-1}X'M^{-1}y, (X'M^{-1}X)^{-1})$ 2. $\phi \sim N((\lambda_y I + \lambda_\phi Q)^{-1}\lambda_y \tilde{y}, (\lambda_y I + \lambda_\phi Q))$ 3. $\lambda_y \sim \Gamma(p/2 + a_y, b_y + 1/2\phi'Q\phi)$ 4. $\lambda_\phi \sim \Gamma(p/2 + a_\phi, b_\phi + 1/2(\tilde{y} - \phi)'(\tilde{y} - \phi))$

5.
$$\rho \sim p(\rho | \phi, y, \lambda_y, \lambda_\phi)$$

where $x \sim \Gamma(a, b)$ means that x follows a Gamma distribution with the form $cx^a e^{-bx}$ where c is a constant. At the end of the sampling step 2 we center the ϕ vector around its own mean following Eberly and Carlin (2000) and Best et al. (1999). The re-centering is equivalent to sampling with the restriction $\sum \phi_i = 0$. Rue and Held (2005) show a general form to sample with linear restrictions and that is equivalent to centering around a mean.

Multivariate Linear Model with MCAR Prior

Next we expand the linear model of Section 5.A to a multivariate setting. The exposition follows Carlin and Banerjee (2003) and Gelfand and Vounatsou (2003).

In this setting we observe J different measurements at each location. That is we use the notation y_{ji} to refer to the j^{th} measurement at location i. We use the notation y_j for $(y_{j1}, \ldots, y_{jp})'$ and Y is a $p \times J$ matrix with columns (y_1, \ldots, y_J) . The same notation is used for the spatial effects ϕ_{ij} and the error terms ϵ_{ij} . That is $\phi_j = (\phi_{j1}, \ldots, \phi_{jp})'$, $\Phi = (\phi_1, \ldots, \phi_J)$ and finally $\epsilon_j = (\epsilon_{j1}, \dots, \epsilon_{jp})', E = (\epsilon_1, \dots, \epsilon_J).$ We observe a common group of N covariates X where $X = (x_1, \dots, x_N)$ and $x_i = (x_{i1}, \dots, x_{ip})'$. Hence we can write

To carry out Bayesian inference we define the following priors

$$p(B) \propto 1$$

$$p(\Sigma) \propto |\Sigma|^{-\frac{v}{2}} e^{-\frac{1}{2}tr\Sigma^{-1}V_{\Sigma}}$$

$$p(\Phi|\Lambda, \Psi) \propto |\Psi|^{-J/2} |\Lambda|^{-p/2} e^{-\frac{1}{2}tr(\Psi\Phi\Lambda\Phi')}$$

$$p(\Lambda) \propto |\Lambda|^{-\frac{v_0}{2}} e^{-\frac{1}{2}tr\Lambda V_{\Lambda}}$$
(5.28)

Above Σ is a $J \times J$ covariance matrix of E and $(E) \sim N(0, \Sigma \otimes I)$; Λ is $J \times J$ and it is the inverse of the covariance matrix between the columns of Φ while Ψ is $p \times p$ and it is the inverse covariance matrix between the rows of Φ . That is, $(\Phi) \sim N(0, \Lambda^{-1} \otimes \Psi^{-1})$.

The form of Ψ might be identical to the form of the Q matrix in the CAR prior. That is $\Psi = (I_{w_{i+}} - \rho W)$ where W and $I_{w_{i+}}$ are defined as before. A second choice for Ψ might be $\Psi = (I_{w_{i+}} - W)$. Carlin and Banerjee (2003) and Gelfand and Vounatsou (2003) use the first form while Gamerman et al. (2003) use the second. A third choice is to define a general form for $\Lambda \otimes \Psi$ as Gelfand and Vounatsou (2003) propose. Gelfand and Vounatsou (2003) propose a form of Q that allows an item (J items) specific ρ parameters. They first define $Q_j = (I_{w_{i+}} - \rho_j W)$ and its Choleski factorization $Q_j = P'_j P_j$. Then they define

$$\Lambda \otimes \Psi = \mathbf{P}'(\Lambda \otimes I_{p \times p})\mathbf{P},\tag{5.29}$$

where **P** is a diagonal matrix with P_j blocks. This last form may allow for a more flexible correlation structure of the Φ parameters. In the application we assume $\rho_j = \rho$ for all j.

Sampling the Multivariate Linear Model with MCAR Prior

If we condition on Φ and define $\overline{Y} = Y - \Phi$ we obtain the traditional multivariate regression model

$$\bar{Y} = X \cdot B + E. \tag{5.30}$$

Given this last expression we can write the density of the model as

$$p(\bar{Y}|X, B, \Sigma) \propto |\Sigma|^{-p/2} e^{-\frac{1}{2}tr(\bar{Y} - XB)'(\bar{Y} - XB)\Sigma^{-1}}.$$
 (5.31)

The joint posterior of B and Σ can be written as

$$p(B, \Sigma | X, Y) = p(Y | X, B, \Sigma) p(B) p(\Sigma)$$

$$\propto |\Sigma|^{-\frac{p+\nu}{2}} e^{-\frac{1}{2}tr\Sigma^{-1}G},$$
(5.32)

where $G = (\bar{Y} - XB)'(\bar{Y} - XB) + V_{\Sigma}$. Furthermore, we can write $G = S + V + (B - \tilde{B})'(X'X)(B - \tilde{B})$ where $S = (\bar{Y} - X\tilde{B})'(\bar{Y} - X\tilde{B})$ and $\tilde{B} = (X'X)^{-1}X'\bar{Y}$. This last form of G allows us to easily integrate out either B or Σ in the last equation and to obtain the posteriors of B and Σ respectively. Therefore

$$p(B|X,Y,\Sigma) \propto |\Sigma|^{-\frac{p+v}{2}} e^{\Sigma^{-1}(B-\bar{B})'(X'X)(B-\bar{B})}$$

$$p(\Sigma|X,Y) \propto |\Sigma|^{-\frac{p+v}{2}} e^{-\frac{1}{2}tr\Sigma^{-1}(V_{\Sigma}+S)},$$
(5.33)

and we can sample B and Σ using these last forms for a matric-variate normal for B and a Inverse Wishart for Σ .

If we condition equation (5.27) on B and we take $\tilde{Y} = Y - XB$ then we have a multivariate regression model

$$\tilde{Y} = \Phi + E \\
_{(p \times J)} + (p \times J), \quad (5.34)$$

and given equation (5.34) we can write the density of \tilde{Y} as

$$p(\tilde{Y}|\Phi,\Sigma) \propto |\Sigma|^{-p/2} e^{-\frac{1}{2}tr(\tilde{Y}-\Phi)'(\tilde{Y}-\Phi)\Sigma^{-1}}.$$
(5.35)

If we use $\phi = (\Phi), y = (\tilde{Y})$ then equation (5.35) can be expressed as

$$p(y|\phi,\Sigma) \propto |\Sigma|^{-p/2} e^{-\frac{1}{2}(y-\phi)'(\Sigma^{-1} \otimes I_{p \times p})(y-\phi)}.$$
 (5.36)

In the same way the prior for Φ can be expressed in vectorized form as

$$p(\phi) \propto |\Psi|^{-J/2} |\Lambda|^{-p/2} e^{-\frac{1}{2}\phi'(\Psi \otimes \Lambda)\phi}.$$
(5.37)

We use the vectorized forms to derive the posterior of ϕ . That is $p(\phi|y, \Sigma, \Psi) \propto p(y|\phi, \Sigma) \times p(\phi)$ and therefore

$$p(\phi|y, \Sigma, \Psi) \propto |\Lambda|^{-\frac{(2p+v_0)}{2}} |\Sigma|^{-p/2} e^{-\frac{1}{2}((\phi-a)'M^{-1}(\phi-a)+S_{\phi})}$$
(5.38)

where $S_{\phi} = y'Hy + a'M^{-1}a$, $M^{-1} = (H + F)$, $H = \Sigma^{-1} \times I$, $F = \Psi \otimes \Lambda$ and a = MHy.

The posterior of Λ can be derived from the third and fourth line of equation (5.28) as follows

$$p(\Lambda|\Phi, Y, \Sigma, \Psi) \propto |\Lambda|^{-\frac{(p+v_0)}{2}} e^{-\frac{1}{2}tr\Lambda(V_\Lambda + \Phi'\Psi\Phi)}.$$
(5.39)

If the form of Ψ contains a ρ or ρ_j parameters Gelfand and Vounatsou (2003) suggest to sample them from a discretized prior. The posterior of the ρ parameters is

$$p(\rho|\Phi, Y, \Sigma, \Lambda) \propto |\Psi|^{-J/2} e^{-\frac{1}{2}tr(\Psi\Phi'\Lambda\Phi)}.$$
(5.40)

In summary we use the following Gibbs steps

1. $\beta | X, \overline{Y}, \Phi, \Lambda, \Psi \sim N((X'X)^{-1}X'\overline{Y}), \Sigma \otimes (X'X)^{-1})$ 2. $\phi | B, X, Y, \Lambda, \Psi \sim N((\Sigma^{-1} \otimes I + \Psi \otimes \Lambda)^{-1}(\Sigma^{-1} \otimes I)y, (\Sigma^{-1} \otimes I + \Psi \otimes \Lambda))$ 3. $\Sigma | Y, B, \Phi, \Lambda, \Psi \sim IW((p + v)/2, V_{\Sigma} + S)$ 4. $\Lambda | \Psi, B, X, Y, \Sigma \sim IW((p + v_0)/2, V_{\Lambda} + \Phi'\Psi\Phi)$ In the chapter we set $V_{\Sigma} = I_3$ and $V_{\Lambda} = I_3$ and $v_0 = 5$ while v = 3 and p = 48. We use 48 states because we leave out Hawaii and Alaska. The matrix Ψ is defined based on the neighborhood structure of the US states where the element Ψ_{kj} takes the value of one when the state k is neighbor of the state j and zero otherwise. We further assume that $\rho_j = \rho$ and we sample this parameter based on the discretized prior described above. Finally, we assume that $\Sigma = \sigma^2 I$ and the β coefficients are equal across technologies.

Nederlandse Samenvatting (Summary in Dutch)

Dit proefschrift richt zich op de analyse van nieuwe of zeer recente marketing gegevens. Hiertoe introduceren we een aantal nieuwe econometrische modellen. We presenteren modellen die nuttig zijn om het volgende te analyseren: (1) het optimale tijdstip voor de lancering van nieuwe en dominante technologieën, (2) de triggers, snelheid en de timing van een substantiële prijsverlaging voor nieuwe producten, (3) de heterogeniteit in preferenties van consumenten die leidt tot specifieke substitutiepatronen in geaggregeerde verkoopgegevens, en (4) locaties die een grote invloed hebben op de verspreiding van nieuwe technologieën. De econometrische technieken die we toepassen zijn divers, maar ze zijn voornamelijk gebaseerd op Bayesiaanse methoden. We maken gebruik van Bayesiaanse mixture modellen, Bayesiaanse variabele selectie technieken, Bayesiaanse spatial modellen en we introduceren een nieuwe Bayesiaanse benadering voor het random coëfficiënten logit model. De gegevens die we analyseren bestaan uit unieke en grote datasets. We bestuderen de prijzen van video-games, de verkopen van video-game consoles, de totale omzet voor specifieke consumentenproducten en online zoekgegevens van Google.

Resumen en Español (Summary in Spanish)

En esta tesis se analizan nuevas bases de datos de mercadotecnia y se presentan nuevos modelos econometricos. Estos nuevos modelos son útiles para analizar (1) el tiempo de lanzamiento óptimo de nuevas tecnologías, (2) los factores que provocan cortes drásticos en los precios de nuevos productos y a la vez la velocidad y el momento en el que ocurren los cortes, (3) la heterogeneidad de los consumidores que determina los patrones de sustitución presentes en datos de ventas agregados, y (4) los mercados influyentes que determinan la difusión de nuevas tecnologías. Los métodos econométricos que se utilizan en esta tesis son diversos pero en su mayoría son métodos Bayesianos. Usamos modelos de mezcla de distribuciones, técnicas Bayesianas de selección de variables, modelos Bayesianos para datos geográficos y proponemos un nuevo enfoque Bayesiano para el modelo logit con coeficientes aleatorios. Los datos que se analizan son precios de videojuegos, ventas de consolas de videojuegos, datos agregados de ventas de productos de consumo y datos de búsqueda *en línea* de Google.

Bibliography

- Albuquerque, P., Bronnenberg, B., Corbett, C., 2007. A spatiotemporal analysis of the global diffusion of ISO 9000 and ISO 14000 certification. Management Science 53 (3), 451–468.
- Assunçao, R. M., 2003. Space varying coefficient models for small area data. Environmetrics 14 (5).
- Banerjee, S., Carlin, B. P., Gelfand, A. E., 2004. Hierarchical modeling and analysis for spatial data. Chapman & Hall/CRC.
- Barnard, J., McCulloch, R. E., Meng, X.-L., 2000. Modeling covariance matrices in terms of standard deviations and correlations, with application to shrinkage. Statistica Sinica 10 (4), 1281–1312.
- Bass, F. M., 1969. A new product growth model for consumer durables. Management Science 15 (5), 215 – 227.
- Bass, F. M., Bultez, A. V., 1982. A note on optimal strategic pricing of technological innovations. Marketing Science 1 (4), 371–378.
- Bass, F. M., Krishnan, T. V., Jain, D. C., 1994. Why the Bass model fits without decision variables. Marketing Science 13 (3), 203–223.
- Bayus, B. L., 1991. The consumer durable replacement buyer. Journal of Marketing 55 (1), 42 -51.
- Bayus, B. L., 1992. The dynamic pricing of next generation consumer durables. Marketing Science 11 (3), 251–265.
- Bayus, B. L., 1994. Optimal pricing and product development policies for new consumer durables. International Journal of Research in Marketing 11 (3), 249–259.
- Bayus, B. L., Jain, S., Rao, A. G., 1997. Too little, too early: Introduction timing and new product performance in the personal digital assistant industry. Journal of Marketing Research 34 (1), 50–63.
- Berry, S., Levinsohn, J., Pakes, A., 1995. Automobile prices in market equilibrium. Econometrica 63 (4), 841–890.
- Berry, S., Pakes, A., 2007. The pure characteristics demand model. International Economic Review 48 (4), 1193–1225.

- Besag, J., 1974. Spatial interaction and the statistical analysis of lattice systems. Journal of the Royal Statistical Society. Series B (Methodological) 32 (2), 192–236.
- Besag, J., York, J., Mollie, A., 1991. Bayesian image restoration, with two applications in spatial statistics. Annals of the Institute of Statistical Mathematics 43 (1), 1–20.
- Best, N. G., Arnold, R. A., Thomas, A., Waller, L. A., Conlon, E. M., 1999. Bayesian models for spatially correlated disease and exposure data. In: Bernardo, J. M., Berger, J. O., Dawid, A. P., Smith, A. F. (Eds.), Bayesian Statistics 6. Proceedings of the Valencia International Meetings on Bayesian Statistics. Oxford University Press: Oxford, pp. 131–156.
- Binken, J. L., Stremersch, S., 2009. The effect of superstar software on hardware sales in system markets. Journal of Marketing 73, 88–104.
- Bodapati, A. V., Gupta, S., 2004. The recoverability of segmentation structure from store-level aggregate data. Journal of Marketing Research 41 (3), 351–364.
- Bradlow, E. T., Bronnenberg, B., Russell, G. J., Arora, N., Bell, D. R., Duvvuri, S. D., Hofstede, F. T., Sismeiro, C., Thomadsen, R., Yang, S., 2005. Spatial models in marketing. Marketing Letters 16 (3), 267–278.
- Bucklin, L. P., Sengupta, S., 1993. The co-diffusion of complementary innovations supermarket scanners and upc symbols. Journal of Product Innovation Management 10 (2), 148 – 160.
- BusinessWeek, 2008a. Apple: Soon to be a mobile gaming force. November 5th. URL http://tiny.cc/i1513
- BusinessWeek, 2008b. Console race: Another lap for Sony. March 13th. URL http://tiny.cc/fepi0
- BusinessWeek Online, Sep. 2007. Apple averts a 'fanboy rebellion'. September 7th. URL http://tiny.cc/72rdr
- BusinessWeek Online, May 2008. iPhone out of stock after price cut. May 12th. URL http://tiny.cc/tnr9u
- Cameron, A., Trivedi, P., 2005. Microeconometrics: Methods and Applications. Cambridge University Press.
- Carlin, B. P., Banerjee, S., 2003. Hierarchical multivariate car models for spatio-temporally correlated survival data. Bayesian statistics 7, 45–64.
- Carlin, B. P., Louis, T. A., 2000. Bayes and Empirical Bayes Methods for Data Analysis. Chapman & Hall/CRC, New York.
- Chatterjee, R., Eliashberg, J., 1990. The innovation diffusion process in a heterogeneus population: A micromodeling approach. MANAGEMENT SCIENCE 36 (9), 1057 – 1079.
- Chintagunta, P. K., Dube, J.-P., Goh, K. Y., 2005. Beyond the endogeneity bias: The effect of unmeasured brand characteristics on household-level brand choice models. Management Science 51 (5), 832–849.

- Chintagunta, P. K., Nair, H. S., Sukumar, R., 2009. Measuring marketing-mix effects in the video-game console market. Journal of Applied Econometrics 24, 421–445.
- Chipman, H., George, E. I., McCulloch, R. E., 2001. The practical implementation of Bayesian model selection. Lecture Notes-Monograph Series 38, 65–134.
- Cho, W., Fowler, J., 2007. Legislative success in a small world. Working Paper .
- Choi, J., Hui, S., Bell, D. R., 2009. Spatio-temporal analysis of imitation behavior across new buyers at an online grocery retailer. Journal of Marketing Research 47 (1), 75–89.
- Christakis, N., Fowler, J. H., 2009. Connected: The surprising power of our social networks and how they shape our lives. Little, Brown and Co., New York.
- Clements, M., Ohashi, H., 2005. Indirect network effects and the product cycle: Video games in the us, 1994-2002. Journal of Industrial Economics 53 (4), 515–542.
- Clyde, M., George, E. I., 2004. Model uncertainty. Statistical Science 19 (1), 81–94.
- Cressie, N., 1992. Statistics for spatial data. Terra Nova 4 (5), 613-617.
- Danaher, P. J., Hardie, B., Putsis, W. P., 2001. Marketing-mix variables and the diffusion of successive generations of a technological innovation. Journal of Marketing Research 38 (4), 501 – 514.
- Dockner, E. J., Gaunersdorfer, A., 1996. Strategic new product pricing when demand obeys saturation effects. European Journal of Operational Research 90 (3), 589–598.
- Dockner, E. J., Jorgensen, S., 1988. Optimal advertising policies for diffusion models of new product innovation in monopolistic situations. Management Science 34 (1), 119–130.
- Dolan, R. J., Jeuland, A. P., 1981. Experience curves and dynamic demand models: Implications for optimal pricing strategies. Journal of Marketing 45 (1), 52–62.
- Doornik, J., 2007. Object-Oriented Matrix Programming Using Ox. Timberlake Consultants Press and Oxford.
- Duan, J. A., Mela, C. F., 2009. The role of spatial demand on outlet location and pricing. Journal of Marketing Research 46 (2), 260–278.
- Durbin, J., Koopman, S., 2002. A simple and efficient simulation smoother for state space time series analysis. Biometrika 89 (3), 603–615.
- Eberly, L., Carlin, B. P., 2000. Identifiability and convergence issues for markov chain monte carlo fitting of spatial models. Statistics in Medicine 19.
- Eliashberg, J., Jeuland, A. P., 1986. The impact of competitive entry in a developing market upon dynamic pricing strategies. Marketing Science 5 (1), 20–36.
- Fast Company Blog, 2009. Sony's Jack Tretton Talks PlayStation 3 and the 10-Year Console Cycle. July 3rd. URL http://tiny.cc/9zoh3

- Feng, Y., Gallego, G., 1995. Optimal starting times for end-of-season sales and optimal stopping times for promotional fares. Management Science 41 (8), 1371–1391.
- Ferguson, M. E., Koenigsberg, O., 2007. How should a firm manage deteriorating inventory? Production & Operations Management 16 (3), 306–321.
- Fernandez, C., Ley, E., Steel, M. F., 2001. Benchmark priors for Bayesian model averaging. Journal of Econometrics 100, 381–427.
- Financial Times, 2004. Sony sets the pace with Japan launch in hand-helds. December 13th.
- Franza, R., Gaimon, C., 1998. Flexibility and pricing decisions for high-volume products with short life cycles. International Journal of Flexible Manufacturing Systems 10 (1), 43–71.
- Fruhwirth-Schnatter, S., 2004. Efficient Bayesian parameter estimation. In: Harvey, A., Koopman, S., Shephard, N. (Eds.), State Space and Unobserved Component Models: Theory and Applications. Press Syndicate and the University of Cambridge, Ch. 7, pp. 123–151.
- Gamerman, D., Lopes, H. F., 2006. Markov Chain Monte Carlo: Stochastic Simulation for Bayesian Inference, 2nd Edition. Chapman & Hall/CRC, New York.
- Gamerman, D., Moreira, A. R., Rue, H., 2003. Space-varying regression models: specifications and simulation. Computational Statistics and Data Analysis 42 (3), 513–533.
- Garber, T., Goldenberg, J., Libai, B., Muller, E., 2004. From density to destiny: Using spatial dimension of sales data for early prediction of new product success. Marketing Science 23, 419–428.
- Gelfand, A. E., Banerjee, S., Gamerman, D., 2005. Spatial process modelling for univariate and multivariate dynamic spatial data. Environmetrics 16 (5).
- Gelfand, A. E., Vounatsou, P., 2003. Proper multivariate conditional autoregressive models for spatial data analysis. Biostatistics 4 (1), 11–15.
- George, E. I., McCulloch, R. E., 1997. Approaches for Bayesian variable selection. Statistica Sinica 7, 339–374.
- Goeree, M. S., 2008. Limited information and advertising in the US personal computer industry. Econometrica 76 (5), 1017–1074.
- Goldenberg, J., Han, S., Lehmann, D. R., Hong, J. W., 2009. The role of hubs in the adoption process. Journal of Marketing 73 (2), 1–13.
- Golder, P. N., Tellis, G. J., 1997. Will it ever fly? modeling the takeoff of really new consumer durables. Marketing Science 16 (3), 256–270.
- Gomez, V., Maravall, A., 2001. A Course in Time Series Analysis. NY: J.Wiley.
- Gupta, D., Hill, A. V., Bouzdine-Chameeva, T., 2006. A pricing model for clearing end-of-season retail inventory. European Journal of Operational Research 170 (2), 518–540.

- Gupta, M. C., Di Benedetto, C. A., 2007. Optimal pricing and advertising strategy for introducing a new business product with threat of competitive entry. Industrial Marketing Management 36 (4), 540–548.
- Haran, M., Hodges, J. S., Carlin, B. P., 2003. Accelerating computation in markov random field models for spatial data via structured MCMC. Journal of Computational and Graphical Statistics 12 (2), 249–264.
- Hastie, T., Tibshirani, R., Friedman, J., 2001. The elements of statistical learning: data mining, inference and prediction. Springer.
- Held, L., Rue, H., 2002. On block updating in markov random field models for disease mapping. Scandinavian Journal of Statistics, 597–614.
- Hofstede, F. T., Wedel, M., Steenkamp, J., 2002. Identifying spatial segments in international markets. Marketing Science 21 (2), 160–177.
- Horsky, D., 1990. A diffusion model incorporating product benefits, price, income and information. Marketing Science 9 (4), 342–365.
- Islam, T., Meade, N., 1997. The diffusion of succesive generations of technology: A more general model. Technological Forecasting and Social Change 56 (1), 49 – 60.
- Islam, T., Meade, N., 2000. Modelling diffusion and replacement. European Journal of Operational Research 125 (3), 551 – 570.
- Jank, W., Kannan, P., 2005. Understanding geographical markets of online firms using spatial models of customer choice. Marketing Science 24 (4), 623.
- Jiang, R., Manchanda, P., Rossi, P. E., 2009. Bayesian analysis of random coefficient logit models using aggregate data. Journal of Econometrics 149 (2), 136–148.
- Joshi, Y. V., Reibstein, D. J., Zhang, Z. J., 2009. Optimal entry timing in markets with social influence. Management Science 55 (6), 926.
- Jun, B. D., Park, Y. S., 1999. A choice-based diffusion model for multiple generations of products. Technological Forecasting and Social Change 61 (1), 45 – 58.
- Kahneman, D., Slovic, P., Tversky, A. (Eds.), 1982. Judgment under uncertainty: Heuristics and biases. Cambridge Univ Press.
- Kalish, S., 1983. Monopolist pricing with dynamic demand and production cost. Marketing Science 2 (2), 135–159.
- Kalish, S., 1985. A new product adoption model with price, advertising, and uncertainty. Management Science 31 (12), 1569–1585.
- Kalish, S., Lilien, G. L., 1983. Optimal price subsidy policy for accelerating the diffusion of innovation. Marketing Science 2 (4), 407–420.
- Kamien, M. I., Schwartz, N. L., 1972. Timing of innovations under rivalry. Econometrica 40 (1), 43 – 60.

- Kim, N., Chang, D., Shocker, A., 2000. Modeling inter-category and generational dynamics for a growing information technology industry. MANAGEMENT SCIENCE 46 (4), 496 – 512.
- Kim, N., Srivastava, R. K., Han, J. K., 2001. Consumer decision-making in a multi-generational choice set context. Journal of Business Research 53 (3), 123 – 136.
- Kim, S.-H., Srinivasan, V., 2001. Research paper no. 1720: A multiattribute model of the timing of buyers' upgrading to improved versions of high technology products. Research Paper Series Stanford University.
- Kim, W. J., Lee, J. D. a., 2005. Demand forecasting for multigenerational products combining discrete choice and dynamics of diffusion under technological trajectories. Technological Forecasting and Social Change 72 (7), 825 – 849.
- Kornish, L. J., 2001. Pricing for a durable-goods monopolist under rapid sequential innovation. Management Science 47 (11), 1552–1561.
- Krishnan, T. V., Bass, F. M., Jain, D. C., 1999. Optimal pricing strategy for new products. Management Science 45 (12), 1650–1663.
- Mahajan, V., Muller, E., 1996. Timing, diffusion, and substitution of succesive generations of technological innovations: The IBM mainframe case. Technological Forecasting and Social Change 51 (2), 109–132.
- Mardia, K., 1988. Multi-dimensional multivariate gaussian markov random fields with application to image processing. Journal of Multivariate Analysis 24 (2), 265–284.
- Morgan, L., Morgan, R., Moore, W., 2001. Quality and time-to-market trade-offs when there are multiple product generations. Manufacturing & Service Operations Management 3 (2), 89.
- Musalem, A., Bradlow, E. T., Raju, J. S., 2006. Bayesian estimation of random-coefficients choice models using aggregate data. Journal of Applied Econometrics 24 (3), 490–516.
- Nair, H. S., 2007. Intertemporal price discrimination with forward-looking consumers: Application to the US market for console video-games. Quantitative Marketing and Economics 5 (3), 239–292.
- Nascimento, F., Vanhonacker, W. R., 1993. Strategic pricing of differentiated consumer durables in a dynamic duopoly: A numerical analysis. Managerial and Decision Economics 14 (3), 193– 220.
- Nevo, A., 2001. Measuring market power in the ready-to-eat cereal industry. Econometrica 69 (2), 307–342.
- Nishida, T., 1983. Alpha status and agonistic alliance in wild chimpanzees. Primates 24 (3), 318–336.
- Nobre, A. A., Schmidt, A. M., Lopes, H. F., 2005. Spatio-temporal models for mapping the incidence of malaria in para. Environmetrics 16 (3), 291–304.
- Norton, J. A., Bass, F. M., 1987. A diffusion theory model of adoption and substitution for succesive generations of high-technology products. Management Science 33 (9), 1069 – 1086.

- Olson, J., Joi, S., 1985. A product diffusion model incorporating repeated purchase. Technological Forecasting and Social Change 27 (4), 385 – 397.
- Padmanabhan, V., Bass, F. M., 1993a. Optimal pricing of successive generations of product advances. International Journal of Research in Marketing 10 (2), 185 – 207.
- Padmanabhan, V., Bass, F. M., 1993b. Optimal pricing of successive generations of product advances. International Journal of Research in Marketing 10 (2), 185–207.
- Park, S., Gupta, S., 2009. Simulated maximum likelihood estimator for the random coefficient logit model using aggregate data. Journal of Marketing Research 46 (4), 531–542.
- Parker, P. M., 1992. Price elasticity dynamics over the adoption life cycle. Journal of Marketing Research 29 (3), 358–358.
- Peterson, R., Mahajan, V., 1978. Multi-product growth models. Research in Marketing 3 (15), 201 231.
- Prasad, A., Bronnenberg, B., Mahajan, V., 2004. Product entry timing in dual distrubition channels: The case of the movie indutry. Review of Marketing Science 2 (1), Article 4.
- Purohit, D., 1994. What should you do when your competitors send in the clones? Marketing Science 13 (4), 392 – 411.
- Putsis Jr, W. P., Balasubramanian, S., Kaplan, E. H., Sen, S. K., 1997. Mixing behavior in cross-country diffusion. Marketing Science 16 (4), 354–369.
- R Development Core Team, 2005. R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria, ISBN 3-900051-07-0.
- Rajan, A., Steinberg, R., Steinberg, R., 1992. Dynamic pricing and ordering decisions by a monopolist. Management Science 38 (2), 240–262.
- Raman, K., Chatterjee, R., 1995. Optimal monopolist pricing under demand uncertainty in dynamic markets. Management Science 41 (1), 144–162.
- Rao, R. C., Bass, F. M., 1985. Competition, strategy, and price dynamics: A theoretical and empirical investigation. Journal of Marketing Research 22 (3), 283–296.
- Reinganum, J., 1981. On the diffusion of new technologies: a game theoretic approach. Review of Economic Studies 48 (3), 395 – 406.
- Robert, C. P., Casella, G., 2004. Monte Carlo Statistical Methods. Springer.
- Robertson, T. S., Eliashberg, J., Rymon, T., 1995. New product announcement signals and incumbent reactions. Journal of Marketing 59 (3), 1 – 15.
- Robinson, B., Lakhani, C., 1975. Dynamic price models for new-product planning. Management Science 21 (10, Application Series), 1113–1122.
- Rogers, E. M., 2003. Diffusion of Innovations. Simon and Schuster: Free Press.

- Rue, H., Held, L., 2005. Gaussian Markov random fields: theory and applications. Chapman & Hall/CRC.
- Sahu, S., Gelfand, A., 1999. Identifiability, improper priors, and gibbs sampling for generalized linear models. Journal of the American Statistical Association 94 (445), 247–254.
- San Francisco Chronicle, 2001. Sega to stop production of Dreamcast. January 31st.
- Schmalen, H., 1982. Optimal price and advertising policy for new products. Journal of Business Research 10 (1), 17–30.
- Shankar, V., Bayus, B. L., 2003. Network effects and competition: An empirical analysis of the home video game industry. Strategic Management Journal 24 (4), 375–384.
- Simon, H., 1979. Dynamics of price elasticity and brand life cycles: An empirical study. Journal of Marketing Research 16 (4), 439–452.
- Sloot, L., Fok, D., Verhoef, P., 2006. The short-and long-term impact of an assortment reduction on category sales. Journal of Marketing Research 43 (4), 536–548.
- Souza, G. C., Bayus, B. L., Wagner, H. M., 2004. New-product strategy and industry clockspeed. Management Science 50 (4), 537–549.
- Tellis, G. J., Stremersch, S., Yin, E., 2003. The international takeoff of new products: The role of economics, culture, and country innovativeness. Marketing Science 22 (2), 188–208.
- Teng, J.-T., Thompson, G. L., 1996. Optimal strategies for general price-quality decision models of new products with learning production costs. European Journal of Operational Research 93 (3), 476–489.
- The Economist, 2004. Hand-to-hand combat video games. December 18th.
- The Herald, 2005. Battle of the thumbs: Game on in hand-held Console War. March 7th.
- The New York Times, 2006. Playstation 3 pushed back for delivery in november. March 16th.
- The Washington Post, 2006. Ramping up the Console Wars. October 14th.
- The Washington Post, 2008. Dusting off the PlayStation Portable. March 2nd.
- Train, K., 2003. Discrete choice models with simulation. Cambridge University Press.
- Trusov, M., Bodapati, A. V., Bucklin, R., 2010. Determining influential users in internet social networks. Journal of Marketing Research (forthcoming).
- Vakratsas, D., Bass, F. M., 2002. A segment-level hazard approach to studying household purchase timing decisions. Journal of Applied Econometrics 17 (1), 49 – 59.
- van den Bulte, C., Joshi, Y. V., 2007. New product diffusion with influentials and imitators. Marketing Science 26 (3), 400–400.
- van Everdingen, Y., Fok, D., Stremersch, S., 2009. Modeling global spillover of new product takeoff. Journal of Marketing Research 46 (5), 637–652.

- van Lawick-Goodall, J., 1973. The behavior of chimpanzees in their natural habitat. American Journal of Psychiatry 130 (1), 1–12.
- Wall, M. M., 2004. A close look at the spatial structure implied by the car and sar models. Journal of Statistical Planning and Inference 121 (2), 311–324.
- Wall Street Journal, 2006. Microsoft studies portable device for videogames. March 21st.
- Waller, L. A., Carlin, B. P., Xia, H., Gelfand, A. E., 1997. Hierarchical spatio-temporal mapping of disease rates. Journal of the American Statistical Association, 607–617.
- Watson, J., 2002. Strategy: An introduction to game theory. W.W. Norton & Company, Inc.
- Wilson, L. O., Norton, J. A., 1989. Optimal entry timing for a product line extension. Marketing Science 8 (1), 1 – 17.
- Wired Magazine, February 2010. Is Apple actually pushing for the \$10 iPad E-Book, after all? February 18th. URL http://tiny.cc/u963w
- Wired.com, September 2007. Four mistakes Apple made with the iPhone price drop. September 10th. URL http://tiny.cc/5m889
- Yang, S., Chen, Y., Allenby, G. M., 2003. Bayesian analysis of simultaneous demand and supply. Quantitative Marketing and Economics 1 (3), 251–275.
- Zhao, W., Zheng, Y.-S., 2000. Optimal dynamic pricing for perishable assets with nonhomogeneous demand. Management Science 46 (3), 375–375.

Author Index

Albuquerque, P. 160, 162, 163, 225 Allenby, G. M. 118, 132, 136, 139, 233 Arnold, R. A. 213, 215, 226 Arora, N. 163, 226 Assunçao, R. M. 211, 225 Balasubramanian, S. 162, 163, 231 Banerjee, S. 171, 210–213, 215, 216, 225, 226, 228 Barnard, J. 119, 127, 140, 225 Bass, F. M. 9, 11-14, 28, 74, 104, 225, 230 - 232Bayus, B. L. 9–11, 23, 28, 30, 35, 67, 73, 104, 225, 232 Bell, D. R. 160, 162, 163, 226, 227 Berry, S. 118, 124, 139, 225 Besag, J. 211, 212, 226 Best, N. G. 213, 215, 226 Binken, J. L. 23–25, 75, 226 Bodapati, A. V. 126, 160, 162, 163, 226, 232Bouzdine-Chameeva, T. 70, 74, 104, 228 Bradlow, E. T. 118, 126, 132, 136, 139, 163, 226, 230 Bronnenberg, B. 11, 160, 162, 163, 225, 226, 231 Bucklin, L. P. 11, 226 Bucklin, R. 160, 162, 163, 232 Bultez, A. V. 104, 225 Cameron, A. 26, 226

Carlin, B. P. 114, 135, 171, 210-213, 215, 216, 225-227, 229, 233 Casella, G. 114, 231 Chang, D. 9, 11, 230 Chatterjee, R. 11, 73, 104, 226, 231 Chen, Y. 118, 132, 136, 139, 233 Chintagunta, P. K. 23, 27, 71-73, 104, 119, 120, 226, 227 Chipman, H. 166, 209, 227 Cho, W. 160, 227 Choi, J. 160, 162, 163, 227 Christakis, N. 160, 227 Clements, M. 23, 71–73, 104, 227 Clyde, M. 209, 227 Conlon, E. M. 213, 215, 226 Corbett, C. 160, 162, 163, 225 Cressie, N. 210, 227 Danaher, P. J. 9, 11, 227 Di Benedetto, C. A. 104, 229 Dockner, E. J. 73, 104, 227 Dolan, R. J. 73, 104, 227 Doornik, J. 110, 227 Duan, J. A. 164, 227 Dube, J.-P. 119, 120, 226 Durbin, J. 120, 129, 130, 134, 140, 227 Duvvuri, S. D. 163, 226 Eberly, L. 213, 215, 227 Eliashberg, J. 11, 73, 104, 226, 227, 231 Feng, Y. 70, 74, 104, 228
Ferguson, M. E. 73, 74, 104, 228 Fernandez, C. 114, 228 Fok, D. 160-163, 168, 169, 232 Fowler, J. 160, 227 Fowler, J. H. 160, 227 Franza, R. 104, 228 Friedman, J. 168, 229 Fruhwirth-Schnatter, S. 138, 228 Gaimon, C. 104, 228 Gallego, G. 70, 74, 104, 228 Gamerman, D. 114, 211, 212, 216, 228 Garber, T. 162, 163, 171, 228 Gaunersdorfer, A. 73, 104, 227 Gelfand, A. 213, 232 Gelfand, A. E. 171, 210-216, 218, 225, 228, 233George, E. I. 166, 208–210, 227, 228 Goeree, M. S. 118, 228 Goh, K. Y. 119, 120, 226 Goldenberg, J. 160, 162, 163, 171, 228 Golder, P. N. 71, 169, 228 Gomez, V. 25, 228 Gupta, D. 70, 74, 104, 228 Gupta, M. C. 104, 229 Gupta, S. 118, 126, 226, 231 Han, J. K. 11, 230 Han, S. 160, 162, 163, 228 Haran, M. 135, 211, 229 Hardie, B. 9, 11, 227 Hastie, T. 168, 229 Held, L. 211, 215, 229, 232 Hill, A. V. 70, 74, 104, 228 Hodges, J. S. 135, 211, 229 Hofstede, F. T. 163, 226, 229 Hong, J. W. 160, 162, 163, 228

Horsky, D. 104, 229 Hui, S. 160, 162, 163, 227 Islam, T. 11, 229 Jain, D. C. 74, 104, 225, 230 Jain, S. 9, 11, 30, 67, 225 Jank, W. 162, 163, 229 Jeuland, A. P. 73, 104, 227 Jiang, R. 118, 119, 123–126, 128, 129, 132, 136, 139, 140, 229 Joi, S. 11, 231 Jorgensen, S. 104, 227 Joshi, Y. V. 9, 10, 35, 160, 229, 232 Jun, B. D. 11, 229 Kalish, S. 73, 104, 229 Kamien, M. I. 10, 21, 22, 30, 229 Kannan, P. 162, 163, 229 Kaplan, E. H. 162, 163, 231 Kim, N. 9, 11, 230 Kim, S.-H. 11, 230 Kim, W. J. 9, 11, 230 Koenigsberg, O. 73, 74, 104, 228 Koopman, S. 120, 129, 130, 134, 140, 227 Kornish, L. J. 104, 230 Krishnan, T. V. 74, 104, 225, 230 Lakhani, C. 73, 104, 231 Lee, J. D. a. 9, 11, 230 Lehmann, D. R. 160, 162, 163, 228 Levinsohn, J. 118, 124, 225 Ley, E. 114, 228 Libai, B. 162, 163, 171, 228 Lilien, G. L. 104, 229 Lopes, H. F. 114, 211, 228, 230 Louis, T. A. 114, 226 Mahajan, V. 9-11, 230, 231

Manchanda, P. 118, 119, 123–126, 128, 129, 132, 136, 139, 140, 229 Maravall, A. 25, 228 Mardia, K. 211, 230 McCulloch, R. E. 119, 127, 140, 166, 208-210, 225, 227, 228 Meade, N. 11, 229 Mela, C. F. 164, 227 Meng, X.-L. 119, 127, 140, 225 Mollie, A. 211, 226 Moore, W. 9, 11, 230 Moreira, A. R. 211, 216, 228 Morgan, L. 9, 11, 230 Morgan, R. 9, 11, 230 Muller, E. 9, 10, 162, 163, 171, 228, 230 Musalem, A. 118, 126, 132, 136, 139, 230 Nair, H. S. 23, 27, 71–73, 88, 104, 227, 230 Nascimento, F. 104, 230 Nevo, A. 118, 230 Nishida, T. 41, 230 Nobre, A. A. 211, 230 Norton, J. A. 9, 10, 12-14, 28, 230, 233 Ohashi, H. 23, 71-73, 104, 227 Olson, J. 11, 231 Padmanabhan, V. 11, 104, 231 Pakes, A. 118, 124, 139, 225 Park, S. 118, 231 Park, Y. S. 11, 229 Parker, P. M. 71, 72, 104, 231 Peterson, R. 11, 231 Prasad, A. 11, 231 Purohit, D. 11, 231 Putsis Jr, W. P. 162, 163, 231 Putsis, W. P. 9, 11, 227

R Development Core Team 27, 110, 231 Rajan, A. 74, 104, 231 Raju, J. S. 118, 126, 132, 136, 139, 230 Raman, K. 73, 104, 231 Rao, A. G. 9, 11, 30, 67, 225 Rao, R. C. 74, 104, 231 Reibstein, D. J. 9, 10, 35, 229 Reinganum, J. 67, 231 Robert, C. P. 114, 231 Robertson, T. S. 11, 231 Robinson, B. 73, 104, 231 Rogers, E. M. 161, 169, 231 Rossi, P. E. 118, 119, 123–126, 128, 129, 132, 136, 139, 140, 229 Rue, H. 211, 215, 216, 228, 229, 232 Russell, G. J. 163, 226 Rymon, T. 11, 231 Sahu, S. 213, 232 Schmalen, H. 73, 104, 232 Schmidt, A. M. 211, 230 Schwartz, N. L. 10, 21, 22, 30, 229 Sen, S. K. 162, 163, 231 Sengupta, S. 11, 226 Shankar, V. 23, 28, 232 Shocker, A. 9, 11, 230 Simon, H. 71, 104, 232 Sismeiro, C. 163, 226 Sloot, L. 168, 232 Souza, G. C. 9, 10, 35, 232 Srinivasan, V. 11, 230 Srivastava, R. K. 11, 230 Steel, M. F. 114, 228 Steenkamp, J. 163, 229 Steinberg, R. 74, 104, 231 Stremersch, S. 23–25, 71, 75, 160–163, 169, 226, 232

Sukumar, R. 23, 27, 71-73, 104, 227 Tellis, G. J. 71, 169, 228, 232 Teng, J.-T. 104, 232 Thomadsen, R. 163, 226 Thomas, A. 213, 215, 226 Thompson, G. L. 104, 232 Tibshirani, R. 168, 229 Train, K. 124, 232 Trivedi, P. 26, 226 Trusov, M. 160, 162, 163, 232 Vakratsas, D. 11, 232 van den Bulte, C. 160, 232 van Everdingen, Y. 160-163, 169, 232 van Lawick-Goodall, J. 8, 233 Vanhonacker, W. R. 104, 230 Verhoef, P. 168, 232 Vounatsou, P. 211, 214-216, 218, 228 Wagner, H. M. 9, 10, 35, 232 Wall, M. M. 211, 233 Waller, L. A. 211, 213, 215, 226, 233 Watson, J. 30, 233 Wedel, M. 163, 229 Wilson, L. O. 9, 10, 233 Xia, H. 211, 233 Yang, S. 118, 132, 136, 139, 163, 226, 233 Yin, E. 71, 169, 232 York, J. 211, 226 Zhang, Z. J. 9, 10, 35, 229 Zhao, W. 104, 233 Zheng, Y.-S. 104, 233

About the Author



Carlos Hernández Mireles received a B.A. in Economics with honors from ITESM in 2001 and he obtained the M.Phil. degree in Marketing from the Erasmus Research Institute of Management (ERIM) in 2006. In November of 2006, he started his Ph.D. research in ERIM and the Econometric Institute. His main research interests are marketing modeling and Bayesian methods. His research has been presented at several international conferences. Prior and during his Ph.D. Carlos worked for Cemex, Unilever and Heineken.

ERASMUS RESEARCH INSTITUTE OF MANAGEMENT

ERIM Ph.D. Series Research in Management

ERIM Electronic Series Portal: http://hdl.handle.net/1765/1

Agatz, N.A.H., *Demand Management in E-Fulfillment*, Promotor: Prof.dr.ir. J.A.E.E. van Nunen, EPS-2009-163-LIS, ISBN: 978-90-5892-200-7, http://hdl.handle.net/1765/15425

Althuizen, N.A.P., Analogical Reasoning as a Decision Support Principle for Weakly Structured Marketing Problems, Promotor: Prof.dr.ir. B. Wierenga, EPS-2006-095-MKT, ISBN: 90-5892-129-8, http://hdl.handle.net/1765/8190

Alvarez, H.L., Distributed Collaborative Learning Communities Enabled by Information Communication Technology, Promotor: Prof.dr. K. Kumar, EPS-2006-080-LIS, ISBN: 90-5892-112-3, http://hdl.handle.net/1765/7830

Appelman, J.H., Governance of Global Interorganizational Tourism Networks: Changing Forms of Co-ordination between the Travel Agency and Aviation Sector, Promotors: Prof.dr. F.M. Go & Prof.dr. B. Nooteboom, EPS-2004-036-MKT, ISBN: 90-5892-060-7, http://hdl.handle.net/1765/1199

Asperen, E. van, Essays on Port, Container, and Bulk Chemical Logistics Optimization, Promotor: Prof.dr.ir. R. Dekker, EPS-2009-181-LIS, ISBN: 978-90-5892-222-9, http://hdl.handle.net/1765/1

Assem, M.J. van den, Deal or No Deal? Decision Making under Risk in a Large-Stake TV Game Show and Related Experiments, Promotor: Prof.dr. J. Spronk, EPS-2008-138-F&A, ISBN: 978-90-5892-173-4, http://hdl.handle.net/1765/13566

Baquero, G, On Hedge Fund Performance, Capital Flows and Investor Psychology, Promotor: Prof.dr. M.J.C.M. Verbeek, EPS-2006-094-F&A, ISBN: 90-5892-131-X, http://hdl.handle.net/1765/8192

Berens, G., Corporate Branding: The Development of Corporate Associations and their Influence on Stakeholder Reactions, Promotor: Prof.dr. C.B.M. van Riel, EPS-2004-039-ORG, ISBN: 90-5892-065-8, http://hdl.handle.net/1765/1273

Berghe, D.A.F. van den, Working Across Borders: Multinational Enterprises and the Internationalization of Employment, Promotors: Prof.dr. R.J.M. van Tulder & Prof.dr. E.J.J. Schenk, EPS-2003-029-ORG, ISBN: 90-5892-05-34, http://hdl.handle.net/1765/1041

Berghman, L.A., Strategic Innovation Capacity: A Mixed Method Study on Deliberate Strategic Learning Mechanisms, Promotor: Prof.dr. P. Mattyssens, EPS-2006-087-MKT, ISBN: 90-5892-120-4, http://hdl.handle.net/1765/7991

Bijman, W.J.J., Essays on Agricultural Co-operatives: Governance Structure in Fruit and Vegetable Chains, Promotor: Prof.dr. G.W.J. Hendrikse, EPS-2002-015-ORG, ISBN: 90-5892-024-0, http://hdl.handle.net/1765/867

Bispo, A., Labour Market Segmentation: An investigation into the Dutch hospitality industry, Promotors: Prof.dr. G.H.M. Evers & Prof.dr. A.R. Thurik, EPS-2007-108-ORG, ISBN: 90-5892-136-9, http://hdl.handle.net/1765/10283

Blindenbach-Driessen, F., Innovation Management in Project-Based Firms, Promotor: Prof.dr. S.L. van de Velde, EPS-2006-082-LIS, ISBN: 90-5892-110-7, http://hdl.handle.net/1765/7828

Boer, C.A., *Distributed Simulation in Industry*, Promotors: Prof.dr. A. de Bruin & Prof.dr.ir. A. Verbraeck, EPS-2005-065-LIS, ISBN: 90-5892-093-3, http://hdl.handle.net/1765/6925

Boer, N.I., Knowledge Sharing within Organizations: A situated and Relational Perspective, Promotor: Prof.dr. K. Kumar, EPS-2005-060-LIS, ISBN: 90-5892-086-0, http://hdl.handle.net/1765/6770

Boer-Sorbán, K., Agent-Based Simulation of Financial Markets: A modular, Continuous-Time Approach, Promotor: Prof.dr. A. de Bruin, EPS-2008-119-LIS, ISBN: 90-5892-155-0, http://hdl.handle.net/1765/10870

Boon, C.T., *HRM and Fit: Survival of the Fittest!?*, Promotors: Prof.dr. J. Paauwe & Prof.dr. D.N. den Hartog, EPS-2008-129-ORG, ISBN: 978-90-5892-162-8, http://hdl.handle.net/1765/12606

Braun, E., *City Marketing: Towards an Integrated Approach*, Promotor: Prof.dr. L. van den Berg, EPS-2008-142-MKT, ISBN: 978-90-5892-180-2, http://hdl.handle.net/1765/13694

Brito, M.P. de, Managing Reverse Logistics or Reversing Logistics Management? Promotors: Prof.dr.ir. R. Dekker & Prof.dr. M. B. M. de Koster, EPS-2004-035-LIS, ISBN: 90-5892-058-5, http://hdl.handle.net/1765/1132

Brohm, R., Polycentric Order in Organizations: A Dialogue between Michael Polanyi and IT-Consultants on Knowledge, Morality, and Organization, Promotors: Prof.dr. G.
W. J. Hendrikse & Prof.dr. H. K. Letiche, EPS-2005-063-ORG, ISBN: 90-5892-095-X, http://hdl.handle.net/1765/6911 Brumme, W.-H., Manufacturing Capability Switching in the High-Tech Electronics Technology Life Cycle, Promotors: Prof.dr.ir. J.A.E.E. van Nunen & Prof.dr.ir. L.N. Van Wassenhove, EPS-2008-126-LIS, ISBN: 978-90-5892-150-5, http://hdl.handle.net/1765/12103

Budiono, D.P., The Analysis of Mutual Fund Performance: Evidence from U.S. Equity Mutual Funds, Promotor: Prof.dr.ir. M.J.C.M. Verbeek, EPS-2010-185-F&A, ISBN: 978-90-5892-224-3, http://hdl.handle.net/1765/1

Burgers, J.H., Managing Corporate Venturing: Multilevel Studies on Project Autonomy, Integration, Knowledge Relatedness, and Phases in the New Business Development Process, Promotors: Prof.dr.ir. F.A.J. Van den Bosch & Prof.dr. H.W. Volberda, EPS-2008-136-STR, ISBN: 978-90-5892-174-1, http://hdl.handle.net/1765/13484

Campbell, R.A.J., *Rethinking Risk in International Financial Markets*, Promotor: Prof.dr. C.G. Koedijk, EPS-2001-005-F&A, ISBN: 90-5892-008-9, http://hdl.handle.net/1765/306

Chen, C.-M., Evaluation and Design of Supply Chain Operations Using DEA, Promotor: Prof.dr. J.A.E.E. van Nunen, EPS-2009-172-LIS, ISBN: 978-90-5892-209-0, http://hdl.handle.net/1765/1

Chen, H., Individual Mobile Communication Services and Tariffs, Promotor: Prof.dr. L.F.J.M. Pau, EPS-2008-123-LIS, ISBN: 90-5892-158-1, http://hdl.handle.net/1765/11141

Chen, Y., Labour Flexibility in China's Companies: An Empirical Study, Promotors: Prof.dr. A. Buitendam & Prof.dr. B. Krug, EPS-2001-006-ORG, ISBN: 90-5892-012-7, http://hdl.handle.net/1765/307

Damen, F.J.A., *Taking the Lead: The Role of Affect in Leadership Effectiveness*, Promotor: Prof.dr. D.L. van Knippenberg, EPS-2007-107-ORG, http://hdl.handle.net/1765/10282

Daniševská, P., Empirical Studies on Financial Intermediation and Corporate Policies, Promotor: Prof.dr. C.G. Koedijk, EPS-2004-044-F&A, ISBN: 90-5892-070-4, http://hdl.handle.net/1765/1518

Delporte-Vermeiren, D.J.E., Improving the Flexibility and Profitability of ICT-enabled Business Networks: An Assessment Method and Tool, Promotors: Prof. mr. dr. P.H.M. Vervest & Prof.dr.ir. H.W.G.M. van Heck, EPS-2003-020-LIS, ISBN: 90-5892-040-2, http://hdl.handle.net/1765/359

Derwall, J.M.M., *The Economic Virtues of SRI and CSR*, Promotor: Prof.dr. C.G. Koedijk, EPS-2007-101-F&A, ISBN: 90-5892-132-8, http://hdl.handle.net/1765/8986

Diepen, M. van, Dynamics and Competition in Charitable Giving, Promotor: Prof.dr. Ph.H.B.F. Franses, EPS-2009-159-MKT, ISBN: 978-90-5892-188-8, http://hdl.handle.net/1765/14526 Dietz, H.M.S., Managing (Sales)People towards Perfomance: HR Strategy, Leadership & Teamwork, Promotor: Prof.dr. G.W.J. Hendrikse, EPS-2009-168-ORG, ISBN: 978-90-5892-210-6, http://hdl.handle.net/1765/1

Dijksterhuis, M., Organizational Dynamics of Cognition and Action in the Changing Dutch and US Banking Industries, Promotors: Prof.dr.ir. F.A.J. van den Bosch & Prof.dr. H.W. Volberda, EPS-2003-026-STR, ISBN: 90-5892-048-8, http://hdl.handle.net/1765/1037

Eijk, A.R. van der, Behind Networks: Knowledge Transfer, Favor Exchange and Performance, Promotors: Prof.dr. S.L. van de Velde & Prof.dr.drs. W.A. Dolfsma, EPS-2009-161-LIS, ISBN: 978-90-5892-190-1, http://hdl.handle.net/1765/14613

Elstak, M.N., Flipping the Identity Coin: The Comparative Effect of Perceived, Projected and Desired Organizational Identity on Organizational Identification and Desired Behavior, Promotor: Prof.dr. C.B.M. van Riel, EPS-2008-117-ORG, ISBN: 90-5892-148-2, http://hdl.handle.net/1765/10723

Erken, H.P.G., *Productivity, R&D and Entrepreneurship*, Promotor: Prof.dr. A.R. Thurik, EPS-2008-147-ORG, ISBN: 978-90-5892-179-6, http://hdl.handle.net/1765/14004

Fenema, P.C. van, Coordination and Control of Globally Distributed Software Projects, Promotor: Prof.dr. K. Kumar, EPS-2002-019-LIS, ISBN: 90-5892-030-5, http://hdl.handle.net/1765/360

Fleischmann, M., *Quantitative Models for Reverse Logistics*, Promotors: Prof.dr.ir. J.A.E.E. van Nunen & Prof.dr.ir. R. Dekker, EPS-2000-002-LIS, ISBN: 35-4041-711-7, http://hdl.handle.net/1765/1044

Flier, B., Strategic Renewal of European Financial Incumbents: Coevolution of Environmental Selection, Institutional Effects, and Managerial Intentionality, Promotors: Prof.dr.ir. F.A.J. van den Bosch & Prof.dr. H.W. Volberda, EPS-2003-033-STR, ISBN: 90-5892-055-0, http://hdl.handle.net/1765/1071

Fok, D., Advanced Econometric Marketing Models, Promotor: Prof.dr. Ph.H.B.F. Franses, EPS-2003-027-MKT, ISBN: 90-5892-049-6, http://hdl.handle.net/1765/1035

Ganzaroli, A., Creating Trust between Local and Global Systems, Promotors: Prof.dr. K. Kumar & Prof.dr. R.M. Lee, EPS-2002-018-LIS, ISBN: 90-5892-031-3, http://hdl.handle.net/1765/361

Gertsen, H.F.M., Riding a Tiger without Being Eaten: How Companies and Analysts Tame Financial Restatements and Influence Corporate Reputation, Promotor: Prof.dr. C.B.M. van Riel, EPS-2009-171-ORG, ISBN: 90-5892-214-4, http://hdl.handle.net/1765/1 Gilsing, V.A., Exploration, Exploitation and Co-evolution in Innovation Networks, Promotors: Prof.dr. B. Nooteboom & Prof.dr. J.P.M. Groenewegen, EPS-2003-032-ORG, ISBN: 90-5892-054-2, http://hdl.handle.net/1765/1040

Gijsbers, G.W., Agricultural Innovation in Asia: Drivers, Paradigms and Performance, Promotor: Prof.dr. R.J.M. van Tulder, EPS-2009-156-ORG, ISBN: 978-90-5892-191-8, http://hdl.handle.net/1765/14524

Gong, Y., Stochastic Modelling and Analysis of Warehouse Operations, Promotors: Prof.dr. M.B.M. de Koster & Prof.dr. S.L. van de Velde, EPS-2009-180-LIS, ISBN: 978-90-5892-219-9, http://hdl.handle.net/1765/1

Govers, R., Virtual Tourism Destination Image: Glocal Identities Constructed, Perceived and Experienced, Promotors: Prof.dr. F.M. Go & Prof.dr. K. Kumar, EPS-2005-069-MKT, ISBN: 90-5892-107-7, http://hdl.handle.net/1765/6981

Graaf, G. de, Tractable Morality: Customer Discourses of Bankers, Veterinarians and Charity Workers, Promotors: Prof.dr. F. Leijnse & Prof.dr. T. van Willigenburg, EPS-2003-031-ORG, ISBN: 90-5892-051-8, http://hdl.handle.net/1765/1038

Greeven, M.J., Innovation in an Uncertain Institutional Environment: Private Software Entrepreneurs in Hangzhou, China, Promotor: Prof.dr. B. Krug, EPS-2009-164-ORG, ISBN: 978-90-5892-202-1, http://hdl.handle.net/1765/15426

Groot, E.A. de, *Essays on Economic Cycles*, Promotors: Prof.dr. Ph.H.B.F. Franses & Prof.dr. H.R. Commandeur, EPS-2006-091-MKT, ISBN: 90-5892-123-9, http://hdl.handle.net/1765/8216

Guenster, N.K., Investment Strategies Based on Social Responsibility and Bubbles, Promotor: Prof.dr. C.G. Koedijk, EPS-2008-175-F&A, ISBN: 978-90-5892-206-9, http://hdl.handle.net/1765/1

Gutkowska, A.B., *Essays on the Dynamic Portfolio Choice*, Promotor: Prof.dr. A.C.F. Vorst, EPS-2006-085-F&A, ISBN: 90-5892-118-2, http://hdl.handle.net/1765/7994

Hagemeijer, R.E., *The Unmasking of the Other*, Promotors: Prof.dr. S.J. Magala & Prof.dr. H.K. Letiche, EPS-2005-068-ORG, ISBN: 90-5892-097-6, http://hdl.handle.net/1765/6963

Hakimi, N.A, Leader Empowering Behaviour: The Leader's Perspective: Understanding the Motivation behind Leader Empowering Behaviour, Promotor: Prof.dr. D.L. van Knippenberg, EPS-2010-184-ORG, http://hdl.handle.net/1765/1

Halderen, M.D. van, Organizational Identity Expressiveness and Perception Management: Principles for Expressing the Organizational Identity in Order to Manage the Perceptions and Behavioral Reactions of External Stakeholders, Promotor: Prof.dr. S.B.M. van Riel, EPS-2008-122-ORG, ISBN: 90-5892-153-6, http://hdl.handle.net/1765/10872 Hartigh, E. den, Increasing Returns and Firm Performance: An Empirical Study, Promotor: Prof.dr. H.R. Commandeur, EPS-2005-067-STR, ISBN: 90-5892-098-4, http://hdl.handle.net/1765/6939

Hermans. J.M., *ICT in Information Services; Use and Deployment of the Dutch Securities Trade, 1860-1970, Promotor: Prof.dr. drs. F.H.A. Janszen, EPS-2004-046-ORG, ISBN 90-5892-072-0, http://hdl.handle.net/1765/1793*

Hessels, S.J.A., International Entrepreneurship: Value Creation Across National Borders, Promotor: Prof.dr. A.R. Thurik, EPS-2008-144-ORG, ISBN: 978-90-5892-181-9, http://hdl.handle.net/1765/13942

Heugens, P.P.M.A.R., Strategic Issues Management: Implications for Corporate Performance, Promotors: Prof.dr.ir. F.A.J. van den Bosch & Prof.dr. C.B.M. van Riel, EPS-2001-007-STR, ISBN: 90-5892-009-9, http://hdl.handle.net/1765/358

Heuvel, W. van den, *The Economic Lot-Sizing Problem: New Results and Extensions*, Promotor: Prof.dr. A.P.L. Wagelmans, EPS-2006-093-LIS, ISBN: 90-5892-124-7, http://hdl.handle.net/1765/1805

Hoedemaekers, C.M.W., Performance, Pinned down: A Lacanian Analysis of Subjectivity at Work, Promotors: Prof.dr. S. Magala & Prof.dr. D.H. den Hartog, EPS-2008-121-ORG, ISBN: 90-5892-156-7, http://hdl.handle.net/1765/10871

Hooghiemstra, R., The Construction of Reality: Cultural Differences in Selfserving Behaviour in Accounting Narratives, Promotors: Prof.dr. L.G. van der Tas RA & Prof.dr. A.Th.H. Pruyn, EPS-2003-025-F&A, ISBN: 90-5892-047-X, http://hdl.handle.net/1765/871

Hu, Y., Essays on the Governance of Agricultural Products: Cooperatives and Contract Farming, Promotors: Prof.dr. G.W.J. Hendrkse & Prof.dr. B. Krug, EPS-2007-113-ORG, ISBN: 90-5892-145-1, http://hdl.handle.net/1765/10535

Huij, J.J., New Insights into Mutual Funds: Performance and Family Strategies, Promotor: Prof.dr. M.C.J.M. Verbeek, EPS-2007-099-F&A, ISBN: 90-5892-134-4, http://hdl.handle.net/1765/9398

Huurman, C.I., *Dealing with Electricity Prices*, Promotor: Prof.dr. C.D. Koedijk, EPS-2007-098-F&A, ISBN: 90-5892-130-1, http://hdl.handle.net/1765/9399

Iastrebova, K, Manager's Information Overload: The Impact of Coping Strategies on Decision-Making Performance, Promotor: Prof.dr. H.G. van Dissel, EPS-2006-077-LIS, ISBN: 90-5892-111-5, http://hdl.handle.net/1765/7329

ERIM Ph.D. SERIES

Iwaarden, J.D. van, Changing Quality Controls: The Effects of Increasing Product Variety and Shortening Product Life Cycles, Promotors: Prof.dr. B.G. Dale & Prof.dr. A.R.T. Williams, EPS-2006-084-ORG, ISBN: 90-5892-117-4, http://hdl.handle.net/1765/7992

Jansen, J.J.P., *Ambidextrous Organizations*, Promotors: Prof.dr.ir. F.A.J. Van den Bosch & Prof.dr. H.W. Volberda, EPS-2005-055-STR, ISBN: 90-5892-081-X, http://hdl.handle.net/1765/6774

Jaspers, F.P.H., Organizing Systemic Innovation, Promotor: Prof.dr.ir. J.C.M. van den Ende, EPS-2009-160-ORG, ISBN: 978-90-5892-197-), http://hdl.handle.net/1765/14974

Jennen, M.G.J., *Empirical Essays on Office Market Dynamics*, Promotors: Prof.dr. C.G. Koedijk & Prof.dr. D. Brounen, EPS-2008-140-F&A, ISBN: 978-90-5892-176-5, http://hdl.handle.net/1765/13692

Jiang, T., Capital Structure Determinants and Governance Structure Variety in Franchising, Promotors: Prof.dr. G. Hendrikse & Prof.dr. A. de Jong, EPS-2009-158-F&A, ISBN: 978-90-5892-199-4, http://hdl.handle.net/1765/14975

Jiao, T., *Essays in Financial Accounting*, Promotor: Prof.dr. G.M.H. Mertens, EPS-2009-176-F&A, ISBN: 978-90-5892-211-3, http://hdl.handle.net/1765/1

Jong, C. de, Dealing with Derivatives: Studies on the Role, Informational Content and Pricing of Financial Derivatives, Promotor: Prof.dr. C.G. Koedijk, EPS-2003-023-F&A, ISBN: 90-5892-043-7, http://hdl.handle.net/1765/1043

Kaa, G. van, Standard Battles for Complex Systems: Empirical Research on the Home Network, Promotors: Prof.dr.ir. J. van den Ende & Prof.dr.ir. H.W.G.M. van Heck, EPS-2009-166-ORG, ISBN: 978-90-5892-205-2, http://hdl.handle.net/1765/1

Keizer, A.B., The Changing Logic of Japanese Employment Practices: A Firm-Level Analysis of Four Industries, Promotors: Prof.dr. J.A. Stam & Prof.dr. J.P.M. Groenewegen, EPS-2005-057-ORG, ISBN: 90-5892-087-9, http://hdl.handle.net/1765/6667

Kijkuit, R.C., Social Networks in the Front End: The Organizational Life of an Idea, Promotor: Prof.dr. B. Nooteboom, EPS-2007-104-ORG, ISBN: 90-5892-137-6, http://hdl.handle.net/1765/10074

Kippers, J., Empirical Studies on Cash Payments, Promotor: Prof.dr. Ph.H.B.F. Franses, EPS-2004-043-F&A, ISBN: 90-5892-069-0, http://hdl.handle.net/1765/1520

Klein, M.H., Poverty Alleviation through Sustainable Strategic Business Models: Essays on Poverty Alleviation as a Business Strategy, Promotor: Prof.dr. H.R. Commandeur, EPS-2008-135-STR, ISBN: 978-90-5892-168-0, http://hdl.handle.net/1765/13482 Knapp, S., The Econometrics of Maritime Safety: Recommendations to Enhance Safety at Sea, Promotor: Prof.dr. Ph.H.B.F. Franses, EPS-2007-096-ORG, ISBN: 90-5892-127-1, http://hdl.handle.net/1765/7913

Kole, E., On Crises, Crashes and Comovements, Promotors: Prof.dr. C.G. Koedijk & Prof.dr. M.J.C.M. Verbeek, EPS-2006-083-F&A, ISBN: 90-5892-114-X, http://hdl.handle.net/1765/7829

Kooij-de Bode, J.M., Distributed Information and Group Decision-Making: Effects of Diversity and Affect, Promotor: Prof.dr. D.L. van Knippenberg, EPS-2007-115-ORG, http://hdl.handle.net/1765/10722

Koppius, O.R., Information Architecture and Electronic Market Performance, Promotors: Prof.dr. P.H.M. Vervest & Prof.dr.ir. H.W.G.M. van Heck, EPS-2002-013-LIS, ISBN: 90-5892-023-2, http://hdl.handle.net/1765/921

Kotlarsky, J., Management of Globally Distributed Component-Based Software Development Projects, Promotor: Prof.dr. K. Kumar, EPS-2005-059-LIS, ISBN: 90-5892-088-7, http://hdl.handle.net/1765/6772

Krauth, E.I., *Real-Time Planning Support: A Task-Technology Fit Perspective*, Promotors: Prof.dr. S.L. van de Velde & Prof.dr. J. van Hillegersberg, EPS-2008-155-LIS, ISBN: 978-90-5892-193-2, http://hdl.handle.net/1765/14521

Kuilman, J., The Re-Emergence of Foreign Banks in Shanghai: An Ecological Analysis, Promotor: Prof.dr. B. Krug, EPS-2005-066-ORG, ISBN: 90-5892-096-8, http://hdl.handle.net/1765/6926

Kwee, Z., Investigating Three Key Principles of Sustained Strategic Renewal: A Longitudinal Study of Long-Lived Firms, Promotors: Prof.dr.ir. F.A.J. Van den Bosch & Prof.dr. H.W. Volberda, EPS-2009-174-STR, ISBN: 90-5892-212-0,http://hdl.handle.net/1765/1

Langen, P.W. de, The Performance of Seaport Clusters: A Framework to Analyze Cluster Performance and an Application to the Seaport Clusters of Durban, Rotterdam and the Lower Mississippi, Promotors: Prof.dr. B. Nooteboom & Prof. drs. H.W.H. Welters, EPS-2004-034-LIS, ISBN: 90-5892-056-9, http://hdl.handle.net/1765/1133

Le Anh, T., Intelligent Control of Vehicle-Based Internal Transport Systems, Promotors: Prof.dr. M.B.M. de Koster & Prof.dr.ir. R. Dekker, EPS-2005-051-LIS, ISBN: 90-5892-079-8, http://hdl.handle.net/1765/6554

Le-Duc, T., Design and Control of Efficient Order Picking Processes, Promotor: Prof.dr. M.B.M. de Koster, EPS-2005-064-LIS, ISBN: 90-5892-094-1, http://hdl.handle.net/1765/6910 Leeuwen, E.P. van, Recovered-Resource Dependent Industries and the Strategic Renewal of Incumbent Firm: A Multi-Level Study of Recovered Resource Dependence Management and Strategic Renewal in the European Paper and Board Industry, Promotors: Prof.dr.ir. F.A.J. Van den Bosch & Prof.dr. H.W. Volberda, EPS-2007-109-STR, ISBN: 90-5892-140-6, http://hdl.handle.net/1765/10183

Lentink, R.M., Algorithmic Decision Support for Shunt Planning, Promotors: Prof.dr. L.G. Kroon & Prof.dr.ir. J.A.E.E. van Nunen, EPS-2006-073-LIS, ISBN: 90-5892-104-2, http://hdl.handle.net/1765/7328

Li, T., Informedness and Customer-Centric Revenue Management, Promotors: Prof.dr. P.H.M. Vervest & Prof.dr.ir. H.W.G.M. van Heck, EPS-2009-146-LIS, ISBN: 978-90-5892-195-6, http://hdl.handle.net/1765/14525

Liang, G., New Competition: Foreign Direct Investment and Industrial Development in China, Promotor: Prof.dr. R.J.M. van Tulder, EPS-2004-047-ORG, ISBN: 90-5892-073-9, http://hdl.handle.net/1765/1795

Liere, D.W. van, Network Horizon and the Dynamics of Network Positions: A Multi-Method Multi-Level Longitudinal Study of Interfirm Networks, Promotor: Prof.dr. P.H.M. Vervest, EPS-2007-105-LIS, ISBN: 90-5892-139-0, http://hdl.handle.net/1765/10181

Loef, J., Incongruity between Ads and Consumer Expectations of Advertising, Promotors: Prof.dr. W.F. van Raaij & Prof.dr. G. Antonides, EPS-2002-017-MKT, ISBN: 90-5892-028-3, http://hdl.handle.net/1765/869

Maeseneire, W., de, *Essays on Firm Valuation and Value Appropriation*, Promotor: Prof.dr. J.T.J. Smit, EPS-2005-053-F&A, ISBN: 90-5892-082-8, http://hdl.handle.net/1765/6768

Londoño, M. del Pilar, Institutional Arrangements that Affect Free Trade Agreements: Economic Rationality Versus Interest Groups, Promotors: Prof.dr. H.E. Haralambides & Prof.dr. J.F. Francois, EPS-2006-078-LIS, ISBN: 90-5892-108-5, http://hdl.handle.net/1765/7578

Maas, A.A., van der, Strategy Implementation in a Small Island Context: An Integrative Framework, Promotor: Prof.dr. H.G. van Dissel, EPS-2008-127-LIS, ISBN: 978-90-5892-160-4, http://hdl.handle.net/1765/12278

Maas, K.E.G., Corporate Social Performance: From Output Measurement to Impact Measurement, Promotor: Prof.dr. H.R. Commandeur, EPS-2009-182-STR, ISBN: 978-90-5892-225-0, http://hdl.handle.net/1765/1

Maeseneire, W., de, *Essays on Firm Valuation and Value Appropriation*, Promotor: Prof.dr. J.T.J. Smit, EPS-2005-053-F&A, ISBN: 90-5892-082-8, http://hdl.handle.net/1765/6768

Mandele, L.M., van der, Leadership and the Inflection Point: A Longitudinal Perspective, Promotors: Prof.dr. H.W. Volberda & Prof.dr. H.R. Commandeur, EPS-2004-042-STR, ISBN: 90-5892-067-4, http://hdl.handle.net/1765/1302

Meer, J.R. van der, Operational Control of Internal Transport, Promotors: Prof.dr. M.B.M. de Koster & Prof.dr.ir. R. Dekker, EPS-2000-001-LIS, ISBN: 90-5892-004-6, http://hdl.handle.net/1765/859

Mentink, A., *Essays on Corporate Bonds*, Promotor: Prof.dr. A.C.F. Vorst, EPS-2005-070-F&A, ISBN: 90-5892-100-X, http://hdl.handle.net/1765/7121

Meyer, R.J.H., Mapping the Mind of the Strategist: A Quantitative Methodology for Measuring the Strategic Beliefs of Executives, Promotor: Prof.dr. R.J.M. van Tulder, EPS-2007-106-ORG, ISBN: 978-90-5892-141-3, http://hdl.handle.net/1765/10182

Miltenburg, P.R., Effects of Modular Sourcing on Manufacturing Flexibility in the Automotive Industry: A Study among German OEMs, Promotors: Prof.dr. J. Paauwe & Prof.dr. H.R. Commandeur, EPS-2003-030-ORG, ISBN: 90-5892-052-6, http://hdl.handle.net/1765/1039

Moerman, G.A., Empirical Studies on Asset Pricing and Banking in the Euro Area, Promotor: Prof.dr. C.G. Koedijk, EPS-2005-058-F&A, ISBN: 90-5892-090-9, http://hdl.handle.net/1765/6666

Moitra, D., Globalization of R&D: Leveraging Offshoring for Innovative Capability and Organizational Flexibility, Promotor: Prof.dr. K. Kumar, EPS-2008-150-LIS, ISBN: 978-90-5892-184-0, http://hdl.handle.net/1765/14081

Mol, M.M., Outsourcing, Supplier-relations and Internationalisation: Global Source Strategy as a Chinese Puzzle, Promotor: Prof.dr. R.J.M. van Tulder, EPS-2001-010-ORG, ISBN: 90-5892-014-3, http://hdl.handle.net/1765/355

Mom, T.J.M., Managers' Exploration and Exploitation Activities: The Influence of Organizational Factors and Knowledge Inflows, Promotors: Prof.dr.ir. F.A.J. Van den Bosch & Prof.dr. H.W. Volberda, EPS-2006-079-STR, ISBN: 90-5892-116-6, http://hdl.handle.net/1765

Moonen, J.M., Multi-Agent Systems for Transportation Planning and Coordination, Promotors: Prof.dr. J. van Hillegersberg & Prof.dr. S.L. van de Velde, EPS-2009-177-LIS, ISBN: 978-90-5892-216-8, http://hdl.handle.net/1

ERIM Ph.D. SERIES

Mulder, A., Government Dilemmas in the Private Provision of Public Goods, Promotor: Prof.dr. R.J.M. van Tulder, EPS-2004-045-ORG, ISBN: 90-5892-071-2, http://hdl.handle.net/1765/1790

Muller, A.R., *The Rise of Regionalism: Core Company Strategies Under The Second Wave of Integration*, Promotor: Prof.dr. R.J.M. van Tulder, EPS-2004-038-ORG, ISBN: 90-5892-062-3, http://hdl.handle.net/1765/1272

Nalbantov G.I., Essays on Some Recent Penalization Methods with Applications in Finance and Marketing, Promotor: Prof. dr P.J.F. Groenen, EPS-2008-132-F&A, ISBN: 978-90-5892-166-6, http://hdl.handle.net/1765/13319

Nederveen Pieterse, A., Goal Orientation in Teams: The Role of Diversity, Promotor: Prof.dr. D.L. van Knippenberg, EPS-2009-162-ORG, http://hdl.handle.net/1765/15240

Nguyen, T.T., Capital Structure, Strategic Competition, and Governance, Promotor: Prof.dr. A. de Jong, EPS-2008-148-F&A, ISBN: 90-5892-178-9, http://hdl.handle.net/1765/14005

Niesten, E.M.M.I., Regulation, Governance and Adaptation: Governance Transformations in the Dutch and French Liberalizing Electricity Industries, Promotors: Prof.dr. A. Jolink & Prof.dr. J.P.M. Groenewegen, EPS-2009-170-ORG, ISBN: 978-90-5892-208-3, http://hdl.handle.net/1765/1

Nieuwenboer, N.A. den, *Seeing the Shadow of the Self*, Promotor: Prof.dr. S.P. Kaptein, EPS-2008-151-ORG, ISBN: 978-90-5892-182-6, http://hdl.handle.net/1765/14223

Ning, H., Hierarchical Portfolio Management: Theory and Applications, Promotor: Prof.dr.
 J. Spronk, EPS-2007-118-F&A, ISBN: 90-5892-152-9, http://hdl.handle.net/1765/10868

Noeverman, J., Management Control Systems, Evaluative Style, and Behaviour: Exploring the Concept and Behavioural Consequences of Evaluative Style, Promotors: Prof.dr. E.G.J. Vosselman & Prof.dr. A.R.T. Williams, EPS-2007-120-F&A, ISBN: 90-5892-151-2, http://hdl.handle.net/1765/10869

Nuijten, I., Servant Leadership: Paradox or Diamond in the Rough? A Multidimensional Measure and Empirical Evidence, Promotor: Prof.dr. D.L. van Knippenberg, EPS-2009-183-ORG, http://hdl.handle.net/1765/1

Oosterhout, J., van, *The Quest for Legitimacy: On Authority and Responsibility in Governance*, Promotors: Prof.dr. T. van Willigenburg & Prof.mr. H.R. van Gunsteren, EPS-2002-012-ORG, ISBN: 90-5892-022-4, http://hdl.handle.net/1765/362

Oostrum, J.M., van, Applying Mathematical Models to Surgical Patient Planning, Promotor: Prof.dr. A.P.M. Wagelmans, EPS-2009-179-LIS, ISBN: 978-90-5892-217-5, http://hdl.handle.net/1765/1 Paape, L., Corporate Governance: The Impact on the Role, Position, and Scope of Services of the Internal Audit Function, Promotors: Prof.dr. G.J. van der Pijl & Prof.dr. H. Commandeur, EPS-2007-111-MKT, ISBN: 90-5892-143-7, http://hdl.handle.net/1765/10417

Pak, K., *Revenue Management: New Features and Models*, Promotor: Prof.dr.ir. R. Dekker, EPS-2005-061-LIS, ISBN: 90-5892-092-5, http://hdl.handle.net/1765/362/6771

Pattikawa, L.H, Innovation in the Pharmaceutical Industry: Evidence from Drug Introduction in the U.S., Promotors: Prof.dr. H.R.Commandeur, EPS-2007-102-MKT, ISBN: 90-5892-135-2, http://hdl.handle.net/1765/9626

Peeters, L.W.P., *Cyclic Railway Timetable Optimization*, Promotors: Prof.dr. L.G. Kroon & Prof.dr.ir. J.A.E.E. van Nunen, EPS-2003-022-LIS, ISBN: 90-5892-042-9, http://hdl.handle.net/1765/429

Pietersz, R., Pricing Models for Bermudan-style Interest Rate Derivatives, Promotors: Prof.dr. A.A.J. Pelsser & Prof.dr. A.C.F. Vorst, EPS-2005-071-F&A, ISBN: 90-5892-099-2, http://hdl.handle.net/1765/7122

Poel, A.M. van der, Empirical Essays in Corporate Finance and Financial Reporting, Promotors: Prof.dr. A. de Jong & Prof.dr. G.M.H. Mertens, EPS-2007-133-F&A, ISBN: 978-90-5892-165-9, http://hdl.handle.net/1765/13320

Popova, V., Knowledge Discovery and Monotonicity, Promotor: Prof.dr. A. de Bruin, EPS-2004-037-LIS, ISBN: 90-5892-061-5, http://hdl.handle.net/1765/1201

Pouchkarev, I., Performance Evaluation of Constrained Portfolios, Promotors: Prof.dr. J. Spronk & Dr. W.G.P.M. Hallerbach, EPS-2005-052-F&A, ISBN: 90-5892-083-6, http://hdl.handle.net/1765/6731

Prins, R., Modeling Consumer Adoption and Usage of Value-Added Mobile Services, Promotors: Prof.dr. Ph.H.B.F. Franses & Prof.dr. P.C. Verhoef, EPS-2008-128-MKT, ISBN: 978/90-5892-161-1, http://hdl.handle.net/1765/12461

Puvanasvari Ratnasingam, P., Interorganizational Trust in Business to Business E-Commerce, Promotors: Prof.dr. K. Kumar & Prof.dr. H.G. van Dissel, EPS-2001-009-LIS, ISBN: 90-5892-017-8, http://hdl.handle.net/1765/356

Quak, H.J., Sustainability of Urban Freight Transport: Retail Distribution and Local Regulation in Cities, Promotor: Prof.dr. M.B.M. de Koster, EPS-2008-124-LIS, ISBN: 978-90-5892-154-3, http://hdl.handle.net/1765/11990

Quariguasi Frota Neto, J., *Eco-efficient Supply Chains for Electrical and Electronic Products*, Promotors: Prof.dr.ir. J.A.E.E. van Nunen & Prof.dr.ir. H.W.G.M. van Heck, EPS-2008-152-LIS, ISBN: 978-90-5892-192-5, http://hdl.handle.net/1765/14785 Radkevitch, U.L, Online Reverse Auction for Procurement of Services, Promotor: Prof.dr.ir. H.W.G.M. van Heck, EPS-2008-137-LIS, ISBN: 978-90-5892-171-0, http://hdl.handle.net/1765/13497

Rinsum, M. van, Performance Measurement and Managerial Time Orientation, Promotor: Prof.dr. F.G.H. Hartmann, EPS-2006-088-F&A, ISBN: 90-5892-121-2, http://hdl.handle.net/1765/7993

Roelofsen, E.M., *The Role of Analyst Conference Calls in Capital Markets*, Promotors: Prof.dr. G.M.H. Mertens & Prof.dr. L.G. van der Tas RA, EPS-2010-190-F&A, ISBN: 978-90-5892-228-1, http://hdl.handle.net/1765/1

Romero Morales, D., Optimization Problems in Supply Chain Management, Promotors: Prof.dr.ir. J.A.E.E. van Nunen & Dr. H.E. Romeijn, EPS-2000-003-LIS, ISBN: 90-9014078-6, http://hdl.handle.net/1765/865

Roodbergen, K.J., Layout and Routing Methods for Warehouses, Promotors: Prof.dr. M.B.M. de Koster & Prof.dr.ir. J.A.E.E. van Nunen, EPS-2001-004-LIS, ISBN: 90-5892-005-4, http://hdl.handle.net/1765/861

Rook, L., *Imitation in Creative Task Performance*, Promotor: Prof.dr. D.L. van Knippenberg, EPS-2008-125-ORG, http://hdl.handle.net/1765/11555

Rosmalen, J. van, Segmentation and Dimension Reduction: Exploratory and Model-Based Approaches, Promotor: Prof.dr. P.J.F. Groenen, EPS-2009-165-MKT, ISBN: 978-90-5892-201-4, http://hdl.handle.net/1765/15536

Rus, D., The Dark Side of Leadership: Exploring the Psychology of Leader Selfserving Behavior, Promotor: Prof.dr. D.L. van Knippenberg, EPS-2009-178-ORG, http://hdl.handle.net/1765/1

Samii, R., Leveraging Logistics Partnerships: Lessons from Humanitarian Organizations, Promotors: Prof.dr.ir. J.A.E.E. van Nunen & Prof.dr.ir. L.N. Van Wassenhove, EPS-2008-153-LIS, ISBN: 978-90-5892-186-4, http://hdl.handle.net/1765/14519

Schaik, D. van, M&A in Japan: An Analysis of Merger Waves and Hostile Takeovers, Promotors: Prof.dr. J. Spronk & Prof.dr. J.P.M. Groenewegen, EPS-2008-141-F&A, ISBN: 978-90-5892-169-7, http://hdl.handle.net/1765/13693

Schauten, M.B.J., Valuation, Capital Structure Decisions and the Cost of Capital, Promotors: Prof.dr. J. Spronk & Prof.dr. D. van Dijk, EPS-2008-134-F&A, ISBN: 978-90-5892-172-7, http://hdl.handle.net/1765/13480

Schramade, W.L.J., *Corporate Bonds Issuers*, Promotor: Prof.dr. A. De Jong, EPS-2006-092-F&A, ISBN: 90-5892-125-5, http://hdl.handle.net/1765/8191

Schweizer, T.S., An Individual Psychology of Novelty-Seeking, Creativity and Innovation, Promotor: Prof.dr. R.J.M. van Tulder, EPS-2004-048-ORG, ISBN: 90-5892-077-1, http://hdl.handle.net/1765/1818

Six, F.E., Trust and Trouble: Building Interpersonal Trust Within Organizations, Promotors: Prof.dr. B. Nooteboom & Prof.dr. A.M. Sorge, EPS-2004-040-ORG, ISBN: 90-5892-064-X, http://hdl.handle.net/1765/1271

Slager, A.M.H., *Banking across Borders*, Promotors: Prof.dr. R.J.M. van Tulder & Prof.dr. D.M.N. van Wensveen, EPS-2004-041-ORG, ISBN: 90-5892-066-6, http://hdl.handle.net/1765/1301

Sloot, L., Understanding Consumer Reactions to Assortment Unavailability, Promotors: Prof.dr. H.R. Commandeur, Prof.dr. E. Peelen & Prof.dr. P.C. Verhoef, EPS-2006-074-MKT, ISBN: 90-5892-102-6, http://hdl.handle.net/1765/7438

Smit, W., Market Information Sharing in Channel Relationships: Its Nature, Antecedents and Consequences, Promotors: Prof.dr.ir. G.H. van Bruggen & Prof.dr.ir. B. Wierenga, EPS-2006-076-MKT, ISBN: 90-5892-106-9, http://hdl.handle.net/1765/7327

Sonnenberg, M., The Signalling Effect of HRM on Psychological Contracts of Employees: A Multi-level Perspective, Promotor: Prof.dr. J. Paauwe, EPS-2006-086-ORG, ISBN: 90-5892-119-0, http://hdl.handle.net/1765/7995

Speklé, R.F., Beyond Generics: A closer Look at Hybrid and Hierarchical Governance, Promotor: Prof.dr. M.A. van Hoepen RA, EPS-2001-008-F&A, ISBN: 90-5892-011-9, http://hdl.handle.net/1765/357

Srour, F.J., Dissecting Drayage: An Examination of Structure, Information, and Control in Drayage Operations, Promotor: Prof.dr. S.L. van de Velde, EPS-2010-186-LIS, http://hdl.handle.net/1765/1

Stam, D.A., Managing Dreams and Ambitions: A Psychological Analysis of Vision Communication, Promotor: Prof.dr. D.L. van Knippenberg, EPS-2008-149-ORG, http://hdl.handle.net/1765/14080

Stienstra, M., Strategic Renewal in Regulatory Environments: How Inter- and Intraorganisational Institutional Forces Influence European Energy Incumbent Firms, Promotors: Prof.dr.ir. F.A.J. Van den Bosch & Prof.dr. H.W. Volberda, EPS-2008-145-STR, ISBN: 978-90-5892-184-0, http://hdl.handle.net/1765/13943

Sweldens, S.T.L.R., Evaluative Conditioning 2.0: Direct versus Associative Transfer of Affect to Brands, Promotor: Prof.dr. S.M.J. van Osselaer, EPS-2009-167-MKT, ISBN: 978-90-5892-204-5, http://hdl.handle.net/1765/1

ERIM Ph.D. SERIES

Szkudlarek, B.A., Spinning the Web of Reentry: [Re]connecting reentry training theory and practice, Promotor: Prof.dr. S.J. Magala, EPS-2008-143-ORG, ISBN: 978-90-5892-177-2, http://hdl.handle.net/1765/13695

Teunter, L.H., Analysis of Sales Promotion Effects on Household Purchase Behavior, Promotors: Prof.dr.ir. B. Wierenga & Prof.dr. T. Kloek, EPS-2002-016-MKT, ISBN: 90-5892-029-1, http://hdl.handle.net/1765/868

Tims, B., Empirical Studies on Exchange Rate Puzzles and Volatility, Promotor: Prof.dr.
 C.G. Koedijk, EPS-2006-089-F&A, ISBN: 90-5892-113-1, http://hdl.handle.net/1765/8066

Tuk, M.A., Is Friendship Silent When Money Talks? How Consumers Respond to Wordof-Mouth Marketing, Promotors: Prof.dr.ir. A. Smidts & Prof.dr. D.H.J. Wigboldus, EPS-2008-130-MKT, ISBN: 978-90-5892-164-2, http://hdl.handle.net/1765/12702

Valck, K. de, Virtual Communities of Consumption: Networks of Consumer Knowledge and Companionship, Promotors: Prof.dr.ir. G.H. van Bruggen & Prof.dr.ir. B. Wierenga, EPS-2005-050-MKT, ISBN: 90-5892-078-X, http://hdl.handle.net/1765/6663

Valk, W. van der, Buyer-Seller Interaction Patterns During Ongoing Service Exchange, Promotors: Prof.dr. J.Y.F. Wynstra & Prof.dr.ir. B. Axelsson, EPS-2007-116-MKT, ISBN: 90-5892-146-8, http://hdl.handle.net/1765/10856

Verheul, I., Is There a (Fe)male Approach? Understanding Gender Differences in Entrepreneurship, Prof.dr. A.R. Thurik, EPS-2005-054-ORG, ISBN: 90-5892-080-1, http://hdl.handle.net/1765/2005

Verwijmeren, P., Empirical Essays on Debt, Equity, and Convertible Securities, Promotors: Prof.dr. A. de Jong & Prof.dr. M.J.C.M. Verbeek, EPS-2009-154-F&A, ISBN: 978-90-5892-187-1, http://hdl.handle.net/1765/14312

Vis, I.F.A., *Planning and Control Concepts for Material Handling Systems*, Promotors: Prof.dr. M.B.M. de Koster & Prof.dr.ir. R. Dekker, EPS-2002-014-LIS, ISBN: 90-5892-021-6, http://hdl.handle.net/1765/866

Vlaar, P.W.L., Making Sense of Formalization in Interorganizational Relationships: Beyond Coordination and Control, Promotors: Prof.dr.ir. F.A.J. Van den Bosch & Prof.dr. H.W. Volberda, EPS-2006-075-STR, ISBN 90-5892-103-4, http://hdl.handle.net/1765/7326

Vliet, P. van, *Downside Risk and Empirical Asset Pricing*, Promotor: Prof.dr. G.T. Post, EPS-2004-049-F&A, ISBN: 90-5892-07-55, http://hdl.handle.net/1765/1819

Vlist, P. van der, Synchronizing the Retail Supply Chain, Promotors: Prof.dr.ir. J.A.E.E. van Nunen & Prof.dr. A.G. de Kok, EPS-2007-110-LIS, ISBN: 90-5892-142-0, http://hdl.handle.net/1765/10418

Vries-van Ketel E. de, *How Assortment Variety Affects Assortment Attractiveness:* A Consumer Perspective, Promotors: Prof.dr. G.H. van Bruggen & Prof.dr.ir. A. Smidts, EPS-2006-072-MKT, ISBN: 90-5892-101-8, http://hdl.handle.net/1765/7193

Vromans, M.J.C.M., Reliability of Railway Systems, Promotors: Prof.dr. L.G. Kroon, Prof.dr.ir. R. Dekker & Prof.dr.ir. J.A.E.E. van Nunen, EPS-2005-062-LIS, ISBN: 90-5892-089-5, http://hdl.handle.net/1765/6773

Vroomen, B.L.K., The Effects of the Internet, Recommendation Quality and Decision Strategies on Consumer Choice, Promotor: Prof.dr. Ph.H.B.F. Franses, EPS-2006-090-MKT, ISBN: 90-5892-122-0, http://hdl.handle.net/1765/8067

Waal, T. de, *Processing of Erroneous and Unsafe Data*, Promotor: Prof.dr.ir. R. Dekker, EPS-2003-024-LIS, ISBN: 90-5892-045-3, http://hdl.handle.net/1765/870

Waard, E.J. de, Engaging Environmental Turbulence: Organizational Determinants for Repetitive Quick and Adequate Responses, Promotors: Prof.dr. H.W. Volberda & Prof.dr. J. Soeters, EPS-2010-189-STR, ISBN: 978-90-5892-229-8, http://hdl.handle.net/1765/1

Wall, R.S., Netscape: Cities and Global Corporate Networks, Promotor: Prof.dr. G.A. van der Knaap, EPS-2009-169-ORG, ISBN: 978-90-5892-207-6, http://hdl.handle.net/1765/1

Watkins Fassler, K., *Macroeconomic Crisis and Firm Performance*, Promotors: Prof.dr. J. Spronk & Prof.dr. D.J. van Dijk, EPS-2007-103-F&A, ISBN: 90-5892-138-3, http://hdl.handle.net/1765/10065

Weerdt, N.P. van der, Organizational Flexibility for Hypercompetitive Markets: Empirical Evidence of the Composition and Context Specificity of Dynamic Capabilities and Organization Design Parameters, Promotor: Prof.dr. H.W. Volberda, EPS-2009-173-STR, ISBN: 978-90-5892-215-1, http://hdl.handle.net/1765/1

Wennekers, A.R.M., Entrepreneurship at Country Level: Economic and Non-Economic Determinants, Promotor: Prof.dr. A.R. Thurik, EPS-2006-81-ORG, ISBN: 90-5892-115-8, http://hdl.handle.net/1765/7982

Wielemaker, M.W., Managing Initiatives: A Synthesis of the Conditioning and Knowledge-Creating View, Promotors: Prof.dr. H.W. Volberda & Prof.dr. C.W.F. Baden-Fuller, EPS-2003-28-STR, ISBN: 90-5892-050-X, http://hdl.handle.net/1765/1042

Wijk, R.A.J.L. van, Organizing Knowledge in Internal Networks: A Multilevel Study, Promotor: Prof.dr.ir. F.A.J. van den Bosch, EPS-2003-021-STR, ISBN: 90-5892-039-9, http://hdl.handle.net/1765/347

Wolters, M.J.J., *The Business of Modularity and the Modularity of Business*, Promotors: Prof. mr. dr. P.H.M. Vervest & Prof.dr.ir. H.W.G.M. van Heck, EPS-2002-011-LIS, ISBN: 90-5892-020-8, http://hdl.handle.net/1765/920

Wubben, M.J.J., Social Functions of Emotions in Social Dilemmas, Promotors: Prof.dr.D. De Cremer & Prof.dr. E. van Dijk, EPS-2009-187-ORG, http://hdl.handle.net/1765/1

Xu, Y., Empirical Essays on the Stock Returns, Risk Management, and Liquidity Creation of Banks, Promotor: Prof.dr. M.J.C.M. Verbeek, EPS-2010-188-F&A, http://hdl.handle.net/1765/1

Yang, J., Towards the Restructuring and Co-ordination Mechanisms for the Architecture of Chinese Transport Logistics, Promotor: Prof.dr. H.E. Harlambides, EPS-2009-157-LIS, ISBN: 978-90-5892-198-7, http://hdl.handle.net/1765/14527

Yu, M., Enhancing Warehouse Perfromance by Efficient Order Picking, Promotor: Prof.dr. M.B.M. de Koster, EPS-2008-139-LIS, ISBN: 978-90-5892-167-3, http://hdl.handle.net/1765/13691

Zhang, X., Strategizing of Foreign Firms in China: An Institution-based Perspective, Promotor: Prof.dr. B. Krug, EPS-2007-114-ORG, ISBN: 90-5892-147-5, http://hdl.handle.net/1765/10721

Zhu, Z., *Essays on China's Tax System*, Promotors: Prof.dr. B. Krug & Prof.dr. G.W.J. Hendrikse, EPS-2007-112-ORG, ISBN: 90-5892-144-4, http://hdl.handle.net/1765/10502

Zwart, G.J. de, *Empirical Studies on Financial Markets: Private Equity, Corporate Bonds and Emerging Markets*, Promotors: Prof.dr. M.J.C.M. Verbeek & Prof.dr. D.J.C. van Dijk, EPS-2008-131-F&A, ISBN: 978-90-5892-163-5, http://hdl.handle.net/1765/12703

MARKETING MODELING FOR NEW PRODUCTS

This thesis addresses the analysis of new or very recent marketing data and the introduction of new marketing models. We present a collection of models that are useful to analyze (1) the optimal launch time of new and dominant technologies, (2) the triggers, speed and timing of new products' price landings, (3) the consumer heterogeneity that drives substitution patterns present in aggregate data, and (4) the influential locations that drive the diffusion of new technologies. The econometric approaches that we apply are diverse but they are predominantly Bayesian methods. We use Bayesian mixture modelling, Bayesian variable selection techniques, Bayesian spatial models and we put forward a new Bayesian approach for the random coefficient logit model. The data that we analyze consist of unique and large datasets of video-game prices, video-game consoles' sales, aggregate sales data for consumer products and Google's online search data.

ERIM

The Erasmus Research Institute of Management (ERIM) is the Research School (Onderzoekschool) in the field of management of the Erasmus University Rotterdam. The founding participants of ERIM are Rotterdam School of Management (RSM), and the Erasmus School of Economics (ESE). ERIM was founded in 1999 and is officially accredited by the Royal Netherlands Academy of Arts and Sciences (KNAW). The research undertaken by ERIM is focused on the management of the firm in its environment, its intra- and interfirm relations, and its business processes in their interdependent connections.

The objective of ERIM is to carry out first rate research in management, and to offer an advanced doctoral programme in Research in Management. Within ERIM, over three hundred senior researchers and PhD candidates are active in the different research programmes. From a variety of academic backgrounds and expertises, the ERIM community is united in striving for excellence and working at the forefront of creating new business knowledge.



ERIM PhD Series Research in Management

Erasmus Research Institute of Management - ERIM Rotterdam School of Management (RSM) Erasmus School of Economics (ESE) P.O. Box 1738, 3000 DR Rotterdam The Netherlands
 Tel.
 +31 10 408 11 82

 Fax
 +31 10 408 96 40

 E-mail
 info@erim.eur.nl

 Internet
 www.erim.eur.nl